AN EXPLORATORY STUDY OF MULTIPLE STAKEHOLDER VIEWS ON HIGHER EDUCATION ENTREPRENEURIAL ECOSYSTEMS IN A RESOURCE-CONSTRAINED ENVIRONMENT

Submitted in fulfilment of the requirements for degree of Doctor of Philosophy

Nilusha Gallage (102364009)

School of Business, Law and Entrepreneurship Swinburne University of Technology

Submitted - October 2022 Approved – January 2023



ABSTRACT

Similar to businesses around the world, higher education institutions are giving importance to their entrepreneurial ecosystems focusing on knowledge commercialisation and student startups. At the same time, there is limited scholarly literature available on higher education entrepreneurial ecosystems developing students for entrepreneurship within more resourceconstrained environments. This is problematic as a lack of entrepreneurial education and support for entrepreneurs reduce the possibilities of start-up success. Given the importance of ecosystems in the higher education context (HEEE), the purpose of this study is to investigate how higher education entrepreneurial ecosystems can influence students' entrepreneurial development in resource-constrained environments.

This study addresses three key topics: how do diverse stakeholders anticipate the continued development of HEEEs in a resource-constrained environment?; what do diverse stakeholders perceive as the contextual factors of HEEEs that could influence students' E&I capabilities in a resource-constrained environment?; and how can diverse stakeholders engage in the factors of HEEEs that could influence students' E&I capabilities in a resource-constrained environment?; E&I capabilities in a resource-constrained environment?; E&I capabilities in a resource-constrained environment. Using exploratory qualitative research, four questions related to continued development, composition and social capital are investigated. An interpretative perspective of diverse views was drawn from 40 interviews held at 2 points in time with 6 stakeholder categories of the entrepreneurial ecosystem representing multi-levels in a resource-constrained environment. Content analysis using NVivo includes qualitative themes and pattern quantification presented in the form of a re-ordered matrix and spectrum diagram.

The interviews indicate that higher education entrepreneurship ecosystems in Sri Lanka are still in the initial phase of co-creation. It is evident that these ecosystems in Sri Lanka have not kept up with pace as they are not actively supporting entrepreneurship, unlike literature well documented in developed countries. In Sri Lanka, higher education entrepreneurship ecosystems suffer from a lack of internal drive from the top management of institutions and external support from the entrepreneurial ecosystem. Findings contribute to literature by extending four academic discussions: (1) HEEEs beyond start-ups to E&I capabilities of students; (2) Beyond the common HEEE factors to contextual factors; (3) Beyond the HEEE factors to actors; and (4) Beyond the composition of HEEEs to social capital and connections. This study brings entrepreneurship and innovation capabilities to HEEE literature advocating for human capital development of entrepreneurs in resource-constrained environments.

The emerging role of higher education entrepreneurial ecosystems is dichotomous, focusing on both entrepreneurship and innovation. Implications and recommendations are formulated, leading in particular to developing entrepreneurship and innovation capabilities of students within higher education entrepreneurship ecosystems in resource-constrained environments. This study contributes empirical findings with a deeper understanding on the concept HEEE that currently lacks theoretical clarification and empirical research in resource-constrained environments. The contribution of this study recognizes a set of interrelated factors and actors of a higher education entrepreneurship ecosystem that can operationally address the challenges in a more resourced constrained environment and how it might bring a focus on developing students' entrepreneurship and innovation capabilities. The findings also open new avenues intersecting with various disciplines including strategic management and design for entrepreneurship researchers in the higher education context.

Keywords – Higher education entrepreneurial ecosystems; student entrepreneurial development; entrepreneurship and innovation capability; stakeholders; resource-constrained environment; Sri Lanka.

DECLARATION

This thesis:

- contains no material that has been accepted for the award to the candidate of any other degree or diploma, with due reference made about this in the text of the examinable outcome;
- to the best of the candidate's knowledge contains no material previously published or written by another person except where due reference is made in the text of the examinable outcome;
- contains less than 100,000 words in length exclusive of bibliography and appendices;
- was professionally edited by Adam Finlay, an IPEd accredited editor, in terms of Standards D and E covered in the Australian Standards for Editing Practice including style of presentation and grammar, with no change to substantive content of the examinable outcome; and
- has met all the requirements of the ethical approval from the Swinburne University of Technology Human Research Ethics Committee under the project 202915-5183.

Nilusha Gallage

5 October 2022

ACKNOWLEDGEMENT

"What's more important, the journey or the destination? Neither. It is the people you travel with."

This was an intriguing quote shared by Professor David Audretsch during my doctoral journey. Travelling with you all makes the destination worthy of the journey and I am grateful to all the people who contributed towards the accomplishment of this PhD thesis.

Submitting this thesis is a significant moment in my life as an academic and in a way, I consider it as my own 'start-up'. To express the meaning of this piece of work to me, I would like to relate to a statement by Kuratko (2005, p. 592) that "entrepreneurial history will judge you, and as the years pass, you will judge yourself, on the extent to which you have used your abilities to pioneer and lead our universities into a new horizon. In your hands is the future of your entrepreneurial world and the fulfillment of the best qualities of your own spirit." To the best of my ability, this is my initiative to move higher education institutions in resource-constrained environments forward. This research on the nexus of higher education entrepreneurial ecosystems and student entrepreneurial development is integrative to my academic career and higher education. I dedicate my career to discovering knowledge and developing students, specifically for entrepreneurship and innovation.

First, I would like to thank Swinburne University of Technology for the tuition fee sponsorship offered to me and funding my doctoral study. Without Professor Christopher Selvarajah accepting my research proposal, this journey would have not set sail. Thank you for giving me this life-changing opportunity, guiding me through the initial phase of my doctoral studies and being a constant support. Dr Viet Le – thank you for taking over as my second principal supervisor, guiding me through the process and offering me the opportunity to be part of the ACERE organising committee. Dr Richard Laferriere – I appreciate you offering me my first academic opportunity as a sessional in Melbourne, sharing your expertise, providing constructive feedback and supporting me continuously. I am deeply grateful for the research insights and continuous support extended by the review panel members including Professor Christine Jubb, Dr Rosemary Fisher and Dr Simon Masli from the School of Business, Law and Entrepreneurship. Thank you for assessing my candidature progress and offering recommendations to shape my research.

Second, I acknowledge all the participants of this study including deans/heads of school, academics, entrepreneurs, angel investors and support professionals in Colombo, Sri Lanka. Without their willingness to participate and their shared perspectives, this thesis would have not been possible to make a contribution to knowledge.

I would like to warmly thank the FBL Research Team and Graduate Research Team for all the support, opportunities, and encouragement. Professor. Timothy Marjoribanks and Professor. Helana Scheepers, I am grateful for the support extended to me during my candidature. I would also like to thank the other teams that supported me – HDR Candidature, Research Ethics, Library, IT and HDR Examinations. You all assisted me and dedicated your time and effort to enable me to continue.

Special sincere heartfelt gratitude to Professor David Audretsch for the inspiration and motivation when I needed it most. I would like to acknowledge Adam Finlay as the editor of my thesis; thank you for the professional editorial services and the support extended.

The path I took as a full-time PhD candidate was paved with love and support by my family and friends. I thank my parents Charlotte Fonseka and Pradeep Polketiyage for making sacrifices and encouraging me to achieve from a BA and MBA through to a PhD. I take pride in hopefully being the first in my generation to hold a PhD. When I received an offer to pursue my doctoral study in Melbourne, my husband sacrificed the successful start-up he passionately built as an entrepreneur in Sri Lanka to become a casual worker in Australia. Thilina Abishek, you always inspire me to dream bigger and work smarter. I am grateful to you for all the support so that I could focus on my studies. I dedicate this PhD thesis to the three of you and without you, the corners of my life would be futile. Finally, I thank Chelsey Taylor for being my friend throughout this journey. In a country where I knew nobody (other than my supervisors), you offered me a shoulder to lean on, shared learnings and became a support pillar.

Personally, this PhD was a mission to disrupt the private higher education industry in Sri Lanka for social impact more than economic benefits. My final words are: life as a PhD candidate is hard and there were times I broke down or I didn't quite understand. Just like a seed can be in the dust, still to sprout and blossom into a flower. At such times, I prayed to God for strength to endure such challenging times. In this moment, when I close the chapter of my doctoral journey, I feel a sense of accomplishment, a sense of recognition and a sense of hope. This is not the end, it's the beginning of a new chapter in my career as an academic and a researcher. Endings, Beginnings.

ABBREVIATIONS

- **E&I** entrepreneurship and innovation
- **GDP** gross domestic product
- **HEEE** higher education entrepreneurial ecosystem
- **SDG** Sustainable Development Goal
- **SME** small-to-medium enterprise
- UK United Kingdom
- US United States of America

CANDIDATURE JOURNEY (MAY 2019 – OCTOBER 2022)

Graduate certificate

Completion of GC-RESIM with High Distinction (Project Management 90%; Innovation and Impact 84%; Qualitative Methods 80% and Quantitative Methods 78%)

Book chapter

Gallage, N., Selvarajah, C., & Laferriere, R. (2022). Ecosystem engagement in entrepreneurship education. In Larios-Hernandez, G., Walmsley, A., & Lopez-Catsro, I. (Eds.), *Theorising Undergraduate Entrepreneurship Education: Pedagogy, Digital Teaching, and Scope*. Palgrave Macmillan.

Conference paper presentation

Gallage, N., Le, V., & Laferriere, R. (2022, February). Higher education entrepreneurial ecosystem for students' entrepreneurial development. [Paper presentation]. Australian Centre for Entrepreneurship Research Exchange (ACERE) Conference, Melbourne, Australia.

Conference participation

2021 – Babson College Entrepreneurship Research Exchange Conference (BCERC)

- 2021 Australian Centre for Entrepreneurship Research Exchange (ACERE) Conference
- 2022 Australian Centre for Entrepreneurship Research Exchange (ACERE) Conference

Research competition

2021 - Finalist from School of Business, Law and Entrepreneurship at the 'Visualise Your Thesis' competition (university level)

Skill development

4-day short course on writing a qualitative PhD facilitated by Australian Consortium for Social and Political Research Inc (ACSPRI), Australia

5-day short course on qualitative data analysis using NVivo facilitated by Australian Consortium for Social and Political Research Inc (ACSPRI), Australia

Research assistant

Assistant to a senior lecturer on multiple research projects from RMIT University

Assistant to a senior lecturer on a research project from Deakin University

Teaching

Sessional tutor at Swinburne University

Sessional lecturer, course coordinator, head tutor and tutor at RMIT University

Head facilitator and course facilitator at RMIT Online

TABLE OF CONTENTS

CHAP	TER 1: INTRODUCTION	1
1.1	Chapter overview	1
1.2	Research background	2
1.3	Practical background	5
1.3.1	Sustainable Development Goals	6
1.3.2	2 National prosperity	6
1.3.3	3 Global start-up economy and ecosystem	7
1.3.4	4 COVID-19 pandemic and entrepreneurship	8
1.4	Research gap and contribution	9
1.4.1	System-level knowledge needs	9
1.4.2	2 Institutional-level knowledge needs	10
1.4.3	3 Individual-level knowledge needs	13
1.5	Research questions and objectives	14
1.5.1	Research development framework	16
1.6	Methodology	17
1.7	Scope and limitations	18
1.7.1	1 Theoretical concept	18
1.7.2	2 Geographic context	19
1.8	Structure of dissertation	20
1.9	Chapter summary	21
CHAP	TER 2: LITERATURE REVIEW	23
2.1	Chapter overview	23
2.2	System view of HEEEs	24
2.2.1	Sub-ecosystem of entrepreneurial ecosystem	24
2.2.2	2 Entrepreneurial initiative of universities	27
2.2.3	3 Co-creation of HEEEs	28
2.3	Higher education entrepreneurial ecosystems	30
2.3.	Emergence and definitions	30
2.3.2	2 Growth of HEEE scholarly work	32
2.3.3	3 Geographic spread of HEEE scholarly work	32
2.3.4	Framing of HEEE scholarly work	34
2.3.5	5 Nexus of entrepreneurship education and entrepreneurial support	36
2.3.0	5 Sub-ecosystems of HEEEs	37

2.4	Continuum of HEEEs	. 40
2.4.	HEEEs for student start-ups	. 41
2.4.2	2 HEEEs for students' entrepreneurial psychology	. 41
2.4.	3 HEEEs for student entrepreneurial development	. 43
2.5	Composition of HEEEs	. 45
2.5.	Factors of higher education entrepreneurial ecosystems	. 46
2.5.2	2 Actors of higher education entrepreneurial ecosystems	. 48
2.6	Stakeholders in HEEEs	. 50
2.6.	Relationship building with ecosystem stakeholders	. 50
2.6.2	2 Stakeholder engagement in HEEEs	. 51
2.7	Chapter summary	. 52
CHAF	TER 3: CONTEXTUALISATION – SRI LANKA	. 54
3.1	Chapter overview	. 54
3.2	Macro-environment outlook of Sri Lanka	. 55
3.2.	History and development	. 55
3.2.2	2 People, culture and communication	. 55
3.2.2	3 Economic growth	. 56
3.2.4	School education system	. 58
3.2.	5 Higher education structure and sector	. 59
3.3	Sri Lanka's progress in sustainable development	. 60
3.4	Importance of quality education and entrepreneurship for youth	. 61
3.5	Challenges faced by the Sri Lankan higher education sector	. 62
3.5.	Lack of quality and relevance	. 62
3.5.2	2 Capacity constraints in higher education	. 63
3.5.	B Demand for private higher education	. 63
3.5.4	4 Government policy interventions	. 64
3.5.	5 Human capital development	. 65
3.6	Entrepreneurship education in Sri Lanka	. 65
3.7	Entrepreneurial ecosystem in Sri Lanka	. 66
3.8	Entrepreneurs' profile and perceived key success factors	. 68
3.8.	Characteristics of start-up entrepreneurs	. 68
3.8.2	2 Reasons for start-up success and failure	. 69
3.9	Chapter summary	. 71
CHAF	TER 4: THEORETICAL FRAMEWORK	. 72
4.1 Ch	apter overview	. 72

4.2	Use of theory in HEEE studies	
4.3	Systems theory	73
4.4	Student involvement theory	75
4.5	Social capital theory	79
4.6	Effectuation theory	81
4.7	Theoretical framework	85
4.8	Chapter summary	
CHA	PTER 5: RESEARCH DESIGN	
5.1 Cł	napter overview	
5.2 M	ethodological congruence	
5.3	Theoretical orientation and interpretative framework	89
5.4	Research approach and methodology	
5.5	Role of the researcher	
5.6	Sample selection	
5.6.	1 Population and participants	
5.6.	2 Sample and sampling	
5.7	Data collection	101
5.7.	1 Research instrument	101
5.7.	2 Participant recruitment	103
5.7.	3 Interview procedure	105
5.8	Data analysis and representation	106
5.8.	1 Data analysis plan	106
5.8.	2 Data analysis process	107
5.8.	3 Data management	108
5.8.	4 Memoing	109
5.8.	5 Codes and themes	110
5.8.	6 Researcher's interpretations	110
5.8.	7 Data representation	111
5.9	Validation and reliability	111
5.10	Research ethics	113
5.10	0.1 Ethical considerations prior to study	113
5.10	0.2 Ethical considerations in preparing to investigate	114
5.10	0.3 Ethical considerations in collecting data	114
5.10	0.4 Ethical considerations in analysing data	115
5.10	0.5 Ethical considerations in reporting data	115

5.10	0.6 Ethical considerations in publishing from the research	115
CHAF	PTER 6: FINDINGS AND ANALYSIS FOR RESEARCH QUESTION 1	117
6.1	Chapter overview	117
6.2	Analysis of diverse stakeholder perspectives on continued development	118
6.3	Continued development of HEEEs	120
6.3.	1 Entrepreneurial mindset	122
6.3.2	2 Entrepreneurial intention	125
6.3.2	3 Entrepreneurship and innovation capabilities	129
6.4	Key roles and contributions of HEEEs	134
6.4.	1 Typology of HEEEs' continued development	134
6.4.2	2 Bridging to the wider entrepreneurial ecosystem	138
6.4.	3 Higher education institutions as catalysts	140
6.5	Chapter summary	142
СНАР	PTER 7: FINDINGS AND ANALYSIS FOR RESEARCH QUESTION 2a	143
7.1	Chapter overview	143
7.2	Analysis of diverse stakeholder perspectives on contextual factors	144
7.3	HEEE factors for students' E&I capabilities in a resource-constrained environmen 147	t
7.3.	1 Entrepreneurial orientation	147
7.3.2	2 Entrepreneurship and innovation education	148
7.3.3	3 Entrepreneurship and innovation research	150
7.3.4	4 Enterprise experiences	153
7.3.5	5 Entrepreneurial networks	155
7.3.0	6 Entrepreneurial support	157
7.4	Perceived importance of HEEE factors	161
7.5	HEEEs through the lens of student involvement theory	164
7.5.	1 Student invests psychological and physical energy	165
7.5.2	2 Student is involved in activities continuously	166
7.5.3	3 Student involvement can be quantitative and qualitative	167
7.5.4	4 Student development is proportionate to student involvement	167
7.5.5	5 Student involvement relates to the effectiveness of activities	167
7.6	Chapter summary	169
CHAF	PTER 8: FINDINGS AND ANALYSIS FOR RESEARCH QUESTION 2b	170
8.1	Chapter overview	170
8.2	Operationalising HEEEs for students' E&I capabilities	171

8.3	Entrepreneurial leadership, entrepreneurial culture and entrepreneurship chair(s) 171
8.3.	1 Entrepreneurial leadership promoting students' E&I capabilities
8.3.2	2 Entrepreneurial culture empowering entrepreneurship within the higher education institution
8.3.	3 Entrepreneurial chairs as change agents 174
8.4	Entrepreneurship curriculum, practical pedagogy and real-world assessments 175
8.4.	1 Entrepreneurship and innovation education for all students 175
8.4.	2 Practical-oriented pedagogy for delivering entrepreneurship and innovation education
8.4.	3 Real-world assessments
8.5	Research-based coursework, research conferences and research-based projects 179
8.5.	1 Research-based coursework
8.5.	2 Research conferences exposing students to scientific knowledge
8.5.	3 Research for start-ups
8.6	Entrepreneurship competitions, internship with start-ups and start-up projects 183
8.6.	1 Entrepreneurship competitions at institution, inter-institutional or national level 183
8.6.2	2 Internship opportunities with start-ups in a range of different industries
8.6.	3 Start-up projects through the institution or external organisations
8.7	Peer engagement, the alumni entrepreneur network and access to external stakeholders
8.7.	Peer engagement at institution and inter-institutional levels
8.7.2	2 Alumni entrepreneur network
8.7.	Access to external stakeholders
8.8	Mentoring, incubators and accelerators
8.8.	1 Mentoring with practitioners including start-up entrepreneurs 192
8.8.2	2 Fully-fledged incubator, internally or with external support 193
8.8.	3 Accelerator programs available privately or nationally 194
8.9	Chapter summary
CHAF	PTER 9: FINDINGS AND ANALYSIS FOR RESEARCH QUESTION 3 198
9.1	Chapter overview
9.2	Analysis of diverse stakeholder perspectives on stakeholder engagement 198
9.3	Social network of stakeholders in HEEEs 201
9.3.	1 Institutional management 201
9.3.2	2 Educators
9.3.	3 Staff representatives
0.2	4 Students 202

9.3.5	Other higher education institutions	. 203
9.3.6	Entrepreneurs	. 203
9.3.7	Angel networks	. 204
9.3.8	Start-up support service providers	. 204
9.3.9	Private companies	. 205
9.3.10	Financial institutions	. 205
9.3.11	Not-for-profit organisations	. 205
9.3.12	Parents	. 206
9.3.13	Mapping social networks in HEEEs	. 207
9.4 S	ocial capital dimensions for entrepreneurial orientation	. 208
9.4.1	Structural social capital for entrepreneurial orientation	. 208
9.4.2	Cognitive social capital for entrepreneurial orientation	. 209
9.4.3	Relational social capital for entrepreneurial orientation	. 210
9.5 S	ocial capital dimensions for entrepreneurship and innovation education	. 211
9.5.1	Structural social capital for entrepreneurship and innovation education	. 212
9.5.2	Cognitive social capital for entrepreneurship and innovation education	. 212
9.5.3	Relational social capital for entrepreneurship and innovation education	. 213
9.6 S	ocial capital dimensions for entrepreneurship and innovation research	. 215
9.6.1	Structural social capital for entrepreneurship and innovation research	. 215
9.6.2	Cognitive social capital for entrepreneurship and innovation research	. 215
9.6.3	Relational social capital for entrepreneurship and innovation research	. 216
9.7 S	ocial capital dimensions for enterprise experiences	. 217
9.7.1	Structural social capital for enterprise experiences	. 217
9.7.2	Cognitive social capital for enterprise experiences	. 218
9.7.3	Relational social capital for enterprise experiences	. 219
9.8 S	ocial capital dimensions for entrepreneurial networks	. 220
9.8.1	Structural social capital for entrepreneurial networks	. 220
9.8.2	Cognitive social capital for entrepreneurial networks	. 221
9.8.3	Relational social capital for entrepreneurial networks	. 222
9.9 S	ocial capital dimensions for entrepreneurial support	. 223
9.9.1	Structural social capital for entrepreneurial support	. 223
9.9.2	Cognitive social capital for entrepreneurial support	. 224
9.9.3	Relational social capital for entrepreneurial support	. 225
9.10 S	takeholder engagement in HEEEs	. 226
9.10.1	Structural configuration of HEEEs	. 226

9.10	0.2 Shared value within HEEEs	. 228
9.10	0.3 Relationships within HEEEs	. 230
9.11	Chapter summary	. 230
CHAI	PTER 10: DISCUSSION AND CONCLUSION	. 232
10.1	Chapter overview	. 232
10.2	Findings for each research question	. 232
10.2	2.1 Beyond student start-ups, entrepreneurial mindset and intention	. 233
10.2	2.2 Beyond the common HEEE factors	. 240
10.2	2.3 Beyond factors of higher education entrepreneurial ecosystems	. 244
10.2	2.4 Beyond the composition of HEEEs	. 250
10.3	Contributions and implications	. 255
10.3	3.1 Contributions to theory	. 255
10.3	3.2 Implications for practice	. 257
10.4	Limitations of the study	. 260
10.5	Future research directions	. 262
10.6	Chapter summary	. 265
REFE	CRENCES	. 266
REFE APPE	CRENCES	. 266
REFE APPE Appen	CRENCES CNDIXES ndix A: Research ethics	. 266 . 322 . 322
REFE APPE Appen A1	CRENCES CNDIXES ndix A: Research ethics Ethics approval email	. 266 . 322 . 322 . 322
REFE APPE Appen A1 A2	ERENCES NDIXES ndix A: Research ethics Ethics approval email Ethics approval certificate	. 266 . 322 . 322 . 322 . 323
REFE APPE Appen A1 A2 A3	CRENCES NDIXES Indix A: Research ethics Ethics approval email Ethics approval certificate Change of chief investigator	. 266 . 322 . 322 . 322 . 323 . 324
REFE APPE Appen A1 A2 A3 A4	CRENCES NDIXES Indix A: Research ethics Ethics approval email Ethics approval certificate Change of chief investigator Final ethics report acceptance email	. 266 . 322 . 322 . 322 . 323 . 324 . 325
REFE APPE Appen A1 A2 A3 A4 Appen	CRENCES NDIXES	. 266 . 322 . 322 . 322 . 323 . 323 . 324 . 325 . 326
REFE APPE Appen A1 A2 A3 A4 Appen B1	CRENCES NDIXES	. 266 . 322 . 322 . 322 . 323 . 323 . 324 . 325 . 326 . 326
REFE APPE Appen A1 A2 A3 A3 A4 Appen B1 B2	CRENCES CNDIXES Indix A: Research ethics Ethics approval email Ethics approval certificate Change of chief investigator Change of chief investigator Final ethics report acceptance email Indix B: Qualitative data collection Interview protocol for internal stakeholders Interview protocol for external stakeholders	. 266 . 322 . 322 . 322 . 323 . 324 . 324 . 325 . 326 . 326 . 327
REFE APPE Appen A1 A2 A3 A3 A4 Appen B1 B2 B3	CRENCES CNDIXES andix A: Research ethics Ethics approval email Ethics approval certificate Change of chief investigator Final ethics report acceptance email andix B: Qualitative data collection Interview protocol for internal stakeholders Participant information statement	. 266 . 322 . 322 . 322 . 323 . 324 . 324 . 325 . 326 . 326 . 327 . 328
REFE APPE Appen A1 A2 A3 A4 A4 B1 B2 B3 B3 B4	CRENCES CNDIXES indix A: Research ethics Ethics approval email. Ethics approval certificate Change of chief investigator. Final ethics report acceptance email. indix B: Qualitative data collection Interview protocol for internal stakeholders. Interview protocol for external stakeholders Participant information statement Consent instrument for authorising employees	. 266 . 322 . 322 . 322 . 323 . 324 . 325 . 326 . 326 . 327 . 328 . 333
REFE APPE Appen A1 A2 A3 A4 A4 B1 B2 B3 B3 B4 B5	CRENCES	. 266 . 322 . 322 . 322 . 323 . 324 . 325 . 326 . 326 . 327 . 328 . 333 . 334
REFE APPE Appen A1 A2 A3 A4 A4 B1 B2 B3 B3 B4 B5 B6	CRENCES CNDIXES Indix A: Research ethics. Ethics approval email. Ethics approval certificate. Change of chief investigator. Final ethics report acceptance email. Indix B: Qualitative data collection . Interview protocol for internal stakeholders. Interview protocol for external stakeholders. Participant information statement. Consent instrument for authorising employees Consent instrument for individual interviewees Email invitation.	. 266 . 322 . 322 . 322 . 323 . 323 . 324 . 325 . 326 . 326 . 327 . 328 . 323 . 334 . 335
REFE APPE Appen A1 A2 A3 A4 Appen B1 B2 B3 B3 B4 B5 B6 Appen	CRENCES CNDIXES adix A: Research ethics. Ethics approval email. Ethics approval certificate. Change of chief investigator. Final ethics report acceptance email. adix B: Qualitative data collection Interview protocol for internal stakeholders. Interview protocol for external stakeholders. Participant information statement. Consent instrument for authorising employees Consent instrument for individual interviewees Email invitation. adix C: Graduate Certificate of Research and Innovation Management	. 266 . 322 . 322 . 322 . 323 . 324 . 325 . 326 . 326 . 327 . 328 . 328 . 333 . 334 . 335 . 336
REFE APPE Appen A1 A2 A3 A4 Appen B1 B2 B3 B3 B4 B5 B6 Appen Appen	CRENCES CNDIXES andix A: Research ethics Ethics approval email Ethics approval certificate Change of chief investigator Final ethics report acceptance email ndix B: Qualitative data collection Interview protocol for internal stakeholders Interview protocol for external stakeholders Participant information statement Consent instrument for authorising employees Email invitation email invitation email invitation email invitation	. 266 . 322 . 322 . 322 . 323 . 324 . 325 . 326 . 326 . 327 . 328 . 333 . 334 . 335 . 336 . 337

LIST OF FIGURES

Figure 1.1 – Chapter 1 outline	1
Figure 1.2 – Research development framework	17
Figure 1.3 – Key interest of research	
Figure 2.1 – Chapter 2 outline	
Figure 2.2 – Countries studied in HEEE literature	
Figure 2.3 – Continents represented by previous HEEE studies	
Figure 2.4 – Participants investigated in previous HEEE studies	
Figure 2.5 – Sub-ecosystems of HEEEs	
Figure 2.6 – Missing link in HEEE studies through the lens of start-ups process	
Figure 2.7 – Domains of the entrepreneurial ecosystem	
Figure 3.1 – Chapter 3 outline	
Figure 3.2 – Astin's student involvement theory and framing to this study	
Figure 3.3 – Framing systems theory to this research	
Figure 3.4 – Framing social capital theory to this study	
Figure 3.5 – Framing effectuation theory to this research	
Figure 3.6 – Theoretical framework	
Figure 4.1 – Chapter 4 outline	54
Figure 4.2 – Population growth in Sri Lanka	
Figure 4.3 – Economic growth (GDP current USD) in Sri Lanka	
Figure 4.4 – Comparison of economic growth (GDP current USD) in Asia	57
Figure 4.5 – Comparison of economic growth (GDP current USD) in South Asia	58
Figure 4.6 – Characteristics of start-up entrepreneurs	69
Figure 5.1 – Chapter 5 outline	
Figure 5.2 – Data analysis spiral	108
Figure 6.1 – Chapter 6 outline	117
Figure 6.2 – Theme structure for continued development of HEEEs	119
Figure 7.1 – Chapter 7 outline	143
Figure 7.2 – Theme structure for research question 2	146
Figure 7.3 – Importance of contextual factors as perceived by diverse stakeholders	161
Figure 7.4 – HEEE adapted to student involvement theory	165
Figure 8.1 – Chapter 8 outline	170
Figure 9.1 – Chapter 9 outline	198
Figure 9.2 – Theme structure for research question 3	200
Figure 10.1 – Chapter 10 outline	232

LIST OF TABLES

Table 1.1 – Overview of research aim, questions and objectives	16
Table 2.1 – Definitions of entrepreneurial ecosystems among scholarly work	
Table 2.2 – Key entrepreneurial ecosystems theoretical frameworks	
Table 2.3 – Definitions of HEEEs among scholarly work	
Table 2.4 – Scholarly work related to HEEEs	
Table 2.5 – HEEEs focusing on student start-ups	
Table 2.6 – HEEEs focusing on students' psychology	
Table 4.1 – Summary of entrepreneurship related indicators	
Table 5.1 – Interpretative framework with associated philosophical beliefs	
Table 5.2 – Research design of prior HEEE studies	
Table 5.3 – Considerations of phenomenological approach	
Table 5.4 – Participants of this research	
Table 5.5 – Participants of prior HEEE studies	
Table 5.6 – Sample of private higher education institutions	100
Table 5.7 – Design of interview protocol	103
Table 5.8 – Participant representation for this study	104
Table 5.9 – Participation of private higher education institutions	105
Table 5.10 – Concepts and attributes for coding	107
Table 5.11 – Case naming system for data management	109
Table 6.1 – Distributions of participants	118
Table 6.2 – Illustrative quotes about the current stage of HEEEs in Sri Lanka	120
Table 6.3 – Reorder matrix of HEEE's continued development	121
Table 6.4 – Key roles and contributions of HEEEs	135
Table 7.1 – Descriptions of initial codes	145
Table 7.2 – Descriptions of aggregated themes	
Table 7.3 – Excerpts on entrepreneurial orientation	148
Table 7.4 – Excerpts on entrepreneurship and innovation education	150
Table 7.5 – Excerpts on entrepreneurship and innovation research	152
Table 7.6 – Excerpts on enterprising activities	154
Table 7.7 – Excerpts on entrepreneurial networks	156
Table 7.8 – Excerpts on entrepreneurial support	159
Table 7.9 – Mapping HEEE factors of existing studies to this research	160
Table 9.1 – Mapping social networks against the HEEE contextual factors	207
Table 9.2 – Social capital dimensions for entrepreneurial orientation	211

Table 9.3 – Social capital dimensions for entrepreneurship and innovation education 2	214
Table 9.4 – Social capital dimensions for entrepreneurship and innovation research	217
Table 9.5 – Social capital dimensions for enterprise experiences	20
Table 9.6 – Social capital dimensions for entrepreneurial networks	23
Table 9.7 – Social capital dimensions for entrepreneurial support	26

CHAPTER 1: INTRODUCTION

1.1 Chapter overview

This PhD study explores the concept of 'higher education entrepreneurial ecosystems' (HEEEs, also known as university-based entrepreneurship ecosystems). The research context is on a South Asian country (i.e. Sri Lanka) given the increasing importance of HEEEs in developing countries with resource-constrained environments. This chapter introduces HEEEs to establish the significance of the topic for the world and research area. The following details the research background complemented by the practical background, identifies the research gaps in the form of knowledge needs, establishes the research objectives, describes the research design followed by the scope and limitations, and outlines the dissertation.

This introduction chapter is organised according to 'Creating A Research Space' (CARS) model by Swales and Najjar (1987), as presented below in Figure 1.1. This structure of the introductory chapter provides a clear pattern, including the situation, the problem and the solution.



Figure 1.1 – Chapter 1 outline

1.2 Research background

The conjunction of entrepreneurship and economic development emphasises the third mission, of teaching, research and entrepreneurship, among higher education institutions (Etzkowitz & Leydesdorff 2020). Higher education institutions have entered a state of transformation due to the increasing expectations of contributions towards entrepreneurship and innovation leading to economic development (Guerrero et al. 2016). More recently, higher education institutions have faced increasing demands for preparing students for future work, particularly following the current COVID-19 pandemic (Bock et al. 2020; Liguori & Winkler 2020; Maritz et al. 2020; Ratten & Jones 2020). Educating the next generation is acknowledged as a natural role of higher education institutions, and skilled graduates are critical outcomes (Wennberg, Wiklund & Wright 2011).

Etzkowitz (1983) defined entrepreneurial universities as higher education institutions considering the possibilities of entrepreneurship and innovation. The emergence of higher education institutions assuming their third mission duties was prevalent in the United States (US), with Massachusetts Institute of Technology, Stanford University and Babson College. This trend spread beyond the US and Europe to other developed and developing countries, including the United Kingdom (UK), Canada (Bramwell & Wolfe 2008), Australia, China (Zhou & Peng 2008), Singapore (Wong, Ho & Singh 2007), Malaysia (Ahmad et al. 2018), Chile (Bernasconi 2005) and Brazil (Amaral, Ferreira & Teodoro 2011). Many countries have now reformed their higher education systems to become catalysts for economic and social development (Gibb & Hannon 2006). Knowledge societies demand the production of human capital to pursue entrepreneurship, drive innovation, increase competitiveness and influence sustainable development (Guerrero, Cummingham & Urbano 2015). Ultimately, in the context of entrepreneurship, higher education institutions hold the purpose of ensuring their students thrive in their endeavours (Audretsch 2014). Governments and scholars argue that higher education institutions must become entrepreneurial for this responsibility and more institutions have embarked on this transformational journey (Clark 2004; Guerrero & Urbano 2014).

In becoming entrepreneurial, higher education institutions are undertaking diverse entrepreneurial pathways, with strategic choices to demonstrate their commitment and involvement in entrepreneurship initiatives within the institution (Klofsten et al. 2019). One of the core entrepreneurial pathways is higher education entrepreneurial ecosystems (HEEEs, also known as university-based entrepreneurship ecosystems). Building on the ecosystem perspective, HEEEs have gained prominence since the early 2000s (Fetters, Greene & Rice 2010). As a contemporary concept, the theoretical development of HEEEs is in the early stages, and the concept of an HEEE is novel and emerging (Hsieh & Kelley 2020; Longva 2021). One of the earliest references to HEEEs is found in the scholarly work by Fetters, Greene and Rice (2010) that discusses pathways to developing HEEEs with wide-ranging missions and resources in different contexts. The same notable research included a series of case studies of six higher education institutions, leading to seven key success factors that support entrepreneurship through entrepreneurship education and entrepreneurial support (Rice, Fetters & Greene 2014). Since then, an increasing number of studies have examined the composition of HEEEs, categorising the factors that form a comprehensive and highly evolved HEEE with contextual relevance (Stam & Spigel 2016). In regard to HEEE literature, there is an increasing critique that studies are producing lists of 'what' an HEEE is rather than explanations of 'how' such HEEEs work and this view can be improved (Longva 2021).

The economist Joseph Schumpeter advocated creative destruction associated with new outcomes and new ways of doing things (Schumpeter 1934). One of the observed limitations of the extant literature is that HEEEs have been largely examined from a static perspective. HEEEs are generally known for generating student start-ups, following the trend in entrepreneurial ecosystems. In 2020, the global start-up economy generated USD3 trillion in value (Startup Genome 2021). While many factors have been garnered about the composition of HEEEs, little is known about the future – the continued development. Scholarly work on HEEEs position them as self-standing ecosystems of a respective higher education institution (Lahikainen et al. 2019). However, higher education institutions are identified as a domain of the entrepreneurial ecosystem (Isenberg 2010) and HEEEs should be considered a subecosystem of the wider entrepreneurial ecosystem, engaging with various stakeholders (Wurth, Stam & Spigel 2021). In this study, a key focus is to explore how HEEEs can develop and evolve considering a broader perspective from diverse stakeholders.

Unlike an ecological system that organically develops, HEEEs are co-created and managed by higher education institutions. Higher education institutions are understood to fulfill their core teaching and research activities while providing a conducive environment for students to explore and exploit ideas for the economic and social development of cities, regions or countries (Klofsten et al. 2019). A potential student entrepreneur requires education and support that develops and reinforces their entrepreneurial capabilities (Rodríguez-López & Souto 2020). The concept of capabilities represents skills, expertise, acumen and knowledge, terms interchangeably used in entrepreneurship literature (Liu, Kulturel-Konak & Konak

2021). Entrepreneurial capabilities involve the ability of an entrepreneur to start and grow a new venture using a combination of resources (Gumsay & Bohne 2018). These entrepreneurial capabilities include the 'know-what' (declarative) and 'know-how' (procedural) along with an understanding of 'why' for each potential entrepreneur (Hagg 2017; Willians-Middleton & Donnellon 2014). However, the development of entrepreneurial capabilities is rarely discussed in the HEEE literature and emerges from this study. Extant studies have examined HEEEs primarily for student start-ups (Allahar & Sookram 2019; Meyer et al. 2020; Miller & Acs 2017; Shil et al. 2020; Wright, Siegel & Mustar 2017) and more recently in preparing students for entrepreneurship through entrepreneurial mindset and intention (Guerrero, Urbano & Gajón 2020; Secundo et al. 2020; Webber, Kitagawa & Plumridge 2020). While it is evident that students are central to HEEEs (Miller & Acs 2017), it is unclear how HEEEs stimulate students' entrepreneurial development.

Unlike in developed countries, co-creating HEEEs in a resource-constrained environment is inherently challenging (Bedő, Erdős & Pittaway 2020). Extending the context to Sri Lanka, it has been shown that, within South Asia, Sri Lanka has one of the highest youth unemployment rates, and its educated youth are disengaged from employment (International Labour Organisation 2020). According to the World Bank (2020) Sri Lanka is a lower-middle income country with a GDP per capita of USD3,852 in 2019, which is similar to other developing countries in South Asia, including India, Pakistan, Bangladesh and Bhutan. Within Asia, Sri Lanka is an emerging economy like Cambodia, Indonesia, Myanmar, the Philippines and Vietnam. Previous studies, such as Kodithuwakku and Rosa (2002), Lin et al. (2013) and de Silva, Uyarra and Oakey (2012), have explored entrepreneurship while characterising Sri Lanka as a resource-constrained environment. This study focuses on exploring the development of HEEEs within a resource-constrained environment.

When designing a research study on HEEEs, the perceptions and experiences of social actors within and related to them are paramount to understanding the concept. Higher education institutions are not just responsible for their missions, but also to engage with stakeholders outside the institution promoting commercial and non-commercial knowledge (Huang-Saad, Duval-Couetil & Park 2018). All stakeholders should be involved in HEEEs (Rice, Fetters & Greene 2014). However, in HEEE literature, the stakeholders who shape and operationalise the environment are rarely heard from as a whole. Most studies have investigated HEEEs among either internal or external stakeholders (Longva 2021; Webber, Kitagawa & Plumridge 2020), with a few early exceptions investigating HEEEs covering both internal and external

stakeholders (Rice, Fetters & Greene 2014; Wright, Siegel & Mustar 2017). In instances when both stakeholder categories were investigated, such studies were based in developed countries, including the US. While diverse stakeholder perspectives are limited in HEEE literature, this research garners broader perspectives, drawing on participation from stakeholders representing the HEEE and the wider entrepreneurial ecosystem. This research explores and interprets multiple views from internal and external stakeholders, including internally, deans/heads of schools, academics/educators, and externally, alumni entrepreneurs, established entrepreneurs, angel investors and support professionals in Sri Lanka. More details on methodology are detailed in Section 1.6.

Understanding the development of HEEEs can improve by researching it as a subecosystem of the wider entrepreneurial ecosystem in a resource-constrained environment, considering broad perspectives from diverse stakeholders. This study intends to add to the current understanding and knowledge of HEEEs by investigating (1) at a systems level to explore their continued development; (2) at an institutional level to determine the contextual factors and mechanisms relevant for developing students in a resource-constrained environment; and (3) at an individual level to discover stakeholder engagement influences on students' entrepreneurial development.

1.3 Practical background

"While having the largest youth labour force in the world, more than half of South Asian youth are not on track to have the education and skills necessary for employment in 2030."

Justin Van Fleet, Executive Director of the Global Business Coalition for Education, shared at the Global Business Coalition for Education forum in 2020 (UNICEF 2020).

The above quote exemplifies the challenge that the education sectors of South Asian countries are currently facing. Justin Van Fleet, Executive Director of the Global Business Coalition for Education, supports the view that the next generation of youth lack self-employment and entrepreneurship-related skills among the skills needed for 21st-century work (UNICEF 2020). Industry research conducted by McKinsey & Company (2020) found that entrepreneurship is a future work skill in the changing world by surveying 18,000 individuals in 15 countries. Under the broad category of leadership, entrepreneurship was an essential skill required by individuals, influencing their career, income and satisfaction. Thus, McKinsey & Company

(2020) suggest that higher education institutions should reform their education systems to focus more strongly on relevant skills including entrepreneurship and ensure lifelong education.

1.3.1 Sustainable Development Goals

In recent years under the United Nations Sustainable Development Goals (SDGs) commitment, 189 countries have strived to achieve 17 goals for sustainable development (United Nations 2020). The 2030 Agenda for Sustainable Development encourages member states to meet these goals, including SDG8 to improve economic growth through youth, job creation, employment, entrepreneurship, and small-to-medium enterprises (SMEs) (United Nations 2020). Therefore, there is considerable enthusiasm for entrepreneurship which is embedded within the national agenda of many countries as a policy and strategy to overcome economic challenges and achieve growth (Cander et al. 2020).

South Asia falls behind other regions, with over 50% of South Asian youth not receiving the education and skills necessary for employment in 2030, meaning these countries are not developing the next generation of youth with the skills essential for the 21st century (UNICEF 2020). This dilemma brings attention to SDG4 aimed at quality education and SDG8 directed at decent work and economic growth. Focusing on quality education, Target 4.4 intends "to substantially increase the number of youth and adults who have the relevant skills for employment, decent jobs and entrepreneurship" (United Nations 2020). At the same time, Target 8.3 promotes "development-oriented policies that support productive activities, decent job creation, entrepreneurship, creativity and innovation" and Target 8.6 strives to "substantially reduce the proportion of youth not in employment, education or training" (United Nations 2020). With less than a decade to achieve these goals and targets by 2030, there is a strong imperative for countries to reap benefits from their youth by investing in the development of their skills, talent and capabilities (UNICEF 2020). This urgency shows that enabling youth, specifically students with relevant capabilities, to pursue entrepreneurial career paths is contemporary, and countries can benefit from research on developing their youth. To a large extent, entrepreneurial capabilities are applied across multiple job roles including selfemployment, hybrid entrepreneurship and intrapreneurship (Alsos et al. 2022). While SDG4 and SDG8 have a reciprocal impact, this research pays attention to quality education.

1.3.2 National prosperity

Countries do not inherit national prosperity: it is created (Acs et al. 2016). In the rush to promote decent work, economic growth and innovation development, scholars and

governments have embraced entrepreneurship (Audretsch et al. 2021) and there is an increasing interest in entrepreneurship by governments at all levels worldwide (Minniti & Lévesque 2008). However, most seminal work on entrepreneurship has been largely associated with developed countries, and the returns of entrepreneurship in developing countries are yet to draw attention (Sautet 2013).

Globally, entrepreneurship is perceived as an engine of economic growth and societal development (Alvarez et al. 2014). At its core, entrepreneurship is recognised as valuable for sustainable development with the opportunity for a country's economic growth and social development leading to national prosperity (Hall et al. 2010). Notable researchers have argued a range of economic and social benefits generated from entrepreneurship, ranging from economic development (Carlsson 2013; Acs et al. 2008; Wennekers & Thurik 1999; Schumpeter 1934); innovation (Wong et al. 2005; Acs & Audretsch 1988; Schumpeter 1934); job creation (NBIA 2014; Parker 2009; Gibb 1996); productivity (van Praag & Versloot 2007); technology transfer and knowledge spillovers (Terjesen & Wang 2013; Acs et al. 2012; Grimaldi et al. 2011); venture creation (Ronstadt 1985); sustainable development (Hall et al. 2010; Brugmann & Prahalad 2007) and community development (Espinoza et al. 2019). Although some studies have confirmed the impact of entrepreneurship on economic performance to be strong and positive (Zoltan et al. 2008; van Praag & Versloot 2007; Wong et al. 2005; Wennekers & Thurik 1999), this correlation continues to be under investigation for further insights (Urbano et al. 2019; Du & O'Connor 2018).

1.3.3 Global start-up economy and ecosystem

In 2020, the global start-up economy was valued nearly at USD3 trillion (Startup Genome 2021). This global start-up economy is not limited to the developed world and in fact is expanding in developing countries, including Sri Lanka (Startup Genome 2021). Despite the challenges faced by the COVID-19 pandemic, the global start-up economy remains large. According to the Global Entrepreneurship Monitor (2021) two shockwaves caused by the pandemic are impacting start-up ecosystems globally: (1) a capital crunch for funding start-ups across the world; and (2) a demand drop in revenue like most operating companies. While the effect of these shockwaves is severe, Startup Genome (2021) claims that post crisis, top start-up ecosystems will continue to perform well and emerging ecosystems need to invest now to arrive at that progress. Post COVID-19 recovery is expected to result in new ecosystems rising. According to the policy advisory of Startup Genome (2021), governments need to take the lead and support start-ups similar to traditional sectors and small businesses. As the start-up

ecosystem grows, the entrepreneurial ecosystem including higher education needs to further develop producing more value (Startup Genome 2021).

Since 2017, Silicon Valley has held the number one position for global start-up ecosystems among 30 ecosystems around the world (Startup Genome 2021). In 2020, Silicon Valley was followed by New York and London in a tied second position. These start-up ecosystems have depth of talent, experience and capital, creating a conducive environment for start-ups. In terms of continent representation within the top global ecosystems, North America is leading followed by Europe and then Asia-Pacific. There is a rise of ecosystems in the Asia-Pacific region, from 20% in 2012 to 30% in 2020 of top ecosystems on the global stand (Startup Genome 2021). The Global Entrepreneurship Monitor (2018) found that among nine conditions that influence entrepreneurs and their businesses in the ecosystem, Asia-Pacific held the highest place in physical infrastructure and lowest in entrepreneurship education. Sri Lanka is recognised among the top 20 global ecosystems for 'bang for buck' and having great potential for its talent (Startup Genome 2021). Among Asia-Pacific countries, Sri Lanka has not yet made it in the top 30 global ecosystems or challenger ecosystems ranked by Startup Genome (2021). However, Sri Lanka ranks second place for its affordable talent while the top three global ecosystems, Silicon Valley, New York and London, ranked 10th position for their talent.

1.3.4 COVID-19 pandemic and entrepreneurship

The COVID-19 pandemic that struck in early 2020 had a catastrophic impact on global societies, businesses and economies. Unprecedented interruptions were caused to people, resources and capital, disrupting the higher education landscape (Liguori & Winkler 2020). Since the pandemic, higher education institutions have faced severe challenges, having lost billions in revenue, and need to re-think their teaching, research and entrepreneurship for sustainable development (Maritz et al. 2020). Emerging scholarly work advocates the value of entrepreneurship and entrepreneurship education as countries strive towards economic recovery (Giones et al. 2020; Liguori & Winkler 2020; Ratten & Jones 2020). Developing entrepreneurship, and governments are implementing policies that support economic and societal development (Álvarez et al. 2014; Minniti & Lévesque 2008). It is ever more important now given the economic recovery after the COVID-19 crisis through entrepreneurship (Ratten 2020; Espinoza et al. 2019). When the pandemic struck, there was a tendency for necessity entrepreneurship and starting a business was identified as the best alternative when jobs are scarce (Global Entrepreneurship Monitor 2021).

1.4 Research gap and contribution

This study belongs to the domain of HEEEs in the field of entrepreneurship. Scholarly work on HEEEs has a close relationship with research on entrepreneurial universities and entrepreneurial ecosystems. There are instances where these domains overlap within the HEEE literature. Research gaps are identified and presented as system-level knowledge needs, institutional-level knowledge needs and individual-level knowledge needs. The expected contribution illustrates the relationship between HEEEs and capability development that should enhance the current discussions in this research field, which might inform higher education institutions on developing their HEEEs and enlighten entrepreneurship scholars on the state of HEEEs in a resource-constrained environment. This study builds on a deep tradition in social science by exploring a link between HEEEs and capability development that remains relatively underrepresented in the area of research.

1.4.1 System-level knowledge needs

Current literature positions HEEEs as a standalone ecosystem (Lahikainen et al. 2019) and is conceived as independent (Autio et al 2014). This ignores the system-ness of HEEEs and disregards one of their main characteristics as an ecosystem. A recent literature review on entrepreneurial ecosystems suggested that future research should explore the complex system of the nature of ecosystems (Wurth, Stam & Spigel 2021). The way forward must not be based on the consideration that HEEEs are self-sufficient ecosystems limited to their institutional environment and isolated from their broader environment. Moreover, Webber, Kitagawa and Plumridge (2020) call for future research to propose a conceptual framework that embeds HEEEs into the wider entrepreneurial ecosystem. HEEEs are emerging to be closely connected and spreading rapidly (Feldman, Siegel and Wright 2019). While respecting the extant literature, this study intends to explore HEEEs with a holistic view.

HEEEs still lack a consistent theoretical foundation and empirical base (Hsieh & Kelley 2020; Longva 2021). The current literature is a collection of studies mostly focused on the composition of HEEEs, specifically the factors that influence student start-ups, entrepreneurial intentions and entrepreneurial mindsets. Emerging research suggests that nested sub-ecosystems link to outputs and outcomes of the entrepreneurial ecosystem (Wurth, Stam & Spigel 2021). In this sense, there is not enough understanding of how HEEEs relate to their wider entrepreneurial ecosystems and whether their outcomes such as student start-ups, entrepreneurial intentions and entrepreneurial mindsets align with the entrepreneurial

ecosystem. A more nuanced understanding of the evolution of HEEEs is needed, taking into consideration that higher education institutions are responsible for a third mission.

The emergence of HEEEs seems to be triggered by the specific outcome of student start-ups (Johnson, Bock & George 2019). In contrast, today's higher education institutions should produce entrepreneurial capital and be catalysts for regional economic and societal development (Audretsch 2014; Guerrero, Cunningham & Urbano 2015). Entrepreneurs and their capabilities are essential to the success of start-ups (Kyndt & Baert 2015). An emerging entrepreneur, such as a student, requires education and support that complements and reinforces their entrepreneurial capabilities by combining different essential ingredients for entrepreneurship (Rodríguez-López & Souto 2020).

With less than 10 years to achieve the SDGs, higher education institutions are striving towards sustainable development. Students represent a proportion of the skilled human capital, and the target is to increase the number of youth with entrepreneurship skills (United Nations 2020). Higher education institutions hold responsibility to critically review their strategies and initiatives within the context of broader stakeholders (Huang-Saad et al. 2018), and since the COVID-19 pandemic, this has become a greater responsibility for higher education institutions, to develop their students for future work (Ratten & Usmanij 2020). In this case, should HEEEs contribute to entrepreneurship beyond students' start-ups, entrepreneurial intentions and mindset? Scholars suggest future research to pay attention to pre-venture capability development to understand the range of capabilities that one should be equipped with to pursue an entrepreneurial career (Gümüsay & Bohné 2018; Rasmussen et al. 2011). In response, more research is needed to investigate how diverse stakeholders anticipate the continued development of HEEEs.

1.4.2 Institutional-level knowledge needs

Within HEEE scholarly work, studies have been largely confined to exploring HEEEs leading to student start-ups, student entrepreneurial intentions and mindset. In this vein, previous studies have explored why some higher education institutions generate more start-ups than others located in developed countries (Di Gregorio & Shane 2003; O'Shea et al. 2008; Shane 2004; Wright et al. 2004; Guerrero et al. 2014; Guerrero & Urbano 2017; Guerrero et al. 2017; Guerrero et al. 2018). Many higher education institutions still focus on delivering entrepreneurship education instead of exercising their wider efforts to focus on student development (Martiz et al. 2020; Birch et al. 2017). A core value proposition for any higher

education institution is to provide quality higher education for their students and universitybased initiatives likely to have long-term sustainability and impact (O'Brien, Cooney & Blenker 2019).

HEEEs are initiatives of higher education institutions in becoming entrepreneurial. According to early literature, higher education institutions have alternative pathways to develop a comprehensive and highly evolved HEEE, as its contextual factors may vary according to its geographic context and institutional characteristics (Rice, Fetters & Greene 2014). Such contextual factors refer to environment-specific factors noted as significant in shaping entrepreneurial activities of HEEEs by Webber, Kitagawa and Plumridge (2020), in contrast to studies focusing on individual and institutional factors. The same study ascertained the importance of contextual factors of higher education institutions shaping students within an HEEE. The challenge is how higher education institutions recognise the suitable factors within their institutional architecture to drive entrepreneurship (Cunningham, Lehmann & Menter 2021). There is a lack of theoretical understanding of HEEE factors, and research has yet to clarify or define these factors (Hsieh & Kelly 2020). In addition to composition, there is little known about how to configure an HEEE to serve its functions (Delanoë-Gueguen & Theodoraki 2021). This calls for research to explore the composition and configuration of HEEEs at an institutional level.

Governments around the world benchmark and attempt to replicate entrepreneurial ecosystems, proven to be successful in other countries, into their development plans and policies (Hruskova & Mason 2020). Such successful environments are impossible to re-create in the context of entrepreneurial ecosystems as successful ecosystems are context-dependent, embracing local conditions and characteristics (Spigel 2016). Therefore, it is understood that entrepreneurship is a social practice, and context is of the essence (Longva 2021; Welter 2011). Context is important for understanding how, why and when entrepreneurship happens as the contextual characteristics and conditions influence entrepreneurs and the way they start and run a new venture (Welter & Smallbone 2011; Welter 2011). Challenges and opportunities vary drastically in different parts of the world, making is impossible to benchmark ecosystems (Volkmann et al. 2021). Differences across the world reflects a more heterogenous student community, which may have varied and complex needs, requiring a tailored ecosystem (O'Brien, Cooney & Blenker 2019). Further, a 'one-size-fits-all' approach that considers higher education institutions as homogeneous is unrealistic, which means there is no

benchmark solution (Degl'Innocenti et al. 2019; Benneworth et al. 2016; Ertuna & Gurel 2011).

HEEEs began in the US and are commonly investigated in developed countries and high-income economies (Kirby 2004). Unlike in developed countries such as the US, UK and Australia, co-creating HEEEs in a resource-constrained environment where factors of a successful HEEE might have gaps is a critical challenge (Bedő, Erdős & Pittaway 2020). For instance, successful HEEE factors of higher education institutions located in Silicon Valley are less favourable for HEEEs in developing countries suffering from lack of resources. Benchmarking HEEE factors found in successful HEEEs of such countries is likely to raise contradictions in resource-scarce environments. As an initial step, Bedő, Erdős and Pittaway (2020) suggested a conceptual framework of an HEEE in a resource-constrained environment. Given the lack of empirical evidence of co-creating HEEEs in a resource-constrained environment, it is important to gain a deeper understanding of their composition and configuration with resource constraints and, in particular, how the contextual factors of HEEEs drive entrepreneurship in such an environment. Taking a context perspective in entrepreneurship research (Audretsch & Belitski 2016), more research is called for to investigate what diverse stakeholders perceive as the contextual factors of HEEEs that could influence students' capability development in a resource-constrained environment.

Although there is emerging literature that contributes to the composition of HEEEs, there is not much research beyond the list of factors such as explanations of 'how' such HEEEs work (Longva 2021). The usefulness of the HEEE depends on an advanced understanding of 'what' the contextual factors are combined with 'how' these factors can be operationalised and 'why' such mechanisms are relevant for a resource-constrained environment. Without such knowledge, higher education institutions are mostly left with HEEE models based in developed countries that cannot be considered to their context due to resource constraints. While an HEEE as a concept itself lacks conceptual clarity (Hsieh & Kelley 2020), more work is required that focuses on the contextual factors, their mechanisms and rationale. In this case, how and why specific contextual factors of HEEEs could influence students' E&I capabilities in a resource-constrained environment. Initially, the focus was on students' entrepreneurial development before E&I capabilities emerged from the first research question.

1.4.3 Individual-level knowledge needs

Although HEEEs, like entrepreneurial ecosystems, include a combination of factors and actors, a majority of the emphasis is on the factors of HEEEs. Ecosystems are a common platform for organisations to create and capture value by interacting with other stakeholders through inclusiveness (Nylund et al. 2021). Despite higher education institutions striving towards contributing to sustainable development for the greater community, the role of stakeholders is understudied (Clauss et al. 2018). A recent study found that a sustainable ecosystem requires important involvement from internal stakeholders in creating collaborative connections (Delanoë-Gueguen & Theodoraki 2020). Future directions call for research to use theories related to stakeholders in HEEE studies influencing students (Guerrero, Urbano & Gajón 2020). Higher education institutions have a responsibility to critical review their HEEE and its strategy within a context of many stakeholders (Huang-Saad, Duval-Couetil & Park 2018).

Higher education institutions and their HEEEs need to consider the wider and different stakeholders involved in being part of the entrepreneurial ecosystem (Ratten & Usmanij 2020). Scholars such as Bischoff (2021) found that stakeholder support and collaboration are crucial for successfully developing entrepreneurial ecosystems. Although higher education institutions creating shared value with various stakeholders have been investigated in different settings, little is known about how stakeholders can support higher education institutions (Cunningham, Lehmann & Menter 2021; Karwowska 2019). Emerging research has found that active participation of stakeholders in entrepreneurship programs is paramount, and ensuring its sustainability and creating a network of stakeholders is fundamental to ensuring student interactions (Galvão et al. 2020). Exploring stakeholders and their engagement could be helpful to understand more than who is or can be included and their relationships within an HEEE (Tejero, Pau & Leon 2019). Sri Lanka does not have adequate institutional frameworks and supportive mechanisms to promote such interactions beyond the institutions (de Silva, Uyarra & Oakey 2012). Future research calls for more investigation to explore how stakeholders can engage in evolving HEEEs as higher education institutions need to work together with other stakeholders (Ratten & Jones 2020; Theodoraki, Messeghem & Rice 2018). In response, this study aims to address the lack of understanding and knowledge about stakeholder engagement in HEEEs.

HEEEs, being relatively nascent and evolving, are understudied, so there is a need for more profound insights into co-creating stronger HEEEs to support students keen to become entrepreneurs (Longva 2021). Different stakeholders of various ecosystems can guide students' intentions and behaviours towards entrepreneurship (Roundy, Bradshaw & Brockman 2018). Entrepreneurship education literature suggests that stakeholder networks need to enable sharing of information, knowledge, experiences and resources for the benefit of students (Galvão et al. 2020). In contrast, current HEEE literature contains little knowledge of the interaction between students and various stakeholders. In particular, there needs to be research on how stakeholders influence students within the HEEE and the value addition for students.

HEEE studies by scholars including Secundo et al. (2020), Guerrero, Urbano and Gajón (2020) and Webber, Kitagawa and Plumridge (2020) have examined how factors prepare students for entrepreneurship by paying attention to individual-level aspects, such as students' entrepreneurial intentions and mindsets. The point of view that the HEEE is a multi-stakeholder environment (Rice, Fetters & Greene 2014) that influences the development of students to pursue entrepreneurship has received considerably less attention among scholarly work. Stakeholders shape the flow of entrepreneurial knowledge and skills obtained by students (Spigel 2017; Rice, Fetters & Greene 2014). Whether it is the home, institution or workplace, for the environment to function effectively as a context for development, it depends on the existence and nature of social interconnections (Bronfenbrenner 1979). More research is needed to explore how diverse stakeholders can engage with HEEEs that could influence students E&I capabilities in a resource-constrained environment.

1.5 Research questions and objectives

This research explores how HEEEs can develop students for entrepreneurship and innovation in a resource-constrained environment. In this case, exploration goes beyond the extant literature that investigated HEEEs for student start-ups, entrepreneurial mindset and intention. This research focuses on students' entrepreneurial development through HEEEs in a resourceconstrained environment. First, the study intends to shed light on the continued development of HEEEs through the multiple views of diverse stakeholders with a deeper understanding of how HEEEs can contribute to their wider entrepreneurial ecosystem. Next, the study aims to provide new knowledge on how higher education institutions can design and evolve an environment (i.e. HEEE) for nurturing students' entrepreneurial development as envisioned by stakeholders. Finally, the study explores the role diverse stakeholders can play within the HEEE, in developing students for entrepreneurship.

Having recognised the above knowledge gaps in Section 1.4, this research shall examine the following research questions:

RQ1: How do diverse stakeholders anticipate the continued development of *HEEEs in a resource-constrained environment?*

RQ2a: What do diverse stakeholders perceive as the contextual factors of HEEEs that could influence students' E&I capabilities in a resource-constrained environment?

RQ2b: How and why do specific contextual factors of HEEEs influence students' E&I capabilities in a resource-constrained environment?

RQ3: How can diverse stakeholders engage in the factors of HEEEs that could influence students' E&I capabilities in a resource-constrained environment?

At the beginning of this study, the focus of the above research questions was on 'student entrepreneurial development' that include entrepreneurial intention, mindset and capabilities. After the data collection and analysis for the first research question, the focus of 'students E&I capabilities' emerged and this led to the revision of research questions including RQ2a, RQ2b and RQ3 refocusing on 'students E&I capabilities'.

What do you want to understand? → Research aim	How can higher education entrepreneurial ecosystems develop students for entrepreneurship in a resource-constrained environment?		
Why do you need to know this? → Research questions	RQ1: How do diverse stakeholders anticipate the continued development of HEEEs in a resource- constrained environment?	RQ2a: What do diverse stakeholders perceive as the contextual factors of HEEEs that could influence students E&I capabilities in a resource-constrained environment? RQ2b: How and why do specific contextual factors of HEEEs influence students E&I capabilities in a resource-constrained environment?	RQ3: How can diverse stakeholders engage in the factors of HEEEs that could influence students E&I capabilities in a resource-constrained environment?
→ Research objectives	To explore how diverse stakeholders anticipate the future of HEEEs in a resource- constrained environment. Specifically, the linkage between HEEEs with the wider entrepreneurial ecosystem at a system level.	To explore what diverse stakeholders perceive the composition of HEEEs that could influence students E&I capabilities in a resource-constrained environment, along with 'how' and 'why. Specifically, key drivers (i.e. contextual factors) for higher education management to implement at an institutional level.	To explore how diverse stakeholders can engage within the HEEE factors that could influence students E&I capabilities in a resource-constrained environment. Specifically, the stakeholder engagement that benefits students at an individual level.

Table 1.1 – Overview of research aim, questions and objectives

Source – adapted from Maxwell (2013, pp. 13–18)

1.5.1 Research development framework

At the beginning of an investigation, it is important to include a framework to guide research through its research questions and objectives, as Creswell and Creswell (2018) suggested. This research focuses on student entrepreneurial development. Student entrepreneurial development within an HEEE should be transformational, where they progress from a student to a graduate with the relevant knowledge and skills suitable for entrepreneurship (Nielsen & Gartner 2017), as depicted in Figure 1.2. This study intends to determine the composition of the HEEE that responds to this transformation in a resource-constrained environment. From which, the research will provide actionable insights evolving into a comprehensive HEEE by higher education institutions.



Figure 1.2 – Research development framework

1.6 Methodology

In this study, the reality is viewed as socially embedded, existing within the mind, and multiple and constantly changing (Grbich 2013). The goal is to seek an understanding of the world in which the problem prevails and to develop knowledge and meaning jointly through interaction between the researchers and the people who live and work within the researched space (Grbich 2013). This purpose recognises the interpretive framework of this study as social constructivism and the researcher inquires by starting with literature then moving to observations from participants to reach results (Creswell & Creswell 2018).

At the developing stage of the theoretical concept, there is a mix of methodologies among HEEE studies. A few studies including the work of Longva (2021), Miller and Acs (2017) and Rice, Fetters and Greene (2014) are qualitative in nature, while other investigations by Guerrero et al. (2020) and Meyer et al. (2020) have used a quantitative method. A recent study by Webber, Kitagawa and Plumridge (2020) adopted a mixed method. Given the prevailing research problem, questions and objectives, there is a need for in-depth exploration of HEEEs. Such a need for exploration and hearing diverse voices directs qualitative research (Creswell & Poth 2018). Qualitative research allows for unexpected development that may emerge as investigation begins from the exploratory research questions, and the outcome is unique models (Grbich 2007).

When exploring HEEEs, the perceptions and experiences of ecosystem actors are paramount in understanding the concept, which they are part of or relate to. Yet these actors that shape and operationalise the environment are yet to be heard as a whole representation. The majority of existing studies have investigated HEEEs either among internal stakeholders
or external stakeholders, with a few early exceptions having investigated HEEEs covering both groups. As one of the initial investigations of HEEEs focused on students' entrepreneurial development, this research plans to gather broader perspectives drawing participants who represent the wider entrepreneurial ecosystem. The qualitative interpretive research garners viewpoints from various stakeholders who can reflect on their real-world experiences and share their observations of working with students and graduates, providing external validity.

Given that literature on entrepreneurial ecosystems emphasises a systems approach (see Table 2.1), this study views HEEEs from a systems perspective. Based on the work on entrepreneurial ecosystems by Stam (2015), Feld (2012) and Isenberg (2010), various stakeholders were identified according to their relevance in the context. This study included deans/heads of schools (human capital domain), academics/educators (human capital domain), alumni entrepreneurs (market domain), established entrepreneurs (market domain), angel investors (financial capital domain) and support professionals (support domain). Taking a systems perspective, various actors of the HEEE and the local entrepreneurial ecosystem were interviewed in their specific roles for this exploratory research. In addition to their involvement in HEEEs, the selection of key external stakeholders was based their interest on supporting young entrepreneurs and contributing to new start-ups. In doing so, this study brings broader perspectives through 40 semi-structured interviews with participants from six diverse stakeholder groups, representing key stakeholders of the HEEE and their entrepreneurial ecosystem. The study is based on the private higher education sector in Sri Lanka, which has minimum involvement from the government.

1.7 Scope and limitations

While this research is focused on a specific problem and plans to contribute to original knowledge, the study has a clear scope. This section defines the scope of investigation in this research, establishing what this study is not about and the boundaries are drawn underlining the theoretical concept and geographic context.

1.7.1 Theoretical concept

This study contributes to the field of entrepreneurship. The theoretical concept under investigation in this research is the 'higher education entrepreneurial ecosystem' of higher education institutions. HEEEs serve a two-fold function of (1) delivering entrepreneurship education and (2) supporting the development start-ups and new ventures (Rice, Fetters & Greene 2014). The research investigates the whole HEEE that includes both functions, as

entrepreneurship education alone will not sufficiently impact student entrepreneurs or entrepreneurial activity as relevant entrepreneurial support is required (O'Brien, Cooney & Blenker 2019). Therefore, this research is not limited to 'entrepreneurship education ecosystems' where scholars explore courses, curriculum, co-curriculum, pedagogy, assessments and educators (Ligouri et al. 2020; Brush 2014) nor is the study restricted to entrepreneurial support that examines incubators and accelerators (Guerrero et al. 2020; Theodoraki et al. 2018). The scope explores HEEEs as a whole, considering their contextual factors and actors in the view of entrepreneurship education and entrepreneurial support. Finally, HEEEs are associated to their entrepreneurial ecosystems that promotes new venture creation and not necessarily innovation systems that encourages innovation outputs.

Further, the growing consensus is that HEEEs co-created and managed by higher education institutions (Wright, Siegel & Mustar 2017) compared to other ecosystems originally formed by students as a movement towards entrepreneurship. Studies related to HEEEs over the last decade have emphasised the university as an institution (Lahikainen et al. 2019; Heaton 2019) and on students (Webber, Kitagawa & Plumridge 2020). The HEEE supports its higher education community including students, alumni and staff with identifying, developing and commercialising entrepreneurial and innovative initiatives (Guerrero et al. 2018). However, this study is focused on HEEEs co-created by higher education institutions for their student community, recognising their importance as youth creating value and contributing to the economy and society.

1.7.2 Geographic context

This study is based on Sri Lanka, an emerging ecosystem where entrepreneurship is a national priority for sustainable development. Sri Lanka has a tiered higher education structure comprising public universities and private higher education institutions providing tertiary education (Dissanayake 2020). The Ministry of Sustainable Development (2018) reported that the Sri Lankan public higher education sector is currently facing several challenges. First, higher education lacks quality, specifically in the line of promoting entrepreneurship and developing students for such a career during their higher education (Ministry of Sustainable Development 2018). Second, limited access to higher education is a challenge for students in Sri Lanka as only 18% of the 20- to 24-year-old population is enrolled in the public higher education sector due to national resource constraints (Ministry of Sustainable Development 2018).

Due to capacity constraints and lack of funding and resources constraining the public sector, there is growing importance on the private higher education section. According to the Sri Lanka Export Development Board (2018) approximately 11,000 students enrol at private higher education institutions for paid undergraduate education every year. These enrolments represent 25% of the total undergraduate enrolments of the public higher education sector. Private higher education institutions deliver undergraduate study programs affiliated to universities in the US, UK, Australia and Canada (Gamage & Wijesooriya 2012). In essence, these private higher education institutions are validated to run the same study programs while providing a similar university experience for their students. Therefore, this study is based on private higher education institutions where HEEEs are relevant, setting a clear boundary for the study.

1.8 Structure of dissertation

This PhD thesis is a cumulation of 10 chapters. They include introduction, literature review, theoretical framework, contextualisation, research design, findings and analysis addressing research question 1, findings and analysis addressing research question 2a, findings and analysis addressing research question 3, and discussion and conclusions.

Chapter 1 introduces the research of this study. The introduction establishes the situation demonstrating that HEEEs are an area of research worthy of investigation, and identifies the problem with specific knowledge needs, research questions and objectives, and the means of arriving at the solution by contributing new understanding and knowledge. Chapter 2 presents a comprehensive review of scholarly work related to HEEEs, including a definitional analysis to establish the consistencies and contrasts within the various versions of HEEE definitions and explanations. This is followed by an integrated review of scholarly work relating to HEEEs that results in the current state of HEEE research, along with academic debates after scanning 67 articles from a total of 991 articles between 2000 and 2020. Chapter 3 offers the context of this study with a detailed overview of Sri Lanka and its higher education sector while framing Sri Lanka as a resource-constrained environment. Chapter 4 provides the theoretical framework for this study including the use of theory in current HEEE studies and the four theoretical lenses. Chapter 5 explains the philosophy and researcher values followed by a justified account of data collection and analysis along with ethical considerations. This chapter illustrates the need for qualitative methodology, justifying the benefits including rich

details and depth of 'how' and 'why' for this research and the area of research. Chapters 6 to 9 present the empirical findings of this study in a systematic and detailed way. Chapter 6 reports findings extracted from interview data that responds to the first research question on continued development of HEEEs, including excerpts and a reorder matrix. Research questions 2a, 2b and 3 were revised after this chapter to focus on students E&I capabilities which emerged in this chapter. Chapter 7 presents and describes the findings supported by excerpts and a spectrum display associated with the second research question. This qualitative research also produced deeper findings on the 'how' and 'why' of HEEEs factors, presented in Chapter 8. Chapter 9 reports findings addressing the final research question on stakeholder engagement and its value for students keen on an entrepreneurial career. Chapter 10 is the closing chapter of this study including a discussion and conclusion together with the theoretical contributions and practical implications. This chapter will also propose future research from the questions that emerged from this research.

1.9 Chapter summary

This introductory chapter gives an insightful overview of the research's theoretical background and shows the increasing importance and growing attention the area of research has recently received. The chapter also positions this study in the field of research, by showing the current literature and what is known so far, clarifying the gap to be covered and discussing the research design. The section on scope and limitation clearly defines the theoretical concept and geographical context in focus. The chapter highlights that HEEEs are an under-researched area, despite having important consequences in the practical world. According to recent work by Delanoë-Gueguen and Theodoraki (2021), Hsieh and Kelley (2020) and Longva (2021), HEEEs are in the early stage of development, with a lack of understanding of their composition and configuration in serving their purpose. Current research mainly emphasises developing HEEEs for producing student start-ups, and more recently, scholars have turned to students' entrepreneurial mindset and intention. Followed by the importance of HEEEs in entrepreneurship, this chapter argues the need for facilitating student entrepreneurial development within HEEEs.

Extant HEEE literature has found a broad list of HEEE factors contributing to the composition of HEEEs. Most studies in the field of HEEE have only focused on student startups, entrepreneurial mindset and intention in developed countries. There is minimum previous research that has investigated HEEE towards student entrepreneurial development within HEEEs, even though there is a great deal of importance expressed on the student knowledge and skill required for taking on an entrepreneurial career. This thesis is one of the empirical studies set to explore the contextual factors and actors of HEEEs through a student development lens in a resource-constrained context. The study is concentrated on Sri Lanka with a focus on undergraduates of private higher education institutions. This research is exploratory and interpretative in nature using a qualitative approach.

To diagrammatically frame this PhD thesis, Luker's (2008) bedraggled daisy is applied below. Figure 1.3 is a Venn diagram representing the sets of eight different elements with the key interest investigated in this study at the centre.



Figure 1.3 – Key interests of this study

CHAPTER 2: LITERATURE REVIEW

2.1 Chapter overview

For decades, scholars have debated on how higher education institutions could play a prominent role in economic development and social change, and there has been great interest in the entrepreneurship and management literature (Klofsten et al. 2019). However, the historical, theoretical and empirical foundations of HEEEs are recent (Longva 2021; Hsieh & Kelley 2020).

The literature review creating the case for this study comprises five distinct sections as systematically structured in Figure 2.1 below. This chapter analyses the current scholarly work to recognise key contributions to HEEEs and determine gaps that exist in the literature using an integrated literature review approach. In doing so, the integrative review of HEEE literature includes 67 articles from a total of 991 articles from three-independent e-databases (Scopus, EBSCOhost and Web of Science) as leading management and entrepreneurship journals. The initial search period was set between 2000 to 2020 to bring together the work within these two decades. In particular, the articles emerged from top tier journals including *Small Business Economics, Entrepreneurship Theory and Practice, Entrepreneurship and Regional Development*, the *Journal of Management* and *Higher Education.* To include the most recent articles in 2021 and 2022 to this literature chapter, a second search was conducted among the same e-databases. Given that HEEE is a contemporary concept, this chapter provides a review underpinning literature in the field, related to and specifically on HEEEs.



Figure 2.1 – Chapter 2 outline

2.2 System view of HEEEs

2.2.1 Sub-ecosystem of entrepreneurial ecosystem

Systematic empirical evidence demonstrates the power entrepreneurial ecosystems have in enhancing entrepreneurship (Audretsch et al. 2021; Stam 2018). The roots of entrepreneurial ecosystems are primarily credited to various scholars including Neck et al. (2004), Cohen (2006), Isenberg (2011), Feld (2012) and Stam (2015). Entrepreneurial ecosystems as a phenomenon have gained enormous popularity within research, practice and policymaking over the last decade (Wurth et al. 2021). This phenomenon was quickly adopted by government and non-government organisations such as the Organisation for Economic Co-operation and Development (OECD) (2021) and the World Economic Forum (2014). The idea is that the entrepreneurial ecosystem surrounding the process supports and provides resources for entrepreneurial individuals igniting entrepreneurship, entrepreneurial ecosystems are the most potent force around the world – in both developed and developing countries (Isenberg 2010).

In early studies, entrepreneurial ecosystems were defined by Cohen (2006, p. 3) as "an interconnected group of actors in a local geographic community committed to sustainable development through the support and facilitation of new sustainable ventures". Since then, several scholars have added new perspectives through their definitions, emphasising different features enhancing the phenomenon (refer Table 2.1). When considering their composition, some definitions are inclined towards the 'factors' of the entrepreneurial ecosystem, while others are focused on the 'actors' (Roundy, Bradshaw & Brockman 2018; Acs et al. 2017; Mason & Brown 2017; Cohen 2006). For this research, the nature of an entrepreneurial ecosystem is inspired by Stam and Spigel (2016, p. 1) and understood as "as a set of interdependent actors and factors coordinated in such a way that they enable productive entrepreneurship within a particular territory". This definition resonates well with this research as it highlights the composition of an entrepreneurial ecosystem, including a combination of factors and actors in a given geographic context.

Table 2.1 – Definitions of entrepreneurial ecosystems among scholarly work

Year, page	Scholars	Definition
2018, p. 5	Roundy, Bradshaw & Brockman	A self-organised, adaptive and geographically bounded community of complex agents operating at multiple, aggregated levels, whose non-linear interactions result in the patterns of activities through which new ventures form and dissolve over time.
2017, p. 50	Spigel	Entrepreneurial ecosystems are combinations of social, political, economic and cultural elements within a region that support the development and growth of innovative start-ups and encourage nascent entrepreneurs and other actors to take the risks of starting, funding and otherwise assisting high-risk ventures.
2017 p. 479	Acs et al.	A National System of Entrepreneurship is the dynamic, institutionally embedded interaction between entrepreneurial attitudes, ability and aspirations, by individuals, which drives the allocation of resources through the creation and operation of new ventures.
2017, p. 120	Kuratko et al.	Entrepreneurial ecosystem as coordinated attempts to establish environments that are conducive to the probabilities of success for new ventures following their launch entrepreneurial ecosystems are focused on creating environments conducive to the success of entrepreneurs and their new venture.
2017, p. 98	Auerswald & Dani	Represents the higher-level infrastructure that enables interactions between the entrepreneurial agents and institutions in the industrial sector They cut across industries and focus on the environment surrounding entrepreneurs – with entrepreneurs and entrepreneurship clearly at the centre.
2017, p. 1	Bruns, Bosma, Sanders & Schramm	Entrepreneurial ecosystem as a multidimensional set of interacting factors that moderate the effect of entrepreneurial activity on economic growth.
2016, p. 2	Audretsch & Belitski	Institutional and organisational as well as other systemic factors that interact and influence the identification and commercialisation of entrepreneurial opportunities.
2015, p. 1765	Stam	The entrepreneurial ecosystem is a set of interdependent actors and factors coordinated in such a way that they enable productive entrepreneurship.
2014, p. 5	Mason & Brown	A set of interconnected entrepreneurial actors (both potential and existing), entrepreneurial organisations (e.g. firms, venture capitalists, business angels, banks), institutions (universities, public sector agencies, financial bodies) and entrepreneurial processes (e.g. the business birth rate, numbers of high growth firms, levels of 'blockbuster entrepreneurship', number of serial entrepreneurs, degree of sell-out mentality within firms and levels of entrepreneurial ambition) which formally and informally coalesce to connect, mediate and govern the performance within the local entrepreneurial environment.
2012, p. 25	Regele & Neck	The interaction of people, roles, infrastructure, organisations and events creates an environment for heightened levels of entrepreneurial activity.
2010, p. 43	Isenberg	The entrepreneurship ecosystem consists of a set of individual elements – such as leadership, culture, capital markets and open-minded customers – that combine in complex ways.
2006, p. 3	Cohen	Sustainable entrepreneurial ecosystems are defined as an interconnected group of actors in a local geographic community committed to sustainable development through the support and facilitation of new sustainable ventures.

There is deep understanding of entrepreneurial ecosystems in terms of their components and configuration. Entrepreneurial ecosystem was characterised by conditions/elements, outputs and outcomes as a system (Stam and Spigel 2016) and are conceived as closely connected (Feldman, Siegel and White 2019). These structural components are consistent in the practical context where the Global Entrepreneurship Monitor draws their conceptual framework including inputs, such as entrepreneurial activity, that lead to outputs and outcomes. This view has transferred from the field of economics, which is about systems that explain outputs and outcomes (Acs et al. 2017). Thus, entrepreneurial ecosystems are developed through the lens of systems theory investigating entrepreneurial ecosystems as a whole rather than a sum of components (Bischoff et al. 2018; Audrestsch & Belitski 2017).

Entrepreneurship scholarly work advocates a holistic approach by focusing on the role of the entrepreneurial ecosystem and their domains to nurture and sustain entrepreneurs (Audretsch et al. 2017; Autio et al. 2014). Among the various entrepreneurial ecosystems, 'human capital' is a common factor (refer Table 2.2). The economist Adam Smith (1776) considered human capital as an asset of economic value, which led to academic interest in the economic effects of human capital. Human capital is considered one of the six domains in Isenberg's (2016) entrepreneurial ecosystem. Human capital represents the talent, including entrepreneurs and individuals skilled with entrepreneurship, that appears as an element among entrepreneurial ecosystems as suggested by Stam (2015), Feld (2012), Cohen (2006) and Neck et al. (2004). Educational institutions are acknowledged as the source of this talent and are connected to the entrepreneurial ecosystem (Feld 2012).

Scholars	Theory	Methodology	Elements/factors of entrepreneurial ecosystems
Stam (2015)	System approach	Conceptual	Networks, leadership, finance, talent, knowledge, support services/intermediaries, demand, physical infrastructure, culture and formal institutions.
Feld (2012)	System approach	Conceptual	Leadership, intermediaries, network density, government, talent, support services, engagement, companies and capital.
Isenberg (2011)	System approach	Conceptual	Policy, finance, culture, supports, human capital and markets.
Cohen (2006)	System approach	Conceptual	Informal network, formal network, university, government, professional and support services, capital services and talent.
Neck et al. (2004)	System approach	Empirical	Incubator organisations, informal and formal networks, university, government, support services, capital, talent pool, large corporations, physical infrastructure and culture.

Table 2.2 – Key entrepreneurial ecosystems theoretical frameworks

2.2.2 Entrepreneurial initiative of universities

The impact of universities can vary considerably within a city, region and country (Spigel & Wright 2015). Although the dominant role of universities is acknowledged as producing qualified employees, these institutions are natural incubators due to their responsibility in teaching, research and contributing to entrepreneurship (Etzkowitz & Kolfsten 2005). Higher education institutions, including universities, have been considered as critical stakeholders of the entrepreneurial ecosystem (Startup Genome 2020). Scholars such as Isenberg (2011), Cohen (2006) and Neck et al. (2004), explicitly listed the university as one of the components in the entrepreneurial ecosystem (refer Table 2.2 above). Entrepreneurial universities are positioned as the engines of the knowledge economy (Nelles & Vorley 2010) that fosters entrepreneurship, shaping the wider entrepreneurial ecosystem (Fischer et al. 2018; Wadee & Padayachee 2017; Miller & Acs 2017; Kuratko 2016; Haase & Lautenschläger 2011; Etzkowitz & Leydesdorff 2000; Drucker 1985). There is strong evidence in developed economies that higher education institutions are the 'hearts' of these environments, being a valuable source of entrepreneurship (Fischer et al. 2018; Miller & Acs 2017; Stam 2009; Etzkowitz 1998). Entrepreneurial universities are one of the most influential ecosystem stakeholders in nurturing new entrepreneurs. This means that entrepreneurial universities have a wider scope, progressing from a traditional knowledge generator to an entrepreneurial ecosystem enabler (Trivedi 2016).

Entrepreneurial universities as a concept can be traced to the early 1980s where they emerged as natural incubators providing support to foster entrepreneurship and innovation for the university community including students, staff and alumni (Guerrero & Urbano 2012). An entrepreneurial university is one that undertakes multiple missions for teaching, research and entrepreneurial initiatives playing a prominent role in socio-economic development (Etzkowitz 2014). In broad terms, entrepreneurial universities have the responsibility of creating entrepreneurial capital including the future entrepreneurial behaviour of students (Audretsch 2014). Universities are expected increasingly to engage in a 'third mission' by becoming entrepreneurial in response to policy implications led by the US and spreading to the UK and Europe (Philpott 2011; Gibb 2005). These universities draw on unique conditions and resource endowments in their environments which influences their capability to be 'entrepreneurial' (Bedő, Erdős & Pittaway 2020).

Entrepreneurial universities are educating students, attracting researchers, encouraging new venture creations, facilitating knowledge transfers, and promoting entrepreneurial culture

to shape and supporting entrepreneurial ecosystems (Miller et al. 2018; Guerrero et al. 2016). As entrepreneurial universities foster a supportive environment that encourages entrepreneurial activity among students and staff, ecosystems are a key element of entrepreneurial universities (Secundo et al. 2020; Hannon 2013; Guerrero & Urbano 2012). In this view, entrepreneurial universities face multiple strategic challenges concerning strategic priorities, contextual factors, and processes such as entrepreneurial teaching and learning and measurement metrics (Cunningham & Miller 2021; Forliano et al. 2021; Klofsten et al. 2019; Etzkowitz et al. 2019). To address these challenges, entrepreneurial universities need to develop their own internal processes, policies, systems and infrastructure (Kirby 2004; Wright, Siegel & Mustar 2017). A 'one-size-fits-all' approach that considers universities as homogeneous institutions is unrealistic as every institution is unique in its environment (Degl'Innocenti et al. 2019; Benneworth et al. 2016).

2.2.3 Co-creation of HEEEs

The role of higher education institutions is emerging to be dichotomous. Higher education institutions contribute to economic growth and social development by designing and developing their ecosystem (Guerrero and Urbano 2016). Within a highly competitive higher education sector, every university seeks a unique approach to entrepreneurship and higher education institutions are considering entrepreneurial pathways to becoming entrepreneurial. Entrepreneurial pathways are strategic choices that higher education institutions undertake to demonstrate their commitment and involvement with E&I initiatives within the institutions are co-creating their own entrepreneurial ecosystems, enacting on the third mission (Guerrero, Urbano & Gajón 2017). HEEEs are co-created as a proactive response to entrepreneurial or education initiatives, or they might emerge as reactive responses to specific gaps in entrepreneurial or education development (Rice, Fetters & Greene 2014).

Babson College was one of the first entrepreneurial universities to be recognised in the literature for their leading initiative in developing an entrepreneurial ecosystem for the university community (Fetters, Greene & Rice 2010). Isenberg (2016) criticised this as 'the creation mistake', meaning ecosystems are not designed or created. However, the growing consensus is that university-based entrepreneurship ecosystems are co-created by entrepreneurial universities along with stakeholders (Wright, Siegel & Mustar 2017). Creating a fully developed comprehensive HEEE takes at least two decades, in a dynamic and non-linear process according to Rice, Fetters & Greene (2014). In 2010, when the concept of university-

based entrepreneurship ecosystems emerged through the work of Fetters, Rice and Greene (2010) based on the Babson Collage's entrepreneurship ecosystem, this was a result of more than 30 years of creating a unique entrepreneurial community for higher education.

Reference made to university-based entrepreneurship ecosystems is multifaceted. Some 17 different terms emerged in the scholarly work referring to the same university-based entrepreneurship ecosystems. These terms are interchangeably used where scholars refer to the same concept and this is likely to lead to a research overlap in the field. When analysing the literature in this review, it came to light that 'university-based entrepreneurship ecosystem' was the most widely represented term in scholarly work. This could be due its roots in the first conceptualisation by Fetters, Rice and Greene (2010). Other references to the ecosystem are entrepreneurial ecosystem', 'university ecosystem', 'university-centred 'university entrepreneurship ecosystem', 'innovation and entrepreneurship ecosystem', 'university business cooperation ecosystem', 'entrepreneurial university ecosystem', 'university environment and support system', 'academic entrepreneurial ecosystem', 'university-driven entrepreneurship ecosystem', 'research university entrepreneurial ecosystem', 'universitywide entrepreneurship ecosystem', 'ecosystem for student start-ups', 'engagement, employability and employment ecosystem', 'higher education entrepreneurial ecosystem', and 'learning and entrepreneurship ecosystem'. Outside of the integrative review, it was found that entrepreneurial ecosystems in the higher education context was emerging to be referred as 'university innovation and entrepreneurial ecosystems'.

The term 'higher education entrepreneurial ecosystem' was suggested by Guerrero, Urbano and Gajon (2017) when investigating the context of an emerging economy, Mexico. HEEEs seems more suitable to contexts where the higher education sector is a representation of universities and other education institutions. Not all countries have mirrored this type of education system. In developing countries, where government funding is limited, higher education is provided by a lower number of public universities and more private higher education institutions. Contrary to this, in developed countries, higher education is dominated by universities and the terminology 'university-based entrepreneurship ecosystem' is suitable.

Given that this study is undertaken in the context of a developing country, key terms used in this research may be different to a developed country. Previous studies, such as by Kasturiratne, Lean and Phippen (2012) on entrepreneurship education in Sri Lanka, used the term 'higher education institution'. Throughout this study, an entrepreneurial university is

referred to as higher education institution and a university-based entrepreneurship ecosystem is referred to as a higher education entrepreneurship ecosystem (HEEE), making it more relevant to the private higher education system in Sri Lanka. As mentioned, HEEE was used by Guerrero, Urbano and Gajón (2017) for a study on HEEEs in Mexico, an emerging economy.

Higher education institutions along with their HEEEs belong to wider entrepreneurial ecosystems and attract resources from stakeholders at local, regional, national and international levels (Webber, Kitagawa & Plumridge 2020). If higher education institutions focus on E&I, the successes and benefits of the transformational process should be profound for the broader community (Matriz et al. 2020; Birch et al. 2017). The link between HEEEs and their wider entrepreneurial ecosystems are significant because students who pursue entrepreneurship affect the entrepreneurial ecosystem through job creation, innovation, knowledge creation and economic dynamism (Maezocchi et al. 2019). The interrelations between entrepreneurial ecosystems are relevant as both ecosystems regulate the entrepreneurial activity that contributes to economic development and social change (Wright, Siegel & Mustar 2017).

2.3 Higher education entrepreneurial ecosystems

Entrepreneurial ecosystems within higher education gained prominence from the work of Fetters, Greene and Rice (2010) and their book *The Development of University-Based Entrepreneurship Ecosystems*. The book provided global insights into framing, designing, launching and sustaining entrepreneurial efforts through HEEEs framed as a support context in which E&I can thrive (Fetters, Greene & Rice 2010). To introduce the topic on HEEE, this section reviews the recent history and evolving definitions, growth of scholarly work, geographic spread, theoretical framing of existing studies, methodology applied, the nexus of entrepreneurship education and entrepreneurial support, the co-creation of HEEEs and the sub-ecosystems of HEEEs.

2.3.1 Emergence and definitions

The sub-ecosystem HEEE has been introduced recently to support entrepreneurship in the context of higher education. Scholars such as Brush (2014) acknowledge the work of Fetters, Greene and Rice (2010) as the initial conceptualisation of HEEEs. This scholarly work was a case study of six universities that led to seven success factors for developing university-based entrepreneurship ecosystems (Rice, Fetters & Greene 2014). Since then, more scholars have

joined the pursuit for factors that form a comprehensive, highly evolved university-based entrepreneurship ecosystem, with contextual relevance (Stam & Spigel 2016).

When theorising about entrepreneurship, scholars Bygrave and Hofer (1991, p. 13) said "good science starts with good definitions". Examining scholarly work found that HEEE lacks a common definition (Bock et al. 2020). The initial study that proposed HEEEs by Greene et al. (2010, p. 2) expressed HEEEs as "multidimensional enterprises that support entrepreneurship development through a variety of initiatives related to teaching, research and outreach", identifying it as a set of organisations. This notion advanced to HEEEs as "an extraordinarily resource-rich, comprehensive and dynamic context for delivering entrepreneurship education and for supporting the start-up and development of new ventures" (Rice, Fetters & Greene 2014, p. 177), which established their context and functions. Being an evolving concept, the theoretical development of HEEE is in its early development stages, where HEEEs are novel and emerging (Longva 2021; Hsieh & Kelley 2020). Drawing on scholarly work, other scholars in the field have expressed HEEEs in a similar way, as shown in Table 2.3.

Year	Scholars	Definition			
2020	Bock et al.	A set of actors that engage with regional and national innovation ecosystems through relationships with the industry			
2020	Webber, Kitagawa & Plumridge	Organisational capabilities, resources and, ultimately, the entrepreneurial knowledge provided by the university, with characteristics associated with both research and teaching missions			
2018	Gomesde et al.	An interconnected network of actors (individuals, groups and organizations) that cooperate in formal and informal ways in order to create value			
2019	Belitski	A context of a specific school, university and entrepreneurial ecosystem, and defined within a specific business environment			
2014	Rice, Fetters & Greene	An extraordinarily resource-rich, comprehensive and dynamic context for delivering entrepreneurship education and for supporting the start- up and development of new ventures			
2010	Fetters, Greene & Rice	Multidimensional enterprises that support entrepreneurship development through a variety of initiatives related to teaching, research and outreach			

Table 2.3 – Definitions of HEEEs among scholarly work

From the above definitions, HEEEs are characterised primarily by the stakeholders (e.g. multidimensional enterprises, set of actors) within a specific context that may contribute to their functioning of delivering entrepreneurship education and supporting start-ups (refer Table 2.3). In contrast, entrepreneurial ecosystems are broadly understood as a set of factors

and actors within a particular territory (Stam 2015). This comparison shows that 'factors' are not explicitly included in the HEEE definitions, although studies investigating entrepreneurship ecosystems in higher education suggest lists of factors (Miller & Acs 2017; Rice, Fetters & Greene 2014). Hence, there is scope to address this gap in the HEEE definition.

2.3.2 Growth of HEEE scholarly work

Although the search period for this literature review began from 2000, the first article that surfaced was dated to 2010. This first article, published by Fetters, Greene and Rice (2010), appears to be a notable contribution to HEEEs. The article is cited by various scholars exploring HEEEs, such as Delanoë-Gueguen and Theodoraki (2021) and Webber, Kitagawa and Plumridge (2020). During the period of 2011 to 2013 inclusive, as well as in 2015, no articles emerged. In 2014, there were two articles and one of these was an extension of the first contribution involving the same scholars – Rice, Fetters and Greene (2014). This article is another noteworthy effort, garnering 42 citations since publication. These two articles seem to have stimulated interest among scholars in this area, as since 2016 there has been a significant growth of scholarly work focused on HEEEs.

2.3.3 Geographic spread of HEEE scholarly work

The majority of articles found during this literature review were based on a single country while a few articles investigated multiple countries (i.e. more than two). Articles related to and on HEEEs covered more than 33 different countries. The roots of entrepreneurship education traced back to the US (Katz 2003) and the spread of publications on HEEE began in the US (Kirby 2004). This trend has continued, where the US remains the most researched geographic context for HEEEs, with over 12 articles published within a decade.

In Figure 2.2, it becomes evident that the extant research is mostly based on developed (high income) countries such as the US, UK and Spain, resulting in an imbalanced spread of HEEE research across the world. This is consistent with the paucity of research on graduate entrepreneurship in developing countries as suggested by Nabi and Linan (2011). Among the results, HEEE-related articles are almost non-existent on developing countries such as Algeria, Bhutan, Cambodia, Ghana, Kenya, Nepal, Sri Lanka, Vietnam and Zimbabwe, which are classified as lower-middle income countries and emerging economies by the World Bank (2020). The lack of research on emerging economies is not rational as entrepreneurship is known to facilitate the developing-to-developed country transition (Masakure, Henson & Cranfield 2009).



Figure 2.2 – Countries studied in HEEE literature

Although the current century is termed the 'Asian Century' (Walmsley 2018), the extant of HEEE research focused on exploring Asian countries is minimal. For instance, this review reports three articles in Malaysia, two in Taiwan and just one article each in Asian countries such as South Korea, Singapore, Indonesia, Bangladesh and the Philippines. However, when classified and analysed as continents, in Figure 2.3, most of the research was found in the Asian countries with 25 of 67 articles, followed by Europe and North America. A high number of 15 Asian countries were studied as the context compared to 10 European countries and four countries in North America. The least representation is reported by South America, Australia and Oceania, and Africa.





2.3.4 Framing of HEEE scholarly work

The integrative review extended to understand the underlying theories of HEEE studies among the 67 articles. Studies have been developed using various theories taking different perspectives. In investigating the composition of HEEE for student start-ups, scholars have applied theories ranging from the resource-based view (Fetters, Greene & Rice 2010), Turner's frontier thesis (Miller & Acs 2017), the triple helix (Shil et al. 2020; Allahar & Sookram 2019) to social network theory (Longva 2021). For students' entrepreneurial intention, studies have undertaken the theory of planned behaviour (Webber, Kitagawa & Plumridge 2020) and the utility-maximising function (Guerrero, Urbano & Gajón 2020).

This review reveals that studies exclusive to the function of entrepreneurship education have involved theories such as stakeholder theory (Bischoff 2018), learning theories (Middleton et al. 2019) and student involvement theory (Bock et al. 2020) to discover the composition of HEEEs.

Lastly, by looking at the current state of HEEE research, it was evident scholars paying attention to entrepreneurial support utilised institutional theory (Lahikainen et al. 2019), social capital theory (Theodoraki, Messeghem & Rice 2018) and the theory of planned behaviour (Guerrero, Urbano & Gajón 2017). There is a trend of multiple theories being used to frame and investigate HEEE studies, even when focusing on the area of study, instead of building on one suitable theory leading to deeper contribution.

Reviews support the need for qualitative, quantitative or mixed studies establishing the research on previous scholarly work (Rocco & Plakhotnik 2009). This review indicates 40% qualitative, 28% quantitative, 21% mixed and 12% conceptual with a high extent of 88% of research being evidence-based publications. Research on HEEEs is predominately undertaken using qualitative methodology and second by quantitative methodology. Although the mixed methodology is known to combine the strengths of qualitative and quantitative methodologies (Johnson, Bock & George 2019; Ricci et al. 2018), mixed methodology is the least conducted by entrepreneurship scholars in the HEEE literature.

Next, the dataset was analysed to determine the most frequently used data collection methods. In most instances, a single method was utilised in the articles while there were some instances when multiple methods were applied, such as two qualitative methods (observations and focus group discussions) or mixed methods of quantitative and qualitative. Although qualitative methodology is most widely practiced by scholars investigating HEEEs, the survey questionnaire (28 of 67 articles) is the most represented among quantitative and mixed-method methodology, closely followed by interviews (26 of 67 articles). As methods, survey questionnaires and interviews are the leading methods among the 67 articles. The use of secondary data, such as public databases, shows more popularity over documents, observations and focus groups in HEEE research. Quantitative methods have only used survey questionnaires while qualitative methods have applied, respectively, interviews, secondary data, documents observations and focus groups.

Among the articles in the review, representation of voice considered in the investigation was determined. Most of the time, multiple groups of participants were investigated rather than a single category of participants such as students or faculty. Although the HEEE context is a multi-stakeholder environment (Belitski & Heron 2016; Autio et al. 2014; Fetters, Greene & Rice 2010), the review reveals that studies have paid greater attention to capture internal views and investigated HEEEs among members of the university. Investigations are most common among faculty, students and the university as participants with 16 of 67 articles each (refer Figure 2.4). Start-up founders and entrepreneurs rank as the second most investigated participants among 10 of 67 articles. External stakeholders such as alumni entrepreneurs are discussed as key contributors of entrepreneurial education and support provision (Meyer et al. 2020), yet their participation is minimal as the unit of analysis, with just three of 67 articles. More studies investigate current students with a focus on the intended behaviour of 'becoming

an entrepreneur'. Considering the extant literature, studies lack the representation of diverse stakeholders in exploring HEEEs.



Figure 2.4 – Participants investigated in previous HEEE studies

2.3.5 Nexus of entrepreneurship education and entrepreneurial support

When analysing scholarly work on HEEE, two dominant lineages emerged – 'entrepreneurship education' and 'entrepreneurial ecosystem'. The first, entrepreneurship education, refers to any education program aimed at developing entrepreneurial knowledge, skills and attitude among students (Fayolle 2006; Katz 2003). Scholars such as Kuratko (2005) argue this entrepreneurship education to be different from typical business education. Following the well-known debate on whether entrepreneurship can be taught, we have established that entrepreneurship education is an accepted discipline perceived to develop an entrepreneurial mindset and create entrepreneurial intention stimulating entrepreneurial behaviour (Kuratko & Morris 2018; Kautonen, van Gelderen & Fink 2015; Rauch & Hulsink 2015; Walter, Parboteeah & Walter 2013; Neck & Greene 2011). With increasing emphasis on entrepreneurship to support growth (Audretsch et al. 2012), entrepreneurship education is recognised as a key driver (Thrane et al. 2016; Katz 2003). This first lineage has developed over time with substantial research (Mason et al. 2020). However, entrepreneurship education

alone does not provide students the full potential towards entrepreneurial outputs such as entrepreneurial knowledge and skills (O'Brien et al. 2019).

Second, an entrepreneurial ecosystem plays the role of supporting the creation, growth and survival of new ventures (Stam & Spigel 2016). An ecosystem is commonly understood as a community of living organisms and non-living components of a respective environment interacting as a system, which is its biological interpretation (Isenberg 2016). Such an entrepreneurial ecosystem within higher education supports student entrepreneurs to start-up, scaleup, survive and succeed (Rice, Fetters & Greene 2014). This second lineage of the entrepreneurial ecosystem being the support system that aids start-ups and accelerates start-ups among students is an emergent research field (Mason et al. 2020). Given the functions of HEEEs set out above by Rice, Fetters and Greene (2014), entrepreneurship education and entrepreneurial support are equally important to represent the whole HEEE. This intersection of entrepreneurship education and entrepreneurial support enact the crucial role HEEEs play in transforming the mindsets, intentions, capabilities and actions of students.

2.3.6 Sub-ecosystems of HEEEs

According to Fetters, Greene and Rice (2010), Babson's entrepreneurial ecosystem had two significant vectors: (1) an entrepreneurial mindset that involved creating career change among individuals; and (2) entrepreneurial thinking and action in established corporations. This understanding later translated into the initial conceptualisation of HEEEs that recognised 'entrepreneurship education' and 'entrepreneurial support' as the two primary functions (Rice, Fetters & Greene 2014). Since then, there has been research on HEEEs as ecosystems and related to functions of HEEEs.

Entrepreneurship education ecosystems on one hand are a central component of the HEEE, including activities associated to curriculum, co-curriculum and research (Brush 2014). On the other hand, is the provision of entrepreneurial support involving networks, incubators and accelerators within higher education (Guerrero, Urbano & Gajón 2017; Theodoraki, Messeghem & Rice 2018). These entrepreneurship support organisations are crucial components in successful entrepreneurial ecosystems (Spigel 2016). According to literature, these are sub-ecosystems of HEEEs (refer Figure 2.5). Table 2.4 identifies empirical studies with a focus on entrepreneurship education or entrepreneurial support. Within the scholarly work, there seems to be more research related to HEEEs including 'entrepreneurship education' or 'entrepreneurial support' rather than on HEEEs as whole ecosystems.



Figure 2.5 – Sub-ecosystems of HEEEs

Scholars	Focus	Theory	Methodology	Context	Factors
Lu et al. (2022)	Entrepreneurship education	n/a	Mixed Interviews, Focus groups, Surveys and Documents	China	Incubation platforms, teacher team building, talent training model, organisational structure and curriculum systems
Liu, Kulturel- Konak & Konak (2021)	Entrepreneurship education	Ecosystem theory	Qualitative Interviews	United States	Entrepreneurship curriculum, entrepreneurial activities and practices, organizational structure, resources, leadership vision, core faculty, and operating mechanism
Bock et al. (2020)	Entrepreneurship education	Student Involvement Theory	Quantitative Surveys questionnaire	Germany	Functional experiences and connecting experiences
Mukesh & Pillai (2020)	Entrepreneurship education	n/a	Quantitative Surveys questionnaire	India	Entrepreneurship promotion activity by HEIs, HEIs attitude towards entrepreneurship, department philosophy on entrepreneurship, HEIs teachers and staff, HEIs support for entrepreneurship, HEIs physical infrastructure and facilities, HEIs financial support for entrepreneurship, governance structure, degree of entrepreneurial education in curriculum, student orientation on entrepreneurship, entrepreneurial teaching methodologies, HEIs ability to connect start-up's with industry, mentoring and coaching programs for entrepreneurs, and extracurricular activity relating to entrepreneurship
Middleton et al. (2019)	Entrepreneurship education	Learning theories	Qualitative Interviews	Spain, Sweden, and the UK	Socialised learning, the role of mentorship, and relevance of previous entrepreneurship education
Lahikainen et al. (2019)	Entrepreneurial support	Institutional theory	Qualitative Interviews	Finland	Cognitive: Incubator, entrepreneurial teams, students, university support Normative: roles, values, ways of working and work norms Regulative: Incentives, patents, resources and support services
Bischoff (2018)	Entrepreneurship education	Stakeholder theory	Qualitative Interviews	19 different European countries	Stakeholder involvement and collaboration
Theodoraki, Messeghem & Rice (2018)	Entrepreneurial support	Social capital theory	Qualitative Interviews	France	Members, ties, networks, specialisation, diversity, leadership, new entrants, barriers, co-evolution, scalability, culture, shared tools/methods/practices, shared vision, common beliefs, stories, legends, climate, trust, rules, standards/customs/traditions, cooperation, complementarity, reorientation, proximity, and roles/functions/services
Guerrero, Urbano & Gajón (2017)	Entrepreneurial support	Theory of planned behaviour	Secondary databases	Mexico	Incubation mechanism: Entrepreneurial university's services – workshops/networking, contact platforms/points, mentoring and coaching programs, seed financial support Entrepreneurial university's resources: financial, physical and human

Table 2.4 – Scholarly work related to HEEEs (by year)

2.4 Continuum of HEEEs

The evolution of HEEEs cater for all students, from potential entrepreneurs with an entrepreneurial intention to those simply keen to know about entrepreneurship. However, the static perspective of an HEEE highlighting student start-ups dominates the current literature with studies by several scholars (Longva 2021; Miller & Acs 2017; Rice, Fetters & Greene 2014; Wright, Siegel & Mustar 2017). Student start-ups are only one aspect of the start-up process according to Duruflé, Hellmann and Wilson (2018). The start-up process for students is identified in three stages: (1) creating entrepreneurial mindset and intentions; (2) developing entrepreneurial knowledge and skills; and (3) building start-ups (Duruflé, Hellmann and Wilson 2018).

Given that HEEEs facilitate the start-up process through entrepreneurship education and entrepreneurial support, current HEEEs studies and models were mapped to this process (refer Figure 2.6). First, HEEE scholarly work has predominantly focused on producing student start-ups and fostering students' entrepreneurial mindset and intentions in recent times. There is no evidence of studies investigating the HEEE influence on students' entrepreneurial development; that is, students' entrepreneurial knowledge and skills. Therefore, student entrepreneurial development is identified as a missing link in HEEE literature. Second, the development of HEEEs is a bottom-up movement where HEEEs have emphasised building student start-ups before considering the antecedents that lead to the end of the start-up process.



Figure 2.6 – Missing link in HEEE studies through the lens of start-ups process

Adapted from Duruflé, Hellmann and Wilson (2018)

2.4.1 HEEEs for student start-ups

Entrepreneurial ecosystems are linked to the increasing rates of start-ups (Audretsch et al. 2017). Scholars such as Johnson, Bock and George (2019) argue that the development of HEEEs seems to be triggered by student start-ups. Considering the trajectory of HEEEs, higher education institutions promote student entrepreneurship through fostering their ecosystems. HEEEs have had their undivided focus on student start-ups since inception and until recent work by Longva (2021). More scholars have investigated HEEEs in different contexts including the US, Caribbean, India, Bangladesh, South Korea and Norway with a focus on student start-ups over the last decade (Allahar & Sookram 2019; Meyer et al. 2020; Miller & Acs 2017; Shil et al. 2020; Wright, Siegel & Mustar 2017). This is the first wave of HEEE focusing primarily on student start-ups (refer Table 2.5).

Scholars	Theory	Methodology	Context	Focus
Liu, Kulturel-Konak & Konak (2021)	Ecosystem theory	QualitativeUSIn-depth interviews		Student start-ups
Longva (2021)	Social network theory	QualitativeNorwayIn-depth interviews		Student start-ups
Meyer et al. (2020)	Stakeholders	Quantitative researchUS, South KoreaSurveyand India		Student start-ups
Shil et al. (2020)	Triple helix	Qualitative research Focus group discussions	alitative research Bangladesh cus group discussions	
Allahar & Sookram (2019)	Triple helix	QualitativeCaribbeanCase studiesSecondary data		Student start-ups
Wright, Siegel & Mustar (2017)	n/a	n/a	Multiple countries	Student start-ups
Miller & Acs (2017)	Turner's frontier thesis	Case study Interviews Observation Documents and media content	US	Student start-ups
Rice, Fetters & Greene (2014)	Resource-based view	Case study Interviews Secondary data (surveys, project data)	US	Student start-ups

 Table 2.5 – HEEEs focusing on student start-ups (by year)

2.4.2 HEEEs for students' entrepreneurial psychology

After a decade since the inception of HEEEs, the dominant focus is observed to have expanded from student start-ups to students' entrepreneurial psychology as the second wave (refer Table 2.6). Although, scholars such as Secundo et al. (2020), Guerrero, Urbano and Gajón (2020) and Webber, Kitagawa and Plumridge (2020) began emphasising students' entrepreneurial mindset and intention through HEEEs recently, the initial conceptualisation of HEEEs underlined entrepreneurial mindset and intention. The main vector of Babson's entrepreneurial ecosystem concentrated on entrepreneurial mindset to create and embrace change and foster a passion for identifying opportunity leading to entrepreneurial intention (Fetters, Greene & Rice 2010). Though students' entrepreneurial psychology was embedded in the early work of HEEEs based on Babson's entrepreneurial ecosystem model, scholars seem to have paid a majority of their intention to student start-ups (Johnson, Bock & George 2019). These studies on students' entrepreneurial mindset and intention within HEEEs were based in the UK, Mexico and Italy (Guerrero, Urbano & Gajón 2020; Secundo et al. 2020; Webber, Kitagawa & Plumridge 2020).

Scholars	Theory	Methodology	Context	Factors	Focus
Webber, Kitagawa & Plumridge (2020)	Theory of planned behaviour	Mixed research Online survey Public databases Institutional documentation	UK	Entrepreneurship education, extracurricular activities (volunteering in enterprise activities and enterprise activities in the private sector) and enterprise experiences (short/intensive programs on entrepreneurship and enterprise skills, enterprise advice sessions)	Students' entrepreneurial intention
Guerrero et al. (2020)	Utility- maximising function	Quantitative research Survey	Mexico	Educational programs and incubators	Students' entrepreneurial intention
Secundo et al. (2020)	Quadruple helix	Case study Observations, documents, interviews and survey	Italy	Intentional flows of knowledge: Seminars, case studies, workshops, business idea competition and entrepreneurs in residence. Unintentional flows of knowledge: Business plan stimulation, students aboard, prototype development, business model canvas design, enterprise project and open innovation challenge	Students' mindset

Table 2.6 – HEEEs focusing on students' psychology (by year)

2.4.3 HEEEs for student entrepreneurial development

Higher education for entrepreneurship focuses on personal development, knowledge, mindset, skills and abilities, or setting up a venture and becoming self-employed (QAA 2012). Lackéus (2015) suggests two distinct views for higher education institutions: a narrow view and wide view. As per the narrow view of entrepreneurship, higher education institutions put emphasis on students becoming entrepreneurs through opportunity identification, business development and self-employment (Lackéus 2015; Fayolle & Gailly 2008). Through the wide view of entrepreneurship, higher education institutions give importance to students becoming entrepreneurial via self-development (Lackéus 2015). The view and approach used profoundly affects the outcomes from higher education institutions. Considering the extant literature of HEEE studies such as Longva (2021), Meyer et al. (2020), Shil et al. (2020), Allahar and Sookram (2019), Wright, Siegel and Mustar (2017), Miller and Acs (2017) and Rice, Fetters and Greene (2014), the prominence is on the narrow view of entrepreneurship where higher education institutions strive for student start-ups where the student undertakes the career option of an entrepreneur. Therefore, the primary focus of entrepreneurship scholars has been on students founding a new venture and there is no doubt that this is an important outcome and fruitful area of research (Burton et al. 2016). However, it is worthwhile to consider whether there should be focus on the transition to entrepreneurship including the entrepreneurial development of the students. Students' entrepreneurial development includes mindset, intention and capabilities.

When co-creating HEEEs, higher education institutions can promote entrepreneurship in unlimited ways (Wright, Siegel & Mustar 2017; Morris & Kuratko 2014). One way is higher education institutions promoting student start-ups as a key part of their third mission role (Mason et al. 2020). Another is by fostering entrepreneurial mindset and intention (Webber, Kitagawa & Plumridge 2020). However, HEEEs have the potential not only to produce student start-ups and encourage entrepreneurial intention but also to develop the skill set of undergraduate students driving successful entrepreneurial activities (Chávez-Tellería et al. 2017). Extant literature emphasises the student entrepreneurial behaviour and student start-ups, but little attention has been paid to students' entrepreneurial development. One of the challenges that higher education institutions face is to encourage, support and facilitate student development and, by focusing on student development, the gains could be transformational and widespread (Birch et al. 2017). Scholars such as Maritz et al. (2020) argue that higher education institutions are not exploring the full potential of entrepreneurship education for student development. Therefore, greater understanding of the influence of HEEEs on student development is imperative to produce potential student entrepreneurs with entrepreneurial capabilities (Middleton et al. 2019).

Should higher education institutions focus on improving the transitions to entrepreneurship better within their HEEE? In other words, smoothen the transition from entrepreneurial mindset and intentions to entrepreneurial knowledge and skills and to student start-ups? When relating to the work by Duruflé, Hellmann and Wilson (2018) on student startups, it seems that higher education institutions tend to develop HEEEs through bottom-up initiatives; several HEEEs studied in works by Allahar and Sookram (2019); Liu, Kulturel-Konak and Konak (2021); Longva (2021); Meyer et al. (2020); Miller and Acs (2017); Rice, Fetters and Greene (2014); Shil et al. (2020) and Wright, Siegel and Mustar (2017) are centred on Stage 3 student 'start-ups'. The trend of HEEEs focusing on student start-ups may have been taken on as higher education institutions are strongly associated with economic development and social progress (Mascarenhas et al. 2017). This is natural as higher education institutions in increasing global competition are striving to contribute to economic growth and social development through knowledge transfer and research commercialisation (Wright 2014). More recently, there is an emphasis in HEEE studies on Stage 1 'entrepreneurial mindsets and intentions' through the work of Guerrero, Liñán and Cáceres-Carrasco (2020), Secundo et al. (2020) and Webber, Kitagawa and Plumridge (2020). With reference to the student start-up process, there is limited literature that shows HEEEs have not yet captured Stage 2, which involves students' entrepreneurial development through knowledge and skills (meaning capabilities). Embedding students in HEEEs is an emerging practice (Siegel & Wright 2015; Wright, Siegel & Mustar 2017), with limited empirical investigations in the same research space (Middleton et al. 2019).

Developing students for entrepreneurship relates to behavioural changes of the individual as well as the outcome of the behaviour (Mitchelmore & Rowley 2010). Student entrepreneurial development in terms of knowledge, skills and attitude are understood as capabilities that is essential to initiate and engage in entrepreneurial activity (Bacigalupo et al. 2016). Entrepreneurial activity may occur later in a former student's career, so building the capability in them increases the understanding and the possibility of finding success in these efforts. Alternatively, it is not just about start-ups but also about being entrepreneurial in existing organisations – intrapreneurs leading change, finding new ways to reach customers, innovating on products and services (Alsos et al. 2022). Limited studies focus on the broader

context of student entrepreneurial development including capabilities associated with HEEEs. The need to address this knowledge gap in understanding HEEEs for students' entrepreneurial development represents the missing link.

Student development is understood as the ways in which a student grows, increasing their capabilities as a result of their higher education experience (Rodgers 1990). Theories of student development take into account the nature of learning, including the different contexts in which learning and development occurs, leading to the well-rounded development of a student (Walker 2008). There is a debate on how development is received differently by male and female students (Belenky 1997; Magolda 1992), however this study does not delve into the gendered differences. In the pursuit of student development, the challenge is not to overwhelm students and to pair activities with appropriate support (Felder 2004).

In terms of context, strong entrepreneurial ecosystems are found in countries such as the US (Bauman & Lucy 2021; Regele & Neck 2012). Different to developed countries, cocreating HEEEs in a resource-constrained environment is a critical challenge (Bedő, Erdős & Pittaway 2020). Benchmarking HEEE factors found in the successful HEEEs of such countries is likely to raise contradictions in resource-scarce environments (Bedő, Erdős & Pittaway 2020). One of the first studies that specifically identified HEEEs in resource-constrained environments was by Bedő, Erdős and Pittaway (2020). Based on the literature, their study suggested a conceptual framework for student start-ups using Stam's (2015) entrepreneurial ecosystem. However, little is known about how higher education institutions can co-create HEEEs in resource-scarce environments. There is limited empirical evidence about how an HEEE can facilitate student entrepreneurial development in a resource-constrained environment.

2.5 Composition of HEEEs

The composition of HEEEs refers to their key ingredients. According to extant literature, there are a few empirical studies on HEEE models that make important contributions addressing the composition of HEEEs. To provide insight into the chosen topic of HEEEs and the wider view of literature, the following is organised into two sections: factors of HEEEs and actors of HEEEs.

2.5.1 Factors of higher education entrepreneurial ecosystems

Given the notion of HEEEs, 'factors' are a key element of their composition. Studies suggest that there is no definite set of factors that enable higher education institutions to support students' entrepreneurial activity (Algieri, Aquino & Succurro 2013; Rice, Fetters & Greene 2014). Among extant literature, there are individual-level factors, institutional-level factors and context-related factors. Previous studies have investigated students' prior entrepreneurial experience (Webber, Kitagawa & Plumridge 2020) and parental influences (Bock et al. 2020; Webber, Kitagawa & Plumridge 2020) as individual factors, which were found to positively influence students in HEEEs. Individual-level factors are critical, including students' attitudes towards entrepreneurship, and these factors are influenced by contextual factors such as entrepreneurial activities in HEEEs by Webber, Kitagawa and Plumridge (2020), and in contrast to studies focusing on individual and institutional factors. As HEEEs are evolving, there is a need to understand the contextual factors that characterise HEEEs and the role that these factors play for theoretical development (Longva 2021; Wright, Siegel & Mustar 2017).

One of the earliest references to HEEEs is found in the scholarly work by Fetters, Greene and Rice (2010) that discusses pathways to developing HEEEs with wide-ranging missions and resources in different contexts. The same study extended into a series of case studies of six higher education institutions in North America, Latin America, Europe and Asia that led to seven key success factors supporting entrepreneurship development through entrepreneurship education and entrepreneurial support (Rice, Fetters & Greene 2014). These seven factors are (1) senior leadership vision, engagement and sponsorship; (2) strong programmatic and faculty leadership; (3) sustained commitment over a long period; (4) commitment of substantial financial resources; (5) commitment to continuing innovation in curriculum and programs; (6) an appropriate organisational infrastructure; and (7) commitment to building the extended enterprise and achieving critical mass. Since this work, an increasing number of studies have examined higher education institutions and categorised the factors that form a comprehensive and highly evolved HEEE, with contextual relevance (Stam & Spigel 2016).

Despite growing scholarly work, the literature overlooks understanding and knowledge for those who seek to promote entrepreneurship in resource-constrained environments (Roundy 2017). Resource-constrained environments may lack factors and interactions between these that are inevitable yet important in ecosystems (Borissenko & Boschma 2017). In contrast to developed countries, co-creating HEEEs in a resource-constrained environment is inherently challenging (Bedő, Erdős & Pittaway 2020). However, countries with resource constraints are no different to other developed countries as developing countries can give a high priority to entrepreneurship but may lack the resources for it (Roundy 2017). Despite the relevance of students' knowledge, skills and attitudes to pursue entrepreneurship, not enough empirical studies have explored the role of contextual factors on student entrepreneurial development in a resource-constrained environment.

Whether entrepreneurship can be taught in a classroom is a common question (Neck & Greene 2011; Henry et al. 2004). However, scholars argue that entrepreneurship should be regarded as any other discipline that can be taught and learned (Neck & Greene 2011; Sarasvathy 2004; Holmgren & From 2005; Drucker 1985). Collins et al. (2004) reveal that higher education institutions do not provide students the entrepreneurial context or environment to groom and develop their entrepreneurial ideas. Mayhew et al. (2017) argued whether higher education institutions can influence student development, specifically innovative capacities, through experiences during higher education. In response, Bock et al. (2020) found that functional experiences (such as resources and physical spaces) and connecting experiences (such as experimental studies, field trials, design projects or internships) have a significant impact on students' innovation capabilities. However, this recent study is more focused on entrepreneurship education exploring student development through a brief list of four aggregated factors. Further, informal factors (such as role models) have a greater influence on students compared to formal factors (such as education and training and support measures) (Guerrero, Urbano & Fayolle 2016).

The empirical study by Bock et al. (2020) found two HEEE factors that are responsible for improving students' innovation capacities at an aggregated level. Scholarly work is criticised for suggesting factors rather than providing explanations of mechanisms on how they work (Longva 2021). This calls for attention to the breakdown of broad HEEE factors investigated as combinations and further exploration on 'how' and 'why' factors work in resource-constrained environments. Formal and informal mechanisms between higher education institutions and stakeholders from wider entrepreneurial ecosystems facilitate teamwork and access to resources (Etzkowitz & Klofsten 2005). These mechanisms of higher education institutions engaged in the third mission can include collaborating with the local environment and anchoring to organisations building a network to exchange tacit knowledge (Bramwell & Wolfe 2008). A holistic ecosystem model that offers support mechanisms along with relevant stakeholders for students leads to access of varied resources and broader social networks (Fuster et al. 2018).

In addition to empirical research, scholars have contributed to HEEE literature by synthesising extant literature as reviews. There are two systematic reviews and one integrative review related to HEEEs from recent years, by O'Brien, Cooney and Blenker (2019), Hayter et al. (2018) and Belitski (2017). In these previous reviews, scholars have concluded with:

- six considerations for expanding university entrepreneurial ecosystems for underrepresented communities (O'Brien, Cooney & Blenker 2019)
- eight independent factors and three dependent outcomes beyond the individual level (Hayter et al. 2018)
- four university-based entrepreneurship ecosystem enablers for knowledge commercialisation and engagement (Belitski 2017).

While these reviews focused on the factors of HEEEs, such as teaching and learning, outreach, culture, resources, stakeholders, networks and infrastructure (O'Brien, Cooney & Blenker 2019; Hayter et al. 2018; Belitski 2017), perhaps they have not fully contributed to HEEE composition.

2.5.2 Actors of higher education entrepreneurial ecosystems

Like any biological ecosystem, an HEEE is a multi-stakeholder environment including a complex network of individuals, groups and organisations that support entrepreneurship education and the creation of start-ups and growth of ventures (Fetters, Greene & Rice 2010; Rice, Fetters & Greene 2014). Stakeholders are the human and social actors of ecosystems that create engagement and dynamism in an environment (Johnson et al. 2019). The successful collaboration of diverse stakeholders attracts a combination of perspectives, experiences, competencies and mindsets that contribute to E&I within the ecosystem (Secundo et al. 2020). Higher education institutions creating shared value with stakeholders have been studied previously in other disciplines but not extensively within an entrepreneurial ecosystem setting for higher education (Karwowska 2019; Bischoff et al. 2018). Extant literature shows a paucity of entrepreneurship education studies that have focused on stakeholders and applied stakeholder or network theories (Galvão et al. 2020). Even fewer studies have explored the interplay of stakeholders in relation to factors and the need to explore how stakeholders can support and collaborate within HEEEs (Longva 2021).

As every HEEE is distinct, each ecosystem consists of a set of actors and factors unique to the higher education institution. Studies by Fetters, Greene and Rice (2010) and Rice, Fetters and Greene (2014) identify a combination of internal and external stakeholders evident among HEEEs of developed economies. Miller and Acs (2017) identify these internal actors as faculty/adjuncts, students, alumni, mentors and external stakeholders as corporations, angel investors / venture capitalists, start-up founders, government and incubators/accelerators. In a recent study, actors of a respective HEEE were classified in four role based domains as Leaders and Governors, Educators, Innovators and Connectors (Chaipongpati, Thawesaengskulthai and Koiwanit 2022). For a broader understanding of stakeholders in the entrepreneurial ecosystem, Isenberg's (2010) domains of the entrepreneurship ecosystem is a suitable framework (refer Figure 2.7). Outside the higher education institution and their HEEE are stakeholders representing the different domains – supports, culture, finance, policy and markets.



Figure 2.7 – Domains of the entrepreneurial ecosystem

Source – Isenberg (2010)

Stakeholder support and collaboration contributes to a strong sustainable entrepreneurial ecosystem (Bischoff 2019). Extant literature emphasises how engaging with

various external actors shapes the development of HEEEs and the significance of managing these external actors to increase entrepreneurial activity (Link & Sarala 2019; Alvedalen & Boschma 2017). For instance, Babson College (in the US) receives external funding for its ecosystem operations, secured from various sources (Hancock 2011). Entrepreneurs and corporates are the most common external stakeholders engaged in sharing the practical essence of entrepreneurship and complementing academic view (Bischoff et al. 2018). However, the role of stakeholders evolves as the ecosystem develops (Colombelli et al. 2019).

2.6 Stakeholders in HEEEs

2.6.1 Relationship building with ecosystem stakeholders

When advancing from the traditional functions of teaching and researching into the role of innovation, higher education institutions took on the triple helix of university–industry– government relations (Etzkowitz & Leydesdorff 2020). A higher education institution becomes a relationship builder that creates a configuration of stakeholders through relationships and these relationships are likely to change throughout the university life cycle (Redford & Fayolle 2014). Different stakeholders within the institution such as management, faculty, students and external stakeholders at local, regional and national levels share synergies in the ecosystem. External stakeholders have grown from industry and government to include more from the entrepreneurial ecosystem. These relationships are networks of various stakeholders from the institution and their external domain (Beliski & Heron 2016). While stable relationships can be critical in the flow of entrepreneurship education (Bischoff et al. 2018), creating a balance in synergies between these stakeholders is complex (Leydesdorff 2000).

Levels of stakeholder involvement and interaction vary from high to low among higher education institutions and stakeholder groups (Bischoff et al. 2018). Perceptions and interests held by stakeholders influence their involvement and contribution to the process of entrepreneurship in higher education (Matlay 2009). Therefore, the involvement of stakeholders must be mutually beneficial and self-sustaining, where they work together with shared efforts in stimulating entrepreneurial activity (Wadee & Padayachee 2017). Such involvement can be geographically constrained, creating a boundary and making it difficult for stakeholders to effectively engage (Acs et al. 2014).

Higher education institutions that go beyond start-ups require stakeholder engagement in different entrepreneurial initiatives (Clauss, Moussa & Kesting 2018). This also calls for recognising that engaging stakeholders is more important than establishing entrepreneurial support for new venture creation (Gibb & Hannon 2006). Students intending to become entrepreneurs or alumni creating a start-up value practical knowledge and are influenced by stakeholders in the external environment (Hayter et al. 2017). The interactions and interconnectedness among various stakeholders may result in a truly entrepreneurial learning experience for students and highlights the dynamic nature of entrepreneurial activity within the ecosystem (Wright, Siegel & Mustar 2017). Entrepreneurs are one of the key stakeholders in HEEEs and are recognised for support in identifying opportunity, offering confidence about the business idea and creating a start-up (Spigel & Harrison 2017). Alumni who become entrepreneurs are known to serve higher education institutions as mentors, investors or donors, and they are also involved in teaching as visiting faculty (Powell & Walsh 2018). This draws on the importance of identifying relevant stakeholders within the HEEE, including external stakeholders from the wider entrepreneurial ecosystem (Wright, Siegel & Mustar 2017).

2.6.2 Stakeholder engagement in HEEEs

Stakeholders can interconnect and interact by collaborating in various factors of an HEEE to promote entrepreneurship. Despite stakeholder management not being a new concept, understanding the variety of stakeholders, their roles, and possibilities for collaboration is critical in developing a well-connected and effective HEEE (Brush 2014). When managing stakeholder engagement in HEEEs, higher education institutions need to be mindful that it is a progressive process that can be planned and phased (Redford & Fayolle 2014). By embedding key stakeholders into the HEEE and engaging them effectively, higher education institutions can promote entrepreneurial activity within their HEEE and contribute to their region and country. HEEE is a collective action and coordinated collaboration between higher education institutions and other stakeholders (Wright, Siegel & Mustar 2017). Similar to entrepreneurship education literature (Bischoff et al. 2018), stakeholder engagement has received less attention in the context of HEEEs.

Higher education institutions and other stakeholders need to invest in promoting entrepreneurship (Prencipe et al. 2020), as well as enhancing the development of HEEEs that nurture the entrepreneurial potential of students (Guerrero et al. 2014). When deciding on various stakeholders, it is significant to determine the importance and prospective contribution of each stakeholder (Redford & Fayolle 2014). Even though literature showcases stakeholders with different interests involved in facilitating entrepreneurship (Galvão et al. 2020), stakeholders keen on collaborating with students and with the higher education institution are a distinct group. Existence and interrelationships of heterogenous stakeholders who share the same goals are essential, as they provide or become non-academic contacts that might not exist otherwise (Hayter 2016).

The core of an entrepreneurial ecosystem is formed by its stakeholders and is characterised by the interactions between their entrepreneurial aspirations, attitudes and abilities (Ács, Autio & Szerb 2014). An entrepreneurial ecosystem takes the form of a dynamic structure that is social and evolves through social networks (Spigel 2017). Entrepreneurs with strong social networks are in a more favourable position to acquire resources than entrepreneurs who are not active in their entrepreneurial ecosystem (Spigel & Harrison 2018). The social element involves resources and these are acquired through social networks. The flow of resources takes place through formal and informal social networks and is significant for the development and emergence of entrepreneurial ecosystems (Cohen 2006). Emerging ecosystems are characterised by sparse social networks. In their growth phase, ecosystems have developed social networks and successful ecosystems are resource rich with dense social networks (Mack & Mayer 2016; Spigel & Harrison 2018).

In a successful ecosystem, relationships can be internal to internal (faculty and student), external to internal (entrepreneur and student) and external to external (alumni to investor) (Powell & Walsh 2018). The dynamics of the environment affect the ecosystem and their outcomes; however, the involvement of stakeholders significantly impacts each stakeholder and the factors of the ecosystem (Godley et al. 2019). Formal and informal connections between ecosystem stakeholders in social networks can contribute to the development of HEEEs and enhance their access to resources. HEEEs are formed with different stakeholders, including the internal community of faculty, staff, students and alumni, and the local community, each with different priorities (Huang-Saad, Duval-Couetil & Park 2018). Further, attention should be paid to how diverse stakeholders can engage within the factors of HEEEs and influence students' entrepreneurial development in a resource-constrained environment.

2.7 Chapter summary

This chapter analysed literature that showed HEEEs have a lineage from entrepreneurial ecosystems and have been developing since their emergence. While definitions for HEEEs are being posited, the prominence is on 'actors' (i.e. stakeholders) of the ecosystem. When comparing this to a common definition of entrepreneurial ecosystems by Stam (2015), it appears that a more complete definition of the HEEE concept would include 'factors and actors'. Further, academic research on HEEEs has grown in a somewhat fragmented nature,

with studies focusing on entrepreneurship education ecosystems and entrepreneurial support. This research will explore the HEEE concept in terms of its composition (i.e. factors and actors) and their linkage to the wider entrepreneurial ecosystem, taking a holistic approach.

The accumulation of HEEE studies in Table 2.5 and 2.6 presents the key scholars, applied theories, methodology, context investigated and findings. More HEEE studies have examined the composition leading to student start-ups and recent HEEEs are working towards student entrepreneurial intention and mindset. With entrepreneurship education and entrepreneurial support included, HEEEs can contribute to students' entrepreneurial development. There is scholarly work on student entrepreneurial development in entrepreneurship education ecosystems through the work by Bock et al. (2020) focused on innovation capabilities. However, there is limited scholarly work found investigating the HEEE factors that could lead to students' entrepreneurial development. Thus, this study investigates HEEEs focused on students' entrepreneurial development, in particular E&I capabilities that emerge from the data.
CHAPTER 3: CONTEXTUALISATION – SRI LANKA

3.1 Chapter overview

During a study, contextualising may occur in the research problem, design, methodology, site, analysis or interpretation (Rousseau & Fried 2001). Contextualisation is central to entrepreneurship research (Welter 2011), with the understanding that ecosystems are unique in the geographic location they are embedded in (Acs et al. 2014). Under this direction, the following chapter focuses on the geographic context for this research (refer Figure 3.1).

According to Walmsley (2018), this is the 'Asian Century', with the prediction of Asian countries to rise. Sri Lanka is one such emerging South Asian country optimistic about harnessing entrepreneurship among its youth through quality education (Ministry of Sustainable Development 2018). The chapter begins with an overview of the socio-economic conditions of Sri Lanka. The bulk of the chapter frames Sri Lanka as a resource-constrained environment through the importance of quality education and entrepreneurship for youth, the country's progress in sustainable development, and challenges faced by the higher education sector. The chapter closes with an overview of entrepreneurship education, entrepreneurial ecosystems, start-up entrepreneurs and what drives them in Sri Lanka. As a country, Sri Lanka does not actively engage and invest in research (de Silva, Uyarra & Oakey 2012) and this chapter is written with the latest available information from public and open sources.



Figure 3.1 – Chapter 3 outline

3.2 Macro-environment outlook of Sri Lanka

Context including conditions and circumstances are not only relevant to a study but contextualising the study also contributes to a deeper level of meaning (Maritz & Brown 2013). Context is significant in this study where there is no ideal HEEE to benchmark entrepreneurship education and entrepreneurial support. In this case, differences in socio-economic conditions and circumstances of countries matter as they influence how entrepreneurs start and run a new venture (Karlsson et al. 2019; Welter et al. 2017). Below is an overview of how Sri Lanka is positioned in terms of history and development, people, culture and communication, economic growth, school education system and the higher education system.

3.2.1 History and development

Sri Lanka is an island nation of 64,630 km² of land located in the Indian Ocean with a geostrategic position (United Nations 2021). Sri Lanka plays a crucial role in the Indian Ocean region and has become important to countries like the US, China, India and Japan due to its naval link between West Asia and South-East Asia. A Dutch, then British, colony and originally named Ceylon, the nation despite is small size has a rich biodiversity with a range of ecosystems ranging from rainforests to grasslands, rivers, wetlands and freshwater bodies and coastal and marine ecosystems (United Nations 2021). Gaining independence in 1948, the nation had limited economic growth as development and industrialisation were limited, Sri Lanka being a colony focused on primary products. A long civil war (1983–2009) further stalled development, and since then the nation has emerged post-war to grow into a developing country. Having successfully achieved the Millennium Development Goals in 2015, Sri Lanka reaffirmed its commitment to the SDGs by aligning national policies and strategies for a country enriched by 2025 (Ministry of Sustainable Development 2018).

3.2.2 People, culture and communication

Known as the pearl of the Indian Ocean, in close proximity to India and Maldives, Sri Lanka is an island home to a population of 23,044,123 people as of July 2021 (refer Figure 3.2) (CIA Online). National censuses show that the people of Sri Lanka belong to four major ethnic communities, being Sinhalese (74.9%), Sri Lankan Tamil (11.2%), Moors (9.2%) and Indian Tamil (4.2%). Sri Lanka is a multi-religious and multicultural country with 70.2% Buddhist, 12.6% Hindu, 9.7% Muslim, 6.1% Roman Catholic and 1.3% Christian (CIA Online). Sri Lankan people primarily speak the national language 'Sinhala', and a smaller proportion

communicate in Tamil as the second national language (Sittarage 2018; Parliament Secretariat 2015). English proficiency is 22% among those over 15 years of age and improving as the government strives to build an English-literate Sri Lanka (Institute of Policy Studies of Sri Lanka 2014). English, the link language, is common among the urban population living in Colombo, where 32.9% speak English, and 34.1% can read and write in English (Department of Census and Statistics 2019; Parliament Secretariat 2015).



Figure 3.2 – Population in Sri Lanka

Source – World Bank (2020)

3.2.3 Economic growth

After the end of a 30-year civil war in 2009, Sri Lanka showed signs of promising economic development, with an average gross domestic product (GDP) growth of 5.8% during 2010–2017 (World Bank 2020). In 2018, Sri Lanka was recognised for climbing to upper-middle income status with a GDP per capita of USD4,102 and earning the highest 57.11 points on the global competitiveness report (World Economic Forum 2018). Sri Lanka's economy is mainly based on the service and industrial sectors, and agriculture and the services sector contributed to growth by 3.6% in 2017 and 4.7% in 2018 (PricewaterhouseCoopers 2019). The following year, the country fell back to being a lower-middle income country with a GDP per capita of USD3,852 in 2019 (World Bank 2020).

Sri Lanka is experiencing a multidimensional crisis followed by the terrorist attacks, the COVID-19 global pandemic and political and economic crisis (World Bank 2022). The economy continued to show weakening signs even before the COVID-19 pandemic and growth averaged only 3.1% between 2017 and 2019 (World Bank 2021). Concurrent challenges in finance, debt, food, fuel, and medication have disrupted the economy and livelihood in Sri Lanka (United Nations Sri Lanka 2022). The economic outlook for Sri Lanka is highly uncertain due to the fiscal and external imbalances (World Bank 2022).

The economy contracted by 3.6% in 2020 (refer Figure 3.3) due to the impact from the pandemic and as the country struggled to control the spread of the virus in 2021. As such, Sri Lanka is now considered a lower-middle income country with a GDP per capita of USD3,682 as at 2020, having declined from USD4,108 in 2017 (World Bank 2020). Sri Lanka's GDP is similar to other developing countries in South Asia, including India, Pakistan, Bangladesh and Bhutan (refer Figure 3.4). Within Asia, Sri Lanka is an emerging economy like Cambodia, Indonesia, Myanmar, Philippines and Vietnam (refer Figure 3.5).



Figure 3.3 – Economic activity (GDP as a fraction of per capita) in Sri Lanka

Source - World Bank (2020)



Figure 3.4 – Economic activity comparison of emerging economies in Asia

Source - World Bank (2020)



Figure 3.5 – Economic activity comparison of developing countries in South Asia Source – World Bank (2020)

3.2.4 School education system

Education in Sri Lanka operates on a 'free education bill' where every child above five years old until university level is entitled to free education (Ministry of Sustainable Development 2018). After the civil war ended in May 2009, the Sri Lankan government focused on improving education aimed at human development. However, these education reforms are criticised for being political responses in a patronage political environment to contain voters' political interest (Little 2011). For many years, any Sri Lankan child could drop out of school at 14 years of age without completing secondary education. This choice was amended recently, making students continue school up to the 'ordinary' level, earning a secondary education certificate (Ministry of Education 2016).

Over the last decades, the public education system has suffered from a lack of funding, resources, facilities and infrastructure, yet young students have showcased their talents by winning internationally recognised awards and medals (Ministry of Education 2016). In 2018, Sri Lanka was recognised by the United Nations as an over-achiever for successfully developing its primary education sector by satisfying all targets, including literacy rates, school enrolments and education completions (United Nations 2020). This achievement was noted for how Sri Lanka stood tall in South Asia and middle-income and high-income countries. There were 10,175 public schools with 4,214,772 students and 98 private schools with 125,669 students as of 2018. The government-funded public schools mainly supplied the demand for primary and secondary education (Ministry of Education 2016).

3.2.5 Higher education structure and sector

The Universities Grants Commission (2019) is the apex body of the higher education system in Sri Lanka, with the responsibility of facilitating the provision of undergraduate, postgraduate and professional education through an integrated system of state universities and higher education institutions. The state sector (also known as the public sector) comprises 36 entities consisting of 16 universities, 17 institutes and three campuses, offering 225 courses of study funded by the government (Universities Grants Commission 2019). Among these 16 universities, the University of Peradeniya has the longest historical roots, and the University of Sri Jayewardnepura has the largest student population (Weerasinghe & Jayawardane 2018).

In addition to the UCG governed public universities, there are non-UCG governed public universities, degree granting institutions, professional institutions and vocational training centres. Private higher education institutions are established and emerging in support of the government policy (Weerasinghe, Jayawardane & Deshani 2016) and some of these institutions are diversified businesses of local large companies. Almost all higher education institutions including in the public sector predominately focus on teaching and academics focus on a singular role (Weerasinghe, Jayawardane & Deshani 2016; de Silva et al. 2012), while the third mission of research and entrepreneurship are functions still emerging.

Due to the free education policy, there is a high degree of dependence on government resources such as funding (de Silva, Uyarra & Oakey 2012). Public universities receive limited funding due to numerous other priorities such as poverty (de Silva, Uyarra & Oakey 2012). Out of 250,000 students who take the advance-level examinations every year, 150,000 qualified for higher education, 92,379 applied for university admission for the academic year 2019/2020 and public universities can only accept 45,000 students (Sri Lanka Export Development Board 2022). The demand for higher education is beyond the supply public higher education can serve, creating the need for private higher education institutions in support.

The last record of the private higher education sector was 47 degree granting institutions across the country (Gamage & Wijesooriya 2011). These private higher education institutions are concentrated mainly in the Western Province including Colombo with a few located in Kandy of the Central Provinance (The World Bank 2019). These private higher education institutions are companies established under the Company Law of Sri Lanka with local or international ownership and funding (Wickramasinghe 2018). The private higher education institutions offer higher education as a service in affiliation with international universities.

Unlike public sector education, these companies are not fully regulated under the Ministry of Higher Education and University of Grants Commission (Wickramasinghe 2018). For example, India enacted the Establishment of and Maintenance of Standards in Private Universities Act in 2003, allowing the University of Grants Commission to regulate their private higher education sector. This standardisation brought in order, quality and relevance while increasing private higher education supply in India (Gamage 2012). Private higher education institutions in Sri Lanka operate like any other private company, with no local regulation or guidance relating to higher education. Due to their independence, these institutions may benefit from this research to become entrepreneurial, evolve their HEEEs to develop student capabilities in entrepreneurship and innovation, and contribute to the national priority on entrepreneurship.

3.3 Sri Lanka's progress in sustainable development

Sri Lanka is one of the 193 United Nations member states committed to the 2030 agenda on sustainable development. The Sri Lankan government follows an institutional plan themed as 'Vision 2025' and aligned to respective SDGs. The plan is governed by the Sustainable Development Act, enacted in 2017 and a Sustainable Development Council to implement the Act (Ministry of Sustainable Development 2018). A national review in 2018 recognised that Sri Lanka had begun progress towards a sustainable and resilient society (United Nations 2020). The most recent update related to the goal for 'quality education' signals as 'on track or maintaining achievement' according to the Sustainable Development Goals Dashboard (2022). However, this is an assessment of only 3 criteria that excludes target 4.4 focused on increasing the youth with relevant skills for entrepreneurship.

The country assessment of gaps towards sustainable development highlighted a lack of entrepreneurship hindering local economic growth and limiting access to international markets (Ministry of Sustainable Development 2018). In terms of work, disengaged Sri Lankan youth who are not in employment, education or training yet actively seeking employment opportunities grew from 24% to 26% (Ministry of Sustainable Development 2018). To address these challenges, the government aims to promote entrepreneurship within the private sectors (Ministry of Sustainable Development 2018). Yet there is no active action plan that is facilitating the promotion of entrepreneurship in the ecosystem including higher education institutions. Further, the Sri Lankan higher education system face challenges relating to quality, relevance and capacity, urging youth to move overseas for better education (Dissanayake

2020). Therefore, there is urge for higher education institutions to improve the experience to develop the next generation of young entrepreneurs, contributing to Sri Lanka's national sustainable development.

3.4 Importance of quality education and entrepreneurship for youth

By region, South Asia is reported to suffer from the highest rate of unemployed youth aged between 15 and 24 years with 18.7% compared to 10.0% in Eastern Asia and 18.7% in South-East Asia countries and the Pacific in 2020 (International Labour Organization 2021). Amidst this crisis, South Asia is projected to have the world's largest youth labour force by 2040, potentially driving economies and societies for national development (UNICEF 2020). Within South Asia, Sri Lanka faces the challenge of the highest youth unemployment of 27.7% (Jayathilake 2020) compared to countries such as India 21.1%, Nepal 19.2% and Maldives 7.6% (International Labour Organization 2020).

Sri Lanka reports overall unemployment rates below 5% in recent years; however, the current downturn in economic activity due to the pandemic is predicted to cause job losses (World Bank 2020). Although the overall unemployment is commendable, youth unemployment has steadily increased over the last few years, from 16.7% in 2011 to 20.8% in 2019 (World Bank 2021). The youth unemployment rate is approximately four times the overall unemployment rate. The proportion of youth disengaged and not in employment, education or training of the total youth population amounts to 26% and is relatively high (Ministry of Sustainable Development 2018). High youth unemployment signifies underutilised human capital, which if employed, could improve economic growth and social development for Sri Lanka (Jayathilake 2020). The high level of youth unemployment has seen an increase in entrepreneurship and start-up activity (Wong et al. 2005).

UNICEF (2020) identified one major obstacle to increasing youth employment as low quality education is widening the skill gap. Sri Lanka's reported high youth unemployment implies a talent mismatch where youth do not participate in the labour market (Ministry of Sustainable Development 2018). Employers claim that they struggle to hire the talent they require as graduates lack the necessary skills for the job (Ministry of Sustainable Development 2018). Young unemployed graduates had the same perception, stating mismatch of knowledge and skills gained from academic qualifications and employer requirements as the second reason for being unemployed (Jayathilake 2020). However, most graduates believed that monetary and non-monetary rewards of available jobs did not match their expectations (Jayathilake 2020). These reasons were confirmed by a different study exploring youth unemployment that recognised one's career aspirations as a motivating factor and insufficient education as a barrier (Dissanayake 2020). Therefore, high youth unemployment and significant underutilisation of youth human capital in affected countries can be alleviated through investment in education and skill development (UNICEF 2020).

The Sri Lankan government had made efforts to address youth unemployment with initiatives at all education levels. For example, it incorporated skill development into the school curricula, introduced a technology stream of study, developed technology faculties at public universities, and encouraged youth to follow vocational training (Ministry of Sustainable Development 2018). 'Vision 25' and 'The Decent Work Country Program' are committed to increasing employment opportunities and re-skilling the labour force to meet socio-economic demands, with a focus on youth (International Labour Organization 2018). However, recent studies investigating unemployed youth in Sri Lanka found that interviewees are unaware of any government support to improve their employability skills and entrepreneurial opportunities (Jayathilake 2020; Dissanayake 2020). Youth with self-employment aspirations have the vision to start and grow a new venture, but they recognise that they lack the knowledge and skills to achieve it successfully (Dissanayake 2020).

3.5 Challenges faced by the Sri Lankan higher education sector

If an investment is made in skill development through education, South Asian countries can build on strong economic growth in the coming decades (UNICEF 2020). The Sri Lankan higher education sector is currently facing several barriers. Some of the key challenges the industry is facing include improve the quality and relevance of education and increase access to higher education (Ministry of Sustainable Development 2018) and they are discussed below. The poor education quality for youth in Sri Lanka, has resulted in a skill gap for 21st-century work.

3.5.1 Lack of quality and relevance

The government recognises the pivotal role that higher education plays in driving economic growth and social development for the nation (Ministry of Sustainable Development 2018). Higher education quality requires improvement, particularly increasing entrepreneurship and employability among youth (Ministry of Sustainable Development 2018). Next, the relevance of education calls for study programs and graduate capabilities to be aligned to national priorities, market needs and employer expectations (Ministry of Sustainable Development

2018). Given the national emphasis for encouraging entrepreneurship and increasing entrepreneurs (Ministry of Sustainable Development 2018), higher education institutions, both public and private, need to reform their education and environments for quality and relevance.

3.5.2 Capacity constraints in higher education

Access to higher education is another challenge for students in Sri Lanka (University Grants Commission 2022). The concern rises from statistics that reveal that only 18% of the 20- to 24-year-old population are enrolled in universities, campuses and institutes of the public higher education sector (Ministry of Sustainable Development 2018). The total tertiary education enrolments of Sri Lanka stand at 21%, and two reasons are identified as the cause of this low number of enrolments and limited access to higher education services (Ministry of Sustainable Development 2018). First, the public sector offers free education and this sector suffers from capacity constraints due to lack of funding and resources. The second reason is limited coordination with the private sector for involvement in the higher education system (Ministry of Sustainable Development 2018). Comparatively, countries such as China enforce a policy that allows public universities to charge fees to cover up to 25% of operating costs, which had a tripling effect on public university enrolments, increasing them from 6.1 million to 19.0 million by 2003 within six years (Gamage 2012).

Approximately 250,000 students take the advance-level examinations every year, and 150,000 of them qualify for higher education (University Grants Commission 2022). Due to the capacity constraints in the public system, more than 47,000 students did not have the opportunity to enrol in public higher education in the academic year of 2019/2020. Nearly 11,000 of these 47,379 students enrolled with private higher education institutions for paid higher education, and 12,000 left for undergraduate education in other countries (Sri Lanka Export Development Board 2018). The public higher education sector enrols a higher number of students and produces a larger number of graduates than the private higher education institutions. However, the public higher education system is constrained in resources, limiting their capacity of service (Ministry of Sustainable Development 2018). Given the capacity constraint faced by public universities, a significant portion of students opt for private higher education.

3.5.3 Demand for private higher education

After decades of the widening gap between higher education demand and supply in Sri Lanka, opportunities to fill the gap have emerged. On the demand side, parents refuse to disadvantage

their child(ren) due to limited free education opportunities and they are willing to pay for higher education creating a demand for private higher education in Sri Lanka (Gamage 2012). This demand is driven by educated parents who influence their children to obtain higher education after completing their secondary education (Dissanayake 2020). For developing countries such as Sri Lanka, private investment in higher education and tuition fees have become imperative (Gamage 2012). This change of preference towards paid higher education increases overall access to higher education while emphasising a growing demand for private higher education.

In Sri Lanka, all private higher education institutions offer internationally awarded undergraduate degrees (Gamage & Wijesooriya 2011). The top 3 popular fields of study are Business/Management, Information Technology and Engineering with Business/Management being the majority choice for higher education (Gamage & Wijesooriya 2011). Sri Lanka is behind the rest of the world in terms of private higher education while the institutions have expanded rapidly globally (The World Bank 2019).

3.5.4 Government policy interventions

Government support may improve entrepreneurship education's efficiency and effectiveness, especially in developing countries (Yu et al. 2017). A study among undergraduates in Sri Lanka found that higher education institution support positively influences student entrepreneurial intentions (Lin et al. 2013). By making entrepreneurship education available to youth, higher education institutions can prepare a new wave of graduate entrepreneurs for socio-economic impact (Volkmann et al. 2009). This explains why entrepreneurship education has received growing recognition, especially among developing countries (Nabi & Linan 2011).

In Asian countries such as China, Taiwan and Singapore, governments have actively enforced policies to accelerate entrepreneurship education (Yu 2018; Yu et al. 2017). For example, the Chinese government and political system have strongly influenced entrepreneurship education. The Chinese government has launched entrepreneurship programs at almost every university, and universities have access to financial support from the government (Yu 2018). Although this practice is common among many Asian nations (Yu et al. 2017), it has not attracted the attention of policymakers in terms of interventions related to the private higher education in Sri Lanka. While public universities play a more active role by offering entrepreneurship education and supporting entrepreneurial initiatives, private higher education institutions may want to take some responsibility to scale up and align their HEEEs to national priorities, including youth entrepreneurship.

3.5.5 Human capital development

The COVID-19 pandemic posed significant challenges to human capital worldwide (Ratten and Jones 2020). Recovery from the pandemic urges a stronger commitment in human capital development which focuses on capabilities and systems to rebuild from the shocks (United Nations Sri Lanka 2022). After the pandemic reversed the human capital development gains, Sri Lanka is experiencing a political and economic crisis (United Nations Sri Lanka 2022). Human capital challenges are exacerbated by the existing lack of quality education and capability development (Ministry of Sustainable Development 2018). Sri Lanka ranks 71st among 174 countries in the 2020 World Bank Human Capital Index as the highest in South Asia. Yet there is significant gap within higher education where Sri Lanka's competitiveness will be affected by the education and capabilities of the workforce in the 21st century (International Finance Corporation 2022). As individual interventions can only go so far, the entrepreneurial ecosystems including higher education institutions need to focus on the human capital challenges (Maritz et al. 2020). As the developing economy must rebuild and reshape their futures, human capital development could support a resiliency recovery and growth opportunities lies in leveraging its human capital (United Nations Sri Lanka 2022). The multiple crisis in the last three years changes many dynamics, however, the study acknowledges the situation and continues to be relevant for Sri Lanka to recover and rebuild. Higher education and their HEEEs are increasingly important for human capital development of youth.

3.6 Entrepreneurship education in Sri Lanka

The trajectory of entrepreneurship education began in the school system when the subject was newly introduced to the Sri Lanka secondary curriculum (Weerasinghe 2020). Entrepreneurship education has a positive influence on the entrepreneurial intention of school children and undergraduate students (Weerasinghe 2020). When explored among public universities, knowledge about entrepreneurship was found to have a strong effect on students' entrepreneurial intentions (Gunawaredane & Weerasinghe 2021).

Among the four public universities with the longest history in Sri Lanka, the University of Sri Jayewardenepura offers a BSc in Entrepreneurship while the University of Moratuwa promotes a MBA in Entrepreneurship and Innovation. Whereas the University of Peradeniya and the University of Kelaniya do not seem to offer any entrepreneurship education programs. This limited availability of entrepreneurship education highlights the disconnect between national needs and higher education offerings. Similarly, entrepreneurship education in the private sector is still emerging with only a few institutions offering first degrees in entrepreneurship. The American National College, affiliated to Northwood University (US), is one of the first private higher education institutions to offer a BA in Entrepreneurship while the Asia Pacific Institute of Information Technology recently launched a BSc in Entrepreneurship and Innovation early 2022, which is awarded by Staffordshire University (UK). Although Business/Management is the most common discipline of undergraduate study, entrepreneurship is yet to become popular.

Sri Lanka as a nation strives towards the SGDs, including Target 4.4 which focuses on increasing the number of youths with relevant skills for entrepreneurship (Ministry of Sustainable Development 2018), and there is evidence that entrepreneurship education positively influences the entrepreneurial intention of undergraduate students (Weerasinghe 2020; Lin et al. 2013). An experimental study in Sri Lanka found that offering entrepreneurship education can add value in multiple ways, including the development of relevant capabilities at undergraduate level (Kasturiratne, Lean & Phiooen 2012). If this is the case, public universities and private higher education institutions should offer study programs in entrepreneurship for youth; however, this is not largely evident. The lack of importance placed on E&I among public universities was explained by the reason that they are mostly unaware of their multiple roles, particularly teaching, research and entrepreneurship (Weerasinghe & Jayewardane 2018). When there is a tendency towards E&I, resource constraints limit their actions (Weerasinghe & Jayewardane 2018). The benefits of entrepreneurship advocated in developed economies and the Western context are found to be less desirable in the unique context of Sri Lanka (Lin et al. 2013). Stakeholders responsible for education are urged to understand the importance of entrepreneurship education and promote the discipline among students in Sri Lanka (Weerasinghe & Jayewardane 2018).

3.7 Entrepreneurial ecosystem in Sri Lanka

Entrepreneurship is a key driver for Sri Lanka's sustainable development (Ministry of Sustainable Development 2018). In global ranking among 137 countries, Sri Lanka is ranked 90th in the Global Entrepreneurship Index for 2018 (Global Ecosystems Dynamics Investigation 2020). SMEs contribute to 52% of the country's GDP and account for 45% of the nation's total employment (KPMG 2020). Sri Lanka claims that its start-up economy is valued at almost USD 60 million (LKR21.8 billion), which has quadrupled in two years (Startup

Genome 2021). Sri Lanka ranks among the top three for affordable talent in the Asia-Pacific region, where talent is a key strength of start-up ecosystems (Startup Genome 2021).

Sri Lanka is ranked 85th in the World Bank's Starting a Business index, which is better than the rankings of other South Asian countries such as India at 136, Nepal at 135 and Bangladesh at 131, among 190 economies (World Bank 2020). This recognition is achieved through various government initiatives to encourage entrepreneurship in Sri Lanka. For example, business registration moved to a fully automated and web-based process called 'eregistration' for quick and convenient service for entrepreneurs forming a new venture (Department of the Registrar of Companies 2020). Second, an initiative called the 'Enterprise Sri Lanka' credit program offers youth subsidised loans amounting to approximately USD250 to USD3,800,000 (LKR50,000 to 750,000,000) to finance business ventures (Department of Development Finance 2018). Third, 'Spiralation', a tech start-up support program, and 'Disrupt Asia', a start-up conference, are government-funded programs for entrepreneurs with start-up exposure through a network of investors, accelerators and mentors (ICTA 2020). The country is becoming attractive as a destination to expand businesses, with international start-ups such as Oyo recently established (PricewaterhouseCoopers 2019).

Sri Lanka is inclined towards technology-driven entrepreneurship as the government strives for a technology-based economy (KMPG 2020). As a result, the highest representation of start-ups is in technology-focused services and products followed by e-commerce and mobile applications (ICTA 2020). Numerous government initiatives have been made across the education sectors, emphasising the development of technical skills for the growing tech start-ups (Ministry of Education 2016; Ministry of Sustainable Development 2018). SLASSCOM, the national chamber for the information technology (IT) industry, estimates Sri Lanka to have 1,000 tech start-ups in 2022. This growth in new ventures has happened even amid the pandemic, as e-commerce platforms are increasingly adopted (KMPG 2020). The Sri Lankan government has imposed a tax-free policy for technology start-ups (Startup Genome 2021). Significant growth is recognised in sub-sectors, including cleantech, agtech and new food sectors (Startup Genome 2021). In addition to tech-based sectors, start-ups have entered industries of consumer durables, apparel, financial services, education, consulting and engineering (KMPG 2020).

Given the indicators in Table 3.1, Sri Lanka appears to rank better than most of the comparable countries in Asia and South Asia. When considering the Global Entrepreneurship

Index ranking for 2018, Sri Lanka is behind India, Philippines and Vietnam and ahead of Indonesia, Cambodia, Myanmar, Pakistan and Bangladesh. In the case of the Starting a Business ranking for 2020, only Myanmar and Pakistan are in front of Sri Lanka. According to a Startup Genome Report (2021), India (Mumbai, 1st), Indonesia (Jakarta, 2nd), Philippines (Manila, between 31 and 40) and Vietnam (Ho Chi Minh City, between 71 and 80) are among the top 100 emerging entrepreneurial ecosystems. The entrepreneurial ecosystem of Sri Lanka benefits from an advantageous geographic location, high literacy levels, technical expertise and internet infrastructure (KPMG 2020). However, Sri Lanka has not still made it on this list or rankings such as Top 30 Global Ecosystems or Challenger Ecosystems by Startup Genome (2021).

	Global Entrepreneurship Index ranking for 2018 (out of 137)	Global Entrepreneurship Index for 2018 (out of 100)	Starting a Business ranking for 2020 (out of 190)	Starting a Business score for 2020 (out of 100%)
Sri Lanka	90	21.9	85	88.2
With Southeast Asia				
Cambodia	113	17.6	187	52.4
Indonesia	94	21	140	81.2
Myanmar	127	13.6	70	89.3
Philippines	84	24.1	171	71.3
Vietnam	87	23.2	115	85.1
With SouthAsia				
India	68	28.4	136	81.6
Pakistan	120	15.6	72	89.3
Bangladesh	134	11.8	131	82.4
Bhutan	n/a	n/a	103	86.4

 Table 3.1 – Summary of entrepreneurship related indicators

Source - Global Entrepreneurship Index (2020) and World Bank (2020)

3.8 Entrepreneurs' profile and perceived key success factors

3.8.1 Characteristics of start-up entrepreneurs

A recent study relating to entrepreneurship was conducted in Sri Lanka to provide an understanding of the characteristics of entrepreneurs (refer Figure 3.6). Sri Lankan entrepreneurs forming start-ups are young (33% in the age category of 25–29 years), educated (73% holding a bachelor's degree or above and 30% are after their first degree), and are predominately male, with an 87% representation (PricewaterhouseCoopers 2019). The bulk of

entrepreneurs are from Computer Science (40%) and Business Management (31%) programs. Their reasons for embarking on a new venture range from having an innovative idea (29%), control/independence (23%), flexibility (19%), financial motivation (15%) and being inspired by others (15%) (PricewaterhouseCoopers 2019). The geographical representation shows that 92% of start-ups are located in the western province, including the Colombo district (PricewaterhouseCoopers 2019).



Figure 3.6 – Characteristics of start-up entrepreneurs

Source – PricewaterhouseCoopers (2019)

3.8.2 Reasons for start-up success and failure

When investigating factors that enable start-ups to succeed in Sri Lanka, young entrepreneurs revealed 'skills and expertise' as the most important (PricewaterhouseCoppers 2020). Other contributing factors were the regulatory environment, pro-entrepreneurship culture, support from family and friends, guidance from experienced entrepreneurs and experts, and role models/successful start-ups (PricewaterhouseCoppers 2020). The most important factor of

'skills and expertise' relates to the top reasons why start-ups fail in Sri Lanka. Although Sri Lanka has a higher rate for starting a business among South Asian countries, start-ups in Sri Lanka have a high failure rate (Gunasekara 2020).

According to investors and mentors, 'lack of market research', 'insufficient business acumen' and 'lack of commitment' are key issues that drive new ventures to fail (PricewaterhouseCoppers 2020). For instance, entrepreneurs design products without considering end-user requirements, launch ventures that do not solve a problem, and limit research to secondary data instead of conducting market research (PricewaterhouseCoopers 2020). These reasons for failure can be identified as individual factors related to entrepreneurs. Thus, entrepreneurship education is critical in developing capabilities, which is the basis for economic growth and societal change (Volkmann et al. 2009). These are clear signs of the need for higher education institutions, both public and private, to develop their students with the capabilities necessary to create self-made career paths and pursue entrepreneurship. It is crucial for Sri Lanka's next generation of undergraduates to have the knowledge, skills and attitudes to create, launch and successfully manage new ventures and transform the country's operating industries. Furthermore, developing capabilities in entrepreneurship not only benefits potential entrepreneurs but also for intrapreneurs in multiple occupations in the corporate sector (Alsos et al. 2022). Therefore, there is scope for an exploratory study how can develop capabilities.

The above sections established an overview of the country context of this research, given its importance for contextualising HEEEs. Further, as stated by Brännback et al. (2007), contextualisation diverges from generalisability. However, empirical findings from this study can be generalised to resource-constrained countries with similar characteristics and conditions as Sri Lanka.

3.9 Chapter summary

This chapter provided a detailed overview of Sri Lanka and its higher education sector. In addition to this study, scholars including Kodithuwakku and Rosa (2002), Lin et al. (2013) and de Silva, Uyarra and Oakey (2012) have previously framed Sri Lanka as a resource-constrained environment when investigating entrepreneurship-related problems. This contextualisation chapter established the socio-economic conditions that allow scholars to relate other developing countries to this study. The importance of quality education and entrepreneurship was highlighted through the prolonged challenge of youth unemployment in Sri Lanka. A key issue that surfaced during a national review of sustainable development was the lack of entrepreneurship impeding local growth and access to international markets (Ministry of Sustainable Development 2018). This same review found key challenges of public higher education sector relating to quality, relevance and capacity. The section on challenges of the higher education section justifies the focus of this study on private higher education institutions. The last section of the chapter offers an understanding on the type of entrepreneurs creating start-ups in Sri Lanka and their individual factors driving start-ups to failure.

CHAPTER 4: THEORETICAL FRAMEWORK

4.1 Chapter overview

In qualitative research, it is suggested that scholars use theories as a lens through which the literature and data are viewed, and avoid the overreliance on theory that may limit the ability to find emerging findings (Collins & Stockton 2018). This chapter illustrates the theoretical scaffolding upon which the study is built and refers to existing theories and debates relevant studies. First, the chapter begins with an overview on the use of theory applied to the HEEE concept and in this study. Second, the four existing theories (student involvement theory, systems theory, social capital theory and theory of effectuation) are discussed to establish their use in relevant studies and the relationships for this study. In these sections, the theoretical framework begins to develop, and the final theoretical framework for this study is illustrated at the end of the chapter. This chapter is followed by a chapter contextualising the research in a resource-constrained environment.



Figure 4.1 – Chapter 3 outline

4.2 Use of theory in HEEE studies

This study contributes to the research domain of HEEEs. As an emerging concept that is theoretical and empirical clarity (Hsieh & Kelley 2020; Longva 2021), prior HEEE studies have drawn on theories and logic from the management, organisational and psychology disciplines. Similar to entrepreneurial ecosystem, HEEEs are perceived to be of multidimensional nature, comprising 'actors' and 'factors'. Thus, this study applies selected theories in a multidimensional approach as the logics of a single theory will not serve the

research aim and objectives. In this case, multiple theories related to entrepreneurial ecosystems, student development and stakeholder engagement were applied to an overarching framework.

Existing studies demonstrate a variety of theories that have been applied to the concept of HEEEs. The resource-based view (Rice, Fetters & Greene 2014), Turner's frontier thesis (Miller & Acs 2017), the triple helix (Shil et al. 2020) and stakeholder theory (Meyer et al. 2020) are some of the theories in HEEE studies. Next, studies taking a focus on entrepreneurship education or entrepreneurial support in the context of HEEEs have used the theory of planned behaviour (Guerrero, Urbano & Gajón 2017; Webber, Kitagawa & Plumridge 2020), social capital theory (Theodoraki, Messeghem & Rice 2018) and institutional theory (Lahikainen et al. 2019). As HEEEs are emerging, different theories have played a role in the conceptual and theoretical development. Further, a single theory was not repeated in another study with the intention of advancing on the findings of previous HEEE studies. In contrast, key scholarly work on entrepreneurial ecosystems is observed to maintain consistency by using systems theory across five different studies during almost a decade of investigation (Stam 2015; Feld 2012; Isenberg 2011; Cohen 2006; Neck et al. 2004). From a theoretical standpoint, it is suggested to reduce the development of alternative models and attempt to integrate cumulative knowledge instead (Shook et al. 2003). Leading from this rationale, this study explores HEEEs using existing theories that are relevant to the key interests of the study.

Development of a highly evolved HEEE can take alternative pathways (Rice, Fetters & Greene 2014). This study focuses on the development of HEEEs towards student capabilities in a resource-constrained environment, through diverse stakeholder perspectives. When establishing the theoretical framework for this study, the research problem, questions and objectives provide a clear rationale for the choice of theories (Grant & Osanloo 2014). Before arriving at the existing four theories discussed in the following sections, other theories are also considered and assessed for their relevance to the study.

4.3 Systems theory

This study involves the intersection of entrepreneurial universities (also known as higher education institutions) and entrepreneurial ecosystems contributing to the research domain of HEEEs. The research problem is drawn from the grand challenge of lack of quality education, where SDG 4.4 targets increasing the number of youths who have relevant entrepreneurship skills by 2030 (United Nations 2020). In response, this study aims to contribute to HEEEs for

capability development in a resource-constrained environment. With the key focus on capability development through the perspectives of diverse stakeholders, the central theory framing this study is systems theory.

The notion of systems, which is as old as European philosophy, explains how organised groups are interrelated parts, influenced by their environment (Von Bertalanffy 1972). In this essence, the statement "The whole is more than the sum of its parts" by Aristotle defines a basic system. At a national level, entrepreneurial ecosystems contribute to economic growth and their factors and actors are interconnected as a system (Acs, Autio & Szerb 2014; Acs, Mickiewicz & Szerb 2018). Entrepreneurial ecosystems draw upon systems theory where scholars explore and examine the development of these systems, their composition and configuration as well as their complex and interconnecting functioning (Stam 2015; Feld 2012; Isenberg 2011; Cohen 2006; Neck et al. 2004). There are few studies investigating the interdependencies of the different factors that form the entrepreneurial ecosystem literature provides a lens through which HEEEs can be viewed from a systems perspective.

Higher education institutions should function within their environment including the ecosystems that are structured with a systematic perspective. First, higher education institutions embrace university-industry-government relations suggested by Etzkowitz and Leydesdorff (2000) where the three stakeholder groups representing academia, business and state associate, creating a national innovation system. These mode 3 higher education institutions are open, highly complex, and non-linear knowledge production systems that demand a strong association with the entrepreneurial ecosystem (Carayannis et al. 2018 p146). Second, entrepreneurial ecosystems represent a set of individuals and organisations including universities, industry and organisations (Carlsson et al. 2002; Mason and Brown 2014). Higher education institutions were recognised as a domain of entrepreneurial ecosystems that represents the integration of these institutions within national systems (Isenberg 2010). These institutions do not operate in isolation and are embedded in specific environments, including entrepreneurial ecosystems where there is collaboration as well as competition for constrained resources (Carayannis et al. 2018). These notions explain how higher education institutions and their HEEEs should be considered as part of the wider entrepreneurial ecosystem.

Entrepreneurial ecosystems have frequently been conceptualised as a system from a systems theory perspective. HEEE literature, on the other hand, has focused on self-sustaining ecosystems with less emphasis on their interdependency with the wider entrepreneurial ecosystem. Studies such as by Autio et al. (2014), Lahikainen et al. (2019) and Rice, Fetters and Greene (2014) positioned HEEEs as self-sustaining ecosystems following the notion that they are independent. However, HEEE began to be conceived as closely connected to their entrepreneurial ecosystems (Feldman, Siegel and Wright 2019). In this study, HEEEs are considered a sub-ecosystem of the wider entrepreneurial ecosystems, the customer of HEEEs is the society in which it is embedded, including the various external stakeholders. Further, higher education institutions as open systems have the notion of exchange with the environment, which is also the ecosystem, as every living system (Von Bertalanffy 1972). Acknowledging the importance of learning and entrepreneurial outcomes meeting the economic and social needs of stakeholders and the ecosystem as shown in Figure 4.3.

4.4 Student involvement theory

Astin (1999) proposed the student involvement theory, which has received academic attention since the mid-1990s (Trowler & Trowler 2010). Student involvement theory emphasises how desirable outcomes by higher education institutions are viewed in relation to how students change and develop as a result of being involved in activities within the institution's environment (Astin 1999). This theory accentuates students' behaviour and has historical roots in literature as a developmental theory for higher education (Lahikainen et al. 2021). As students' involvement is a crucial pillar to the higher education experience, the student involvement theory is known as a useful theory to help the management of higher education institutions to design a more effective learning environment for better student involvement and engagement (Astin 1999). This benefit of the student involvement theory resonates well with the research that intends to direct higher education institutions on designing and developing their HEEEs in a resource-constrained environment, in consideration of diverse stakeholder perspectives.

Student involvement theory places the importance on students and students being actively involved in the learning process within the higher education environment (Astin 1999). While some scholars argue that the only way to make students more entrepreneurial is by taking

a learning-by-doing approach (Lackéus 2015), this theory helps higher education institutions recognise how to motivate students to get involved. Developmental theories generally tend to primarily focus on the developmental outcome whereas student involvement theory is unique as it shifts the concentration to the behavioural process that facilitates student development (Astin 1999). As this theory emphasises the 'how' of student development, framing the study with this theory at the centre allows the exploration of the overarching aim of how can HEEEs develop students for E&I in a resource-constrained environment.

Key scholarly work in the entrepreneurial ecosystem literature including Stam (2015), Feld (2012), Isenberg (2011), Cohen (2006) and Neck et al. (2004) demonstrates the consistent use of systems theory, building on theory for entrepreneurial ecosystems. The core principle of the student involvement theory is also based on the similar characteristics of systems theory, being inputs, environment and outcomes with five basic assumptions about students' involvement (Astin 1999). First, Astin (1999) argues that student involvement refers to an investment of psychological and physical energy. Second, student involvement is continuous along a continuum and distinct for each student. Third, student involvement may be qualitative and quantitative. Fourth, student development is proportionate to student involvement. Fifth, student involvement relates to the effectiveness of activities. Such students devote their energy to studying, spend their time physically on campus, actively participate in various student organisations and frequently interact with stakeholders such as peers and educators (Astin 1999). These five propositions are discussed with empirical data in Chapter 7.

Astin's (1999) student involvement theory enables researchers to explore student development within the higher education environment. This study is framed using Astin's (1999) student involvement theory that enables the exploration of student development in the HEEE context. A recent study in Germany investigated how entrepreneurship education ecosystems shaped the innovative capabilities of undergraduates using this theory at the centre (Bock et al. 2020). This study can be identified as one of the first to associate ecosystems in higher education to student development. Bock et al. (2020) found that students' innovation capabilities are influenced by contextual factors including functional experiences and connecting experiences. While this is a noteworthy contribution, this investigation was within the context of an entrepreneurship education ecosystem, focused on entrepreneurship education and not scoped to HEEEs for entrepreneurship education and entrepreneurial support. Building on the initial findings of Bock et al. (2020), this study applies student involvement theory as

the core theoretical frame and investigates at a deeper level through qualitative research with diverse stakeholders that would enumerate student development within HEEEs.

The student involvement theory is expressed as a progressive flow of inputs \rightarrow environment \rightarrow outcome (refer Figure 4.2). First, the theory gives prominence to students and their existing characteristics as they enter higher education (Astin 1999). With this input, it is possible to evaluate how student characteristics such as demographics, background and previous experiences play a role in the development process (Astin 1999). In previous studies, students' existing characteristics, such as their demographics, prior E&I experiences and parents' E&I experiences, have been investigated (Webber, Kitagawa & Plumridge 2020). However, these individual-level factors are beyond the scope of this research, therefore it is understood that input is the student entering higher education for the purpose of obtaining a bachelor degree. Next, the theory brings attention to the environment of the higher education (Astin 1999). Factors and mechanisms of the HEEE can be considered as activities that students involve and experience. Last, the theory leads to outcomes that students achieve in the form of student characteristics, knowledge, attitudes and values at the point of graduation (Astin 1999), which is considered as capabilities development in this study.

This theory frames the study by enabling exploration of how HEEEs can develop students for E&I in a resource-constrained environment. According to this framing (refer Figure 4.2), student involvement theory garners student development including capabilities, addressing how HEEEs can develop the student during their study period.



Figure 4.2 – Astin's student involvement theory and framing to this study

Sourced and adapted from Astin (1999)



Figure 4.3 – Framing systems theory to this study

4.5 Social capital theory

Social networks draw from the notions of systems theory with the assumption that stakeholders (also known as actors) are embedded in the environment and should be explored holistically. Systems theory emphasises the interconnectedness of stakeholders within a system and the importance of these actors interacting and collaborating in a system (Freeman 1984). This can be related to entrepreneurial ecosystems and HEEEs, where stakeholder groups form social networks interconnected in the ecosystem. When conceptualising social networks for this study, stakeholder groups are found in the HEEE and wider entrepreneurial ecosystems, directly or indirectly influencing student E&I capabilities (refer Figure. 4.4)

Social network theory is a network theory important for theoretical development in the field of entrepreneurship (Stuart & Sorenson 2005; Elfring & Hulsink 2003; Jack & Anderson 2002). This theory suggests that stakeholders construct their social environment by building relationships and is commonly used to explore stakeholder interactions within and between ecosystems (Hayter et al. 2016; Clarysse et al. 2014). A few entrepreneurial ecosystem studies, such as by Neumeyer et al. (2019), Pittz et al. (2019) and Neumeyer and Santos (2018), have applied social network theory, where stakeholders were examined among various ecosystem factors and their connectivity.



Figure 4.4 – Framing social capital theory to this study

In contrast, little association is made to network theories in HEEE literature. A recent study framed social network theory to explore social networks in the student-to-entrepreneur transition within the HEEE (Longva 2021). This study found certain factors including curricular activities, co-curricular activities, infrastructure and incubators are perceived as important for student start-ups by student entrepreneurs, educators and support actors in the ecosystem. The same study also established that students gain access to ideas, resources and identity through their social network within the HEEE. Yet scholars including Lahikainen et al. (2019), Miller (2011) and Rice, Fetters and Greene (2014) have identified HEEEs more as independent systems. In contrast, Longva (2021) argues that the HEEE needs to integrate and interact with the wider entrepreneurial ecosystem to function well, just like in a biological community of organisms.

This study explores HEEEs in a resource-constrained environment and in such environments, networks have been identified as underdeveloped and social capital is scarce (Bedő, Erdős & Pittaway 2020). Social capital refers to resources that emerge from structure and networks which are exploited for specific actions (Lin 1999). Entrepreneurs with social capital from diverse networks can benefit from support through their connections, which results in becoming successful with their start-ups (Song et al. 2021). Social capital is embedded within a network of mutual connections and social capital theory involves a network of social relationships forming a valuable resource for the conduct of affairs (Anderson & Jack 2010). Social capital theory and social network theory are closely related as both theories resonate on benefits gained from members within the social network (Ramos-Rodríguez et al. 2010).

In using social capital theory, scholars such as Theodoraki, Messeghem and Rice (2018) argue that all three dimensions of social capital – structural, cognitive and relational – are relevant for the HEEE to function effectively. Taking an emphasis of incubators within the HEEE, it was found that the structural dimension improves access to resources, the cognitive dimension strengthens the relationship among stakeholders, and the relational dimension increases the complementarity and trust of stakeholders (Theodoraki, Messeghem & Rice 2018). The same study found that developing and applying the three social capital dimensions has potential to improve benefits for stakeholders. For this reason, this study investigates how stakeholder engagement can be beneficial for students and their capability development, in particular, using social capital dimensions.

Networks are found to be important for knowledge ecosystems in terms of gaining competitive advantage (Clarysse et al. 2014). Social capital created through these networks provide unique resources and capabilities within the environment (Cruz, Howorth & Hamilton 2013). Scholars such as Theodoraki, Messeghem and Rice (2018) and Hayter et al. (2018) suggest higher education institutions be explored using social network and capital theories when investigating their ecosystems. Prior studies including by Theodoraki, Messeghem and Rice (2018) are focused on HEEEs, with an emphasis on incubators. The conceptualisation of this research inquires as to how diverse stakeholders engage within factors of HEEEs for students' capability development. Through a lens of social capital through social networks within HEEEs, this research explores the ways stakeholders can interact and benefit students within the HEEE. In particular, it shows how stakeholder engagement can develop students, the potential entrepreneurs. While student involvement theory is central to the theoretical framework, social capital is a multidimension being explored.

4.6 Effectuation theory

Effectuation theory, by Sarasvathy (2001), is a decision-making logic that explains how entrepreneurs think, make decisions, behave and act entrepreneurially by creating start-ups and new ventures. Jiang and Ruiling (2019) related effectuation to the 'black box', which is the mind of an individual where effectuation takes place. The concept of the black box is common in marketing literature, where it is referred to as the brain of a customer/consumer (Kotler 1967). Sarasvathy (2001) investigated expert entrepreneurs to understand how they discover and capture new opportunities in their entrepreneurial journey. Effectuation expresses a logic of thinking including a set of means and selecting between possibilities that can be created with that set of means and control (Sarasvathy 2008). A main criticism of effectuation is that it is not a theory (Arend et al. 2015). Sarasvathy (2008) articulated that effectuation is a logic of entrepreneurial action rather than a theory to be tested and proved.

Effectuation refers to new venture creation where entrepreneurs begin the entrepreneurial process and career goals are based on the means that involve 'who I am', 'what I know' and 'whom I know' (Sarasvathy 2001). This effectuation logic appeals to different audiences including the education setting (Frank & Landström 2016) and has been discussed as a practical approach to teaching entrepreneurship to students (Lackéus 2015). Potential entrepreneurs ask three questions related to personal identity (who I am), knowledge (what I know) and networks (whom I know) which leads to 'What can I do?' to image the possibilities

of creating a new venture. These are individual-based antecedents where the means including 'what I know' and 'whom I know' are directly related to the research questions and are explored, whereas the 'who I am' question is beyond the scope of this study.

Effectuation logic is an interesting way to frame research in the field of entrepreneurship, presenting a new perspective (Frank & Landström 2016). Efforts to incorporate effectuation in entrepreneurship education literature are apparent with the focus of capability building (Towers et al. 2020). A recent study encouraged future research in entrepreneurship to connect the theory of effectuation with other psychological theories, helping to build effectuation as a burgeoning theory (Zhang et al. 2021). When using effectuation, career goals such as becoming an entrepreneur emerge by developing potential courses of action that are based on means, such as what they know (Sarasvathy 2001). A review on the principal topic of effectuation by Grégoire and Cherchem (2019) suggested that new research should map the 'means' held among potential entrepreneurs during the early stages of E&I efforts. Mapping the means of potential entrepreneurs allows us to pay attention to the personality of the entrepreneur, an important view that is often overlooked when focusing on what entrepreneurs create (Sarasvathy 2022). This 'human action' direction is pursued in this study, where student capability development is associated with 'means', connecting the HEEE to the effectuation process (refer Figure 4.5).

Capabilities of entrepreneurs are central to the creation and success of start-ups (Gümüsaya & Bohnéc 2018). Higher education institutions are capable of developing student entrepreneurial talent with their HEEEs (Rice, Fetters & Greene 2014). When examining expert and novice entrepreneurs, it was found that students who are beginners use a predictive framework as effectuation, rather than experts who pay less attention to predictive information (Dew et al. 2019). In the event of learning to become an entrepreneur, effectuation is a relevant logic when developing entrepreneurial capabilities (Fayolle & Gailly 2008) and facilitating opportunity recognition (Sarasvathy 2001) for would-be entrepreneurs. However, effectuation is means driven for taking enterprising action without limiting to becoming an entrepreneur (Kautonen, Gelderen & Fink 2015). One study applied effectuation in the HEEE context, examining some factors including curricular, co-curricular activities and financial support (Shirokova et al. 2017). However, more research on HEEEs needs to apply the effectuation logic (Belitski 2019). In conceptualising effectuation for this study, there is an opportunity to explore capability development within the HEEE related to the means as an input of the

entrepreneurial process. This may allow us to understand the interdependencies of HEEEs in the wider entrepreneurial ecosystem.



Figure 4.5 – Framing effectuation theory to this research

Sourced and adapted from The Society for Effectual Action (2018)

4.7 Theoretical framework

The theoretical framework for this study is constructed using the natural order of theory building – that is, 'input \rightarrow process \rightarrow output (Grégoire & Cherchem 2019) – and draws on the logics from four existing theories: systems theory, student involvement theory, social capital theory and effectuation theory. While student involvement theory serves as the central logic of this study, paying attention to student development within HEEEs, systems theory conceptualises the wider entrepreneurial ecosystem. Next, social capital theory focuses on the interactions and benefits to and from stakeholders within the HEEE, with particular emphasis on students. In this way, student involvement theory and social capital theory work together, demonstrating the underlying logic of stakeholder ties within HEEE composition. Using effectuation theory, the HEEE is extended, bridging the gap between the development process and the entrepreneurial process, where an individual with the available means may become a successful entrepreneur.



Figure 4.6 – Theoretical framework

4.8 Chapter summary

This chapter scaffolds the relevant theories this research is built on. The theoretical background of this research draws from three domains: entrepreneurial ecosystems; entrepreneurial universities and HEEEs. The four theories applied to the theoretical framework were selected based on its purpose for the research. The central theory, student involvement theory, accentuates how students change and develop as they participate in the higher education environment, in this case the HEEE, during their learning process. Next, systems theory highlights the environment HEEEs belong to and explains the association to the wider entrepreneurial ecosystem. Social capital theory emphasises stakeholder relationships and engagement that create benefits and opportunities for other stakeholders including students. Finally, the theory of effectuation explains the potential entrepreneur beginning the decision-making process towards entrepreneurship based on their available means. Collectively, these four theories provide the lenses for this study to explore the development of HEEEs, their composition and stakeholder engagement towards students' capability development.

CHAPTER 5: RESEARCH DESIGN

5.1 Chapter overview

This chapter aims to describe the design of this study. The organisation of the chapter in Figure 5.1 is adapted from the research framework by Creswell and Creswell (2018), which guides researchers to plan research. When designing this study, the following essential questions were answered: (1) What was done to achieve the research aim and objectives? (2) Why was a particular methodology chosen over others? (3) When was data collected and from whom? (4) What tools were used to collect data and why? (5) What tools were used to analyse data and why? (6) What ethical considerations were considered? How this study was carried out will be discussed below.



Figure 5.1 – Chapter 5 outline

5.2 Methodological congruence

Researchers tend to start with a research problem, examine related literature, pose research questions, gather and analyse data, and proceed to findings (Creswell & Poth 2018). This study was carried out in the same process. Within this process, there should be methodological congruence where the purpose, research questions and methods are interrelated, leading to a cohesive study (Morse & Richards 2002). Further, Maxwell (2013) advances an interactive approach to research design with the goal of creating coherent research. Therefore, the researcher was mindful of the interconnection of the different sections and began the research design by identifying the research problem that needs investigation, advancing the purpose of the study and specifying the research questions to be addressed (Creswell & Poth 2018).

This study explores the development and composition of HEEEs. Among previous studies, HEEEs are positioned an initiative by higher education institutions in their endeavour to become more entrepreneurial by generating student start-ups and fostering student entrepreneurial intention and mindset. This research investigates HEEEs for student entrepreneurial development including E&I capabilities in a resource-constrained environment. At a global level, countries are striving to increase the number of youths with relevant skills to pursue entrepreneurship in line with the United Nations SDGs. At a local level, Sri Lanka as a developing country is challenged by the high start-up failure rate associated with entrepreneur-related capabilities. The research aims to understand the composition, factors and actors of HEEEs for facilitating E&I capabilities in a resource-constrained environment, in order to provide actionable insights on designing learning environments at an institutional level.

Research questions explore how diverse stakeholders anticipate the continued development of HEEEs in a resource-constrained environment; what they perceive as the composition of HEEEs that could influence students' E&I capabilities in a resource-constrained environment, and how and why. The findings of this study intend to advance the theoretical understanding of HEEEs and benefit the management of higher education institutions, stakeholders of the wider entrepreneurial ecosystem, students, future entrepreneurs and entrepreneurship scholars. The above problem, purpose and questions lay the foundation for decisions related to the research design (Creswell & Poth 2018).

5.3 Theoretical orientation and interpretative framework

Higher education is driven by scientific research for its progress (Juhl & Buch 2019). Entrepreneurship and its investigated subfields of study including HEEEs is a social science concept (Urbano et al. 2019; Malecki 2018; Hoppe 2016; Audretsch et al. 2015; Karataş-Özkan et al. 2014; Etzkowitz 2013; Karataş-Özkan 2011). Scholars in social science view and investigate social phenomena shaped by philosophical assumptions (Grbich 2007). It is argued that philosophical assumptions go unsaid in research (Slife & Williams 1995). However, it is important to identify the use of abstract ideas and beliefs that represent the assumptions and explanations as this informs the practice of research (Creswell & Creswell 2018).

While philosophical assumptions are brought to the research as 'worldviews' by the researcher, theoretical orientations are found in the extant literature and taken to the research as a 'theoretical lens' (Creswell & Poth 2018). As a study, this research is a theoretically oriented project informed by theories in literature which were discussed in Chapter 4 The combination of four lenses – systems theory, student involvement theory, social capital theory and effectuation theory – provide foundations of inquiry to the study. When using existing theories in this research, framing theories can be understood as 'a spotlight' or 'a coat closet' (Maxwell 2013). In this study, the intention of illuminating what the researcher sees and shed light on the areas of interest demonstrates the metaphor 'theory is a spotlight' (Maxwell 2013). The use of social science theoretical lenses identifies that this study underpins a theoretical orientation which operates more at a practical level (Creswell & Poth 2018).

When adapting a theoretical orientation, there are paradigm interpretive frameworks including post-positivism, social constructivism, transformation and postmodernism that inform a qualitative study (Creswell & Poth 2018). In this study, reality is viewed as socially embedded and existing within the mind, where it is multiple and constantly changing (Grbich 2009). This study is cause-driven research investigating a real-world situation and problem relating to higher education institutions being responsible for contributing to sustainable development and economic recovery through entrepreneurship (Maritz et al. 2020; Giones et al. 2020; Liguori & Winkler 2020; Ratten & Jones 2020; Cander et al. 2020). Furthermore, these higher education institutions are expected to alleviate national challenges such as increasing youth employment through skill development (Dissanayake 2020; World Bank 2020; Department of Census and Statistics 2018; Ministry of Sustainable Development 2018; Senarath et al. 2017). The researcher seeks to understand the world in which the problem prevails and to develop knowledge and meaning jointly through interaction between the
researcher and the people who live and work within that space (Grbich 2009). This intent identifies the interpretive framework of this research as social constructivism with its ability to provide a deeper understanding of the concept studied by exploring and interpreting the lived experiences of the participants (Creswell & Poth 2018). A social constructivism paradigm involves pluralistic, open-ended, interpretative and contextualised viewpoints on reality (Creswell & Miller 2000). The philosophical assumptions of social constructivism are summarised in Table 5.1 to outline the nature of reality, how reality is known, the role of the researcher and the approach to inquiry.

Interpretative	Ontological	Epistemological	Axiological	Methodological
framework	beliefs	beliefs	beliefs	beliefs
Social constructivism	Multiple realities are created through the lived experiences and interactions with others	Reality is created between the researcher and the participants shaped by individual experiences	Researcher's role and values are acknowledged and honoured	Qualitative inquiry using methods such as interviews, and observing and analysing texts

Table 5.1 – Interpretative framework with associated philosophical beliefs

Source – Creswell and Poth (2018, p. 35)

Social constructivism is an interpretative framework that shapes how the researcher views the problem and interprets the data (Creswell & Poth 2018). Reality is constructed through individual perceptions and experiences shared between the researcher and the participants of the study and contributing to multiple realities (Creswell & Creswell 2018). For instance, when addressing the second research question, the researcher seeks the complexity of views related to contextual factors rather than narrowing the meanings into a few factors that influence student entrepreneurial development. When seeking answers for the research questions, the researcher relies as much as possible on participants' views of the situation (Creswell & Poth 2018). The outcome of knowledge is considered a social product that is subjective and constructed based on the shared culture of the researcher and participants (Grbich 2009).

5.4 Research approach and methodology

Scholarly work in entrepreneurship represents predominantly qualitative and quantitative research (Landström et al. 2012). When conducting research, there seem to be concerns that researchers compromise and opt for convenience and quick turnaround rather than well-

designed research to develop knowledge that is academically and empirically cumulative or practical and applicable (Miller 2011). Through the review of scholarly work on HEEEs (refer Table 5.2), it is evident that HEEEs have been predominately investigated using either qualitative or quantitative inquiry.

Scholars	Theory	Research design	Context	Focus
Longva (2021)	Social network	Qualitative inquiry	Norway	Start-ups
	theory	In-depth interviews		
Webber,	Theory of	Mixed research	UK	Entrepreneurial
Kitagawa &	planned	Online survey		intention
Plumridge (2020)	behaviour	Public databases		
(2020)		Institutional		
		documentation		
Guerrero,	Utility-	Quantitative inquiry	Mexico	Entrepreneurial
Urbano & Gaión (2020)	function	Survey		intention
Mever et al	Stakeholders	Ouantitative inquiry	US South Korea	Start-ups
(2020)		Survey	and India	Start aps
Shil et al.	Triple helix	Oualitative inquiry	Bangladesh	Start-ups
(2020)		Focus group discussions	2 angraati	
Secundo et al.	Quadruple helix	Mixed research	Italy	Entrepreneurial
(2020)		Case study		mindset
		Observations, documents,		
		interviews and survey		
Allahar &	Triple helix	Qualitative inquiry	Caribbean	Start-ups
Sookram		Case studies		
(2019)		Secondary data		
Wright, Siegel	n/a	n/a	Multiple countries	Start-ups
& Mustar				
(2017) Miller & Acs	Turner's frontier	Qualitative inquiry	US	Start_ups
(2017)	thesis	Case study		Surrups
		Interviews observations		
		documents and media		
Rice, Fetters	Resource-based	Case study	US	Start-ups
& Greene	view	Interviews and secondary		-
(2014)		data		

Table 5.2 – Research design of prior HEEE studies

This study is exploratory in nature, inquiring with 'how' questions focused on the development of HEEEs, specifically on their composition and stakeholder engagement. Qualitative research is suggested as most appropriate when a research problem needs to be explored (Creswell & Creswell 2018). For this research, exploration is crucial to arrive at the

continued development of HEEEs as a part of the entrepreneurial ecosystem, the contextual factors through the perceptions of diverse stakeholders of the entrepreneurial ecosystem, and the opportunities for stakeholder engagement within an HEEE. Using qualitative research, researchers can conduct a complex and deeper exploration of a problem and concept (Creswell & Poth 2018). The breadth and depth of this understanding can only be established by investigating relevant stakeholders directly and giving them the opportunity to share their experiences through qualitative study (Creswell & Poth 2018). In recent times, scholars have begun to widely accept qualitative research by producing more studies using qualitative approaches and generating new theories shaping the understanding of theoretical knowledge (Gehman et al. 2018).

When aiming for a complex study, researchers may recognise an approach to the qualitative inquiry from five qualitative approaches: narrative research, phenomenological research, grounded theory, ethnographic research and case study research (Creswell & Creswell 2018; Tracy 2010). This study exploring HEEEs for student entrepreneurial development draws on psychology and education disciplines. The central theory of the study, student involvement theory, is based on students' psychology and developmental influences during their higher education journey. Further, the research focus revolves around understanding the essence of the HEEE and the research problem involves describing the essence of a lived HEEE experience by diverse stakeholders with a student perspective. Therefore, the qualitative approach that best fits this research type and needs is phenomenological research (Creswell & Poth 2018).

Entrepreneurship scholars suggest that phenomenological studies are useful in contributing to the theoretical development of concepts through new insights and practical occurrences (Ratten & Miragaia 2019). Further, Pret, Shaw and Dodd (2016) emphasise that research with a phenomenological approach enables scholars to enhance existing knowledge by suggesting new research avenues. Given that entrepreneurship and their sub-fields including entrepreneurship education and HEEEs are in developing stages, entrepreneurship scholars such as Ratten and Miragaia (2019) and Thrane et al. (2016) have taken and recommend a phenomenological approach in this field.

The phenomenological approach includes a strong philosophical component and involves exploring and understanding the rich meaning of a concept held by a group of individuals through their lived experiences leading to a universal essence (Creswell & Creswell 2018). Description of this essence of the experience of the concept becomes the phenomenology and several approaches to phenomenology (transcendental phenomenology, existential phenomenology and hermeneutical phenomenology) are highlighted in the literature (Creswell & Poth 2018). This study involves research that begins with a theory, and this approach refers to hermeneutical phenomenology (Creswell & Poth 2018). In hermeneutical phenomenology, research is based on theory and uses the interpretation of participants' experiences and perspectives (Grbich 2009). Therefore, this study undertakes a combination of a deductive approach that is theory driven and an inductive approach that is data driven (Creswell 2013). Table 5.3 summarises the considerations of the phenomenological approach that justify its applicability for this research and directs upcoming research design decisions.

Qualitative approach	Nature of disciplinary origins	Research focus	Unit of analysis	Data collection and analysis
Phenomenological	Philosophy,	Understanding the	Studying several	Interviews with
approach	psychology and	essence of the	individuals who	individuals and
	education	experience	have shared the	analysing data for
			experience	significant themes

 Table 5.3 – Considerations of phenomenological approach

Source – Creswell and Poth (2018, p. 104)

5.5 Role of the researcher

In qualitative research, the researcher is considered important. In social constructivism, the researcher and their background are recognised as they are involved in interpretating participant perceptions and experiences of the world (Creswell & Poth 2018). As individual values of the researcher are honoured in social constructivism, Saunders et al. (2018) suggest a statement of values to position the researcher. Consequently, I present my statement of personal values that underpins this study.

"Over the last 14 years, I have been an academic by profession and passion. Because of my penchant for the students' growth through higher education, I explored a new area of study while marketing has been my forte. Staying true to my values, I am committed to supporting students to have a positive transformation by receiving the relevant higher education experience to pursue their preferred careers. As a researcher, I intend to contribute to knowledge and advance the understanding of higher education entrepreneurial ecosystems, for the benefit of higher education institutions which can ensure in an economic and social impact. I undertake this PhD journey to advance my academic career and as a personal achievement."

Further, Kuratko (2005, p. 592) stated "entrepreneurial history will judge you, and as the years pass, you will judge yourself, on the extent to which you have used your abilities to pioneer and lead our universities into a new horizon. In your hands is the future of your entrepreneurial world and the fulfillment of the best qualities of your own spirit." Guided by the advice of Kuratko, this is the extra-textual frame of this study which demonstrates the knowledge the researcher has obtained and through which the world is viewed (Grbich 2009; McLachlan & Reid 1994). In my position as the researcher, I intend to leverage my experience to advocate for greater responsible management in the higher education industry. This study is a timely, relevant initiative, and collectively the work aims to advance higher education institutions and scholarships.

5.6 Sample selection

5.6.1 Population and participants

Studies involving ecosystems raise the question of what the most suitable unit of analysis is: the country, a state, a city or something like an incubator (Miller & Acs 2017). When deciding on the unit of analysis, the research aim and questions need to be considered (Saunders et al. 2018). This study is centred on ecosystems of higher education institutions (HEEEs). Fetters, Greene and Rice (2010) refer to an HEEE as a multi-stakeholder environment and emerging definitions identify HEEEs as a set of interconnected groups of actors committed to entrepreneurship (Bock et al. 2020; Rice, Fetters & Greene 2014). As a study underlining a phenomenological approach, it is most suitable for the research to involve studying a group of heterogeneous individuals who have shared experience (Creswell & Poth 2018).

Literature echoes those stakeholders that form the entrepreneurial community within its environment and enable higher education institutions to recognise their group of interconnected individuals and organisations when co-creating an HEEE (Brush et al. 2017). Scholars distinguish HEEEs as a complex system of collaborative links between salient stakeholders and have posited it as important to include these stakeholders as key informants when engaging in research on facilitating entrepreneurship through higher educational institutions (Belitski & Heron 2017). All stakeholders should be involved in entrepreneurial activities of successful HEEEs and should be invited to develop the HEEE community (Greene, Mole & Storey 2004; O'Brien, Cooney & Blenker 2019). Lived experiences of diverse stakeholders relating to entrepreneurship education influence students towards entrepreneurship and beyond within the ecosystem (Fetters, Greene & Rice 2010).

However, the central theory framing this study is student centric. HEEE studies have included students (Longva 2021; Secundo et al. 2020; Webber, Kitagawa & Plumridge 2020) and even been based only on student perspectives (Guerrero, Urbano & Gajón 2020). Using students as the unit of analysis is criticised to some extent in the literature due to the lack of external validity (Mortensen et al. 2012). Some scholars argue that students lack real-world experiences hence their responses might not make the best data (Peterson 2001). Due to this view, alumni entrepreneurs who were students of participating higher education institutions were included in this study. These alumni entrepreneurs experienced HEEEs and are now active within the broader entrepreneurial ecosystem. Their perspectives are considered as stronger reflections of the past experience compared to current students who are critised for their lack of external validity. While alumni entrepreneurs may not be current students, their views are of the HEEE at a given point in time include reflections of the reality of the HEEE at the time of completion. Therefore, it is rational to explore HEEEs through a group of diverse stakeholders who have experienced and observed the concept and can relate to the student experience. The system of an entrepreneurial ecosystem, of which higher education institutions are a part, refers to diverse stakeholders as the participants for this research.

Higher education institutions represent the key domain of 'human capital' as institutions offering academic degrees or entrepreneurship training in an entrepreneurial ecosystem (Isenberg 2010). In this case, the population consists of higher education institutions and their HEEEs in Colombo, Sri Lanka. These higher education institutions are privately owned companies offering undergraduate study programs affiliated with international universities. When exploring HEEEs in a resource-constrained environment, the most relevant stakeholders to the context are recognised as deans/heads of schools, academics/educators, alumni entrepreneurs, established entrepreneurs, angel investors and support professionals (refer Table 5.4). These stakeholders are involved in the design and delivery of HEEEs.

Deans/heads of schools and academics/educators of higher education institutions can be categorised as internal stakeholders. External stakeholders include alumni entrepreneurs, established entrepreneurs, angel investors and support professionals. These internal and external stakeholders represent the domains of human capital, markets, finance and support in an entrepreneurial ecosystem (Isenberg 2010). Stakeholders for the 'policy' domain are intentionally not investigated in this research as the government in Sri Lanka plays a minimal role in private higher education institutions, where such institutions operate as private businesses collaborating with foreign universities to offer higher education through accredited study programs.

Type of stakeholder	Domain of entrepreneurship ecosystem	Stakeholder group	Rationale for stakeholder selection
Internal stakeholders	Human capital (educational institutions)	Deans/heads of schools	Deans and heads of school in leadership positions managing schools of study and taking decisions on the respective HEEEs
		Academics / educators	Academics/educators teaching entrepreneurship or related units within the undergraduate degree and engaging with students within the various initiatives of the HEEE.
External stakeholders	Market (networks)	Alumni entrepreneurs	Undergraduates who have founded a start-up after their graduation and the start-up is less than 3.5 years old. These alumni entrepreneurs have experienced HEEEs within the last 3- 4 years and hold stronger reflections of the HEEE experience as students.
		Established entrepreneurs	An entrepreneur with more than 3.5 years in profit-oriented business. Such entrepreneurs have engaged with HEEEs through various initiatives and support student entrepreneurs.
	Finance (financial capital)	Angel investors	Individuals who provide capital for a start-up in exchange for ownership equity or convertible debt. These investors have interest in young entrepreneurs and seek for new business ideas to invest in.
	Support (support professionals)	Support professionals	Individuals who are mentoring, coaching or providing support to young entrepreneurs. Such professionals interest in young entrepreneurs and extend their support voluntary.

Table 5.4 – Participants of this research

5.6.2 Sample and sampling

The above section identities the participants of this research as deans/heads of schools, academics/educators, alumni entrepreneurs, established entrepreneurs, angel investors and support professionals. A range of internal or external stakeholders has been investigated in previous HEEE studies (refer Table 5.5). Previous HEEE work by Rice, Fetters and Greene (2014) and Miller and Acs (2017) explored HEEEs through a combination of internal and external stakeholders. However, more recent studies on HEEEs by Webber, Kitagawa and Plumridge (2020) and Guerrero, Urbano and Gajón (2017) have given prominence to the

student population. Further, the focus of these previous studies is different to the newly explored student entrepreneurial development in this study. The sample for this study involves six salient stakeholders representing HEEEs and their entrepreneurial ecosystems.

Scholars	Theory	Research design	Participants	Context
Longva (2021)	Social network theory	Qualitative inquiry In-depth interviews	15 student entrepreneurs, educators and support actors	Norway
Webber, Kitagawa & Plumridge (2020)	Theory of planned behaviour	Mixed research Online survey, public databases and institutional documents	1,210 students	UK
Guerrero, Urbano & Gajón (2020)	Douglas and Shepherd's utility- maximising function	Quantitative inquiry Survey	8,948 students	Mexico
Meyer et al. (2020)	Stakeholders	Quantitative inquiry Survey	Representatives from entrepreneurship centres of 5 universities	US, South Korea and India
Shil et al. (2020)	Triple helix	Qualitative inquiry Case study Focus group discussions	Faculty members and administrative officials of 1 university	Bangladesh
Secundo et al. (2020)	Quadruple helix	Mixed research Case study Observations, documents, interviews and survey	Professors, mentors, educators and students of 1 university	Italy
Allahar & Sookram (2019)	Triple helix	Qualitative inquiry Case studies Secondary data	2 universities	Caribbean
Wright, Siegel & Mustar (2017)	n/a	n/a	n/a	Multiple countries
Miller & Acs (2017)	Turner's frontier thesis	Case study Interviews, observation, documents and media content	32 internal and external stakeholders	US
Rice, Fetters & Greene (2014)	Resource-based view	Case study Interviews Secondary data (surveys, project data)	Internal and external stakeholders of 6 universities	US

Table 5.5 – Participants of prior HEEE studies

The first two stakeholder groups are deans/heads and academics/educators attached to private higher education institutions in Colombo, Sri Lanka. According to Walter et al. (2011), stratified sampling is effective for studies focused on entrepreneurship education in which the researcher may first identify the higher education institutions. To ensure high quality and representativeness, a stratified random sample of nine private higher education institutions was drawn and is shown in Table 5.6. The strata were set as private higher education institutions:

(1) offering an undergraduate in entrepreneurship or teaching entrepreneurship within undergraduate degrees of other disciplines;(2) offering three or more fields of study; and(3) located in Colombo, Sri Lanka.

Institution	Element of entrepreneurship	Departments / Fields of study
Institution A	Entrepreneurship modules	Business, computing and law
Institution B	Entrepreneurship taught within modules	Business, computing and health
Institution C	Entrepreneurship degree	Business, computing, hospitality and psychology
Institution D	Entrepreneurship taught within modules	Business, computing and law
Institution E	Entrepreneurship modules	Business, computing and education
Institution F	Entrepreneurship taught within modules	Business, computing, engineering and science
Institution G	Entrepreneurship taught within modules	Business, engineering and law
Institution H	Entrepreneurship modules	Architecture, business, computing, engineering, humanities and science, law, hospitality and culinary
Institution I	Entrepreneurship taught within modules	Business, computing, engineering and science

 Table 5.6 – Sample of private higher education institutions

The rest of the four stakeholder groups are individuals from start-ups, business ventures or organisations including alumni entrepreneurs, established entrepreneurs, angel investors and support professionals. Purposive sampling is commonly used for studies in the field of entrepreneurship including by Lahikainen et al. (2019), Theodoraki et al. (2018) and Krueger et al. (2000). For this group, stakeholders were drawn using purposive sampling (Saunders et al. 2018) by selecting key informants with interest in student entrepreneurs, higher education and HEEEs. This purposeful selection of participants was based on their interests and characteristics that can contribute to the research problem and study (Creswell 2007). Although this sampling technique is argued to be researcher biased, purposive sampling was deemed suitable as the selection was based on clear criteria of stakeholder groups identified in Table 5.4 (Ranga et al. 2016; Small 2009; Marshall 1996).

With reference to the sample size, there is no specific number suitable for qualitative research. However, the number of participants can be decided based on the nature of the population. If the research population is of a homogeneous nature, six to eight interviews are adequate and if the population is more heterogeneous then 12 to 20 is acceptable (Saunders &

Townsend 2016). Further, in general, 20 to 30 individuals are considered acceptable for a qualitative study (Creswell & Creswell 2018). A key work on HEEEs by Miller and Acs (2017) based on the University of Chicago interviewed 32 individuals who had an interest and were pivotal in start-ups created by students at US colleges. Therefore, it was deemed suitable for qualitative data collection to engage 30 to 40 participants given the heterogeneous population of the study. As an approximation, it was initially planned to complete interviews with 36 stakeholders with an equal representation of six participants per stakeholder group. It was understood that only some higher education institutions would be willing to participant in this study. Thus, the study sought input from six deans/heads and six academics/educators from participating higher education institutions, six alumni entrepreneurs, six established entrepreneurs, six angel investors and six support professionals. During data collection, the sample increased to ten alumni entrepreneurs resulting in 40 interviews. Scholars suggest continuing data collection until saturation where there are no new ideas, themes or findings (Francis et al. 2010) and the same applies for this study.

5.7 Data collection

5.7.1 Research instrument

As qualitative research, the study calls for gathering consensus through methods such as interviewing, observing and analysing texts (Creswell & Poth 2018). However, as research with a phenomenological approach, data collection involves in-depth interviews with diverse individuals who have experienced the concept (Grbich 2009). Compared to observations and secondary records, interviews are social interactions between the researcher and participant where knowledge is constructed (Creswell & Creswell 2008). HEEE studies discussed in the literature review including work by Longva (2021), Secundo et al. (2020), Miller and Acs (2017) and Rice, Fetters and Greene (2014) have used interviews as the tool for qualitative data collection. Specifically, the University of Chicago study employed semi-structured interviews among individuals involved in creating student start-ups at US colleges (Miller & Acs 2017). As an exploratory study, semi-structured interviews are suggested as the best fit (Saunders et al. 2018) where interviews are an attempt to understand the world from the participant viewpoint to explore the meaning of their experiences (Freeman et al. 2007). Therefore, semi-structured interviews were selected as the instrument for this study.

As exploratory research, the interview questions were designed as an open inquiry allowing for free discussion during data collection (Creswell & Creswell 2018). The interview

protocol was designed according to the research questions of the study (Creswell & Poth 2018). Each research question aimed to explore the key concepts and ask questions in a simple manner that participants could understand. This design protocol (refer Figure 5.7) allowed the researcher to gather participants' perspectives on HEEEs that addressed the research aim and questions. Therefore, the interview protocol was developed in four sections:

Section A: Participant and start-up/organisation demographics

Section B: Continued development of HEEEs

Section C: HEEE factors

Section D: Stakeholder engagement

The interview protocol was designed in a structured format with four sections: basic information about the interview, an introduction of the interviewer and research, a series of questions, and closing instructions (Creswell & Creswell 2018). As data collection invited participation from a diverse range of stakeholders, two sets of interview protocols (refer Appendix A1 and A2) were designed to best suit; (1) deans/heads of school and academics/educators from higher education institutions; and (2) alumni entrepreneurs, established entrepreneurs, angel investors and support professionals from the entrepreneurial ecosystem. During any given interview, the order of questions and flow differed according to the participant and direction of the conversation.

In such interviews, the total number of questions should range between five and ten, which are prepared in advance and used consistency during interviews (Creswell & Creswell 2018). The interview protocol for this research encompassed three categories of interview questions: opening questions, content questions and closing questions.

Focus and research questions	Explored concepts	Interview protocol	Section
Higher education entrepreneurial ecosystems	Domain and role	Personal and start- up/organisation demographics	Section A
How do diverse stakeholders anticipate the continued development of HEEEs in a resource-constrained environment?	HEEE and link to entrepreneurial ecosystem	Development of HEEEs and contribution to wider entrepreneurial ecosystem	Section B
What do diverse stakeholders perceive as the composition of HEEEs that could influence students' E&I capabilities in a resource-constrained environment? How and why do specific contextual factors of HEEEs influence students' E&I capabilities in a resource- constrained environment?	HEEE factors aligned to entrepreneurship education and entrepreneurial support	Activities currently conducted within HEEEs and suggested initiatives for the future	Section C
How can diverse stakeholders engage within the factors of HEEEs that could influence students' E&I capabilities in a resource-constrained environment?	Stakeholders within HEEEs and outside from the wider entrepreneurial ecosystem	Stakeholders presently involved in HEEEs and opportunities for better engagement	Section D

Table 5.7 – Design of interview protocol

5.7.2 Participant recruitment

Data collection for this study was conducted in two phases. Recruitment of participants for this study began in September 2020 and data collection continued until November 2020. Participant recruitment was also conducted in two phases: (1) participants within higher education institutions; and (2) participants outside in the wider entrepreneurial ecosystem. At first, the expected number of participants was 36 stakeholders, however, there was a high level of interest from the invited participants. After the first round of theory-driven analysis, a follow-up round of interviews was held between August and September 2021 in search of deeper understanding. This second round of interviews recruited new participants, bringing the total to 40 participants; the breakdown of participants is shown in Table 5.8. A higher number of alumni entrepreneurs participated in the study compared to the other stakeholder groups. There were no withdrawals from any participant after accepting the invitation to participate or completing the interview.

Stakeholder category	Domain of entrepreneurial ecosystem	Stakeholder group	Description / Criteria	Number of recruited interviewees
Internal stakeholders	Human capital	Dean / Head of school	Deans and heads of school in leadership positions managing schools of study	6
		Academic / Educator	Academics/educators teaching entrepreneurship or related units within the undergraduate degree	6
External stakeholders		Alumni entrepreneur	Graduates who have founded a start-up after their graduation and the start-up is less than 3.5 years old	10
		Established entrepreneur	An entrepreneur with a profit- oriented business more than 3.5 years old	6
	Finance	Angel investor	Individuals who provide capital for a start-up in exchange for ownership equity or convertible debt	6
	Support	Support professional	Individuals who are mentoring, coaching or providing support to young entrepreneurs	6

Table 5.8 – Participant representation for this study

First, the email addresses of the higher education institution sample were extracted from their respective institution websites and the deans/heads of school were contacted via email. The introductory email for the purpose of participant recruitment (refer to Appendix A6) was sent to the identified deans/heads outlining the purpose of the study in brief and inviting them to participate, as suggested by Saunders et al. (2018). This email also included the participant information statement as an attachment (refer to Appendix A3) explaining the research project, its rationale, interests, risks and benefits, consent, privacy and confidentiality, research output and contact information. If the invitee responded communicating their willingness to participate in the research, consent was requested from the participant. Two consent forms were used in the case of higher education institutions, one for authorising employees to be filled by deans/heads when nominating staff, and the other for individual adults by deans/heads and academics/educators (refer to Appendix A4 and A5).

Three out of nine deans/heads declined the invitation, and six deans/heads from higher education institutions accepted the invitation to participate in the study (refer Table 5.9). In addition to the deans/heads who participated, they nominated two or three potential

academics/educators who were suitable for the study and these academics/educators were contacted separately in the listed order. In all cases, the first contacted academic/educator was willing to participate in the study.

Institution	Entrepreneurship education	Response to research invitation
Institution A	Entrepreneurship modules	Accepted
Institution B	Entrepreneurship taught within modules	Declined
Institution C	Entrepreneurship degree	Declined
Institution D	Entrepreneurship taught within modules	Accepted
Institution E	Entrepreneurship modules	Declined
Institution F	Entrepreneurship taught within modules	Accepted
Institution G	Entrepreneurship taught within modules	Accepted
Institution H	Entrepreneurship modules	Accepted
Institution I	Entrepreneurship taught within modules	Accepted

Table 5.9 – Participation of private higher education institutions

Second, the remaining participants, including alumni entrepreneurs, established entrepreneurs, angel investors and professionals, were identified and contacted via LinkedIn. The researcher identified the suitable participants by scanning their posts for involvement in HEEE-related activities and who might be able to share their perceptions/experiences related to the research. The identified participant was sent a brief message regarding the research and if they responded with a willingness to participate then communication was continued via email. The majority of the contacted participants were willing to participate and approximately one participant per stakeholder group did not respond to the LinkedIn message. Similar to the deans/heads and academics/educators, these participants were also forwarded the participant recruitment email with the participant information sheet. As they communicated their interest to participate in the study, their consent was requested in the same email using the consent form for individuals. To schedule the interview, a calendar invite was sent to all participants along with a Zoom meeting link, making it easier for the participant to calendarise the interview within their schedule.

5.7.3 Interview procedure

Interviews are purposeful discussions between the researcher and the participant built around the area of research to gather data relevant to the study (Saunders et al. 2018). The procedure

for preparing and conducting interviews as suggested by Creswell and Poth (2018) was used for this qualitative data collection. Each one-to-one interview was planned for a duration of approximately 30 to 45 minutes, depending on the availability and willingness of the interviewee. For comparison, a study exploring entrepreneurial universities conducted interviews with a 20- to 60-minute duration with students, faculty and management of a Finnish entrepreneurial university (Lahikainen et al. 2019). Consent was confirmed prior to starting each interview and interviews took between 30 and 60 minutes. Each interview was completed within the scheduled appointment and did not face disruptions or require re-scheduling. The interviews could not take place in the natural setting of the institution, start-up or organisation due to COVID-19 pandemic restrictions and as all participants were working from home. As an adequate recording procedure, all interviews were electronically recorded via Zoom with the consent of the participant (Creswell & Poth 2018).

5.8 Data analysis and representation

5.8.1 Data analysis plan

Research studies following the phenomenological approach, such as this study, are based on a theory and use the interpretations of participants to arrive at meanings about a concept (Grbich 2009). Interview data from this study presents lived experiences of participants describing an external reality (Silverman 2022) and this type of data is commonly analysed using content or thematic analysis to identify, analyse and report patterns (Braun & Clarke 2006). Both analytical techniques are applied in entrepreneurship literature according to the purposes served. While a thematic analysis facilitates an interpretative perspective of the data and creation of themes in a theoretical sense, a content analysis can further display data in frequency counts per category of content (Pistrang & Barker 2012; Bichoff, Volkmann & Audretsch 2017). The strength of content analysis is quantifying the themes to reach a consensus among the participants' responses to offer a better description of the findings (Saldaña 2021). As quantifying makes qualitative data more credible, a content analysis was applied in this study, similar to other studies (Grégoire & Cherchem 2018; Theodoraki, Messeghem & Rice 2018; Reyes 2016; Hannon, Collins & Smith 2005). To maintain the richness of the interview data, a set of relevant themes was first identified for each research question in this study before quantifying themes, where necessary (Frank & Landström 2016).

This study with its phenomenological approach was analysed using a combination of theory-driven and data-driven approaches (Creswell 2013). At the beginning of the study, a

theoretical background was established and an a priori framework developed using literature. The first step of analysis took a theory-driven approach using the initial dataset to ensure the validity of the study by establishing that theories and concepts represented the key interest of the study (Whittemore et al. 2001). For this deductive approach, this study utilised a list of codes pre-defined according to the research questions, theories and concepts that emerged from the literature review (Creswell & Poth 2018). Table 5.10 outlines the concepts and their attributes used for coding the themes pre-determined from literature.

Concept	Attributes
Development of HEEEs	Student start-ups, entrepreneurial mindset, entrepreneurial intention and entrepreneurial ecosystem
Composition of HEEEs	Entrepreneurial education and entrepreneurial support
Stakeholders of HEEEs	Internal stakeholders and external stakeholders

Table 5.10 – Concepts and attributes for coding

The next stage of data analysis involved generating themes from the analysis of significant statements or excerpts that provide an understanding of how the participants experienced the concept (Creswell & Poth 2018). These themes represent clusters of meaning that led to developing textual and structural descriptions (Creswell & Poth 2018). This analysis involves an inductive approach using a bottom-up technique and building themes from the data (Creswell & Creswell 2018). Within the social constructivism and hermeneutical phenomenology approaches, the 'making sense' process is used for identifying patterns and themes that lead to insights shaping the concept of interest (Creswell & Creswell 2018; Gehman et al. 2018; Tracy 2010). The sensemaking process includes three steps: (1) creation – identify cues and patterns; (2) interpretation – make sense of lived experiences; and (3) enactment – to recreate the story (Sutton & Staw 1995). Within this study, creation is evident in the three empirical chapters of findings, and interpretation and enactment is presented in the discussion chapter. More information on data analysis is discussed in Section 5.8.2.

5.8.2 Data analysis process

A qualitative researcher enters the data analysis process with a large amount of raw data and intends to exit with an account of findings (Creswell & Poth 2018). Once data is collected or during data collection, findings are custom-built or, rather, choreographed by the researcher using analytic procedures in the field (Huberman & Miles 1994). For this study, the data

analysis spiral was followed, where the researcher applied certain analytic strategies at each spiral to arrive at specific analytic outcomes (Creswell & Poth 2018). The five step data analysis spiral process is similar to the data analysis process out by Braun and Clarke (2006, p. 35), including the steps of "familiarising yourself with your data, generating initial codes, searching for themes, reviewing themes, defining and naming themes, and producing the report". Although this analysis process was originally used in psychology research, scholars have previously applied it to studies related to entrepreneurship (Ratten & Miragaia 2020), entrepreneurship education (Donald et al. 2018; Pittaway & Cope 2007) and HEEEs (Allahar & Sookram 2019). The data analysis spiral used in this study is presented in Figure 5.2.



Figure 5.2 – Data analysis spiral

Source – Creswell and Poth (2018)

5.8.3 Data management

At the beginning of the analysis process, all 40 interviews were transcribed verbatim (word for word) using Otter.ai. This helped to organise the data and create digital files where each interview transcript ranged between five and ten pages of transcription. Once the interview transcripts were ready, the data was input to the software NVivo for content analysis. Next, a project was created and stored on NVivo preparing a file for data management. Scholars commonly use NVivo to facilitate the content analysis of interview data as the software offers

transparency, flexibility, data coding and data retrieval (Corbin & Strauss 2015). Therefore, the data management software was selected based on the pre-decided analysis procedure.

The project in NVivo followed a case naming system, where each transcription was created as a case with an identifiable code. For instance, a transcription of a dean/head of school was named using the identifier DH with a number to represent participant and the first case was identified as DH1. This naming system was followed for all 40 cases of the research as shown in Table 5.11.

Stakeholder group	Participant code	Description
Dean / Head of school	DH1 to DH6	Deans and heads of school in leadership positions managing schools of study
Academic / Educator	AC1 to AC6	Academics/educators teaching entrepreneurship or related units within the undergraduate degree
Alumni entrepreneur	AE1 to AE10	Graduates who have founded a start-up after their graduation and the start-up is less than 3.5 years old
Established entrepreneur	EE1 to EE6	An entrepreneur with a profit-oriented business more than 3.5 years old
Angel investor	AI1 to AI6	Individuals who provide capital for a start-up in exchange for ownership equity or convertible debt
Support professional	SP1 to SP6	Individuals who are mentoring, coaching or providing support to young entrepreneurs

Table 5.11 – Case naming system for data management

5.8.4 Memoing

After creating cases in the project, the researcher continued to get a sense of the qualitative data (Creswell & Poth 2018). During this second spiral (refer Figure 5.2), each transcript was read twice, the first time to improve the readability of the content as software-generated transcriptions tend to have inaccuracies due to pronunciation differences by participants with English as their second language. Reading the full transcription, a second time helped make sense of the participant perspectives before breaking down the interview into parts (Creswell & Poth 2018). The second reading took longer as the researcher spent time putting down reflection notes in the form of a memo for each case/transcription. Memoing is a way of recording the researcher's ideas, insights and interpretations of the case, created separately on NVivo yet linked to the case, which is a transcription of a participant (McNeil 2021). This step

helped to organise the data and resulted in an audit trail that can be retrieved when needed, which also acts as a validation strategy (Creswell & Poth 2018).

5.8.5 Codes and themes

After becoming familiar with the dataset, the third spiral involved describing, classifying and interpreting the data (Creswell & Poth 2018). To start coding, the data were identified and labelled according to the initial codes pre-defined, as the theory-driven approach was formed by categorising small sections of data that prevailed relevant to the analysis and labelling the core ideas. When creating the second set of codes, the researcher inquired about the statements appropriate for the relevant research question. However, because of winnowing the data, not all data of a qualitative study is used, and some data was discarded (Creswell & Poth 2018).

Next, the researcher proceeded to 'lean coding' where data was reviewed to identify expanded codes which led from the first coding. Regardless of the project size, a final code list of 25 to 30 codes is suggested for a research project, leading to five or six aggregated themes (Creswell & Poth 2018). At the end of this spiral, a list of codes was finalised along with their descriptions, which set boundaries for each code (inclusion and exclusion criteria), resulting in a codebook for the project (Creswell & Poth 2018).

5.8.6 Researcher's interpretations

As a qualitative study with a social constructivism framework, the researcher acknowledges that she interpreted the gathered data, and this interpretation flows from her own personal, cultural and historical experiences (Creswell & Creswell 2018). For this reason, the researcher's reflexivity was established earlier in this chapter to understand her experiences and background (refer Section 5.4). Through the data analysis, the intention is to interpret and analyse the meaning that participants have about HEEEs as part of their world (Creswell & Poth 2018). This 'sense-making' requires the researcher to be critical and creative when considering the categories, patterns and themes generated by the analysis (Creswell & Poth 2018). Therefore, interpreting qualitative data involved conceptualising beyond the codes and themes to establish a larger meaning for the data. In doing this, the researcher turned to existing literature and models linking her interpretations to research previously developed by scholars (Creswell & Poth 2018). Further, initial conceptualisations for the research questions were presented to the research supervisors and subject experts to obtain feedback on whether interpretations were rational (Creswell & Poth 2018).

5.8.7 Data representation

Analysing data and concluding the interviews was an iterative and non-linear process for almost two years. In the final spiral, the researcher presents the findings from the study in the form of text, tables or figures (Creswell & Poth 2018). The representation of data was decided based on the nature of the research question addressed. The sought outcomes are the explanatory concepts and models where uniqueness is preferred and wide generalisation is avoided (Grbich 2009). The three research questions were answered with text and figures/tables. The text shows the multiple perspectives in the form of the researcher's interpretations supported with relevant excerpts from the interviews. For instance, in reporting contextual factors for the second research question, the researcher used the block-and-file approach to segment excerpts into columns with headings and present data as a table instead of separate entries (Grbich 2009). This text led to a qualitative visual of the findings through suitable data visualisation to broaden the understanding of findings (Henderson & Segal 2013). The visual aid demonstrated data more concretely through displayed patterns and possible comparisons visually represented that text cannot offer (Creswell & Poth 2018).

For all research questions, a distinct content analysis led to findings in the form of text and quantified themes. The data visualisations for each research question varied according to the nature and purpose of the question, and these visualisations displayed at theme level have moderate complexity. In research question one, a reorder matrix demonstrated the three outcomes found in participants' responses and displayed the level of importance for each outcome perceived by each participant. In this case, the reorder matrix listed the themes derived from the interviews and quantified them with the associated level of importance as perceived by the participants of the study (Henderson & Segal 2013). A spectrum display was used for the second research question to display qualitative data, as in Slone (2009). The spectrum displayed all six themes (i.e. contextual factors) and indicated whether each individual participant/case discusses the respective theme in their interview, which quantified the responses (Henderson & Segal 2013).

5.9 Validation and reliability

An ongoing academic debate is taking place on the validation and reliability of qualitative research (Creswell & Poth 2018). Among many perspectives on validation and reliability in qualitative research, this study holds the perspective that validation is an evolving process for assessing the accuracy of findings as best constructed by the researcher and participants

(Creswell & Poth 2018). This choice of perspective is based on the suggestion that in qualitative research with an interpretative approach such as this study, validation is a judgement of trustworthiness or goodness in the research (Augen 2000).

This perspective of validation and reliability was translated into strategies and techniques applied in this study as suggested by Augen (2000) and Creswell and Poth (2018). Augen (2000) suggested two types of validation for interpretive research; ethical validation and substantive validation. Ethical validation requires research to provide practical solutions to research problems with a generative promise. In doing so, this study found new understandings; raised new possibilities for HEEEs; opened new questions for entrepreneurship scholars and stimulated new dialogue through publication and presentation. This validation is of transformative value that may lead to action and change within higher education (Augen 2000).

Substantive validation refers to the researcher's understanding of the research topic, understanding from other sources and the documenting of this understanding in the study (Augen 2000). The compilation of interpretations was recorded in the thesis and publications for others such as the supervisors, panel, reviewers of this study and entrepreneurship scholars to determine the trustworthiness of meanings arrived at the end of the doctoral study. During the three years of this research, early chapters have been improved at three points with advanced understanding of the researcher's own topic. The findings for the research questions are intended to resonate with the audiences as compelling and convincing (Augen 2000). Through these strategies and techniques, ethical validation and substantive validation is achieved for this research.

A few more strategies and techniques for validation and reliability suggested by Creswell and Poth (2018) were followed. First, the researcher corroborated evidence from multiple stakeholder groups. Interviews gathered perceptions and experiences from six stakeholder groups representing the HEEEs and wider entrepreneurial ecosystem, with more than five participants per group (refer Sections 5.6). Second, the findings presented in chapters 6 to 9 establish multiple views of statements confirming and disconfirming prior literature. Third, to reduce researcher bias, the values and experiences that the researcher brought to the research are stated in Section 5.5 (researcher and reflexivity) and her interpretations in the research in Section 5.8.5 (researcher's interpretations).

Fourth, the researcher spent considerable time in the context of study prior to and during the study to familiarise herself with the context and participants (see Section 5.5). Fifth, for the purpose of auditing, the researcher captured the analysis through a digital audit trail via NVivo that enables examination of the process and the findings. Sixth, memos were created with rich, thick descriptions that helped during the analysis linked to the transcription (refer Section 5.8.3). Seventh, this research received reviews by internal supervisors and panel during the three years, while associated publications underwent external review by subject experts who were familiar with the concepts explored. From the nine procedures suggested for validation and reliability (Creswell & Poth 2018), a majority of seven have been observed in this research with validity lenses through the researcher, participant and reviewer.

5.10 Research ethics

Ethical considerations in research are paramount and a common misconception is that ethical issues surface only during data collection (Creswell & Creswell 2018). Research studies in Australia are regulated by the National Statement on Ethical Conduct in Human Research to ensure that the study is managed in adherence with legal, ethical and professional frameworks. For this study, ethics was anticipated through all stages of the qualitative study, from the beginning of the study to reporting and publishing the research findings (Creswell & Poth 2018).

5.10.1 Ethical considerations prior to study

To adher to the National Statement on Ethical Conduct in Human Research, all ethical issues and considerations relating to the university were considered prior to commencing any data collection. Swinburne University facilitated the process of ethical scrutiny to create awareness of the relevant ethics issues and plans to mitigate them according to the type of study. In preparation, the draft ethics application for qualitative research was initially reviewed by both supervisors. Then, the improved application was reviewed by the research advisor of the School of Business, Law and Entrepreneurship, and developed prior to its submission. A wellreviewed human research ethics application including details of research team, administration, project details, data management plans and other additional documents was submitted to the Swinburne Ethics Committee for institutional review (Project ID 2915). Ethics approval for this study was granted in September 2020 and the study commenced following the completion of ethical review (refer Appendix B). This study was conducted according to the Swinburne University Code of Ethics 2018, ensuring that the research was conducted responsibly and committed to specified guidelines, principles and relevant responsibilities.

5.10.2 Ethical considerations in preparing to investigate

At the beginning of the study, potential research participants were sent an invitation email as participant recruitment informing them of the purpose of the study. The participant information statement was also attached in this initial email for invited participants to access a comprehensive understanding of the research prior to accepting the invitation to participate. In both instances, it was clearly communicated that participation was voluntary. For instance, when deans/heads of school nominated two to three potential academics/educators suitable to participate in the study, they were contacted in the listed order and the invitation email for participant recruitment stated that participation was voluntary and there was no undue risk involved. This was to remove the possibility of any perceived coercion by being nominated by their managers, the deans/heads of the higher education institutions.

5.10.3 Ethical considerations in collecting data

If an invited participant did not respond to the first email invitation, they were not sent followup emails, as the study is voluntary. Only in the instance when the invited participant responded to the invitation with interest or willingness to participate was communication continued, including requesting for their consent. This was to avoid potential participants being pressured to participate or provide consent and to maintain voluntary participation in the study. The second email to the participant requested probable dates/times for the interview or if this was already provided then confirmation or an alternative of the interview date/time along with the request to provide consent prior to the interview. In most cases, signed consent was received and when needed kind reminders were sent to ensure that consent was obtained before the interview date. There were no sensitive populations such as children involved in the study that required special provisions.

During the online interviews, the researcher began with a self-introduction and the purpose of the study to build trust among the parties (Creswell & Poth 2018). Next, the researcher confirmed the consent received and permission to record the interview before starting the questions although signed consent was previously obtained. When the interaction between the participant and the researcher began, the researcher as interviewer remained within the research scope and interview questions. A few interviews naturally had a power imbalance and power was observed mainly among participants as experts on the area of interest. The

power imbalance was respected and only when needed the control of the interview was taken by the researcher over the line of questioning to ensure relevance yet avoid bias (Grbich 2009).

5.10.4 Ethical considerations in analysing data

In the attempt to respect the privacy of participants, each transcript from the interviews was labelled using a code in NVivo instead of the participant's name or organisation. This masking of names characterised each case by a stakeholder group and participant number with a composite profile (Creswell & Poth 2018). As analysis took place, small segments of data from the interview were assigned to codes that were emerging and within this process, the researcher collated diverse perspectives for each code with a combination of positive and contrary statements. This helped in developing a complex understanding of the issue addressed and gathering multiple perspectives for the next stage (Creswell & Poth 2018).

5.10.5 Ethical considerations in reporting data

It was evident that collected data held some sensitive information regarding higher education institutions in Sri Lanka. For instance, there were employee views on what was not being done or supported, although this was not the intention of the study. When reporting data, the researcher gave focus to honestly reporting data addressing the research questions, without disclosing any information that potentially could cause harm to participants in the present or future (Creswell & Poth 2018). Findings from the investigation were presented in clear communication and appropriate language for the intended audience of this study (Creswell & Poth 2018). Each research question was met with a distinct section of analysis and findings including a set of telling excerpts from the interviews and a relevant data visualisation to make a composite point of view. There were multiple perspectives respecting every participant's opinion and findings that complemented or contradicted existing literature.

5.10.6 Ethical considerations in publishing from the research

The sought outcome from this study is to fulfill the researcher's academic qualification in Doctor of Philosophy. In addition, the researcher and the supervisors published and intend to publish findings from the study. However, to honour the literature and findings of the study, key sections of the dissertation were published and planned for publication instead of aiming to 'piecemeal' multiple publications of smaller parts (Creswell & Poth 2018). In this way, some findings were and will not be used for more than one publication in the same genre. As it is important to share the curated knowledge with participants of the study and the wider audience that could apply this knowledge, there is a plan for more suitable publications such as articles

in newspapers and business magazines sharing the practical information. This helps to tailor the reporting of the study to diverse academic and practical audiences.

5.10 Chapter summary

This chapter discussed the research design of this study in detail. This study explores HEEEs for students E&I capabilities in a resource-constrained environment. Given the research focus and needs, this study undertook an interpretivist perspective to the research of HEEEs with a social constructivism paradigm and hermeneutical phenomenology approach. As the meaning of data is interpreted by the researcher with this study, this chapter acknowledged the researcher's role and values on the research process. The participants represented the HEEEs and the wider entrepreneurial ecosystem. Internal stakeholders (deans/heads of schools and academics/educators) represent the higher education institutions while external stakeholders include alumni entrepreneurs, established entrepreneurs and support professionals. Upon data collection of 40 semi-structured interviews, a combination of theory-driven and data-driven analysis was conducted. Data collection and data analysis followed processes suggested by Creswell and Poth (2018) for high quality research. All research design decisions were based on procedures, features and considerations of a study undertaking social constructivism and hermeneutical phenomenology approaches. Two types of validation for interpretive research (ethical validation and substantive validation) were embedded into the research process as suggested by Augen (2000). In this investigation, there were several challenges and these were acknowledged. Finally, the research obtained ethics approval from the Swinburne University of Technology Human Research Ethics Committee and this chapter concluded with the ethical considerations prior to the study, during the study and after the study. The next chapter is the first empirical chapter, addressing research question one.

CHAPTER 6: FINDINGS AND ANALYSIS FOR RESEARCH QUESTION 1

6.1 Chapter overview

This qualitative interpretative research conducted a series of 40 semi-structured interviews. The first research question explores multi-stakeholder perspectives on how diverse stakeholders anticipate the continued development of HEEEs in a resource-constrained environment. Having established that HEEEs are still emerging in the field of entrepreneurship, this question focuses on the development of the concept. The chapter, structured as in Figure 6.1, includes the existing condition of HEEEs in a resource-constrained environment and views on their continued development as perceived by internal stakeholders who represent the HEEE itself and external stakeholders from the wider entrepreneurial ecosystem.



Figure 6.1 – Chapter 6 outline

In this chapter, the perspectives on the continued development of HEEEs are first presented using a reorder matrix representing each participant and their views of the importance of the three HEEE pathways. This reorder matrix is supported with data extracts of multi-stakeholder perspectives on entrepreneurial mindset, entrepreneurial intentions, and E&I capabilities. Next, the chapter reports on the key roles and different contributions of HEEEs, followed by two sections on bridging to the entrepreneurial ecosystem and higher education institutions as catalysts. These two sections contribute to the first research question of this study on the continued development of HEEEs, through the views of stakeholders.

6.2 Analysis of diverse stakeholder perspectives on continued development

This qualitative research aimed to explore HEEEs: the continued development of HEEEs, their composition and configuration including contextual factors, and stakeholder engagement within HEEEs in a resource-constrained environment. To gather broad coverage of viewpoints, the research investigated diverse stakeholders including deans/heads of schools and educators of higher education institutions, alumni entrepreneurs, established entrepreneurs, angel investors and support professionals. Two rounds of interviews were conducted, resulting in a total of 40 participants (refer Table 6.1).

Internal stakeholders	External stakeholders
Total participants ($n = 12$)	Total participants $(n = 28)$
Deans/heads of school $(n = 6)$	Alumni entrepreneurs $(n = 10)$
	Established entrepreneurs $(n = 6)$
Academics/educators $(n = 6)$	Angel investors $(n = 6)$
	Support professionals $(n = 6)$

Table 6.1 – Distributions of participants

A content analysis was conducted to arrive at findings addressing the first research question. The section presents the three-level coding data structure resulting from the inductive approach of analysing perspectives on how diverse stakeholders anticipate the continued development of HEEEs in Sri Lanka (refer Figure 6.2). By exploring the interview data and identifying themes within multi-stakeholder perspectives, three pathways of HEEEs emerged, representing the development of HEEEs. The first round of interview data was coded for all three pathways while understanding deepened through the second interview data that found roles and contributions of HEEEs. To arrive at the three aggregated themes, internal and external stakeholders were questioned on their lived experiences and observations of HEEEs using the below interview questions:

- What do you think about current HEEEs among private higher education institutions?
- How do you expect these HEEEs to develop in the future?
- What should higher education institutions focus on in their HEEEs?

During the analysis process, first-level codes were broadly noticed within the dataset. Having identified patterns within the participants' experiences, three sets of codes were initially labelled and coded to data. These initial codes are based on general opinions such as 'stimulate beliefs and thought process towards entrepreneurship'. Then, this initial code was expanded to more abstract concepts with a theoretical background such as 'goal orientation', 'prior knowledge' and 'social influence'. For the last level, coding led to the aggregated themes, and the first theme was 'entrepreneurial mindset'. In this case, the continued development of HEEEs in a resource-constrained environment is related to mindset, intentions and capabilities. Figure 6.2 shows a thematic structure of the synthesised themes and codes along with the relationships between them.

RQ1: How do diverse stakeholders anticipate the continued development of HEEEs in a resource-constrained environment?



Figure 6.2 – Theme structure for continued development of HEEEs

6.3 Continued development of HEEEs

Interview questions related to the first research question asked participants to share their views and opinions about the current stage of HEEEs and their future development. Internal stakeholders including deans/heads and academics/educators spoke about the present learning environment and activities offered in their HEEE. Alumni entrepreneurs reflected on their recent HEEE experience during their higher education journey. Other external stakeholders such as established entrepreneurs, angel investors and support professionals shared their current associations with HEEEs to explain their existing conditions and suggest progress needed in HEEEs. All participants, internal and external, hold positive opinions of HEEEs and value their potential towards entrepreneurship for students.

Both internal and external stakeholders shared a consensus that HEEEs of private higher education institutions in Sri Lanka are still in their early stage of development (refer Table 6.2). Despite entrepreneurship being a national priority in Sri Lanka, entrepreneurship as a strategic choice is relatively new within the private higher education sector. Existing HEEEs are observed to be in their introduction stage, somewhat focusing on fostering entrepreneurial mindsets and encouraging intentions among students. According to the statements made by internal stakeholders, HEEEs are more organically formed, with minimal drive from senior management. Students and educators are promoting entrepreneurship within existing initiatives, while well-planned initiatives are still taking form. External stakeholders such as angel investors and support professionals agree that HEEEs are still emerging and evolving, whereas the entrepreneurial ecosystem is in more of a growth stage with initiatives set up, connections made, and results recognised.

Interviewee	Data extracts
DH5, a chief academic officer	"that it's still it's [HEEEs] in the very early stage"
AC4, an academic teaching an entrepreneurship unit	"Entrepreneurship ecosystems among the higher education bodies in Sri Lanka is still an evolving area"
AE3, a young entrepreneur of a fitness start-up	"So I guess, in its [HEEE] early stage, it is more about promoting entrepreneurship among students like getting into their minds"

Table 6.2 – Illustrative quotes about the current stage of HEEEs in Sri Lanka

Important		Somewhat important		Not discussed
	Interviewee	Entrepreneurial mindset	Entrepreneurial intention	Entrepreneurial & innovation capabilities
Internal stakeholders	Dean / Head of school 1			
	Dean / Head of school 2			
	Dean / Head of school 3			
	Dean / Head of school 4			
	Dean / Head of school 5			
	Dean / Head of school 6			
	Academic / Educator 1			
	Academic / Educator 2			
	Academic / Educator 3			
	Academic / Educator 4			
	Academic / Educator 5			
	Academic / Educator 6			
External stakeholders	Alumni entrepreneur 1			
	Alumni entrepreneur 2			
	Alumni entrepreneur 3			
	Alumni entrepreneur 4			
	Alumni entrepreneur 5			
	Alumni entrepreneur 6			
	Alumni entrepreneur 7			
	Alumni entrepreneur 8			
	Alumni entrepreneur 9			
	Alumni entrepreneur 10			
	Established entrepreneur 1			
	Established entrepreneur 2			
	Established entrepreneur 3			
	Established entrepreneur 4			
	Established entrepreneur 5			
	Established entrepreneur 6			
	Angel investor 1			
	Angel investor 2			
	Angel investor 3			
	Angel investor 4			
	Angel investor 5			
	Angel investor 6			
	Support professional 1			
	Support professional 2			
	Support professional 3			
	Support professional 4			
	Support professional 5			
	Support professional 6			

Table 6.3 – Reorder matrix of HEEE's continued development

6.3.1 Entrepreneurial mindset

The consensus from this empirical study is that HEEEs should foster entrepreneurial mindset among students. Entrepreneurial mindset is explained as a 'way of thinking' (Ireland, Hitt & Sirmon 2001) and the way entrepreneurs 'think, reason, make decisions and set goals' (Baron 2014) that is different from non-entrepreneurs (Davis, Hall & Mayer 2016). Such an entrepreneurial mindset will enable potential entrepreneurs to move from exploration to exploitation in the entrepreneurship process (Ucbasaran et al. 2003).

Six out of 12 (50%) internal stakeholders expressed the importance of fostering an entrepreneurial mindset among students while 17 out of 28 (60%) external stakeholders also expressed the same (refer Table 6.3). Some stakeholders pointed out that entrepreneurial mindset should be the most important priority of HEEEs, while others suggest that entrepreneurial mindset and E&I capabilities are equally important. For example, one internal and one external stakeholder shared the consensus that mindset as equally important as capabilities:

Interviewee DH2, a head of school: "So if I were to rank these three, I would say, capabilities is one and mindset is equally important."

Interviewee AE6, a co-founder of a men's fashion start-up: "I think it's a combination of the things that these ecosystems should focus on. So, there is the intention. But the mindset or the entrepreneurial mindset is most important. And the capabilities of course."

Data extracts related to entrepreneurial mindset from the interviews explain a range of reasons for the importance of entrepreneurial mindset within an HEEE. In most interviews, stakeholders spoke of entrepreneurial mindset as the way of thinking about entrepreneurship as a career choice and a start-up. Further, higher education institutions being entrepreneurial results in a broad range of activities including the development of an entrepreneurial mindset (Etzkowitz & Klofsten 2005).

When considering the environment, entrepreneurship as a career choice among students is still not commonly accepted by the education system and wider society. Higher education institutions tend to teach students to become proficient employees instead of successful entrepreneurs (Solomon 1989). The larger belief among these systems is for students to complete their undergraduate degrees and continue into corporate jobs. Therefore, both internal and external stakeholders argue that students should be exposed to alternative career options such as entrepreneurship and break through such fixated thinking.

Interviewee DH4, a head of academic affairs: "Because end of the day, we don't want these sorts of box minded students going out there, thinking that the only option is corporate life. So, I think that's where the mindset needs to change, and we as higher education institutions have a significant role to play. If you give them [students] the exposure and platforms during the academic period, I think their minds will open to these trends and identify their own strengths."

Interviewee AC2, an academic in charge of an entrepreneurial initiative: "Most of the students come to us, having the mindset that we must finish the degree and we have to do a job. They are not open to anything entrepreneurial, accepting risks and trying something new. So, I think we need to step in here. I think mindset is critical, developing the mindset required for a student because I personally believe mind drives you for whatever the things you do. With the right mindset, they are bound to take on entrepreneurship as a career and create jobs."

Interviewee AE9, an alumni entrepreneur with a tech start-up: "So I would say, being able to help students build that mindset by putting them on the spot, validating their ideas, and showing them alternate paths to the same goal without breaking the spirits, of course. Students need to build that mindset of being entrepreneurial rather than no I cannot I do not want as excuses."

Interviewee AI5, an angel investor: "So my simple answer is mindset. I think mindset is perhaps the most important thing that education institutions must develop. And when I say mindset, they must be given a realistic understanding of what entrepreneurship means."

Fostering an entrepreneurial mindset among students implies a higher education experience that constitutes values, beliefs, attitudes and emotions associated with E&I. At the same time, promoting an entrepreneurial mindset among students can affect society. Creating an entrepreneurial mindset among students and educators can become a priority in contributing to societal development (Gibb et al. 2009). Given the impact of the COVID-19 pandemic, higher education institutions need to incorporate an entrepreneurial mindset within their HEEE as entrepreneurship has become a necessity more than a choice for some youth. These entrepreneurs identify business opportunities and generate innovative ideas, not because they want to launch a new business but as an alternative for their survival (Maritz et al. 2020). An external stakeholder from the wider entrepreneurial ecosystem spoke about how mindset plays a role in becoming an entrepreneur. Interviewee AI3, an angel investor, said:

"First of all, you must really start thinking about that and it's a mind game. I mean, many entrepreneurs become entrepreneurs, because they are in a mind space where they understand that the current solution is not good. Right? And they can create something better. They take the risk of going out and doing it. Right now, how did Zoom start? I mean, I'm sure you know, the story of Zoom ..."

An entrepreneurial mindset can stimulate beliefs and open the boxed thought process towards entrepreneurship. A variety of internal and external stakeholders emphasised that entrepreneurial mindset should be given key priority within HEEEs by higher education institutions, which can lead to students developing as entrepreneurs. Where students share an entrepreneurial learning experience and receive support towards entrepreneurship, students can gain an entrepreneurial mindset. For example, one internal and one external stakeholder shared the consensus that higher education institutions and their HEEEs are responsible for fostering an entrepreneurial mindset among their student community.

Interviewee AC1, an academic teaching entrepreneurship: "First of all, there must be a kind of a psychological or mindset change. So, I would say, when it comes to that mindset change, there are many factors that will impact this transformation. But out of the factors, I would say that our ecosystems including education and network plays a huge role."

Interviewee SP4, an external mentor in entrepreneurship: "As an element, education institutions give them the basics such as education to understand the entrepreneurship subject matter. But the entrepreneurship mindset, now that's a totally different, I would say. A lot of mindset change is necessary, because launching a start-up has not yet become the acceptable thing by the society and by the education system. I would say, an institution should instil the spirit for this entrepreneurial mindset."

Efforts within the HEEE facilitate advancing the mindset when it is driven by internal stakeholders including deans/heads and academics. Higher education institutions have found effective strategies for creating an entrepreneurial mindset through entrepreneurship education (Etzkowitz & Leydesdorff 2000; Fayolle 2013). Further, HEEEs hold the aim of developing entrepreneurial awareness and mindsets among students through linkages and a variety of

activities including business idea presentations, open innovation challenges, enterprise projects and business games (Secundo et al. 2020). To develop potential entrepreneurs among students, HEEEs must create entrepreneurial mindsets through psychological elements that orient feelings, intentions, motivations, and behaviours of students.

6.3.2 Entrepreneurial intention

There is a mixed result about HEEE encouraging entrepreneurial intention among students. Entrepreneurial intention refers to the state of mind that influences attention towards creating a new venture, at some point in time (Tomy & Pardede 2020; Thompson 2009). Some stakeholders only expect HEEEs to play a pivotal role in fostering entrepreneurship and supporting students to take up a viable career where the intention and decision to become an entrepreneur is voluntary. Others perceive that higher education institutions should promote entrepreneurship and influence the entrepreneurial intention.

Only four stakeholders per category expressed the importance of promoting an entrepreneurial intention among students (refer Table 6.3). Most stakeholders have different views on promoting entrepreneurial intentions among students within HEEEs. Entrepreneurial intention is understood as a career choice to becoming an entrepreneur and launching a startup. One stakeholder highlighted that no student plans to become an entrepreneur and that entrepreneurship takes its own course of action according to circumstances.

Interviewee AI3, an angel investor: "Nobody intends to be an entrepreneur, it's just you go down a certain path. Right ..."

Some stakeholders suggest entrepreneurial intention as a separate priority that higher education institutions should have for their HEEEs, while others were of the viewpoint that fostering mindsets and intentions happen together. Entrepreneurial higher education institutions bear a third mission oriented towards transforming mindsets, intentions and actions of students and academic programs are being designed to influence entrepreneurial intentions of students (Guerrero, Urbano & Gajón 2020; Nabi & Liñán 2011; Pittaway & Cope 2007). For example, four internal stakeholders and one external stakeholder emphasised that higher education institutions and their HEEEs should promote entrepreneurial intentions among students.
Interviewee DH3, a head of school: "There are foundations for entrepreneurship in our learning environment and students are developing awareness and intention, so that they know that this is also a career choice for them."

Interviewee DH2, a head of school: "I would rank that [intention] last, but because I think we don't necessarily have to create the intention among students. I think because that self-made intention needs to come from within students. I think there is very little that we [institutions] can do in terms of creating that, that intention, but mindset is a different thing."

Interviewee AC3, an academic teaching entrepreneurship: "Education institutions, we have a key role to play here. In terms of, you know, building that mindset and creating that intention through awareness."

Interviewee AC5, an academic teaching a career planning module: "Networks and partnerships are crucial for it, I mean, for private education institute to have because otherwise, what happens is students are only limited to their degree program and the institution walls. So that's not enough when it comes to, you know, creating that mindset and intention. Because you know it is the most important thing."

Interviewee AE6, a co-founder of a men's fashion start-up: "I think it's a combination of the things that these ecosystems should focus on. So, there is the intention. But the mindset or the entrepreneurial mindset is most important. And the capabilities of course."

When entrepreneurial intention is being considered on its own, it is identified to be of least importance compared to mindset and capabilities due to two reasons. First, both stakeholder categories, internal and external, point out that having an entrepreneurial intention is individual and students should be given the freedom to make that career choice, if they wish to embark on a self-made journey. Like other career choices, becoming an entrepreneur is driven by the individual's passion, traits and characteristics. Second, internal and external stakeholders further indicate that entrepreneurial intentions among students are strongly influenced by their parents and the wider community they live in. Parents have control of their children's future and others, including friends, tend to set expectations for these students. One HEEE study found that entrepreneurial behaviours of parents positively affect students' entrepreneurial intentions (Webber, Kitagawa & Plumridge 2020).

Interviewee SP5, a support professional working with incubators and accelerators: "Students nowadays they have that intention to become entrepreneurs. Whereas I guess during our time with it was more about you know, career focused like, where do we want to work? So yes, intention is quite important, but I feel it is a more personal thing. "

Interviewee EE3, a second-generation entrepreneur: "I think career intentions are heavily influenced by parents and social trends. While this is the case, students may go about having intensions based on their confidence and capabilities. I think there is a limit to institutions creating intentions, but they can influence those intentions through the education and various experiences offered."

Rather than considering a career pathway led by education or qualification, youth in Sri Lanka intend to become entrepreneurs based on their experiences (Dissanayake 2020). Stakeholders stated that life events that take place over time are likely to trigger an individual's aspiration for an entrepreneurial career. For example, children are commonly involved in parental family business and such experiences can cause youth to take the same career path. Studies in entrepreneurship have found that the relationship of a family's E&I history has an influence on students' entrepreneurial propensity (Bock et al. 2020). Another influence is prior experiences with social problems leading to social entrepreneurial intentions (Hockerts 2015). Potential entrepreneurs with experience in a recycling program or a charity program are more likely to be driven towards solving a social problem, through an entrepreneurial initiative. Therefore, prior experiences are a key influence on youth towards launching start-ups as they are experienced compared to ones with no similar experience (Shirokova et al. 2017).

Another viewpoint that came to light in this research was that students may intend to become entrepreneurs by being influenced by their broader environment. In addition to students who want to follow their parents' career path and pursue entrepreneurship, some students intend to launch start-ups and become entrepreneurs because of disruptive events. Economic crisis is a factor when youth are motivated to emerge from economic crisis and survive hardship by embarking on entrepreneurship (Dissanayake 2020). Due to the prolonged civil war followed by economic instability in Sri Lanka, a portion of educated students intend to start small ventures out of necessity for their survival and hope for a better life. To mitigate the critical effects of disruptive events, entrepreneurs engage in entrepreneurship and innovation (Posen et al. 2018). The COVID-19 pandemic is considered as one of the worse disruptions in

recent times, causing an unprecedented challenge for people around the world, especially in developing countries. Being challenged by such economic hardship or disruptions can lead to 'necessity' entrepreneurs (Maritz et al. 2020).

Interviewee AI1, an angel investor: "When living in a developing country like Sri Lanka, potential entrepreneurs including these students naturally develop an attitude to cope up with changes in the environment. For us these changes include business uncertainty and economic crisis. We are still going back and forth when political parties changes and now we are in the worst economic crisis Sri Lanka as ever faced right? This triggers them [students] to take control."

The various viewpoints shared by internal and external stakeholders emphasise that entrepreneurial intentions of students are predominately influenced by micro (e.g. home background) and macro environments, leaving higher education institutions to focus on supporting career aspirations more than encouraging intentions. This shared consensus in the research contrasts to a large portion of literature that articulates the role of higher education institutions and their HEEEs in promoting entrepreneurial intentions among students. In Mexico, an emerging economy, a study found results on HEEEs facilitating employability options including entrepreneurship education reinforcing graduate work effort and incubators reducing graduate risk aversion (Guerrero, Urbano & Gajón 2020).

Higher education institutions should become catalysts for economic and societal development and produce entrepreneurial capital (Audretsch 2014; Guerrero, Cunningham & Urbano 2015). In this vein, stakeholders claim HEEEs focus on creating an entrepreneurial mindset and supporting students as students develop entrepreneurial intentions within collective dynamics mostly outside the higher education institutions. To support students towards entrepreneurship, HEEEs can influence students' self-efficacy by developing their E&I capabilities relevant for an entrepreneurial career. A range of HEEE factors with the involvement of diverse stakeholders may influence a student's development and ability to become an entrepreneur. When students who are keen to pursue entrepreneurship seek to develop internal capability and obtain support, it should be found within the higher education environment. Students with E&I capabilities are likely to feel confident that they are prepared for entrepreneurship as a career and have the necessary skills to launch a new start-up as an entrepreneur.

6.3.3 Entrepreneurship and innovation capabilities

Stakeholders held the strong view that HEEE should pay attention to the human capital developed more than intention or mindset. They emphasise that HEEEs should develop students to solve problems, innovate solutions, handle uncertainty, collaborate with others, leverage resources, mitigate risks and make informed decisions. Entrepreneurial capabilities refer to knowledge and skills related to the entrepreneurial process, judgmental abilities and decision making related to entrepreneurial action, and social skills and networking abilities (Alsos et al. 2022). E&I capabilities is a new construct that emerged in this study, although it is not new in entrepreneurship education and entrepreneurship education ecosystems.

Eight out of 12 (67%) internal stakeholders and 21 of 28 (75%) external stakeholders also expressed the importance of developing E&I capabilities among students (refer Table 6.3). Thus, the broad consensus among internal and external stakeholders is that developing students' capabilities should be the priority within the HEEE. Given that HEEEs are in an introductory stage among higher education institutions in Sri Lanka, stakeholders suggest that priorities should be elevated to capabilities from mindset and intention as progression. These capabilities are understood as a combination of knowledge, skills and attitude needed to successfully launch a start-up and be an entrepreneur.

Stakeholder calls for higher education institutions to pay more attention to developing capabilities among students were twofold. First, some external stakeholders including entrepreneurs and investors reflected on their experiences and argued for the need for students to develop the abilities to identify, evaluate and exploit opportunities within the dynamics of a resource-constrained environment. Second, internal and external stakeholders agree that they observe that some young entrepreneurs fail in entrepreneurship due to a lack of relevant capabilities. Thus, the emphasis on capability development was based on these two dominant views held among both stakeholder categories.

Interviewee DH5, a chief academic officer: "It is, I think, it will be all three. But I think more than intention. For me, I think, as a private Higher Education Institute, what we need to kind of focus on is the mindset, and perhaps more importantly, the entrepreneurial capabilities."

Interviewee AC4, an academic teaching an entrepreneurship unit: "So from my observations, obviously, we need to give them [students] the knowledge on entrepreneurship. But see, even if you're educated about entrepreneurship, you really

need the capabilities, so I think entrepreneurial capabilities is important. When you're talking about starting your own company, it's less about knowledge, but more about your capabilities in being an entrepreneur and your capabilities to become an entrepreneur."

Interviewee AE7, an alumni entrepreneur: "Education is important, I mean, from school to higher education. All levels of education are important. But that education setting with so many other things must develop the entrepreneurial capabilities within innovation and build discipline in students who want to turn out as entrepreneurs, at least for me that was it."

Interviewee AE10, an alumni entrepreneur who founded a start-up in fashion: "Now students are being exposed to successful entrepreneurs, so that mindset is starting to develop. Because they see a lot of entrepreneurs today, but it is the capabilities students are lacking. And that is where I think private higher education institutions should really focus on."

Entrepreneurial higher education institutions play a fundamental role in knowledgebased entrepreneurial economies (Acs et al. 2009; Audretsch 2014; Link & Sarala 2019) through the development of entrepreneurial capabilities and the creation of conducive environments for students to pursue entrepreneurship (Kirby 2004; Mian et al. 2016; Morris et al. 2017). From the interviews, it was understood that academics, to some extent, felt inadequate in developing capabilities of students as they highlighted that senior management needs to accept and commit to preparing students for entrepreneurship as an alternative career. Although all educators emphasised that business/management students should learn about entrepreneurship, they pointed out that some higher education institutions did not offer a single entrepreneurship-based module in the degree program and entrepreneurship was introduced to students through other disciplines. This unavailability of core and elective entrepreneurship modules has a chain reaction where students from other study programs such as computing and law also lack the opportunity to gain knowledge on entrepreneurship.

Interviewee AC6, an academic: "I think it is the dean or the head that needs to push and be there, mainly to make sure that these initiatives are sustained to give that commitment to developing entrepreneurial capabilities in students. See, as academics, we can't do that on our own." Academic experiences must deliver knowledge to students by creating a conducive environment for entrepreneurial opportunity and innovative breakthroughs (Fayolle 2013; Guerrero et al. 2016; Bergmann et al. 2016). When stakeholders identified capabilities as a priority that higher education institutions should commit to and involve within their HEEEs, they were clear on the type of capabilities. While more internal stakeholders highlighted entrepreneurial capabilities, external stakeholders emphasised capabilities in E&I. The suggestion for E&I was supported by the notion that innovation is a capability that potential entrepreneurs must possess. Further, emphasis was made on Sri Lanka's national interest in harnessing technology-based entrepreneurship driven by innovation. Therefore, stakeholders encourage higher education institutions to understand the critical role played by E&I capabilities in the entrepreneurial process and in the success of entrepreneurs.

Interviewee DH2, a head of school: "So if I were to rank these three, I would say, entrepreneurial capabilities were one and mindset is equally important."

Interviewee AC5, an academic teaching a career planning module: "Along with creating this mindset, we need to give these students these entrepreneurial capabilities, we don't have in our students. We could give them the research skills and the deeper insights into you know, what it takes to be an entrepreneur and what you need to go through in starting a new venture. "

Interviewee AE1, an alumni entrepreneur who founded tech start-ups: "Capabilities in entrepreneurship and innovation, of course, is number one, that is what they need to be doing. That is what we need to be pushing for them to develop these capabilities. Because that is what is lacking. Because I think intention and mindset, if we look at current students today, they do have an intention. They also may have the mindset to a certain extent ... "

Interviewee EE1, a second-generation entrepreneur and co-founder of a global brand: "I think factors that inhibit entrepreneurship is that when we are a little too close, and it forces it to be a little open ended, we need to go beyond theory and let people imagine you do need to encourage their free spirit, you need to empower people to go out and teach their ideas to do, too. You must empower your students to be creative and create innovations. Essentially in entrepreneurship, if you're going out and creating something, give them what is required to be innovative thinkers and give them enough knowledge to create that foundation that allows them to create from and then grow." External stakeholders further elaborated on capabilities being the underlying ability to execute ideas and strategies. More precisely, entrepreneurs, investors and support professionals stated that students should be equipped with execution capabilities as they develop within HEEEs. To improve their preparedness for the entrepreneurial process, students need to develop their abilities in launching a start-up and being an entrepreneur. In doing so, these stakeholders emphasised that students should be given the opportunity to learn by doing in an HEEE. For example, three external stakeholders emphasised developing execution capabilities and not just any capabilities relevant for entrepreneurship.

Interviewee AI1, an angel investor: "So talking about the most important thing, these execution capabilities. Starting from problem solving capabilities, or deconstructing the problem that people face, everything understanding the market and industry to from being comfortable in working, building relationships ... "

Interviewee AI2, an angel investor: "If a student wants to start a company today, the cost of starting a company is zero. You don't really need to spend anything to start a company, even to get sales, you don't really need to spend much. If you understand how to operate digitally, you can start a business and you can grow that business. That requires your capabilities to execute. But I don't see that stuff happening in Sri Lankan higher education institutions."

Interviewee EE6, a social entrepreneur: "If I take myself or myself as an example, how I came about doing what I'm doing today as an entrepreneur, the first key thing would be passion. This passion is cultivated within me and of course it's very personal. But to implement my passion and launch my start-up I turned to my capabilities. If I had what it took to become an entrepreneur."

Often, entrepreneurship programs offered by higher education institutions relate to new venture creation and small business management while educating students 'about' entrepreneurship instead of developing students 'for' entrepreneurship with the relevant capabilities (Kirby 2004). In this case, in contrast to internal stakeholders who emphasised the former, external stakeholders highlight the latter being the need for HEEEs to develop capabilities 'for' E&I. External stakeholders including entrepreneurs, investors and support professionals pointed out that students keen to pursue an entrepreneurial career would benefit from a variety of capabilities.

Some entrepreneurs appreciated the theoretical knowledge about entrepreneurship by asserting that it lays the foundation that capabilities are built on. A few other entrepreneurs highlighted the importance of self-management capabilities as potential entrepreneurs. Investors were most keen on practical capabilities, such as opportunity seeking and problem solving, to successfully launch a start-up. Support professionals who undertake the role of mentors to young entrepreneurs stated that developing emotional skills, such as self-awareness, is vital for entrepreneurial success. While literature highlights the emotional challenges that entrepreneurs encounter during their entrepreneurial journey, the efforts of education should equip entrepreneurs with the emotional skills they need to thrive (Aly, Audretsch & Grimm 2021). Input on the variety of E&I capabilities recognised elements beyond the balanced skill set of hard and soft skills.

Interviewee SP5, a support professional working with incubators and accelerators: "*I* think that beyond entrepreneurial knowledge, these students can really benefit from emotional intelligence. Emotional skills will leverage their problem solving which is a something they do as an entrepreneur and help them with being social as they navigate through their networks."

Interviewee SP6, a mentor for young entrepreneurs: "To be a leader, to be an entrepreneur one must have self-awareness. Understanding where they stand emotionally will help students to become and be entrepreneurs. This distorts perceptions and creates awareness of others and importantly the world out there helping them making better decisions day in and out."

Interviewee AI1, an angel investor: "Being an entrepreneur and living life an entrepreneur is no walk in the park. You are faced with challenges and roadblocks on the journey. Having emotional skills will make young entrepreneurs handle situations and difficult conversations. Especially take control of their own emotions that can get in the way."

When entrepreneurs engage in entrepreneurship, they are especially influenced by their cognition, emotions and behaviour. However, one aspect that goes unnoticed is the high levels of affective intensity when engaging with entrepreneurship that includes self-awareness and management. Understanding one's own strengths, emotions, motivation, integrity, control and regulation, confidence and wellness are the key areas that came to light in the interview data.

All these are identified as significant by stakeholders, since they impact the development of E&I capabilities that translate to being entrepreneurial.

6.4 Key roles and contributions of HEEEs

6.4.1 Typology of HEEEs' continued development

Interviews with internal and external stakeholders of HEEEs canvassed a range of aspects dealing with the development of HEEEs, their composition and configuration, and ecosystem engagement for this study. In addition to the current stage of HEEEs reported in Section 6.3, the first set of interview questions probed the working of the HEEEs, any observed gaps or inadequacies with respect to their functioning, and opportunities for HEEEs to contribute to the wider entrepreneurial ecosystem. Responses to these questions were themed into five key roles: influential, developmental, networking, entrepreneurial and regional (refer Table 6.4). HEEEs in their evolved form offer multiple supports for their community including students (Fuster et al. 2018) and must realise their contribution to the wider entrepreneurial ecosystem. On the surface, according to the labels these roles may seem typical, however the contribution can be found in the uniqueness related to the resource-constrained context encompassed in the roles.

Not learning about similar entrepreneurial start-ups, not understanding oneself, not connecting with people in and related to entrepreneurship, not having a mentor/advocate, and not knowing what is happening in the real world are key challenges that stakeholders understand that students face within HEEEs. HEEEs are meant to deal with these issues through their roles and contributions to develop students for entrepreneurship. In the context of entrepreneurship, higher education institutions bear the responsibility of ensuring their students thrive in their endeavours (Audretsch 2014).

Key role	Type of role	Contribution / Expectation
Influential role	Leading	Foster an entrepreneurial mindset among students
	Guiding	Encourage entrepreneurial intentions among students
Developmental role	Governing	Build HEEEs with a system
	Facilitating	Develop students with capabilities relevant for entrepreneurship and innovation
Networking role	Community building	Connect students with other students
	Inspiring	Connect students with alumni entrepreneurs
	Mentoring	Connect students with external stakeholders
	Coaching	
	Consulting	Connect students with specialised experts
Entrepreneurial role	Experimenting	Provide support to generate student start-ups
		Connect to points of support such existing incubators and accelerators
	Expanding	Provide programs and support to create spin-offs from research
Regional role	Coordinating	Align goals with national priorities
	Talent sourcing	Lead human capital development
		Provide entrepreneurial talent as a source

Table 6.4 – Key roles and contributions of HEEEs

The most common role played by HEEEs is the role of 'influential'. In this study, some stakeholders perceived that HEEEs play a role in fostering an entrepreneurial mindset and encouraging entrepreneurial intentions among students. The influential role was understood as an activity that was being undertaken to some extent by the HEEEs of higher education institutions. Although given their current influential role, stakeholders outside the institutions such as entrepreneurs and investors stated that HEEEs need to invest more effort by strengthening their dynamics within. Educators confirm that they actively work on the mindset and career intentions of students; however, they highlighted that HEEEs lack the institutional support in playing a more influential role that demands a stronger entrepreneurial drive. Senior leadership and their long-term commitment are considered significant for the effective functioning of ecosystems (Allahar & Sookram 2019; Bischoff, Volkmann & Audretsch 2017; Yu et al. 2017; Rice, Fetters & Greene 2014). Through entrepreneurial leadership, such leaders can remove barriers and carry out entrepreneurial change in a particular context (Gibb, Haskins & Robertson 2013). In this way, HEEEs can benefit from goal setting, strategy development, providing leadership and long-term commitment. As a resource-constrained environment, entrepreneurial leadership along with governance of HEEEs was perceived as far more important. Therefore, it is essential to direct attention to the much needed reorientation of leadership among higher education institutions.

Interviewee DH2, a head of school: "So we have the heads and the deans who are leading this entrepreneurial initiative, who are responsible to a large extent, and they must do it within their capacity. Then we have the academics and the educators ... Without the support of the deans, such initiatives would fail within a private education institution. So you definitely need commitment from the deans to sustain those initiatives. Because even if you have alumni, entrepreneurs, and other experts who in my opinion can contribute towards developing these entrepreneurs, no higher education institution can contribute to the same if you don't have the support and commitment from the top. If senior management does not strongly believe in the need for creating that mindset and developing student entrepreneurs, they will continue to push students for cooperate jobs."

Interviewee AC3, an academic teaching entrepreneurship: "We encourage students for entrepreneur. Right now. We don't have that leadership promoting leadership although it is getting prominent as a country. We have our school-level objectives aligned to entrepreneurship. But the fact is, what's really happening at operational level, I do have a question mark on that. Because we are limited when they try to do things because there is limited support from the top. In a country like ours, we need our leaders to realise our national priorities and work together for those, even as a private company right?"

While the influential role was the most apparent, the other four roles of developmental, networking, entrepreneurial and regional were expectations of HEEEs from diverse stakeholders. With respect to the developmental role, an explicit expectation surfaced from the interviews that students need to be entrepreneurially developed with the capabilities and not just prepared academically. The developmental role of nurturing students with capabilities relevant for entrepreneurship was not evident within HEEEs in Sri Lanka as one would expect given that higher education institutions are entrusted with the responsibility of human capital development. Analysis of data extracts suggests that developing capabilities of students goes beyond theoretical knowledge and practical skills, including an element of self-awareness and management as an entrepreneur. For this, higher education institutions must be open to the environment and draw input from stakeholders in the entrepreneurial ecosystem on issues and

trends. All stakeholders should be involved in developing HEEEs (Rice, Fetters & Greene 2014) and the development of capability among students needs to be a collaborative effort, especially when faced with resource constraints.

Expectations associated with the networking role were strongly represented in the interview data. An HEEE is expected to act as a network facilitator for students, enabling them to make connections with a variety of relevant stakeholders of the HEEE and its wider entrepreneurial ecosystem. Stakeholders pointed out that students have limited access to others and must be given the opportunity to be exposed beyond their HEEE and institution. Given the importance of social influence, it was suggested that students can benefit from building immediate relationships with students from the same and other institutions, and alumni entrepreneurs. Another point about networking was students being connected to the purpose of mentoring and consulting. Students also seek the views of others and societal trends in their development as potential entrepreneurs. Higher education institutions should build communities through their HEEEs to advocate for entrepreneurship and support student E&I capabilities. For resource-constrained environments, it is of utmost importance for stakeholders within HEEEs and in the wider entrepreneurial ecosystem to work closely and they are stronger together.

Student start-ups launched through higher education institutions and their HEEEs have become an important contribution to economic development (Di Gregorio & Shane 2003; Wright, Siegel & Mustar 2017). Being a resource-constrained environment, the role of HEEEs being entrepreneurial is almost non-existent in Sri Lanka. In this role, HEEEs should act as a hothouse for all things entrepreneurial with various factors and diverse stakeholders, motivating students to engage in entrepreneurship through student start-up projects and enabling spin-offs through research. Such a role would mean that HEEEs reach their potential sustaining the momentum for entrepreneurial initiatives. Currently, higher education institutions and their HEEEs have not played a role in such an entrepreneurial initiative and have almost no noteworthy example as evidence. Some external stakeholders pointed out that this entrepreneurial role would be an ultimate outcome for HEEEs to achieve but higher education institutions are not just there yet. If necessary HEEE factors are not present due to resource constraints, mapping and creating points of connections to support is a means of providing a pathway for students who are interested in start-ups. Given resource constraints of HEEEs, they see possibilities for start-ups projects and spin-offs through engagement with existing incubators and accelerators, and sponsorships from individuals and organisations in the wider entrepreneurial ecosystem.

Regional roles emerged primarily from the views of external stakeholders as a responsibility that current HEEEs are lacking. In the context of private higher education institutions in Sri Lanka, it became evident that there is a disconnect between the private higher education sector and national governance. At the institutional level, goals and objectives are less aligned with priorities of the country and operate more like a private business in education. At a national level, government bodies should attract the commitment from HEEEs and communicate expectations from the private higher education sector. HEEEs along with their institutions are expected to lead human capital development and provide entrepreneurial talent for the economy and entrepreneurial ecosystem. Based on these views, HEEEs should consider their regional role and scale up their commitment and leadership. In this case, HEEEs can make the education and experience of students more relevant to their country's needs, synthesise with national development and be in line with the world today.

The emergence of these HEEE roles add depth and breadth to the understanding of the development and functioning of HEEEs. Although the contributions from the five roles may have little radical new insights, the composite understanding of the various roles provides indepth knowledge of how HEEEs can serve, as an entrepreneurial initiative and for the wider entrepreneurial ecosystem. Table 6.4 presents the five roles of HEEEs with their associated type of role and contributions/expectations, drawing on the synthesis of interview data relating to HEEEs' continued development. As described, it is a process of evolving in a systematic manner and the five roles can be understood as HEEE development in a resource-constrained environment. This acts as an inspiration for higher education institutions and stakeholders in the wider entrepreneurial environment on how to advance the HEEE and grow together.

6.4.2 Bridging to the wider entrepreneurial ecosystem

Generally, HEEEs are considered to be self-sustaining ecosystems and in this sense they are not part of a wider entrepreneurial ecosystem, although they interact with regional stakeholders (Lahikainen et al. 2019; Miller & Acs 2017). However, this research recognises the HEEE as a sub-ecosystem of the wider entrepreneurial ecosystem being a system of ecosystems (Wurth, Stam & Spigel 2021). Viewpoints from both internal and external stakeholders highlight that HEEEs and entrepreneurial ecosystems may be complementary, interconnected and independent in how they elevate to reach each other and as they relate to each other. In their ongoing development, HEEEs cannot be separated from their wider entrepreneurial ecosystem and need to embrace support from the entrepreneurial community in delivering their key roles, including the development of E&I capabilities. To evolve into a fully-fledged HEEE, they must open up to their regional milieu and specifically its entrepreneurial ecosystem.

Especially in such resource-constrained environments, entrepreneurial initiatives within the HEEE can benefit from collaborating with the wider entrepreneurial ecosystem including stakeholders, such as other higher education institutions, entrepreneurs, investors and incubators. Although HEEEs are associated with a specific higher education institution, the various members share the same goal within a local geographic community. Formal and informal networks and connections between HEEEs and ecosystem stakeholders can enhance access to resources and contribute to optimal configuration (Theodoraki, Messeghem & Rice 2018). Stakeholders from the wider entrepreneurial ecosystem were clear and consistent that HEEEs need to build a layer of external stakeholders in addition to senior management, educators, students and staff who share the same goals and priorities. In this case, there is a need for HEEEs to create a bridge with their wider entrepreneurial ecosystem that can result in resource sharing. For a resource-constrained environment, the agenda of an HEEE should be formulated and revised constantly for the purpose of sustaining synergy and commitment for mutual benefit from various stakeholders.

Some HEEEs may emerge as a proactive response to new economic development or education initiatives while other HEEEs may be reactive responses to specific gaps in economic development or education resources (Rice, Fetters & Greene 2014). Therefore, stakeholders argue that higher education institutions play the role of a hub organisation in the wider entrepreneurial ecosystem where their HEEEs depend on collaboration and coordination between autonomous yet linked stakeholders. In a resource-constrained environment, HEEEs are characterised by this crucial role that leads to creating value for connected ecosystems. In doing so, higher education institutions need to find a balance between academic and market logic, as well as the logic of the entrepreneurial ecosystem, which consists of the entrepreneurial-market logic and community logic (Roundy 2017). The entrepreneurial-market logic includes guiding actions in the pursuit of creativity, innovation and development of new business models (Roundy 2017). When doing so, HEEEs can expose students to entrepreneurial-market logic through various factors and mechanisms including study programs, research and projects and thereby develop their capabilities in E&I. Further,

community logic will bring a community focus among students resulting in solving problems through E&I initiatives while helping the community.

6.4.3 Higher education institutions as catalysts

Given that HEEEs are an entrepreneurial initiative of higher education institutions, these institutions play a crucial role in their development. Higher education institutions claim that they have made significant investments, such as in their study programs and learning environments. However, stakeholders at the other end engaging with graduates question the relevance and usefulness of programs and efforts made by higher education institutions. To some extent, higher education institutions are managing their HEEEs poorly and do not have sufficient support to fulfill their function of entrepreneurship education and entrepreneurial support. They seem to have ignored creating the community value that should guide their HEEEs towards E&I initiatives for students. Continued development of HEEEs must address this relevance and resource constraints by developing the E&I capabilities of students.

One way to think about the continued development of HEEEs is the source of development itself. According to the interview data, self-employment is valued by academics but not by academia in Sri Lanka. The senior management and larger community are still fixated on preparing students for corporate jobs. Other stakeholders including entrepreneurs, investors and support professionals criticised this fixation, bringing a focus to the real promise of higher education. Academia's fixation on corporate careers for students can restrict the continued development of their HEEEs, leading to lack of relevant capabilities among students. Thus, external stakeholders are urging higher education institutions to reconsider their priorities and strategies by emphasising the needs of students and the broader society. HEEEs should focus on their inner environment, nurturing students and paying attention to the specific needs of students, to develop them for different careers, most importantly their chosen career. Below is an extract of an interview that relates to this notion:

Interviewee EE7, a social entrepreneur: "These institutes should ask themselves whether they want to produce students for corporate jobs and supply employers or help students create jobs as new employers. They need to really ask who they are serving. Is it the employers and job market or it the students. My understanding is that their responsibility is students and should focus on giving the best for its students and now focus on things like how many students got employed after graduation, how quickly that was and pitch the highest paid jobs. Instead, they have a responsibility to expose students to different career path including entrepreneurship."

Higher education institutions are urged to understand the difference between preparing corporate-fit employees and developing students keen to pursue an entrepreneurial career. External stakeholders find that entrepreneurs have a greater propensity for innovation as it is part of the entrepreneurial process. Creating an innovative solution that solves a problem and starting a new business is inherent in entrepreneurship. It is only one side of the coin to develop students for entrepreneurship. It is another not to foster creativity and innovation that will support the entrepreneurial journey.

When considering 'what' is stopping higher education institutions from developing students' E&I capabilities, answers point mostly to the scarcity of resources. The higher education institutions that co-create and manage HEEEs suffer from two scarcities: financial resources and human resources. Both are important for the effective functioning of HEEEs. If HEEEs bridge the gap, they can attract resources from the wider entrepreneurial ecosystem including alumni entrepreneurs, other higher education institutions, financial institutions, angel investors, private companies, support professionals and even government organisations. Just as entrepreneurship education and entrepreneurial support can equip students with the relevant capabilities to master the challenges in entrepreneurship, so can resources make HEEE environments conducive to entrepreneurship. More on this is found in Chapter 9, on stakeholder engagement.

Only two out of six higher education institutions seem to have mapped institutional goals and objectives against the United Nations SDGs and commitment to entrepreneurial initiatives for their students. Higher education institutions participating in sustainable development highlight that leadership plays a more important role than governance, and stronger leadership results in more efficient outcomes from entrepreneurial initiatives (Goldstein & Glaser 2012). External stakeholders point out that more higher education institutions need to champion the integration of entrepreneurship education and entrepreneurial support in their HEEEs and empower students with capabilities in E&I. Unless higher education institutions act as catalysts with responsibility and invest in the development of their HEEEs, their value to students and society is at risk. Such a transformation might show students that higher education institutions are invested in their education needs and are valued. Thereafter, higher education institutions need to identify the composition of their HEEE that

contributes to students E&I capabilities. More on this is found Chapter 7, on the composition of HEEEs.

6.5 Chapter summary

This chapter presented the findings relevant to the first research question of this study. The findings resonate with a multi-stakeholder perspective on the development of HEEEs in a resource-constrained environment. Diverse stakeholders shared their viewpoints on the status of HEEEs in the private higher education section and how they anticipate the continued development of HEEEs. A three-level content analysis found the aggregated themes of entrepreneurial intentions, entrepreneurial mindsets and E&I capabilities. E&I capabilities is a new construct that emerged from this study. Stakeholder views on these three pathways are graphically represented in the reorder matrix in Figure 6.2, which highlights their perceived level of importance for each pathway. While student start-ups, entrepreneurial intentions and entrepreneurial mindsets are more common in HEEEs of developed countries, there was a contrasting finding on the development of HEEEs in a resource-constrained environment. In the developing context of Sri Lanka, there was a strong need for developing the E&I capabilities of students. This study suggests the key roles and contributions of HEEEs by identifying five specific roles of influential, developmental, networking, entrepreneurial and regional. Having drawn out views on the continued development of HEEEs, the chapter closes with two sections on the broader role of HEEEs bridging to their entrepreneurial ecosystem and higher education institutions as catalysts.

CHAPTER 7: FINDINGS AND ANALYSIS FOR RESEARCH QUESTION 2a

7.1 Chapter overview

This empirical chapter relates to the second set of research questions about what diverse stakeholders perceive as the contextual factors of HEEEs that could influence students E&I capabilities in a resource-constrained environment. While prior studies have garnered lists of factors for HEEEs in developed countries, such factors may work differently in a resource-constrained environment (Bedő, Erdős & Pittaway 2020). The findings from this set of research questions recognise the contextual factors for students' E&I capabilities that involves developing E&I capabilities. Higher education management can co-create and evolve their HEEEs, including the design of their learning environments, by using these factors at an institutional level. The structure of this chapter is presented in Figure 7.1.



Figure 7.1 – Chapter 7 outline

Due to the increasing critique of studies resulting in a growing suite of what an ecosystem is rather than discussing how it works (Longva 2021), as an attempt for advancement, the findings addressing the second research question are analysed and presented in two parts. The first focuses on 'what' contextual factors diverse stakeholders perceive as the composition of HEEEs towards a pathway for students' E&I capabilities. The second focuses on 'how' these contextual factors translate into mechanisms in operationalising an HEEE for students' E&I capabilities and why. The following chapter is based on part one, the contextual factors, and the next chapter address the second part.

7.2 Analysis of diverse stakeholder perspectives on contextual factors

Within the two pathways identified, scholars have contributed to the composition of HEEEs by establishing the factors that impact student start-ups, and students' entrepreneurial mindset and intention (Longva 2021; Miller & Acs 2017; Rice, Fetters & Greene 2014; Webber, Kitagawa & Plumridge 2020). To contribute to the evolution of HEEE composition, one of the second research questions explore contextual factors that diverse stakeholders perceive as leading to students' E&I capabilities.

The second research question involves contextual factors leading to students' E&I capabilities within HEEEs in a resource-constrained environment, through the perspective of internal and external stakeholders. In search of themes relating to research question 2a, a comprehensive content analysis was conducted. As discussed in the research design, all 40 interview transcripts were fed into NVivo to identify themes and produce an account of findings. For this analysis, a problem-to-outcome approach was taken where data was organised into three levels: (1) the prevailing problem, such as lack of institutional direction; (2) various mechanisms suggested by diverse stakeholders to address the problem, such as senior leadership, entrepreneurial culture and responsible staff; and (3) the outcome as a contextual factor, being entrepreneurial orientation. As the study investigated a resource-constrained context, Sri Lanka, it was meaningful to undertake a problem-to-outcome approach for the analysis. Interviewees spoke about problems, barriers and challenges (refer Table 7.1) related to contextual factors justifying why they made their suggestions. This content analysis involved two coding levels that resulted in aggregated themes (Figure 7.2), which are defined for better understanding in Table 7.2.

The themes that emerge help to answer the research questions of a study (Creswell & Creswell 2018). According to Figure 7.2, six aggregated themes emerged from the 40 interviews; each theme represents a contextual factor relating to entrepreneurship literature and is discussed below. In answering the second research questions, ecosystem stakeholders perceive that HEEEs can develop students' E&I capabilities through entrepreneurial orientation, E&I education, enterprising experiences, E&I research, entrepreneurial networks and entrepreneurial support. The contextual factors are reported with excerpts from the interviews with diverse stakeholders. The qualitative data was edited to improve readability and is not reported as verbatim (Jones et al. 2015; Rasmussen & Borch 2010). In Section 7.2, a block-and-file approach was utilised to present meaningful groupings of excerpts in columns as this approach supported better clarity rather than presenting separate excerpts (Grbich 2009).

Table 7.1 – Descriptions of initial codes

Initial codes	Descriptions
Lack of institutional	National priorities versus organisational priorities
direction	Strategic objectives versus operations
	Preference on employees versus self-employment careers
Lack of business acumen	No/limited entrepreneurship courses/units
	More theory and less practical knowledge
	Standard assessments
Lack of research skills	Less emphasis on research in coursework
	No/limited opportunities for research projects
	Limited scientific knowledge as new knowledge
Lack of enterprising	Less opportunities for participating in competitions
exposure	Internships with multinational companies versus start-ups internships
	No/limited opportunities for start-up projects
Lack of connections	Less social interaction among like-minded peers interested in
	entrepreneurship
	Seek for confidence and validation from senior/alumni entrepreneurs
	It is not 'what' but 'who' you know outside the institution
Lack of support	Need support to develop identities as potential entrepreneurs
	Need resources to launch entrepreneurial initiatives
	No/limited support to accelerate start-ups and develop their capabilities

Table 7.2 – Descriptions of aggregated themes

Aggregated theme	Description
Entrepreneurial orientation	Entrepreneurial leadership, entrepreneurial culture and responsible staff that drive institutional efforts promoting entrepreneurship and innovation
Entrepreneurship and innovation education	Curriculum, pedagogy, and assessments that enable teaching and learning entrepreneurship and innovation for students to develop their knowledge
Enterprising experiences	Outside the classroom activities including business idea competitions, internships with start-ups and start-up projects that allow students to gain exposure
Entrepreneurship and innovation research	Various research initiatives including research-based coursework, research-based projects, and research conferences for students to improve research skills
Entrepreneurial networks	Contacts and interpersonal relations between students-students and students-external stakeholders to access and develop connections
Entrepreneurial support	Services and infrastructure including mentoring, incubators, and accelerators to support students develop capabilities and become potential entrepreneurs

RQ2a: What do diverse stakeholders perceive as the contextual factors of HEEEs that could influence students E&I capabilities in a resource-constrained environment? RQ2b: How and why do specific contextual factors of HEEEs influence students E&I capabilities in a resource-constrained environment?



Figure 7.2 – Theme structure for research question 2

*See tables 7.1 and 7.2 above for descriptions of initial codes and aggregated themes.

7.3 HEEE factors for students' E&I capabilities in a resource-constrained environment

7.3.1 Entrepreneurial orientation

The concept of entrepreneurial orientation refers to the configuration of processes, practices and policies that offer insights into the creation of entrepreneurial actions and decisions (Lumpkin & Dess 2015; 1996). Entrepreneurial orientation was initially associated with superior performance of an organisation (Bell 2019; Rauch et al. 2009) and entrepreneurial orientation continues to be useful in the strategic orientation of organisations (Lumpkin & Pidduck 2021). In this case, entrepreneurial orientation influences the management driving the organisation towards entrepreneurship (Poon et al. 2006; Dess & Lumpkin 2005).

Most of the stakeholders observed that higher education institutions lack institutional direction and development towards entrepreneurship. Academics/educators pointed out the need for senior leadership to promote entrepreneurship within the institution. Some 71% of external stakeholders including established entrepreneurs, angel investors and support professionals are of the view that higher education institutions can benefit from entrepreneurial orientation by aligning better to teaching, research and entrepreneurship.

As shown in Table 7.3, both internal and external stakeholders perceive that higher education institutions need entrepreneurial orientation in co-creating HEEEs to develop students' E&I capabilities. This shared consensus leads to recognising entrepreneurial orientation as a contextual factor for HEEEs in resource-constrained environments. In this case, entrepreneurial orientation is found to be necessary for actively driving institutions in resource-constrained environments. Entrepreneurial leadership, entrepreneurial culture and entrepreneurship chair (staff) are identified as mechanisms for operationalising entrepreneurial orientation; these are discussed in the next chapter.

There have been research efforts to investigate entrepreneurial orientation in a variety of contexts, including cities (Feldman 2014; Brown & Mason 2013). Scholars such as Lumpkin and Pidduck (2021) argue that entrepreneurial orientation in new contexts must capture what it means for organisations to be entrepreneurial, and characteristics of entrepreneurial orientation may differ. Among studies such as Webber, Kitagawa and Plumridge (2020), Bell (2019) and Olutuase et al. (2018) entrepreneurial orientation has been related to entrepreneurial intention to understand what makes students more malleable to behave entrepreneurially. None of the identified HEEE studies explicitly discussed entrepreneurial orientation as a contextual

factor, except an early study that suggested senior leadership, vision and sponsorship as success factors of HEEEs in developed countries (Rice, Fetters & Greene 2014).

Interviewee AI3 , an angel investor working with young entrepreneurs:	Interviewee AC5 , an academic teaching a career planning unit:	Interviewee PS4 , an external mentor in entrepreneurship:
"So, there's Babson, Stanford, there's MIT, you know, if you look at the US universities and compare with Sri Lanka. What do we lack? From my observations and experiences, we lack people from the top enforcing entrepreneurship or bringing about that change into the institute. If senior management does not value or embrace entrepreneurship, how can we expect the future generation to be aware of entrepreneurship and what entrepreneurship is about? So where do they get the knowledge from if not from the institute? For example, Babson has a focused entrepreneurial program and they have done that for a very long time. But I don't know of many private institutions in higher education that offer entrepreneurship as a degree. There is one or two."	"During their education journey, students look for people they can relate to and learn from. Of course, we are there as lecturers. I have come across a couple of students who are interested in becoming entrepreneurs. So, they look for that person they can talk to, someone who can answer their questions, give them direction and learn everything entrepreneurship. Sometimes it is not the lecturers they look for. They want to know how it is like being an entrepreneur, how to improve themselves for it and things like that."	"Especially in a country like Sri Lanka, it is not just about teaching entrepreneurship as a module or course. I believe that institutes should focus on developing that spirit, as well as that mindset, and you know, encourage an entrepreneurial culture within the student population. Of course, these shared values should come from the top and spread within the institute. This is mainly because entrepreneurship can happen at any point in their careers and life. And this is the attitude we need among students, our future youth. You know some youth today have a 9am to 5pm full time job and work on a start up from 5pm to 9pm. But this is just one instance. There are others who change gears from a full- time employee to a self-made employer after a few years of working for somebody else."

Table 7.3 – Excerpts on entrepreneurial orientation

7.3.2 Entrepreneurship and innovation education

Compared to general education, entrepreneurship education is focused on stimulating entrepreneurial knowledge and skills while promoting entrepreneurship as an alternative career (Verheul et al. 2001). Entrepreneurship education literature has been categorised into three types: education 'about', 'for' and 'through' (Caird 1990). The association between educational attainment and entrepreneurship suggests that better education increases entrepreneurial activity by strengthening entrepreneurial intentions (Poschke 2013; Martinez et al. 2010; Fayolle et al. 2006). Entrepreneurship education can develop the ability and attitude of students towards entrepreneurship (Volkman et al. 2009; Hansemark 1998). Research has established that education empowers students for an entrepreneurial career (Katz 2003; Kuratko 2003;

Meyer 2001). HEEEs as an initiative of higher education institutions being entrepreneurial are primarily focused on delivering entrepreneurship education (Rice, Fetters & Greene 2014).

All stakeholders agreed that entrepreneurship education is important, however there is mixed rationale for this importance. Like entrepreneurial orientation, the importance of E&I education was heard from academics/educators more than deans/heads of schools. It may seem natural for academics/educators in entrepreneurship to validate the significance of entrepreneurship as a field of study. In contrast, alumni and established entrepreneurs recognised the need for E&I education based on prevailing insufficient business acumen and lack of relevant skills among entrepreneurs, which leads start-ups towards failure. Some 66% of external stakeholders supported the contextual factor of E&I education. They view that education lays a good foundation for students in undertaking an entrepreneurial career and becoming a potential entrepreneur.

Ecosystem stakeholders are confident that capabilities of students can be developed through E&I education. Like representatives from higher education, external stakeholders such as alumni and established entrepreneurs also consider E&I education as a factor critical for students to develop and successfully engage in entrepreneurship (refer Table 7.4). Curriculum, pedagogy and assessments emerged as mechanisms for E&I education and they are discussed with respect to how they can be operationalised within HEEEs in the next chapter.

Among the identified HEEE studies, entrepreneurship education is one of the contextual factors that is constant in each HEEE (Longva 2021; Webber, Kitagawa & Plumridge 2020; Guerrero, Urbano & Gajón 2020; Meyer et al. 2020; Shil et al. 2020; Secundo et al. 2020; Allahar & Sookram 2019; Wright, Siegel & Mustar 2017; Miller & Acs 2017; Rice, Fetters & Greene 2014). In line with Longva (2021) entrepreneurship education is essential to the HEEE. Through HEEEs, higher education institutions contribute to E&I and not just entrepreneurship (Webber, Kitagawa & Plumridge 2020). This research also found the need to investigate education for E&I. This is in line with analysing entrepreneurial and innovative ecosystems as an independent phenomenon (Guerrero & Urbano 2019).

Interviewee EE1, a second-	Interviewee AE5, an alumni	Interviewee AC1, an academic
generation entrepreneur and co-	entrepreneur of a start-up:	teaching entrepreneurship:
founder of a global brand:		
"I've always looked at education, especially higher education, as 'you learn to learn'. Education gives you the skill set, and it gives you the knowledge. For example, what they learn gives students the knowledge to structure a start- up, to structure their thoughts, to structure business plans, and it gives students a widened knowledge about the different aspects or different spectrums within a business. Most importantly, it gives students the necessary tools to keep learning, absorbing information, and present information and so on, which is integral to running a	"As an entrepreneur, the challenge is not coming up with a solution. The challenge is many times understanding a problem deep enough. Beyond the superficial surface level. So, for that, you must have particular knowledge and skills through the course one is taking. Now institutes must make sure students get the theoretical knowledge and more practical skills like problem solving and creative thinking from the assignments they do. This is where the real capabilities develop in my understanding."	"If you look at the nature of entrepreneurship, it involves finding an innovative solution to a particular problem. This practical thought process can be developed or supported through assessments to arrive at practical outcomes. Some assessments push students to arrive at more than a written report and be involved in doing projects. These activities support to get the so-called entrepreneurial attitude into their mind and body. From such assessments, they learn from their mistakes too."
business or creating a brand."		

7.3.3 Entrepreneurship and innovation research

Research can be theoretical and applied across the discipline of entrepreneurship with the primary focus of dissemination of findings and access to data (Brush 2014). Higher education institutions should continue to emphasise their core activities, specifically research including theoretical and applied research, to develop students' E&I capabilities. The importance of E&I research for HEEEs was less common among internal stakeholders with just a 20% representation. In contrast, external stakeholders highlighted applying research among start-ups and developing research skills as a distinguishing capability for potential entrepreneurs. Stronger views of E&I research as a contextual factor HEEEs were from alumni and established entrepreneurs. For instance, interviewee SP3, a support professional who is an external mentor in entrepreneurship, commented:

"Research is very very crucial in entrepreneurship. Because information and data are key. Back in the day, we have heard about gut-feeling in taking decisions. But things have changed, and anything related to the product, market, competitors, environment and especially a new start-up is research driven. I have seen start-ups fail when they are built on poor research, especially market research. So, it's an important skill in planning, launching, and managing a start-up. Even coming up with innovative solution is linked to research. I would say research skills is among the top five skills for a successful entrepreneur. For student entrepreneurs, research is the starting point of their start-up."

To successfully develop students' E&I capabilities, a research experience is deemed essential by practitioners. Excerpts in Table 7.5 demonstrate the core intuition among academic and non-academic stakeholders, crediting the importance of using new knowledge in their startup and fostering the ability to research among students. Research may benefit students at various points during their entrepreneurial journey right from the beginning of a start-up through its commercialisation to its development. From this, E&I research is revealed as an HEEE contextual factor. Interviewees suggested research in terms of research-based coursework, research conferences and research-based projects within HEEEs. The three mechanisms for E&I research are further discussed relating to operationalising HEEEs in the following chapter.

In the late 19th century, an academic revolution introduced research into the university mission, adding to the primary responsibility of teaching (Etzkowitz & Leydesdorff 2000). In relation to entrepreneurship, research involves testing the hypothesised truth and arriving at new knowledge to support decisions for entrepreneurs (Brush 2014). US campuses operate in open and resource-rich environments where research is an asset providing impactful outcomes and research collaborations are instrumental to reputation and engagement to the outside world (Miller & Acs 2017). A strong ecosystem requires resources for research to develop the entrepreneurial knowledge and skills of students (Mason & Brown 2014). Leading research-driven institutions are behind the stories of successful ventures such as Google and Yahoo from Stanford and Facebook from Harvard, where the understanding is that research is a key resource for potential entrepreneurs (Walter et al. 2013).

Table 7.5 –	- Excerpts on	entrepreneurship	and innovation	research
-------------	---------------	------------------	----------------	----------

entrepreneur with tech start-ups:ac"Research plays a major role in""entrepreneurship. Even though Ithcome from a tech backgroundcdand my start-ups are technologypdbased. I have had to drive inWinformation about everything.thUnfortunately, I wish I was morethcompetent in research throughthdeveloping my research skills.anStudents nowadays should doexresearch through theirth	academic officer: "When I joined this institute, they had an annual research conference for all schools to participate, students and staff. We have an entrepreneurship theme for the conference. For	mentor in entrepreneurship: "At the end of a degree, a student has done three-four years of coursework. Students can learn more when the coursework is linked with the
"Research plays a major role in "" entrepreneurship. Even though I th come from a tech background co and my start-ups are technology pa based. I have had to drive in W information about everything. th Unfortunately, I wish I was more th competent in research through th my education as I feel I am still di developing my research skills. an Students nowadays should do ex research through their th assignments and participate in the	"When I joined this institute, they had an annual research conference for all schools to participate, students and staff. We have an entrepreneurship theme for the conference. For	"At the end of a degree, a student has done three-four years of coursework. Students can learn more when the coursework is linked with the
come from a tech backgroundcomeand my start-ups are technologypabased. I have had to drive inWinformation about everything.thUnfortunately, I wish I was morethcompetent in research throughthmy education as I feel I am stilldideveloping my research skills.anStudents nowadays should doexresearch through theirth	conference for all schools to participate, students and staff. We have an entrepreneurship theme for the conference. For	years of coursework. Students can learn more when the coursework is linked with the
research through their th	his, students present posters of their research from their dissertation or business project and winners are selected by external judges. I remember	real world and requires them to use research to to attempt the coursework. Like when mini cases are used to set the scene but to actually answer the questions students need to do
research events. Because when you are an entrepreneur, you need to know everything related to your business functions, industry trends, everyday challenges, and even what's happening internationally. That's what improves your knowledge and understanding to run the show when one is in this entreprenergy and the second	nere was a student who did her research dissertation on entrepreneurial intention among undergraduates, and she was keen to understand the increasing trend in this new career path. So, such attempts are driven by their future behaviour."	their own research. I believe this is a way to inbuild research into their learning, so students develop that research-based attitude by searching and using information for managing a start-up. This is for entrepreneurship related units but not just for these. Be it management, marketing or finance, it is still important for student to know how to attempt neglewoold workhows "

While some scholars argue that students engaged in research are more likely to possess superior knowledge and skills to identify opportunities and develop ideas (Ucbasaran et al. 2008; Arenius & De Clercq 2005), there is a view that the impact of entrepreneurship research on such students is limited (Neck et al. 2011). Among the 10 identified HEEE studies, research is only suggested as a contextual factor in the study by Miller and Acs (2017), alongside factors such as education and extracurricular activities. In this study, external stakeholders pointed out that students can learn and develop from research and knowledge, scientific and practical, should not be discounted in the HEEE. This emphasis is based on the view that lack of market research skills is one of the key reasons why start-ups fail in Sri Lanka, especially in the first two years of inception.

Compared to internal stakeholders who did not express the importance of research within the HEEE strongly, external stakeholders related experiences that highlighted the need for start-up founders to be data driven and empowered with research skills. Angel investors and support professionals including mentors indicated that some start-up founders have made less consideration of the customer and their need for the product due to lack of market research.

This led to a major shortcoming where such start-ups are not solving a problem in the market and suffer from not being able to attract customers for the new business. Often, founders who failed in their start-up addressed problems that are interesting to solve with a market offering or based a start-up on their capabilities and passion rather than serving a market need. These external stakeholders claim that start-up founders spend the least amount of time exploring the market and understanding unmet customer needs as doing research does not seem to be part of their system. Alumni and established entrepreneurs argued that secondary market research in Sri Lanka is limited or almost non-existent and conducting primary research is expensive.

7.3.4 Enterprise experiences

Enterprise experiences relate to undertaking various projects including volunteering and internships (Webber, Kitagawa & Plumridge 2020) and this study shows mixed results for this contextual factor. While extracurricular activities are popular among higher education institutions, internal stakeholders hold a moderate view of enterprise experiences. In Sri Lanka, lack of exposure to enterprise experience is a pressing concern according to external stakeholders. They suggest that HEEEs should offer enterprise experiences and not merely co-curricular and extracurricular activities that enrich student learning experiences. Some 74% of external stakeholders highlighted the factor of enterprise experiences, with alumni and established entrepreneurs advocating strong opinions. Unlike co-curricular and extracurricular activities that emerged in this study is specific to entreprise experiences a strong pillar of HEEEs that enable students to acquire practical applications of entrepreneurship in real or almost real situations.

Activities outside the classroom are opportunities for enterprise experiences, allowing students to engage in collective learning where they can learn from the wider community. Table 7.6 presents snippets from interviews where multiple stakeholders express the value of enterprise experiences. Engaging in enterprise activities not only helps students to start thinking about what interests them and what their strengths are but to apply their knowledge in practical situations and learn from those experiences. Thus, enterprise experiences are perceived as a contributing factor of HEEEs by diverse stakeholders. Higher education institutions should create opportunities for students to reach beyond the institution to engage with industry and entrepreneurial communities.

Most of the HEEE studies identified in this study include various activities representing enterprise experiences (Longva 2021; Webber, Kitagawa & Plumridge 2020; Shil et al. 2020; Secundo et al. 2020; Meyer et al. 2020; Miller & Acs 2017; Wright, Siegel & Mustar 2017; Rice, Fetters & Greene 2014). In HEEE literature, enterprise activities including business plan competitions (Shil et al. 2020; Secundo et al. 2020; Meyer et al. 2020; Miller & Acs 2017; Wright, Siegel & Mustar 2017; Rice, Fetters & Greene 2014), volunteering (Webber, Kitagawa & Plumridge 2020), work experiences or internships (Shil et al. 2020; Meyer et al. 2020; Webber, Kitagawa & Plumridge 2020; Miller & Acs 2017) and start-up projects (Longva 2021; Secundo et al. 2020; Wright, Siegel & Mustar 2017) have been found to affect start-ups and entrepreneurial intention. While volunteering did not strongly emerge from this study, entrepreneurial intention. While volunteering did not strongly emerge for this study, entrepreneuring intentions (business idea, venture), internships with a start-up and start-up projects are common mechanisms that stakeholders suggest for students to gain a valuable enterprise experience helping them develop capabilities for E&I through the HEEE. These three mechanisms for enterprising experiences within HEEEs are further discussed in the next chapter.

Interviewee EE5, an	Interviewee EE4, an award-	Interviewee AE4, an award-
entrepreneur of a fashion start-	winning entrepreneur:	winning social entrepreneur:
up:		
"During their education students should have access to competitions. They will get more confidence, meet new people, and network and even get inspired. When they compete, they become independent and resilient preparing them for future situations. I can say that competitions were helpful in my experiences. For me, such competitions makes students aware of the knowledge they need to bring a start-up together. But I don't see much of it being organised."	"Institutes should put students into the real-life situations like internships to work for entrepreneurs and start-ups where they have to do things and learn by doing. This is where students will understand their limitations and recognise their skills. If they sit in a classroom, listen to a lecture and do an activity they will not necessarily discover their strengths and weaknesses because they are not facing real life situations."	"Yeah, young students need personality building sort of experiences. In addition to teambuilding, I feel we lack the confidence to go out there and you know, pitch an idea, you know, just you have the drive as an entrepreneur, they're afraid to even ask a question. Right? So, we are very backward compared to other cultures. Participating in mini projects can break these barriers and develop the right attitude among students, especially if they get involved during higher education."

Table 7.6 – Excerpts on enterprising activities

Enterprise experiences such as business competitions are an enduring enabler for students (Morris & Pryor 2013). For example, the University of Chicago launched the New Venture Challenge, a business competition for students, with a vision to encourage students to

learn about entrepreneurship and acquire a skill set from their experience. In two years since its inception, the business competition has attracted such significant entrepreneurial interest that it needed to take a much larger form, prompting the entrepreneurship centre (Miller & Zoltan 2017). Giving students a meaningful opportunity to practise and experience is critical to encouraging interest in entrepreneurship (Barr et al. 2009) and creating enthusiasm within the ecosystem (Fetters, Greene & Rice 2010).

7.3.5 Entrepreneurial networks

Both groups of internal and external stakeholders expressed the important influence of entrepreneurial networks in capability development. There was a high level of recognition in entrepreneurial networks among internal and external stakeholders. Deans/heads of school and academics/educators acknowledge the importance of entrepreneurial networks for students to pursue a career in entrepreneurship and develop necessary capabilities. Almost 70% of external stakeholders, alumni entrepreneurs and angel investors perceive that students lack connections beneficial for their entrepreneurial career. They explained that entrepreneurial networks enable students to become a part of an entrepreneurial community and see themselves as potential entrepreneurs.

Entrepreneurial networks contribute to entrepreneurial capabilities including human, financial and technical capacities (Jack, Doff & Anderson 2008). Such networks not only offer students opportunities to build connections and relationships with peers, alumni and other stakeholders but also the leverage they need for entrepreneurship. The excerpts in Table 7.7 indicate how these networks and relationships contribute to student's E&I capabilities through internal and external stakeholders. Ecosystem stakeholders identify entrepreneurial networks as a contextual factor of HEEEs where entrepreneurial networks inspire, facilitate and promote entrepreneurship while engaging students with other stakeholders. Interviewees from this study suggest entrepreneurial networks in the form of peer engagement, alumni networks and access to external stakeholders; these are further discussed in the next chapter, on operationalising networks in HEEEs.

Entrepreneurs are a result of their social environment (Anderson & Miller 2003). When in the presence of successful and experienced entrepreneurs, the community, including students, receive inspiration for entrepreneurship as a career option, and the absence of such connections discourages new venture creation (Gnyawali & Fogel 1994). Interviewee EE4, an award-winning entrepreneur, commented on how entrepreneurial networks among students, alumni and external stakeholders influence students for entrepreneurship:

"During my higher education, we didn't have clubs or communities but outside the institute, I was a member of the Model United Nations Future World Leaders Summit. Up to date, I am still connected to many people I got to know and use those relationships and the friendships that I built for the benefit of my business. If we network, all these people come up in life. Especially in Sri Lanka, everybody knows each other. It's not about what you know, it's whom you know, in our part of the world because the systems are not in place properly and resources are not widely available. Because of this, these networks, interactions and relationships give budding entrepreneurs the leverage."

Interviewee AC3 , an academic teaching an entrepreneurship-related unit:	Interviewee SP5 , an external mentor in entrepreneurship:	Interviewee EE3 , a second generation entrepreneur:
"Currently, we have all one-off events where an entrepreneur will come for a talk, share his or her entrepreneurial story, take questions from students and leave inspiring the students. But this is not enough so we created an entrepreneurial club among current students. We believe it is useful to connect aspiring students, peers from various fields of study, network on a going basis and build a supportive community of potential entrepreneurs. This way they meet regularly, share knowledge, and produce outcomes a club and individuals. Such a networking arrangement facilitates 'learning by doing' taking students beyond aspiration and equipping them with practical knowledge to launch a start-up at some stage."	"Networking is an essential part of entrepreneurship and entrepreneurial networks with people outside the institute have various benefits for students. Students can talk about ideas and validate their business. Even ask the 'how' questions from people who have successfully started ventures or assisted in the process. There are other benefits also like tips and tricks, training opportunities, research and more. These can impact a students' behaviour and not mere intention. One may even meet potential partners, investors and mentors through external networks. I have actually learnt a great deal and benefited from my networks. Your network becomes your social capital."	"Education is not just about going to university and learning the theory of some subject. But the exposure which is way beyond. Meeting people from the outside world, interacting with them, getting to know about the society, people's behaviour, lifestyles and learning about environments. It's really important for students to learn about the society and learn from the broader community. This Is the kind of social learning and attitude that students should receive to prepare themselves for something as entrepreneurship."

 Table 7.7 – Excerpts on entrepreneurial networks

Networks offer higher education institutions the ability to collaborate with stakeholders including industry, government and others in the external environment (Yi & Uyarra 2018; Hayter 2013; Anderson & Jack 2010). The dynamic interconnectivity of stakeholders within

the ecosystem is argued to add energy to the development process of students (Wraae & Thomsen 2019; Kolb & Kolb 2005). When learning, the entrepreneur experiences, reflects and theorises and their thoughts have a 'social character' (Pavlica et al. 1998). This contrasts to learning approaches such as by Kolb (1984) that view the learner as 'an intellectual Robinson Crusoe' isolated from fellow beings. Since then, research has begun to acknowledge the influence of entrepreneurial networks in knowledge development and transfers in entrepreneurial learning and network-centred learning (Collinson & Shaw 2001). Therefore, an integral building block of the education process is the complex network of relationships including various ecosystem stakeholders such as alumni, entrepreneurs and more (Taylor & Thorpe 2004).

Most of the HEEE studies identified various forms of entrepreneurial networks (Longva 2021; Meyer et al. 2020; Shil et al. 2020; Secundo et al. 2020; Allahar & Sookram 2019; Wright, Siegel & Mustar 2017; Miller & Acs 2017). In the HEEE study by Longva (2021), it was suggested that students develop their social contacts and ties within the HEEE during their education journey. Her study showed that students access professional knowledge and advice through social networks with peers and faculty. This is consistent with literature that demonstrates that access to information is a key benefit of a well-developed network (Elfring & Hulsink 2003; Jenssen & Koenig 2002). Further, students tend to develop their identities and build self-confidence through their networks in the HEEE they are engaged in (Longva 2021). The development of a framework related to entrepreneurial learning would require and benefit from relevant networks (Taylor & Thorpe 2004).

7.3.6 Entrepreneurial support

Entrepreneurial support positively impacts on developing student E&I capabilities that have the potential to lead to entrepreneurship (Lichtenstein & Lyons 2001). Entrepreneurial support involves supporting students to start and scale their business through advice, training and resources (Hruskova & Mason 2020). Entrepreneurial support recognises the challenges that entrepreneurs face in the entrepreneurial process and any lack of skills and resources (Spigel 2016). The help offered by entrepreneurial support organisations goes beyond the traditional perspective that includes professional services offered for business development, accounting, law and compliance (Feldman 2001).

As with E&I education, all 10 identified HEEE studies included different forms of entrepreneurial support in the proposed HEEEs (Longva 2021; Webber, Kitagawa & Plumridge

2020; Guerrero, Urbano & Gajón 2020; Meyer et al. 2020; Shil et al. 2020; Secundo et al. 2020; Allahar & Sookram 2019; Wright, Siegel & Mustar 2017; Miller & Acs 2017; Rice, Fetters & Greene 2014). There was an almost 40% representation for entrepreneurial support among internal stakeholders and 60% from external stakeholders. Deans/heads hold the view that entrepreneurial support is important; however, they expressed that their institutions require external support from individuals, organisations and the government to provide support services for their students. With the lack of support currently available, external stakeholders including alumni entrepreneurs, angel investors and support professionals advocate that entrepreneurial support can leverage students as potential entrepreneurs, with their entrepreneurial ideas and careers.

While there is much emphasis on entrepreneurship education, there is more involved in supporting students towards entrepreneurship. Entrepreneurial support from individuals and organisations is a critical factor of entrepreneurial ecosystems in supporting entrepreneurs to start, consolidate and scale up their businesses (Hruskova & Mason 2020). Entrepreneurial support positively impacts the development of entrepreneurial capabilities among students (Lichtenstein & Lyons 2001). These support services are a valuable addition that higher education institutions can provide through their HEEEs to leverage entrepreneurial ideas and the careers of students. Interviewee SP4, an external mentor in entrepreneurship, shared:

"I'm a firm believer in mentoring, coaching, incubators, accelerators and so forth. Creating support services for students is helpful and required especially in developing countries. I think private institutions have a huge responsibility to create these platforms and provide students that support. This is the support; institutes should offer their students to really develop for entrepreneurship, engage in entrepreneurship and get their hands dirty."

Many interviewees advocated that entrepreneurial support could develop students for a self-made career in becoming entrepreneurs. Table 7.8 reveals some excerpts about how entrepreneurial support can encourage students' entrepreneurial ideas and intentions for the creation and growth of start-ups through E&I capabilities. In these views, there is entrepreneurial support that higher education institutions can offer. Interviews from this research suggest entrepreneurial support includes mentoring with an entrepreneurship expert, incubators and accelerators; these mechanisms are discussed in the next chapter, on how to implement HEEEs in resource-constrained environments.

A study on an institutional support system confirmed that entrepreneurial support has a positive effect on the entrepreneurial intention of students (Bazan 2018). Similar to E&I education, all 10 identified HEEE studies include different forms of entrepreneurial support in the proposed HEEEs (Longva 2021; Webber, Kitagawa & Plumridge 2020; Guerrero, Urbano & Gajón 2020; Meyer et al. 2020; Shil et al. 2020; Secundo et al. 2020; Allahar & Sookram 2019; Wright, Siegel & Mustar 2017; Miller & Acs 2017; Rice, Fetters & Greene 2014). This use of entrepreneurial support is consistent with the observation made in the literature review that HEEEs are a nexus of entrepreneurship education and entrepreneurial support. Scholars including O'Brien, Cooney and Blenker (2019) argue that entrepreneurship education on its own does not offer students the full potential towards E&I including developing capabilities.

Interviewee PS1, an external	Interviewee DH5, a chief academic	Interviewee AE2, an alumni
mentor and coach:	officer:	entrepreneur:
"You see accelerator programs gives students expert advice. Whether it's on business pitching, lean models, managing teams, attracting resources and others. Students who get this knowledge and experience have the ability to improve their scalable start-ups, ideas and concepts to market-ready product and services with more successful start-ups."	"Large investment has taken place over the last few years for our incubator centre. We have a massive space with all the infrastructure, probably the largest in the private higher education sector. After the graduation, students can work in the facility, create prototypes, and test them. Engineering students work on such projects with a staff member's supervision. The idea behind is to push their behaviour towards entrepreneurial initiatives. So, this is in addition to the co-working space which is available for them to run their office during the early days. There is a reception that handles all administration work for them. It is a great deal of support with a full range of services for them at almost no cost."	"Our institutes had mentoring by academics but they did touch upon professional development and stuff, but not much practically related to careers during my time. To make things more hands-on, mentoring should be organised as one to one or in small batches of students. These should be mentors who can share expert knowledge. Mentors can open discussion to get students thinking and students can ask questions to get their thoughts and ideas off ground. So that students develop a more practical attitude and become aware of hands-on things."

Table 7.8 – Excerpts on entrepreneurial support

In summary, six contextual factors of HEEEs for students' E&I capabilities in a resource-constrained environment came to light through this research. After reporting the findings above, these six contextual factors are compared to the HEEE literature. According to Table 7.9 below, it is evident that extant HEEE studies have investigated some contextual factors towards intentions, mindset and start-ups. Briefly, the most common are mechanisms of entrepreneurship education and entrepreneurial support and the least popular are

mechanisms of entrepreneurial orientation and entrepreneurial research. The majority of HEEE studies have suggested HEEEs include mechanisms of entrepreneurship education, enterprise experiences, entrepreneurial networks and entrepreneurial support. However, none of the existing studies investigated or proposed a comprehensive model including all six contextual factors in HEEEs for students' E&I capabilities in a resource-constrained environment. This study proposes an HEEE model including the six contextual factors through multiple views of internal and external stakeholders: (1) entrepreneurial orientation; (2) E&I education; (3) E&I research; (4) enterprise experiences; (5) entrepreneurial networks; and (6) entrepreneurial support.

HEEE studies	Focus of HEEE	Entrepreneurial orientation	Entrepreneurship and innovation education	Entrepreneurship and innovation research	Enterprise experiences	Entrepreneurial networks	Entrepreneurial support
Longva (2021)	Student start-ups				Ø	Ø	Ø
Webber, Kitagawa & Plumridge (2020)	Students' intention		Ø		Ø		Ø
Guerrero, Urbano & Gajón (2020)	Students' intention		Ø				Ø
Meyer et al. (2020)	Student start-ups		Ø		Ø	Ø	V
Shil et al. (2020)	Student start-ups		Ø		Ø	Ø	V
Secundo et al. (2020)	Students' mindset		Ø		Ø	Ø	Ø
Allahar & Sookram (2019)	Student start-ups		Ø			Ø	V
Wright, Siegel & Mustar (2017)	Student start-ups		Ø		V	V	Ø
Miller & Acs (2017)	Student start-ups		V	V	Ø	Ø	Ø
Rice, Fetters & Greene (2014)	Student start-ups	Ø	Ø		Ø		V

Table 7.9 – Mapping HEEE factors of existing studies to this research

*Note – The above 10 HEEE studies have not explicitly identified the six contextual factors but rather some mechanisms that represent these factors. For example, Rice, Fetters and Greene (2014) identify senior leadership, which in this study is a mechanism of entrepreneurial orientation.

7.4 Perceived importance of HEEE factors

Following the discussion of the aggregated themes, which are presented as contextual factors, it is important to quantitatively summarise the findings, a process which makes qualitative research more creditable (Silverman 2022). The above section suggested that diverse stakeholders perceive *entrepreneurial orientation*, *entrepreneurship and innovation education*, *enterprising experiences*, *entrepreneurship and innovation research*, *entrepreneurial networks* and *entrepreneurial support* as contextual factors that lead to the pathway of developing students' E&I capabilities in a resource-constrained environment. To summarise the data, a spectrum display is used to identify the factors that each stakeholder discussed in their respective interview (Henderson & Segal 2013). The spectrum display visualises qualitative data at the aggregated theme level associated with the level of importance (Slone 2009).



DH Dean/Head of School and AC Academic/Educator, AE Alumni entrepreneur, EE Established entrepreneur, AI Angel investor and SP Support professional

Each \bullet (dot) is identified as a point

Figure 7.3 – Importance of contextual factors as perceived by diverse stakeholders
The spectrum display in Figure 7.3 shows a summary of data collected during the interviews on how internal and external stakeholders perceive HEEE composition, specifically contextual factors. The outer labels DH1 to SP6 represent the 40 participants of the study from six different stakeholder groups. The six layers of the display stand for the six contextual factors that emerged through the study. Each dot in the display articulates that a specific participant indicates the contextual factor as important for developing students' E&I capabilities. Both groups of stakeholders, internal and external, recognised all six contextual factors and their mixed views on the six contextual factors were discussed above.

In total, the six contextual factors appear equally important, with 29, 27, 25, 27, 26 and 31 points, respectively. Entrepreneurial support (31) is the contextual factor with the greatest number of points indicating the highest importance, followed by entrepreneurial orientation (29). Internal and external stakeholders perceive that students can benefit from engaging with the broader environment and develop their E&I capabilities better with additional support from outside the institution. However, external stakeholders, specifically angel investors and support professionals, point out that for this collaboration to happen effectively, higher education institutions need to embrace entrepreneurship, make it a strategic choice and drive it actively internally through their HEEE. All six academics that participated in this study emphasised the same, sharing that their institutions can improve from being more entrepreneurial. However, more than half of the alumni entrepreneurs did not indicate the importance of entrepreneurial orientation for HEEEs. To some extent, alumni entrepreneurs believe that students organically develop HEEEs as getting top management to drive entrepreneurship is less likely.

When comparing the two dominant functions of HEEEs, entrepreneurial support (31) ranks as more important than E&I education (27). This trend could be because some of the external stakeholders including angel investors and support professionals are of the view that students can gain practical knowledge and skills from non-formal education such as receiving one-toone mentorship, participating in an incubator and accelerator program. Students are perceived to develop deeper E&I capabilities by sharing dialogue with experts on validating business ideas, creating prototypes in facilities and meeting like-minded peers, rather than learning in classrooms. Guidance from expert entrepreneurs and support professional along with supporting infrastructure are factors that enable start-ups to succeed in the resource-constrained environments. However, the external stakeholders agree that potential entrepreneurs benefit from entrepreneurship and innovation education that gives them the foundations. Internal and external stakeholders broadly recognise E&I education as an important contextual factor within HEEEs. Internal stakeholders, particularly academics naturally give importance to their service of imparting knowledge and as a result perceive the need for entrepreneurship education. Further, these academics lack formal education and exposure in entrepreneurship and perceive the need for potential entrepreneurs to receive it. External stakeholders expressed stimulating entrepreneurial knowledge and skills as essential for young entrepreneurs to understand the uniqueness of start-ups, compared to other forms of business. E&I education lays a strong foundation for students in undertaking an entrepreneurial career and becoming a potential entrepreneur, like any other profession.

E&I education and enterprise experience are rated equally important with 27 points. Although it is the same degree of importance for both contextual factors, mixed view are shared by different stakeholders. Majority of internal and external stakeholders are of the opinion that E&I education and enterprise experience are equally important (14). Some internal and external stakeholders believe only E&I education is more important (11) and only two stakeholders highlighted enterprise experience (1 – internal, 1 – external). The consensus is that both, education and exposure is essential for developing student's E&I capabilities within the HEEE in a resource-constrained environment. The common perception held is that education on its own is inadequate for E&I capabilities as students need the opportunity to apply what they learn and develop from successful and unsuccessful experiences. Stakeholders also noted that some students might come from entrepreneurial families and already have some level of capability through a sound background from childhood.

Internal and external stakeholders express that entrepreneurial network (26) is salient for HEEEs and leads to E&I capabilities The understanding is that students should have the opportunity to engage with a combination of stakeholders, inside and outside the higher education institutions, to develop for entrepreneurship and innovation. Students not only develop E&I capabilities through entrepreneurial networks but also build their resource repository by engaging with various stakeholder that promote and support entrepreneurship. Further, such networking opportunities allow students to build social capital which increases their likelihood of becoming an entrepreneur and advancing the entrepreneurial process. However, social networks of HEEEs are still in early stages of development in this resourceconstrained environment and most networking is ad hoc interactions, which are less impactful on students E&I capabilities. The least common contextual factor is identified as entrepreneurship and innovation research (25). However, a little over 60% of the participants strongly emphasised the importance of research skills for potential entrepreneurs. More than internal stakeholders of institutions, external stakeholders hold the importance of E&I research linking it to their experiences with entrepreneurs. Angel investors and support professionals highlight that startups tend to suffer from insufficient market research due to lack of research skills among entrepreneurs. Such start-ups are more likely to fail in solving a market problem and serving a market need. Primary research is rarely commissioned to support decision making and second research is not readily available. External stakeholders emphasied that potential entrepreneurs need to understand the importance of data-driven decision-making, to maintain sustainability and avoid basing business decisions on emotions and passion only.

In the end, there should be no factor limiting a student's ability to pursue an entrepreneurial career. Knowledge in entrepreneurship and innovation, an enterprising skillset, entrepreneurial community and continuous support, are key for student E&I capabilities.

7.5 HEEEs through the lens of student involvement theory

According to the psychologist Bronfenbrenner (1979), the most immediate and influential environment affecting an individual are the activities they are engaged in. Active involvement in or even exposure to such activities can inspire one to undertake similar experiences or develop one's capabilities (Bronfenbrenner 1979). The importance is that student development involves a change and can change what an individual feels, thinks and does (Bronfenbrenner 1979). This principle applies to higher education institutions and stakeholders involved in developing students through HEEEs. Considering the HEEE through the lens of student involvement theory places the HEEE in the middle of the model, replacing environment with the HEEE. Figure 7.4 shows that the HEEE behaves as the environment of a higher education institution. Students are involved and experience higher education during their undergraduate degree, leading to the desired outcome.



Figure 7.4 – HEEE adapted to student involvement theory

Adapted from Astin (1999)

The student involvement theory helps us to understand how the HEEE acts as the environment and the efforts of higher education institutions can translate into student development (Astin 1999). The HEEE of a higher education institution can be analysed through the five postulates of student involvement theory: (1) student invests psychological and physical energy; (2) student is involved in activities continuously; (3) student involvement can be quantitative or qualitative; (4) student development is proportionate to student involvement; and (5) student involvement relates to the effectiveness of environment activities (Astin 1999). The last two propositions are salient to co-creating effective HEEEs for students and are subject to empirical evidence from the specific higher education institution (Astin 1999).

7.5.1 Student invests psychological and physical energy

The first proposition of the student involvement theory is that a highly involved student devotes psychological and physical energy to studying, taking part in activities, undertaking tasks, and interacting with peers, academics and others in the higher education environment (Astin 1999). This proposition is illustrative of student involvement and there can be many forms of participation in an HEEE. Through this study, management of higher education institutions are exposed to six contextual factors (entrepreneurial orientation, E&I education, E&I research, enterprising experiences, entrepreneurial networks, and entrepreneurial support) that impact students' E&I capabilities within an HEEE in a resource-constrained environment (refer Figure 7.2). The operationalisation of these factors in HEEEs can benefit from collaboration of stakeholders from the wider entrepreneurial ecosystem.

Higher education institutions can offer opportunities for students to invest their energy in highly generalised activities such as entrepreneurship-oriented lectures and highly specialised activities such as an internship with a start-up or mentoring by an entrepreneurship expert. These activities can vary according to student preferences and participation and should be regularly evaluated. Students should have access to various activities to engage and invest their psychological and physical energy in during their higher education.

7.5.2 Student is involved in activities continuously

The second proposition of the student involvement theory is that a student commits to activities along a continuum (Astin 1999). External stakeholders such as alumni and established entrepreneurs perceive that students should be involved in entrepreneurship-related activities from the first year of their undergraduate studies. This means students should have opportunities to be involved in new and ongoing activities during their studies. In terms of specialised learning, most higher education institutions do not offer entrepreneurship undergraduate degrees in Sri Lanka due to the low demand for it in terms of student numbers. In such a situation, students should have the ability to take elective units in entrepreneurship within their undergraduate degree. A dean from a higher education institution stated that their business management students can take an entrepreneurship-related unit every year as an elective during the three-year degree. However, this was evident in only one of the six institutions. Educational programs with entrepreneurship electives within the HEEE may attract more students and even contribute to higher levels of self-employment. In response to the increasing interest in entrepreneurship among students, higher education institutions can consider offering minors or graduate certificates in entrepreneurship.

As for activities outside the classroom, it is the view of external stakeholders that higher education institutions should provide various activities, namely internships with start-ups, mentoring programs with experts, research-based projects and start-up projects, to move students from receiving inspiration to developing capabilities. Any activity should be relevant to the student in the degree and should have continuity for the student to engage during the whole higher education timeline. Using the contextual factors discussed in Section 7.2, higher education institutions can design and develop their HEEEs for continuous involvement. Within the continuum of activities, stakeholders from the wider entrepreneurial ecosystem can engage and play different roles that range from inspiring entrepreneurship to providing expert feedback. However, institutions must understand that students may hold different degrees of involvement in each activity. This highlights the need for higher education institutions to continuously observe student participation in the respective HEEE.

7.5.3 Student involvement can be quantitative and qualitative

The third proposition of student involvement theory is that student involvement in an activity can be measured quantitatively or qualitatively (Astin 1999). This means that quantitative or qualitative measures can evaluate student involvement in various activities within HEEEs. The number of hours a student partakes in a start-up project is a quantitative measure of student involvement. On the other hand, how well the student translates learnt theories and applies knowledge in a start-up project is more qualitative. While internal stakeholders emphasised the number of activities and hours that students participate, external stakeholders were keen on knowledge application. Although quantitative involvement matters, it is qualitative involvement in HEEE factors that lead to E&I capabilities. When higher education institutions co-create their HEEEs and take related decisions, it is necessary to understand that student involvement in a specific activity can be both quantitative and qualitative.

7.5.4 Student development is proportionate to student involvement

The fourth proposition of student involvement theory is that student learning and personal development are directly proportional to the quantity and quality of student involvement in the activity (Astin 1999). This condition implies that student development associated with any activity of the HEEE is related to the student's involvement in that activity. For instance, students' E&I capabilities can improve by taking up networking with external stakeholders or joining an incubator or accelerator. Both internal and external stakeholders expressed that there is a growing interest in entrepreneurship-related activities among students. If students are more engaged in activities that appeal to them, this involvement can lead to their E&I capabilities in HEEEs can result in a higher level of E&I capabilities. Therefore, higher education institutions can increase or improve opportunities for student involvement through the institution and by exploiting the support of external stakeholders.

7.5.5 Student involvement relates to the effectiveness of activities

The final proposition of student involvement theory is that the effectiveness of any activity directly relates to the capacity of that activity to increase student involvement (Astin 1999). This proposition suggests that student involvement depends on the activity. The evaluation of activities within an HEEE can depend on various measures. However, the judgement of success is largely held by the students. If an activity is useful or successful, then the activity generally attracts the involvement of more students. For example, guest lectures and industry

involvement were popular initially, but now with the emergence of entrepreneurship, the focus of students is more on practical activities. While educators strongly believe in the value of entrepreneurship education, external stakeholders suggest that it establishes the foundation for an entrepreneur and students develop a variety of skills only by being involved in activities associated with enterprise experiences, entrepreneurial networks and entrepreneurial support. Entrepreneurship scholars can examine the effectiveness of various HEEE factors and their mechanisms among students or alumni entrepreneurs within a particular resource-constrained context. However, there is a caution that some activities in an HEEE are mandatory as part of higher education, while others are voluntary and based on student interests.

The above are propositions for higher education institutions to consider in co-creating and evolving HEEEs for the developmental influences in their learning environment for students. The more involved the student is in the HEEE, the greater the student's E&I capabilities will be. This shows that HEEEs, through the lens of student involvement theory, consider behavioural processes that facilitate students E&I capabilities, rather than simply focusing on the factors that enable development. It is important for higher education institutions to understand this value creation. However, value creation tools are not so common in education but represent a promise for higher education (Lackéus 2015). Among such tools, design thinking is applicable for this case, which is another field of study that could be contextualised to an education setting to support student learning and development (Lackéus 2015). Further, the need to understand the needs presented by the wider entrepreneurial ecosystem, to recognise students engaging and investing their psychological and physical energy, to continuously observe student participation and to evaluate the effectiveness of activities in HEEE is similar to the design thinking approach.

Design thinking has made its way to entrepreneurship and entrepreneurship education literature but not so much in HEEE scholarly work. Design thinking is a collective iterative process of being creative and inspired by the world, brainstorming for activities that might help and testing these activities (Brown 2008). This process involves going beyond 'what is' and 'what should be' to 'what might be' (Dunne & Martin 2006). Design thinking–led HEEEs require higher education institutions to understand their customers (students and parents) and the society (wider entrepreneurial ecosystem), collaborate with stakeholders that play an important role in the process and continuously improve. Designing HEEEs using design thinking resonates with a student-centred approach for understanding their needs and refining the learning processes, similar to developing entrepreneurial pedagogy (Huq & Gilbert 2017).

Higher education institutions can design their HEEEs along with stakeholders from the wider entrepreneurial environment as this may develop the resourcefulness they lack.

7.6 Chapter summary

This chapter analysed and presented one part of the findings associated with the second research question that focused on 'what' contextual factors of an HEEE can develop students' E&I capabilities, through multi-stakeholder perspectives. A content analysis was conducted for this question that took a problem-to-outcome approach identifying common codes and themes. This section is one of the first empirical steps to present findings on HEEEs for students' E&I capabilities in a resource-constrained environment. This study proposes six contextual factors through multiple views of internal and external stakeholders: (1) entrepreneurial orientation; (2) E&I education; (3) E&I research; (4) enterprising experiences; (5) entrepreneurial networks; and (6) entrepreneurial support. After reporting the findings, this chapter encompassed a section discussing HEEEs through the lens of student involvement theory. The second part of the findings is presented in the next chapter emphasising the operationalising HEEEs for students E&I capabilities.

CHAPTER 8: FINDINGS AND ANALYSIS FOR RESEARCH QUESTION 2b

8.1 Chapter overview

This empirical chapter is a continuation of the findings and analysis for the second set of research questions. Having understood 'what' contextual factors diverse stakeholders perceive as the composition of HEEEs towards students' E&I capabilities, this chapter extends to 'how' and 'why'. This chapter presents and discusses findings related to how and why specific contextual factors of HEEEs influence students' E&I capabilities in a resource-constrained environment.

A critical challenge in co-creating HEEEs is that many factors of a successful HEEE in a resource-constrained environment would have gaps in such contexts (Bedő, Erdős & Pittaway 2020). According to the perceptions of the internal and external stakeholders interviewed in this research, there are several gaps within HEEEs that challenge higher education institutions to function effectively. To advance our understanding, this chapter goes beyond 'what' contextual factors presented in Chapter 7 to reporting 'how' and 'why'. Therefore, the next section directs higher education institutions to concerted efforts in developing E&I capabilities of students within HEEEs. The chapter outline is shown in Figure 8.1.



Figure 8.1 – Chapter 8 outline

8.2 Operationalising HEEEs for students' E&I capabilities

There is an increasing critique in the emerging HEEE literature that it is producing lists of 'what' an HEEE is over explanations of 'how' an HEEE can be implemented (Longva 2021). There is little consensus and clarity regarding which specific actions can support higher education institutions in implementing contextual factors within an HEEE in a resource-constrained environment (Bedő, Erdős & Pittaway 2020). HEEE factors for developing students' E&I capabilities in a resource-constrained environment are identified in Chapter 7 and these contextual factors are explained through the current gaps of HEEEs and suggested mechanisms below. Understanding how the identified six contextual factors can be implemented in the learning environment is important for co-creating HEEEs and developing students' E&I capabilities. Thus, this section provides clarity on how an HEEE can be co-created in a resource-constrained environment by proposing specific actions through consensus from internal and external stakeholder groups.

8.3 Entrepreneurial leadership, entrepreneurial culture and entrepreneurship chair(s)

A combination of internal and external stakeholders who participated in this research emphasised entrepreneurial leadership, entrepreneurial culture and entrepreneurship chairs as mechanisms to adopt an entrepreneurial orientation within an HEEE; these are discussed below. Entrepreneurial orientation is known for strategic alignment of organisations with their roots as a firm-level unidimensional strategy (Lumpkin & Dess 2015). When adopting an entrepreneurial orientation, internal stakeholders including academics, administration, staff and students are likely to synergise towards a common vision and play a key role in supporting the higher education institution's entrepreneurial agenda (Klofsten et al. 2019). This entrepreneurial orientation spreads across departments and within the higher education institution, building an atmosphere and culture characterised by proactiveness, innovativeness and risk-taking (Todorovic et al. 2011). Leadership, culture and change agents in sync can facilitate the development of students' E&I capabilities within an HEEE in a resourceconstrained environment.

8.3.1 Entrepreneurial leadership promoting students' E&I capabilities

The role of higher education institutions facilitating entrepreneurial-driven economic growth bears the responsibility of more than generating student start-ups and providing leadership for creating entrepreneurial capital (Audretsch 2014). Senior leadership and their long-term

commitment are considered a core factor for HEEEs (Allahar & Sookram 2019; Bischoff, Volkmann & Audretsch 2017). Building a successful HEEE requires the commitment and engagement of senior leadership, typically at the dean level and through to entrepreneurial senior leaders (Rice, Fetters & Greene 2014). Scholars Yu et al. (2017) applied to senior leadership as suggested by Rice, Fetters and Greene (2014) in a study of higher education institutions in Singapore and Taiwan. Through this research, it was found that institutions in Singapore had senior leadership committed to entrepreneurship through HEEEs while institutions in Taiwan did not appear to have such a commitment (Yu et al. 2017).

When questioned on key stakeholders of an HEEE, a dean of a higher education institution, interviewee DH2, said that:

"So, we have the heads and the deans who are leading this entrepreneurial initiative, who are responsible to a large extent, and they must do it within their capacity. Then we have the academics and the educators ... Without the support of the deans, such initiatives would fail within a private education institution. So, you definitely need commitment from the deans to sustain those initiatives. Because even if you have alumni, entrepreneurs, and other experts who in my opinion can contribute towards developing these entrepreneurs, no higher education institution can contribute to the same if you don't have the support and commitment from the top. If senior management does not strongly believe in the need for creating and developing student entrepreneurs, they will continue to push students for cooperate jobs."

Both internal and external stakeholders that participated in this research agree that senior leaders of higher education institutions in Sri Lanka should actively drive entrepreneurship in a top-down approach.

An empirical study conducted across the US, UK, Ireland and Australia found that senior leadership plays a key 'park ranger' role in setting the higher education institution towards entrepreneurship (Thomsen, Muurlink & Best 2018). This means that the senior management sets the cultural tone, authorises initiatives and allocates resources (Thomsen, Muurlink & Best 2018). The commitment and engagement by senior leaders can be expressed in certain ways. For example, in the declaration of strategic intent such as the vision expressed with importance to entrepreneurship, such as the strategic vision of the National University of Singapore as "Toward a Global Knowledge Enterprise" (Yu et al. 2017). Further, it can be expressed in the allocation of resources, infrastructure and stakeholders by senior leadership, which can vary from low to deep (Brush 2014). In a resource-constrained environment, senior leaders must understand the business benefits and impacts for individuals and society related to developing students' E&I capabilities within HEEEs. This understanding will extend to communicating initiatives to employees with a purpose and building a strong entrepreneurial orientation.

8.3.2 Entrepreneurial culture empowering entrepreneurship within the higher education institution

The creation of an HEEE is a key element of entrepreneurial higher education institutions and this happens through the entrepreneurial culture (Secundo et al. 2020). Higher education institutions with successful HEEEs consider fostering an entrepreneurial culture as vital (Allahar & Sookram 2019). A higher education institution's entrepreneurial culture plays a crucial role in not only harnessing the willingness of potential entrepreneurs of creating a start-up but also the willingness of other stakeholders like staff, investors and mentors working with potential entrepreneurs (Spigel 2016). Thus, scholars suggest that entrepreneurial culture influences how the HEEE and their actors navigate and engage, which impacts efforts (Bock et al. 2020; Huyghe & Knockaert 2015).

When promoting an entrepreneurial culture, it is necessary to recognise that it is senior leaders that set the tone of entrepreneurial culture at a higher education institution (Thomsen, Muurlink & Best 2018). Interviewee AI5, an angel investor, shared:

"At university level, the culture must support it. If one joins a higher education institution today if that entrepreneurial culture does not exist, then there is no way for that student to really learn about entrepreneurship, be interested in such a career or develop the necessary capabilities. So, the culture of these institutions has to drive their ecosystems. If you look at some of the popular universities that have successfully produced entrepreneurs and successful entrepreneurs are the ones that have that entrepreneurial culture nested within."

As the culture and values of a higher education institution often orients their students, this in turn impacts their ability to recognise entrepreneurship. Thus, senior management needs to take leadership and commit to an entrepreneurial culture. In the effort to transform into an entrepreneurial culture, there can be student-led initiatives but more significant are institutional-led initiatives (Wright, Siegel & Mustar 2017).

A recent study by Bischoff (2021) highlighted the importance of local/regional entrepreneurial culture in creating sustainable entrepreneurial ecosystems. A mentor in entrepreneurship, interviewee SP4, said that:

"As an active member of the local entrepreneurial ecosystems I can say that there is a lot going on and people are doing a lot of things for young entrepreneurs and startups. Recently, I went to a pitching competition as a judge and there was one student who was given an offer on the spot. So, I think there is an increasing entrepreneurial culture and support at a national level. But now higher education institutions need to build up the same entrepreneurial culture within their space."

This excerpt reinforces that advocating a strong entrepreneurial culture within the HEEE is important. The promotion of an entrepreneurial culture through strategic actions enables higher education institutions to adapt to their environment (Guerrero et al. 2014). Entrepreneurial leaders must build a shared entrepreneurial culture by doing things that highlight the importance of E&I, and the relevance of developing capabilities among students.

8.3.3 Entrepreneurial chairs as change agents

Higher education institutions are key enablers of economic growth and social development and they act as agents of change in society (Klofsten et al. 2019). Studies have explored how higher education institutions with an entrepreneurial mission in developing countries act as change agents in sustainable development (Wakkee et al. 2019). To be effective, this extends to staff of higher education institutions, requiring them to act as change agents. Change agents can be members of senior management, academic staff or staff within the institution. For instance, staff responsible for entrepreneurial activity within a higher education institution can be identified as entrepreneurship 'chairs' and they are positioned to drive actions promoting entrepreneurship (Rice, Fetters & Greene 2014). Entrepreneurship chairs were suggested in an early study by Rice, Fetters and Greene (2014) but do not seem to resurface in the other identified HEEE studies. Such staff delegated with the responsibility to drive entrepreneurship within the higher education institutions. Interviewee DH2, a head of school, commented:

"We have two entrepreneurship pillars appointed for the role of fostering the entrepreneurial spirit among students. Basically, to promote entrepreneurship our students. The e-club is one of their responsibilities as well. Monthly they have to report what has been achieved and discuss upcoming initiatives with the management." Further, interviewee DH4, a head of academic affairs, shared:

"So, we have what is called an entrepreneurship circle and a staff member is put in charge of this initiative. Students from the different programs are part of it. Initially it was a handful of students but now we have a large group of students actively working in this circle. The chair is responsible for various outcomes at the end of the financial year. Students launch start-ups through this circle but of course these are small side hustles like a cupcake business promoted through Facebook."

Having such internal role models will gradually build an entrepreneurial culture of rewarding E&I (Gibb, Haskins & Robertson 2013). These employees are an essential part of the institutional infrastructure for managing and advancing HEEE efforts (Rice, Fetters & Greene 2014).

In resource-constrained environments, senior management must identify potential change agents, share the purpose and assign responsibility for promoting E&I. This new role can complement the current duties and responsibilities of existing staff beyond heads of departments and academics. Such change agents can champion building internal teams and collaborating with stakeholders from the wider entrepreneurial ecosystem as well.

8.4 Entrepreneurship curriculum, practical pedagogy and real-world assessments

Multiple stakeholders discussed how entrepreneurship curriculum, pedagogy and assessments should be addressed in delivering E&I education within the HEEE. Entrepreneurship education is a key contributing factor that supports entrepreneurial activities within the entrepreneurship education ecosystem, strengthening the business environment of a country (Regele & Neck 2012). Higher education offers an incentive to attract potential entrepreneurs through study programs that are most relevant for markets and best ensure success of new start-ups in the future (Bauman & Lucy 2021). According to the literature, entrepreneurship and entrepreneurship education are on the rise in the US (Bauman & Lucy 2021; Regele & Neck 2012); however, this may differ in a resource-constrained environment such as Sri Lanka. With changes in the environment, so have expectations of education changed.

8.4.1 Entrepreneurship and innovation education for all students

Curriculum, including program and content, is a key building block nested in the ecosystem when universities facilitate entrepreneurship (Brush 2014; Kuratko 2005). Scholars seem to view entrepreneurship education through a narrow lens resulting in a fragmented snapshot of

entrepreneurship curriculum (Shane 2003). Whether it is instilling entrepreneurship through means of a core module, elective unit or a major, a great deal must be focused on developing the entrepreneurship curriculum (Fetters, Greene & Rice 2010). According to interviewee PS6, a mentor in entrepreneurship:

"You can't become an entrepreneur just because you learn about entrepreneurship. So, it is better to have entrepreneurship as options for any study program where they can get the entrepreneurial knowledge and skills no matter what they are studying. Entrepreneurship and innovation should go hand in glove with Computing, Fashion, Law or basically any area for that matter. Like entrepreneurship within dentistry or entrepreneurship within agriculture. So that they can consider a business or become an entrepreneur in their area or expertise."

The shared view among external stakeholders is that entrepreneurship should be offered in the form of core, elective and zero credit units to all undergraduates, including business and non-business students.

In 1970, only 16 higher education institutions in the US offered entrepreneurship education and today more than 2,300 US colleges and universities offer courses in entrepreneurship (Bauman & Lucy 2021). Unlike in the US, where there is an increase of entrepreneurship majors offered, none of the six participating higher education institutions offer an undergraduate degree in entrepreneurship. A head of a business school, interviewee DH6, said:

"We don't offer a degree in entrepreneurship, but our students can opt to progress into an entrepreneurship bachelors with a partner university overseas if they are going international to complete the degree."

The common decision not to offer an undergraduate degree in entrepreneurship is due to the low number of students wanting to enrol in an entrepreneurship degree for undergraduate studies. Further, it is also the perception of external stakeholders that, more than an undergraduate degree in just entrepreneurship, entrepreneurship should be embedded into the curriculum of various study fields.

8.4.2 Practical-oriented pedagogy for delivering entrepreneurship and innovation education

Since the mid-1990s, teaching entrepreneurship has become a challenge for universities (Carlsson et al. 2013), and contributions from scholars of diverse fields of study have been published. Studies report that potential entrepreneurs learn by copying, from experiment, by problem-solving, from opportunity taking, and from making mistakes where learning involves theorising, acting, experiencing and reflecting (Taylor & Thorpe 2004). However, there is no consensus or a universal pedagogy on teaching entrepreneurship (Fayolle & Gailly 2008). Teaching entrepreneurship has challenged old pedagogies by recognising new ways of teaching and learning (Hoppe 2016). When planning the pedagogical side of a unit or course, it is always good to add an element of surprise for students as predictability means students become bored in the classroom (Fiet et al. 2000). A future research direction suggests the need for research on pedagogy focused on developing capabilities of potential entrepreneurs (Neck & Greene 2011).

Entrepreneurial learning seems to be growing in an alternative way where entrepreneurship modules are delivered through mixed theory–practice learning (Belitski & Heron 2017; Hoppe 2016). Interviewee PS1, an external mentor and coach in entrepreneurship, said:

"The more academic it is, the less I believe in it, the more professional or more practical it is, it is definitely valuable to student entrepreneurs. You know, when teaching I believe it should be more integrated with theoretical knowledge, practical skills and self-reflection. These are important for the future entrepreneurs which is not the same as businessmen we had from decades ago."

A head of a business school, interviewee DH6, added:

"So, these students receive the knowledge within the degree but, you know, it is really the attitude towards entrepreneurship. We always encourage the student to take up initiative and give them confidence. We are not limiting ourselves to the traditional pedagogical, you know. It is combination of teaching approaches but mostly collaborative teaching to engage students with others. When teaching, students learn from each other as well." Scholars such as Lourenco et al. (2013) advocate 30/70 pedagogy design where a session should consist 30% of the time to instruct/review theory, concepts and tools and impart theoretical knowledge, and 70% of the time for practical learning through a video, case/story or guest speaker and then practical application through individual or group activities.

8.4.3 Real-world assessments

When suggesting entrepreneurship education frameworks at the university level, scholars have identified one of the key facets as assessments (O'Brien et al. 2019; Maritz & Foley 2018; Maritz & Brown 2013; Fayolle & Gailley 2008). Assessments in the forms of testing concrete behaviour of individual students or groups are best supported, rather than being merely written examinations and reports, in the entrepreneurship discipline (Hoppe 2016; Gibb 2002). Learning through experimental work involves trial and error and encourages entrepreneurial projects (Hoppe 2016; Gibb 2002). Hence, entrepreneurial learning should be designed to have suitable assessments for effective evaluation and measurement of learning outcomes (O'Brien et al. 2019).

Learning is shifting from the traditional lecture-based to increasing practice-based sessions for more experience of entrepreneurship, where students pursue learning activities or projects in classrooms or incubation centres on campus, or outside on projects with start-up entrepreneurs (Neck & Greene 2011; Gibb 2002). Therefore, entrepreneurship, like any creative activity, is best learned through hands-on practice (Johannisson & Madsen 1997). It is argued the entrepreneurial capabilities are developed when people learn from experience where they are involved in and mostly practice, for example, problem-solving skills (Pittaway & Cope 2007). This is in line with interviewee AI1, an angel investor, who stated:

"If you look at the nature of entrepreneurship, it involves finding an innovative solution to a particular problem. This practical thought process can be developed or supported through assessments to arrive at practical outcomes. Assessments should be designed to push students to arrive at more than a written report and be involved in doing activities or projects. These activities support to get the so-called entrepreneur attitude into their mind and body. From such assessments, they learn from their mistakes too."

Educators may want to engage students in real-world assessment, taking a hands-on approach to learning.

The gaming industry has evolved, and games entertain the player and inspire action, effect attitudinal change or instil lessons (Keitt 2009). Digital tools such as online games expose students to real business situations and challenges in a virtual world, aligning them to learn while they play (Volkmann et al. 2009; Pink 2006). Universities such as Babson have experimented with and embedded various educational games into the entrepreneurship curriculum and toolbox, allowing students to practice entrepreneurship (Regele & Neck 2012; Harmeling & Sarasvathy 2011). A study that investigated over 2,000 students using more than 100 stimulation games in Germany revealed that stimulation games have high learning effects and suggest their use in entrepreneurship education (Huebscher & Lendner 2010). While engaging in games facilitates experiential learning, it also allows assessing how students act and adopt entrepreneurial behaviours based on their entrepreneurial knowledge and experiences (Kriz & Auchter 2016; O'Connor 2013).

Entrepreneurs who founded tech start-ups, such as interviewee AE1, said:

"Entrepreneurs are known to have a unique mindset and way of thinking. Some of the guys in one of my start-ups are gamers and the way they think is amazing. They are quick, adaptive and team oriented. So, having things like online games to evaluate students is a new trend, right? It is more practical than a written exam or report. People have scored As and got first class degrees, but they don't necessarily become successful entrepreneurs. It's about thinking ability and skills."

One such video game supported students to think like entrepreneurs under conditions of uncertainty and risk (Neck et al. 2011). This means online stimulation games need to be carefully selected based on the learning outcomes that need to be accomplished (Fox et al. 2018). However, there is an argument that while simulation games can be useful, there could be real-life tests that could evaluate entrepreneurship education (Ratten & Usmanij 2020).

8.5 Research-based coursework, research conferences and research-based projects

Stakeholders are of the view that research-based coursework, research conferences and research-based projects are initiatives that higher education institutions can offer to develop research skills within an HEEE. Higher education institutions striving to fulfil the three core activities of teaching, research and entrepreneurship and adopt a coordinated strategic alignment across these critical activities attempt to provide a conducive environment for their students, staff and university community (Guerrero et al. 2014; Kirby et al. 2011). To this end, research is a salient factor that combines with teaching and entrepreneurship in harmony for

the emergent phenomenon with a positive impact on economic and social development (Etzkowitz 2014). An HEEE study based on a traditional research higher education institution is identified as the one study that suggested research as a key HEEE factor when compared with the extant literature (Miller & Acs 2017).

8.5.1 Research-based coursework

There is academic debate on whether there is a positive impact of research on developing students for entrepreneurship (Mason & Brown 2014; Brush 2014; Walter et al. 2013; Neck et al. 2011). External stakeholders of this study advocate research for student E&I capabilities in congruence with Mason and Brown (2014), Brush (2014), Ucbasaran et al. (2008) and Arenius and De Clercq (2005). Interviewee SP5, an external mentor in entrepreneurship, said:

"Research skills is so important. You should have information about your own product, competing products, quality and everything. That's a basic knowledge from research. Next you should have knowledge about the market and industry. Then the broader environment including new laws, economic situation, technology developments. Research is crucial in entrepreneurship and students should be taught how to research, especially problem-based research and use information and data in businesses. This will sharpen their problem solving and critical thinking. This is something that can be embedded in students through assessments they attempt during the study program."

External stakeholders suggest developing research skills and knowledge among students for E&I using research-based coursework within HEEEs.

Scholars such as Rothaermel et al. (2007) argue that higher education institutions should engage more in applied research when becoming more entrepreneurial. Similarly, external stakeholders suggest prioritising applied research for developing students for E&I, allowing them to generate practical solutions. Applied research uses accumulated theories, knowledge and methods for a specific purpose or problem (Brush 2014). Interviewee AC5, an academic teaching an entrepreneurship-related unit, stated:

"... students get exposed to research through their assessments. Units such as Marketing and Finance have assessments that require students to research, primary and secondary. Marketing assignments tend to be case study based where students end up suggesting solutions for a problem that a product or brand faces. Even for their capstone project they do research. For example, students select a problem or opportunity they are interested in, conduct research on it and write their project. We give our students the freedom to select a more academic orientated through a dissertation or a business plan. Students who are keen on entrepreneurship go for the business plan option. The purpose is to allow them to apply research in different contexts, get in-depth knowledge of the area and come up with a practical solution."

Therefore, students should be given the task to apply research in assignments across the different fields of study and not be limited to entrepreneurship-related units within an undergraduate degree.

8.5.2 Research conferences exposing students to scientific knowledge

The trajectory of entrepreneurship education resulted in the first major academic research conference on technical entrepreneurship in 1970, followed by the first entrepreneurship research conference and publication in 1981 by Babson College (Katz 2003). The Babson College Entrepreneurship Research Conference was a major effort by the university, dedicated to increasing the understanding of entrepreneurship by encouraging early-stage research on entrepreneurship and creating a useful networking opportunity for global entrepreneurship scholars and researchers (Fetters, Greene & Rice 2010). Given the benefits of researching entrepreneurship, entrepreneurship research conferences and centres were identified as a factor of HEEEs (Rice, Fetters & Greene 2014).

To understand how research conferences develop students, views from various stakeholders were considered. Research conferences are possible platforms to help students link theory and practice and develop related skills, and excerpts from interviews resonate with the relevance of research conferences for students. Interviewee DH5, a chief academic officer, said:

"We host a research symposium, and this is the third consecutive year it is running. All schools of the institute contributes, and entrepreneurship is one area covered. Of course, for this is for traditional academic interest but we cannot discount the value it offers for students. For a potential entrepreneur it is more about new knowledge and linking the theory to practical situations. It is a chance for students to get involved in researching problems that companies and industries face and arrive at real world solutions through their papers."

Further, interviewee AC5, an academic teaching entrepreneurship, commented:

"We don't have a research committee, centre, or conference in-house within the institute. But our students and staff attend research conferences outside. There are external conferences, and we receive invitations for these. So, we make sure that our students participate and contribute. Although these are more academic conferences, students can contribute to knowledge or get exposed to problem solving, presentation skills and research skills for entrepreneurship."

In this study, the value of research conferences is recognised by internal stakeholders with the view that students are knowledge producers. However, research conferences are not opposed by external stakeholders. The benefit is perceived to be more indirect but not irrelevant to developing student E&I capabilities.

8.5.3 Research for start-ups

There can be various research projects encouraged within the HEEE, including research projects to develop prototypes (Rasmussen & Borch 2010). In this study, the emphasis is on research-based projects with entrepreneurs and their start-ups. Interviewee EE4, an award-winning social entrepreneur, said:

"I worked with some students from a design institute on a project and it was a research project for them. Instead of coming up with solution in-house I reached out this institute. The task was for students to propose a packaging solution for a new organic start-up I was working on and I was quite happy with their suggestions that we selected one of it and implemented it. Students had done good research, found the right materials, how to source these materials locally, and suggested sustainable ideas for the packaging solution. This is the best way for them to get hands-on research experience by working with projects."

Through such research-based projects, students become exposed to applied research by conducting market research and arriving at innovative solutions for specific problems or opportunities. Moreover, students become open to the broader environment including markets, industries and ecosystems outside the institution.

Previous research suggests that an entrepreneur's research skills influence their decisions (Di Greorio & Shane 2003). Interviewee EE5, an entrepreneur of a fashion start-up which become an international brand, stated:

"When you're looking at entrepreneurship, and how research can support that, research becomes so important for budding entrepreneurs. When students work on research projects for start-ups or for a company, it really takes their inner strengths out and puts them on a trial. So especially when you are starting off, they need to know how to make the most of research and use that information wisely in starting and scaling."

Student involvement in various research-related projects can inspire and develop them to create a self-made career path as a potential entrepreneur. For this initiative, higher education institutions can connect with entrepreneurs and offer them access to students as a provision of talent that can be deployed for research. Such research project work, like researching business start-up opportunities or designing suitable packing for a new product, can not only provide 'hands-on' activities, but can also engage external stakeholders, such as start-up entrepreneurs, thus extending the students' perspectives (QAA 2018).

8.6 Entrepreneurship competitions, internship with start-ups and start-up projects

Entrepreneurship-related competitions, internships with start-ups and start-up projects emerged as specific actions to operationalise enterprise experiences for students within the HEEE, through interviews with internal and external stakeholders. Diverse stakeholders suggested these mechanisms as specific actions to offer enterprise experiences for students within the HEEE. However, enterprise experiences are not limited to competitions, programs, projects, forums and events (Brush 2014). Students can learn entrepreneurship through a combination of curricular, co-curricular and extracurricular activities (Bischoff et al. 2017). These activities complement and extend students' academic knowledge through a wide range of activities while supporting student and graduate entrepreneurship (Al-Dajani et al. 2014; Rae et al. 2012). Extant HEEE studies such as by Webber, Kitagawa and Plumridge (2020) found that enterprise experiences such as internships have more influence on shaping students' entrepreneurial intentions compared to formal education in a classroom.

8.6.1 Entrepreneurship competitions at institution, inter-institutional or national level

Giving students significant opportunity to practise and experience is critical to encouraging interest in entrepreneurship (Barr et al. 2009) and creating enthusiasm among peers within the HEEE (Fetters, Greene & Rice 2010). Interviewee DH1, a dean of academic affairs, stated:

"Couple of our students won at the social innovation competition which is an external event organised at a national level with participation drawn from institutes across the island. After becoming winners at the competition, their idea was funded and led to a social start-up. I feel that this achievement gave these students the confidence to consider a career as an entrepreneur and the winners spread some excitement for new students to get inspiration from."

Multiple stakeholders including deans, academics and entrepreneurs articulated that students gain confidence to consider entrepreneurship as a career and develop capabilities they need for entrepreneurship by participating in related competitions. Interviewee AC5, an academic teaching an entrepreneurship-related unit, added:

"... there are very less competitions when it comes to entrepreneurship. But students participate in various competitions like business idea organised by external association or at national level. Irrespective of the nature of the competition, I think it motivates students to step forward, perform and excel. Most competitions have attractive rewards too. But mostly they are all opportunities to get experience, showcase their skills and build their aptitude."

With the limited availability of business idea competitions internally, some external stakeholders are of the view that higher education institutions should direct students towards inter-institutional or national competitions.

There is debate about the effectiveness of business plan competitions (Honig & Karlsson 2010). Scholars of six HEEE studies suggested business plan competitions for students interested in entrepreneurship (Shil et al. 2020; Secundo et al. 2020; Meyer et al. 2020; Miller & Acs 2017; Wright, Siegel & Mustar 2017; Rice, Fetters & Greene 2014). Interviewee DH5, a chief academic officer, shared:

"Innovation and entrepreneurship of our students is displayed at the annual competition. Where they display their novel ideas and prototypes to corporates and industry. Through this event, students have received the opportunity to join and work with companies who were interested in their innovation. This experience has empowered students through trial and error before actually venturing into their own start-up."

Further, the benefits of such competitions vary, from an actionable business plan, to coaching and training, cash rewards, seed funding, consultancy services, networking opportunities and commercialisation (Grimaldi et al. 2011).

8.6.2 Internship opportunities with start-ups in a range of different industries

An internship is considered a methodology for teaching entrepreneurship (Hasse & Lautenschläger 2011) and internship with a start-up is deemed essential for experiential entrepreneurship education (Fayolle et al. 2006). Scholars argue that a greater emphasis is needed on experiential learning, such as through internships with start-ups and other hands-on activities that involve students interacting with entrepreneurs (Volkmann et al. 2009). For instance, in response to the Asian financial crisis of the late 1990s, the National University of Singapore expanded its remit by introducing several entrepreneurial initiatives including internships in start-ups for its students (Etzkowitz 2014).

Sri Lanka does not have a long history of internships within a study program in the private higher education sector and degrees have been predominantly focused on learning through the set curriculum. Interviewee SP2, a mentor in entrepreneurship, commented:

"Nowadays more universities tend to offer more internship to students but these are internships to work for multinational companies like Unilever, Nestle. But in my view, is that internships can be with start-ups or even SMEs. Students can work for start-ups, developing ventures and understand the grassroots level of a business. These experiences become helpful when students have to set up their own company, craft the organisational model, validate their proposition and develop the go to market strategy. That would be the best way of learning."

This benefit of start-up experience leading to start-up success is in line with an HEEE study by Meyer et al. (2020) that looked at interest in students taking up work in start-ups or growing venture companies.

A few scholars suggest that HEEEs should include work experiences or internships (Shil et al. 2020; Meyer et al. 2020; Miller & Acs 2017). A study on entrepreneurial capabilities in Vietnam found that students should be mainly trained in entrepreneurship through practical experience during internships when compared with thematic courses, curricular subjects, extracurricular activities and projects commissioned by enterprises (Devetag et al. 2020). However, the general practice among higher education institutions in Sri Lanka is promoting

standard internships with multinational companies. Interviewee DH1, a dean of academic affairs, said:

"... one small example is when Mr Merrill Fernando came for a seminar to talk about Dilmah and his entrepreneurial story of founding a global tea brand. We ended the seminar with new product ideas for Dilmah and a group of students proposed tea flavoured cereal. He thought that's very innovative, and he liked the idea so much that those students were invited for a factory visit. Some of them ended up doing an internship with him. These are the different life experiences which help them drive their entrepreneurial spirit and ambitions forward."

Thus, external stakeholders consisting of entrepreneurs and mentors are of the view that higher education institutions should offer students internship opportunities with start-ups in a range of different industries.

8.6.3 Start-up projects through the institution or external organisations

A start-up project is a funded project designed to enable students to start a new venture. A study by Jones et al. (2015) evaluated 'Beta', a start-up project for undergraduate students that grants GBP1,500 of seed funding to students for initiating the start-up process. The purpose of the 'Beta' start-up project was for participating students to complete the project with an economically viable business that offers promising self-employment post-graduation (Jones et al. 2015). With the growing importance of start-up projects, scholars suggest higher education institutions should provide a start-up project with initial funding, to integrate a second level of funding (Jones et al. 2015) and even to replace student placements with a start-up project (Matt & Schaeffer 2018).

During the interviews an external mentor in entrepreneurship, interviewee PS6, said:

"Even evaluating business proposals, hiring for myself, or even when I'm hiring for my portfolio companies, I do look whether students have engaged in any projects, and especially related to start-ups. Because then they are likely to have a better understanding of how do you perform in a team, understanding their role in that team and even leading the team, so that the overall objective becomes accessible, that's exposure."

Such views establish the need for higher education institutions to encourage students to try start-up projects that contribute to getting experience and advancing skills.

Start-up projects have been found to affect students' start-ups and entrepreneurial mindset in HEEE studies (Longva 2021; Secundo et al. 2020; Wright, Siegel & Mustar 2017). Interviewee SP1, an external mentor and coach in entrepreneurship, illustrated:

"I strongly believe participating in various projects is very important. I encourage my son to take part in any opportunity of start-up-related projects through university because I have learnt that through projects one cultivates various elements such as leadership, teamwork, confidence, resilience, problem solving and more. I know that I did. I know others have. Most importantly, it is that attitude that grows in you to seize opportunities to develop yourself. This is one thing that makes students ready for entrepreneurship and even the modern society."

In France, the government took the initiative to provide cash prizes between EUR5,000 and EUR20,000 to 53 out of 600 students with a promising start-up project (Wright, Siegel & Mustar 2017). While government support in such start-up projects for students of private higher education institutions is almost non-existent, external stakeholders believe that higher education institutions should offer it through internal schemes or collaborate with external organisations for this purpose.

8.7 Peer engagement, the alumni entrepreneur network and access to external stakeholders

An entrepreneur is increasingly recognised as a social animal, working socially, and engaging with and in the social milieu (Jack et al. 2008). Entrepreneurial networks create a social environment for students to connect and engage with people who may support their entrepreneurial careers (Mueller 2011). Within higher education, networking can provide access to information, support, resources and even help one to acquire capabilities from others (Tomy & Pardede 2020). Resource-constrained environments are likely to lack dense entrepreneurial networks (Bedő, Erdős & Pittaway 2020), thus it is important to bridge the connections and connectivity within the HEEE and between the HEEE and external stakeholders. The concept of an HEEE proposes interconnectedness with all stakeholders to perform in the entrepreneurial environment (Shil et al. 2020). Multiple stakeholders highlighted peer engagement among students, an alumni network of entrepreneurs and access to external stakeholders as ways for students to develop connections through their HEEEs.

8.7.1 Peer engagement at institution and inter-institutional levels

Networking among peers across departments and schools was found to be a key initiative influencing the increase in creating start-ups at Massachusetts Institute of Technology (Roberts & Eesley 2011). Students networking with each other build a network channel for continuous engagement, breaking the traditional practice of campuses being active only during term (Belitski & Heron 2016). This peer engagement can take various forms within higher education and an entrepreneurs' club is one of them (Allen & Liberman 2010). Research suggests that social norms and the interactions of a network may have significant impact on a student towards entrepreneurship (Krueger et al. 2000). The collective view of stakeholders, as expressed below, is that an entrepreneurs' club is a supportive community for aspiring entrepreneurs.

Interviewee EE4, an award-winning social entrepreneur, said:

"When I was studying, we didn't have opportunities like to engage with other students much. I think that something like a club or community will draw people who are interested in entrepreneurship and even promote entrepreneurship. These communities will allow people to share their ideas and feelings with similar students. And then I think, it will encourage people towards entrepreneurship and even come up with good ideas, maybe. So, I think is good for students to mix with other students from other programs and even other institutes. This interaction and exchange between students help the process and even have better understanding of it."

Interviewee DH6, a head of a business school, shared:

"For networking, we have a few clubs established which are all student led clubs, actually. One is student activity club. Then the Toastmasters club, and we have an entrepreneurs' club. If you look at these clubs, all of them provide exposure indirectly and directly for entrepreneurship. They can expand their interests and thinking. It starts from the thinking pattern when you meet the like-minded entrepreneurial people. As part of the club, they engage with the community for different purposes or even address different problems. Through these clubs, they network but they also develop their strengths and skills during their involvement and effect their behaviour."

Such networking can result in acquiring competencies and knowledge from others that contribute to students' entrepreneurial self-efficacy (Tomy & Pardede 2020).

An entrepreneur club, also known as an eClub, is designed for undergraduate students who are keen on entrepreneurship and are considering a self-made career by starting or buying a start-up or business (Allen & Liberman 2010). Members of the entrepreneur club join forces to help each other achieve their individual goals and produce networking events with alumni entrepreneurs, other successful entrepreneurs, investors, topic experts and other stakeholders (Allen & Liberman 2010). Although the benefits are clear, not everyone is willing to share knowledge, and this raises the question raises of which mechanisms can engage network collaboration (Ratten & Usmanji 2020).

Higher education institutions in the US put great emphasis on experimental learning through internal communities such as student clubs and communities (Finkle et al. 2013). Moreover, student communities such as the National University of Singapore (NUS) are considered powerful via word-of-mouth marketing, spreading the entrepreneurial spirit (Ho et al. 2018). Another study has acknowledged the role of student clubs such as NUS Entrepreneurship Society, National Taiwan University Entrepreneur Association, and University of Southern California eClub (Yu et al. 2017; Allen & Liberman 2010). This network influence is validated where student clubs are included as a building block of the HEEE model, suggested in work by Rice, Fetters and Greene (2014). However, there is the argument that student clubs draw low commitment compared to an incubator facility encompassing faculty, alumni and entrepreneurship (Brush 2014).

8.7.2 Alumni entrepreneur network

Networks can be used to exchange knowledge and experiences for entrepreneurial learning and build entrepreneurial skills (Belitski & Heron 2017). Alumni of the University of Southern California represent a significant importance to its HEEE, where they give back to the entrepreneurial student body through start-up funding, jobs, internships and time (Allen & Lieberman 2010). Scholars argue that alumni entrepreneurs and their start-ups are an available asset for any higher education institution that play a major role in developing potential entrepreneurs, identifying opportunities and forming new start-ups (Miller et al. 2017; Siegel & Wright 2015). Further, it is an important issue for higher education institutions to engage their students with alumni start-ups and SMEs bridging connections to the local entrepreneurial ecosystem (Webber, Kitagawa & Plumridge 2020).

A study by Longva (2021) suggests that students develop their social capital within their HEEE. Interviewee DH5, a chief academic officer, shared:

"We have a large pool of alumni entrepreneurs and we bring them together through something called IGNITE. It is an alumni network that operates as a forum and collaborate for various initiatives. It's an alumnus led body consisting of alumni entrepreneurs that involve current students in events and skill development programs. Their interactive monthly discussions are a safe place for students to engage, ask questions and learn entrepreneurship outside of the classroom through workshops and other activities."

This alumni networking is similar to the University of Southern California, which hosts an alumni network day every month that is well-attended by 300 to 400 individuals with an interest in entrepreneurship leading to start-ups (Allen & Lieberman 2010). Other examples are the Babson alumni network, which was the first step to creating a network focused on entrepreneurship within its HEEE (Fetters, Greene & Rice 2010) and the Massachusetts Institute of Technology's high-level alumni network (Ribeiro et al. 2018).

Resources are a key driving force of the entrepreneurial process and alumni entrepreneurs play an indirect role in a network providing resources to current students (Bauman & Lucy 2021; Rasmussen & Wright 2005). For current students to receive this benefit, initiatives must be put in place within an HEEE. Alumni entrepreneurs themselves may lobby the higher education institution to establish such entrepreneurship initiatives and be willing to provide funding and advice to get initiatives within the HEEE organised and running (Meyer et al. 2020; Rice, Fetters & Greene 2014). Interviewee AC5, an academic teaching an entrepreneurship-related unit, said:

"Since we are somewhat of a new institute, our alumnus is not large in numbers. But among graduated students, some of them have their own start-ups and businesses. They have start-ups in various area such as fashion, baking and even music and they want to interact with the current students. Our alumni entrepreneurs gets involved with institute as well as the current students. If we have an annual event they would bring in sponsorships. If a student wants any information, we connect them to our alumni. This way current students have some access to an alumni network of entrepreneurs to reach out. But all this is currently rather informal." In addition to organically growing an alumni entrepreneur network, another approach is through an incubator of the higher education institution, which attracts a loyal group of alumni (Theordoraki et al. 2018).

8.7.3 Access to external stakeholders

Involvement of external stakeholders on programs is vital for the co-creation of an HEEE (Wright, Siegel & Mustar 2017). Even within the HEEE, entrepreneurship succeeds when external stakeholders are embedded in the entrepreneurship education ecosystem, catering to aspiring entrepreneurs (Isenberg 2010). Research has found that direct involvement of external stakeholders from the entrepreneurial ecosystem can guide students on their entrepreneurial process by identifying opportunities and transferring knowledge (Secundo et al. 2020). Scholars such as Allahar and Sookram (2019) suggest that higher education institutions that face a relatively slow development of their HEEEs can accelerate growth to a more fully-fledged HEEE through stronger collaboration with external stakeholders.

External to a higher education institution, other primary stakeholders in the entrepreneurial process are alumni entrepreneurs, start-up entrepreneurs, investors (angel investors, venture capitalists, crowd-source funding), mentors, professional service providers, corporations and government officials (Meyer et al. 2020). These external stakeholders may be involved in various activities of the HEEE, including entrepreneurship education or extracurricular and other activities (Bischoff et al. 2018). Interviewee SP2, a mentor in entrepreneurship, shared:

"Of course, the role and ownership are on the institutes per se and they need to have a setup for collaborating with external stakeholders for activities which can benefit its students. It could be to create inspiration through experience sharing, validating a business idea, or even finding seed capital. There are plenty of external stakeholders who are willing to contribute and help students. So external stakeholders need to be attracted and connected to students to support students with advice and resources. But this needs to be structured".

Therefore, higher education institutions need to create the opportunities for external stakeholders to engage with students. External stakeholders can be involved and play a role in all six identified HEEE factors. The possibilities of this involvement will be further discussed in Chapter 9.

8.8 Mentoring, incubators and accelerators

One of the attributes of a successful start-up community is support services (Feld 2012). Entrepreneurship support is identified as a key function of HEEEs along with entrepreneurship education (Rice, Fetters & Greene 2014). Types of entrepreneurial support can include a variety of broad, functional and specialised services (Hruskova & Mason 2020). Generally, entrepreneurial support is meant to provide training and resources to potential entrepreneurs and their start-ups (Spigel 2016). A study found that students perceive and desire educational support the most, followed by concept development support and finally business development support from higher education institutions in Australia and Europe (Kraaijenbrink et al. 2010). Higher education institutions can offer a continuum of entrepreneurial support activities depending on the needs of their students; these can range from early stage to development stage (Wright, Siegel & Mustar 2017). Mentoring, incubators and accelerators are the three mechanisms that emerged from the interviews with various stakeholders as suggestions for higher education institutions in Sri Lanka.

8.8.1 Mentoring with practitioners including start-up entrepreneurs

Leading higher education institutions have developed a sustainable network of entrepreneurship mentors to deploy as experienced advisors for specific students (Meyer et al. 2020). The National University of Singapore recognises mentoring as one of the mechanisms that strengthens its HEEE (Rice, Fetters & Greene 2014). One HEEE study found that developing entrepreneurs within the higher education environment requires appropriate mentoring for potential entrepreneurs (Shil et al. 2020). In another instance, an incubator or an accelerator of a higher education institution can offer mentoring for its participants as a key service (Allahar & Sookram 2019; Wright, Siegel & Mustar 2017).

Active mentoring provides every student with support to solve business problems through an expert (Shil et al. 2020). In contrast, mentoring through experienced mentors among the investigated higher education institutions tend to take the form of seminars and workshops, which addresses students in general. Interviewee DH5, a chief academic officer, stated:

"Currently we offer students mentoring through seminars and workshops in small groups of students. You know, we discuss topics like how to become a successful entrepreneur? how to pitch a business idea?, how to design a lean start-up?. Of course, these are ad-hoc, but these sessions have been useful for students to gain practical training to setup a new start-up and any advice that improves their awareness. Some students participate because they are keen to take a step forward and want information and others are more of collecting information type. Either way the mentoring sessions conducted by externals give students something more related to entrepreneurship."

Compared to this practice, external stakeholders suggest mentoring to be organised on an ongoing, one-to-one basis using entrepreneurship-experienced mentors.

In developed countries like the US, Canada and Europe mentoring is considered a free service that is extended voluntarily (Meyer et al. 2020). This is similar to the Sri Lankan context where there are experienced mentors who are willing to step up and mentor potential entrepreneurs of higher education institutions. Interviewee PS4, a mentor in entrepreneurship, said:

"I am working on a mentorship initiative, but I find it very difficult. Maybe it is this Sri Lankan context here. Even going into the private higher education sector for free and offering our resources and time to help students, I still find a lot of barriers. Our future youth can benefit from a mentor who will elevate their potential and achieve their goal. In my process, we train and develop the mentee also how to receive the best. But higher education institutions need to give us access to do this service."

This highlights that a higher education institution can develop their internal mentorship program and collaborate with external mentors to come onboard for the benefit of their students.

8.8.2 Fully-fledged incubator, internally or with external support

Business incubators were first established in the 1950s, and became popular, quickly spreading worldwide, primarily in developing countries (Voisey et al. 2013). Establishing an on-campus incubator for students and graduates not only facilitates new start-ups but also upgrades a higher education institution from an isolated to a networked institution, creating relationships with private and public sector stakeholders interested in entrepreneurship (Etzkowitz 2002). Other studies relate incubators within the higher education institution to create successful start-ups by students and graduates (O'Brien & Cooney 2019; Breznitz & Zhang 2019; Etzkowitz 2013), and still others argue that hosting an incubator guarantees an influence on the rate of start-up activity (Di Gregorio & Shane 2003).

Entrepreneurial universities have accepted the variety of benefits that a business incubator can offer to students/graduates, staff and institutions, acknowledging that it can

initiate and promote student entrepreneurship (Al-Dajani et al. 2014; Rasmussen & Borch 2010). Interviewee DH2, a head of school, shared:

"We have a dedicated Incubation Centre which was established a few years ago as an entrepreneurial arm. This was established with the purpose to empower students and support their innovative projects with commercial value. It is fully funded by the institute itself and not many institutes can do this on their own. Students are given the opportunity to work on their entrepreneurial projects through the fully equipped incubator at no cost. I think the most important aspects is the sharing of knowledge that happens within the incubator. These students also receive training opportunities related to entrepreneurship like workshops and seminars. Services of the incubator are accessible to all students regardless of their year of study."

A university business incubator is an essential element of entrepreneurial support for HEEEs (Theodoraki et al. 2018). An academic teaching an entrepreneurship-related unit in a different institution, interviewee AC3, said:

"We offer an in-house incubator for our students and students from any field of study can use its services. Student can run their business in this incubator for two years without any overhead costs including rent, energy, wi-fi and so on. They can grow their business during this time and phase out into a place of their own. Engaging in the incubator can impact their attitude towards a self-made career. This incubator is monitored and supported by staff, and it is our way of supporting our students to successfully launch their ventures."

Three of six institutions do not currently have an internal or access to an external incubator. In this case, higher education institutions are unable to fund and operate an internal incubator and should reach out for external support. Internal and external stakeholders collectively perceive that incubation centres develop students' E&I capabilities and help them with a new start-up through fully-fledged facilities.

8.8.3 Accelerator programs available privately or nationally

An accelerator is a functional entrepreneurial support activity (Spigel 2016). Having a standalone accelerator with committed resources and infrastructure signifies high commitment to entrepreneurship by a higher education institution (Brush 2014). Student interest in entrepreneurship is influenced by supportive infrastructure such as an accelerator (Huang-

Saad, Fay & Sheridan 2017). Students with start-up ideas look for opportunities to move into incubator and accelerator facilities. Interviews confirmed that there were no accelerators attached to any of the participating higher education institutions.

Accelerators are organisations that emerged in response to the shortcomings of incubators and aim to accelerate successful start-ups and venture creation through an intensive program for a limited duration (Wright, Siegel & Mustar 2017). Interviewee PS4, an external mentor in entrepreneurship, added:

"Creating a facility like an incubator with working space, professional services, workshops and skill training is helpful. But it can also be a bigger step like an accelerator program. Students get to commercialise their idea or scale-up the business through a 10–12-week accelerator program. I think private institutions have a responsibility to create these platforms and provide students that intensive coaching. Also, on the other hand, if they are really good, they can secure seed funding through the program and that's a big relief for new entrepreneurs. This is the support; institutes should offer their students to really engage in and get their hands dirty."

External stakeholders including entrepreneurs and mentors expressed that accelerator programs help students to grow as entrepreneurs and develop their start-ups.

According to the HEEE study by Miller and Acs (2017), an accelerator can be part of the local entrepreneurial ecosystem which higher education institutions reach out to for specialised services. Interviewee EE1, a second-generation entrepreneur and co-founder of a global brand, revealed:

"External stakeholders bring in practical knowledge and resources that can be used in different ways. Institutes commonly invite us as speakers for a one-time show. But they can also make these stakeholders a part of an accelerator program. I am currently part of an accelerator and there is some intense work that's being done. I have witnessed how young entrepreneurs move to the next level with the support of more experienced entrepreneurs who have failed and succeeded. They do not leave the program with the same attitude for sure. But this accelerator is not attached to an institute and I haven't heard of an institute that runs an accelerator program." In the case where a higher education institution is unable to host and operate an accelerator program in-house, students should be directed to such programs externally through the institution.

8.9 Chapter summary

This chapter presented the second half of findings for the second research question. Sections 8.2 to 8.6 provided the analysis and findings for codes that emerged during the content analysis on how HEEEs can operationalise mechanisms towards students E&I capabilities. Eighteen different mechanisms are suggested by internal and external stakeholders towards six contextual factors of HEEEs. The contextual factors and corresponding mechanisms are:

- Entrepreneurial orientation commit senior leadership to developing student E&I capabilities, foster entrepreneurial culture within the higher education institution, and empower staff with the responsibility of entrepreneurial initiatives
- (2) Entrepreneurship and innovation education offer E&I education to all undergraduate students, apply a practical-oriented pedagogy for delivering E&I education, and engage students in real-world assessments
- (3) Entrepreneurship and innovation research encourage students to practice applying research through coursework, expose students to acquiring scientific knowledge through research conferences, and offer opportunities for students to research startup projects
- (4) Enterprise experiences create access to entrepreneurship competitions at institution, inter-institutional and/or national level, offer internship opportunities with start-ups in a range of different industries, and provide start-up projects through the institution or external organisations
- (5) Entrepreneurial networks engage students with peers interested in entrepreneurship at institution and inter-institutional levels, connect students to alumni who have succeeded and failed in entrepreneurship, and create ways for students to connect with various external stakeholders
- (6) Entrepreneurial support offer students mentoring with practitioners including start-up entrepreneurs, provide students with a fully-fledged incubator, and, internally or with external support, leverage students through accelerator programs available privately or nationally.

These mechanisms lead to explanations of how higher education institutions can implement contextual factors for the development of students' E&I capabilities. This section of findings is underrepresented in HEEE literature and advances knowledge and understanding of operationalising HEEEs in resource-constrained environments. Given the resourceconstrained environment, such mechanisms encourage HEEEs to evolve beyond their standalone nature and connect with external stakeholders in wider entrepreneurial ecosystems.
CHAPTER 9: FINDINGS AND ANALYSIS FOR RESEARCH QUESTION 3

9.1 Chapter overview

This chapter relates to the last research question, exploring how diverse stakeholders engage in the factors of HEEEs that could influence students E&I capabilities in a resource-constrained environment. Findings from this research question intend to determine avenues for higher education institutions to foster stakeholder engagement to influence students E&I capabilities within the context of HEEEs. When designing or revising their HEEEs, higher education institutions need to identify relevant stakeholders, understand their shared value and recognise the relationships that contribute to developing students' capabilities for E&I. This knowledge enables institutional management to co-create and evolve their HEEE to understand stakeholder engagement within the contextual factors that benefit the E&I capabilities of students. The structure of this chapter is presented in Figure 9.1.





9.2 Analysis of diverse stakeholder perspectives on stakeholder engagement

Through multi-stakeholder perspectives, the final research question explores stakeholder engagement within the six contextual factors leading to student development in E&I within HEEEs. In the first round of interviews, data were coded for stakeholder engagement. During the second round of interviews, data related to social capital dimensions emerged as opinions related to structural, cognitive and relational social capital. In the process of analysing and interpreting the data, a three-level data structure was developed, moving from interview data to theoretical interpretation (refer Figure 9.2).

Coding was attached to participant opinions and viewpoints, defining the initial codes and making sense of the data gathered. These initial codes are based on general aspects such as 'stakeholders' reflecting on their experiences with students, higher education institutions, HEEEs and entrepreneurship. Then, initial codes were consolidated into expanded codes based on more abstract concepts of related theory. For example, 'internal and external stakeholders' were identified as 'ecosystem actors'. Finally, these expanded codes became the aggregated themes relating to social capital theory, and specifically social capital dimensions. In this case, 'ecosystem actors' related to structural social capital.

In this sense, higher education institutions can use 'ecosystem actors', 'network ties' and 'ecosystem configuration' to develop HEEE structures to empower stakeholder engagement and thus 'structural social capital' is the first dimension. Within 'cognitive social capital', various forms of shared value including 'shared goals and purpose', 'shared knowledge and understanding', 'shared values and culture', 'shared language and narratives' and 'shared identity' identified the resources that provide the system of meaning among different stakeholders in the HEEE. Finally, 'relational social capital' is a combination of parameters including 'obligations and expectations', 'trust and respect', 'norms' and 'reciprocity' that represent the nature of quality of relationships between stakeholders in the HEEE.

RQ3: How can diverse stakeholders engage in the factors of HEEEs that could influence students E&I capabilities in a resource-constrained environment?



Figure 9.2 – Theme structure for research question 3

In this chapter, the social network is reported first, suggesting the 'actors' that can collaborate with the HEEE for developing students' E&I capabilities. The next section describes the structural, cognitive and relational social capital related to the six HEEE contextual factors. In the final section, stakeholder engagement is presented, paying attention to structural configuration of HEEEs, shared value and relationships within HEEEs. These three sections answer the final question on how diverse stakeholders can engage within the HEEE, contributing to students E&I capabilities.

9.3 Social network of stakeholders in HEEEs

Participants shared their opinions on their social networks; that is, who was currently involved in HEEEs and who they perceived should be involved as key stakeholders of HEEEs. Based on the content analysis, 12 categories of stakeholders important to HEEEs in a resourceconstrained environment were identified. These categories could be further understood as internal and external stakeholders that emerged as the social network. Internal stakeholders are institutional management, educators, staff representatives and students, while external stakeholders are other higher education institutions, entrepreneurs, alumni, financial institutions, angel networks, service support providers, corporates, not-for-profit organisations and parents. Some of these stakeholders are common to HEEEs, while a few, such as other higher education institutions and private companies, are unique to the resource-constrained context. Developing E&I capabilities requires the involvement of the larger community and wider entrepreneurial ecosystem (Bischoff, Volkmann & Audretsch 2018).

9.3.1 Institutional management

If higher education institutions are to use their HEEEs to become entrepreneurial, there needs to be an entrepreneurial drive from the top: namely, senior management. A strong viewpoint among external stakeholders was that management positions within institutions need to commit and lead entrepreneurship. In essence, higher education institutions need their management to set objectives and endorse strategies that promote an entrepreneurial culture and develop students' capabilities for E&I. Institutional management including chief executive officers, department heads and deans/heads of school are key stakeholders in HEEEs who act as role models, initiating, advocating and leading E&I.

9.3.2 Educators

Educators continue to be significant stakeholders of HEEEs where internal and external stakeholders value entrepreneurship education. This research included educators in entrepreneurship, confirming that all six higher education institutions had academics responsible for delivering entrepreneurship or subjects related to entrepreneurship. A fundamental challenge for students and institutions is that their educators may have work experience but not experience in entrepreneurship. Educators may need to come from the field of entrepreneurship research while perhaps not having been an entrepreneur, they have a deeper understanding of the field. External stakeholders including entrepreneurs and investors suggest that educators should have entrepreneurship experience, even when delivering theory-related content to students. Further, there was consensus that educators should play an active role in other contextual factors such as entrepreneurial orientation, E&I research and entrepreneurial support in addition to the education element.

9.3.3 Staff representatives

In addition to senior management and educators, staff can promote entrepreneurship among students. For example, an industry liaison manager could be involved in value-added services. These staff representatives bring in additional effort to empower entrepreneurship initiatives within HEEEs. Two out of six higher education institutions have this practice in place, where such staff oversee various internal activities involving students. Among the six contextual factors of HEEEs, staff representatives can contribute to managing student participation in entrepreneurship competitions outside the institution (enterprise experiences), administering student communities, organising networking events (entrepreneurial networks), facilitating mentoring with entrepreneurship mentors and leading incubator programs (entrepreneurial support). This includes breaking away from limiting students to internal initiatives and giving them access to external programs and platforms with the support of staff representatives.

9.3.4 Students

While HEEEs serve the higher education community including academics, staff and alumni, students tend to be the central focus of higher education institutions. A portion of youth enter higher education institutions to pursue a career that inspires and interests them after undertaking an undergraduate degree. Higher education institutions in Sri Lanka predominately focus on teaching students to become proficient employees to take on corporate careers, instead of successful entrepreneurs. However, educators pointed out that there is a growing population

of students keen on entrepreneurship in their classes. These students represent a fraction of the Sri Lankan competitive talent and can be potential entrepreneurs that go on to the entrepreneurial ecosystem. In this case, students undergo a transition through their experience in the HEEE, developing their E&I capabilities.

9.3.5 Other higher education institutions

Higher education institutions tend to collaborate with other institutions in Sri Lanka for various events including sports and entertainment. However, higher education institutions with an entrepreneurship focus that participated in the study did not express any affiliations with other institutions or public universities for entrepreneurial activities such as competitions, networking events and incubators or accelerators. In some instances, public universities in Sri Lanka have access to more resources and is discussed later in this chapter. A higher education institution or public university with incubator facilities can share their resources with students from other institutions. Another higher education institution active with their network of potential student entrepreneurs can connect with similar students and network with other institutions. Higher education institutions can collaborate and support each other in resource-constrained environments.

Further, as higher education institutions in Sri Lanka closely work with international universities offering affiliated study programs, external participants expressed the desire to strengthen HEEEs through these relationships. These international collaborations can play an important role since universities from resource-rich environments can share resources and opportunities with affiliated partners to some extent. Although their best practices may not work as well for HEEEs in resource-constrained environments, such cooperation can expose students beyond their local settings. Other higher education institutions, local and international, joining forces towards developing students with capabilities can strengthen their respective HEEEs.

9.3.6 Entrepreneurs

Most participants emphasised the importance of entrepreneurs for HEEEs, identifying them as key stakeholders with a strong influence on developing students' capabilities in E&I. Their role in the HEEE is particularly important due to their experience in entrepreneurship and strong ties with the entrepreneurial ecosystem. The interviews revealed that entrepreneurs could be alumni, start-ups or established entrepreneurs. However, it was pointed out that entrepreneurs in the early or growing stage of their ventures should be in close in contact with students. These

entrepreneurs may have more ability to engage and see greater value compared to mature and established entrepreneurs whose start-up experience is likely to be from more than five years ago. Another view from multiple participants was that students should receive exposure from entrepreneurs of various paths, irrespective of whether they have had a higher education and if they have experienced some form of entrepreneurship and have experience to share. Currently, higher education institutions mostly welcome mature entrepreneurs who have a success story to share with students for inspiration.

9.3.7 Angel networks

Interaction between higher education institutions and angel networks are almost non-existent within this HEEE context. Angel investors are individuals or companies who provide financial capital for a start-up or SME, usually in exchange for ownership equity or convertible debt. Participants from angel networks express their strong willingness to work with higher education institutions to sustain their HEEEs and not just develop their students. HEEEs can benefit from various provisions of support from angel networks including E&I education, entrepreneurial networks and entrepreneurial support. Investors from angel networks evaluate business pitches presented by potential entrepreneurs and their capabilities to pursue the entrepreneurial journey. Angel investors are useful expert advisors on developing curriculum and assessments for entrepreneurship, and are resource personnel for networking activities, mentoring, incubators and accelerators. In addition to their formal duties, investors claim that they attend various entrepreneurship-related competitions organised by third parties to evaluate new business ideas and business pitches but rarely interact with institutions.

9.3.8 Start-up support service providers

Start-up support service providers can be highly diverse and are identified as support professionals who offer advisory services and organisations with infrastructure, including fully functional co-working spaces and incubators. Support services can be sponsored, voluntary or paid depending on the arrangement and requirement. Although such start-up support service providers are beneficial for higher education institutions in resource-constrained environments, there is no widespread linkage between the two parties. Higher education institutions appeal for such entrepreneurial support within their HEEE but this need is not limited to only one factor. These support service providers can play an important role in making students aware of important areas that they should pay attention to such as company statutory requirements,

regulatory requirements and governance. Without this practical knowledge, potential entrepreneurs lack the capability to secure funding and investments for their start-ups.

9.3.9 Private companies

Currently, there is minimal linkage between HEEEs and private companies although higher education institutions have partnered with companies for other educational purposes. Both internal and external participants pointed out the need for support from companies for HEEEs to effectively function in resource-constrained environments. Companies in this case can be from a wide range of industries including large multinationals that can offer resources to HEEEs focused on students' E&I capabilities. This stakeholder group does not include start-ups and SMEs.

Private companies can support HEEEs with various contextual factors including E&I research, enterprise experiences and entrepreneurial support. HEEEs can work with companies for mutual benefit while developing students' E&I capabilities. It came to light that some top multinationals in Sri Lanka have introduced entrepreneurial support programs for selective higher education institutions, but one challenge was that some support was exclusive to the HEEE. In this case, higher education institutions are urged to support a proposal that aligns with the respective company intending to build long-term relationships.

9.3.10 Financial institutions

None of the higher education institutions reported interactions with financial institutions for the purpose of supporting their HEEE. Financial institutions of private and public origin can collaborate directly with higher education institutions or indirectly with support through funding. This financial support can be in the form of sponsorships, seed funding and loan schemes. Such support can offer students programs various enterprise experiences and entrepreneurial support, while developing E&I capabilities. Financial institutions are known for various initiatives like financial aid for good causes and acts of corporate social responsibility. In addition to funding, financial institutions sometimes provide guest speakers for financial-related topics in start-ups and panel members for entrepreneurship competitions.

9.3.11 Not-for-profit organisations

Non-for-profit organisations and public universities are working together to accelerate entrepreneurship within higher education. However, private higher education institutions that participated in the study did not express any linkages with non-for-profit organisations for the purpose of promoting entrepreneurship among their students. By nature, private higher education institutions have limited contact with such organisations. Both internal and external participants expressed that HEEEs can gain from external support towards the national priority of developing students with E&I capabilities.

As HEEEs are part of the wider entrepreneurial ecosystem, a focus on creating local and national partnerships with non-for-profit organisations is considered significant. Such organisations include chambers of commerce, and chambers of young Sri Lankan entrepreneurs and youth business Sri Lanka. Some external participants expressed that higher education institutions are not open for engagement in most instances and should initiate formal communications with such non-profit organisations.

9.3.12 Parents

Some entrepreneurs and support professionals argue that more higher education institutions should offer relevant entrepreneurship education by identifying different undergraduate groups. For instance, in Sri Lanka it is common for the younger generation to follow their parents and join the family business, resulting in next-generation entrepreneurs. These family businesses want to grow by launching new products, entering new markets and even diversifying. Their children with E&I capabilities, next-generation entrepreneurs having the ability to bring innovation to their family business are likely to lead towards higher performance and revenue. Such family business-oriented parents seek suitable entrepreneurship education for their children. However, most business degrees offered are limited to business management and business administration. These degrees include a minimal extent of entrepreneurship such as one subject.

Participating higher education institutions give minimal importance to parents within their HEEEs as they only appear to be key target audiences of marketing communications. In this case, HEEEs lack consideration of external influences such as parents of students. Parents are identified as an influence on the entrepreneurial intention of students (Webber, Kitagawa & Plumridge 2020). Further, family social capital is found to play an important role for startups in resource-constrained environments (de Brito, Lenz & Pacheco 2022). Through this study, parents are recognised as a key stakeholder group that hold high interest in their children's education and play a decision-making role in their children's career choices. Higher education institutions can benefit from engaging with parents and keeping them satisfied. Thus, this appears as another point where HEEEs are disconnected from their wider entrepreneurial ecosystems.

9.3.13 Mapping social networks in HEEEs

The previous chapter highlighted six contextual factors as the composition of HEEEs that could influence student E&I capabilities in a resource-constrained environment. As discussed above, the 12 identified stakeholder groups play an important role and contribute to HEEEs by igniting inspiration, sharing knowledge, providing resources, offering advice or information, and leading E&I capabilities among students. Each stakeholder group of the social network is mapped against the six contextual factors of HEEEs in resource-constrained environments below in Table 9.1. It is apparent from the table that internal stakeholder needs to drive entrepreneurship within the HEEE, educators can be involved beyond E&I education, and external stakeholders can leverage enterprise experiences, entrepreneurial networks and entrepreneurial support for students. Students are involved in each contextual factor as they experience the HEEE.

	Entrepreneurship orientation	Entrepreneurship & innovation education	Entrepreneurship & innovation research	Enterprise experiences	Entrepreneurial networks	Entrepreneurial support
1. Institutional management	V					
2. Educators	M	V	V			V
3. Staff representatives	V			\checkmark	V	V
4. Students	M	V	V	\checkmark		V
5. Other higher education institutions				\checkmark		V
6. Entrepreneurs		V	V	\checkmark		V
7. Angel networks		V			V	V
8. Financial institutions				\checkmark		V
9. Support service providers						V
10. Companies			Ø			
11. Not-for-profit organisations						
12. Parents		V				

Table 9.1 – Mapping social networks against the HEEE contextual factors

 \blacksquare identifies that a stakeholder is relevant for a respective HEEE factor

While a social network primarily emphasises an institution's network, social capital theory sees this network as a source of resources highlighting the value of relationships (Hayter

2013). Social capital is argued as the glue that binds social networks and the lubricant that energises network interactions (Powell & Smith-Doerr 1994). Further, social capital increases the likelihood of an individual becoming an entrepreneur and advances the entrepreneurial process (Baron & Markman 2003; Davidsson & Honig 2003; Sahasranamam & Nandakumar 2020). In social capital theory, social capital has been described as 'structural', 'cognitive and 'relational'. The structural dimension of social capital involves the people of the network that shape the structure and influence outcomes (Anderson & Jack 2002). The cognitive and relational dimensions seem similar, however cognitive refers to shared knowledge and understanding, whereas relational relates to affective behaviour among people (Anderson & Jack 2002). The three social capital dimensions of social capital theory are relevant for an HEEE to function effectively (Theodoraki, Messeghem & Rice 2018). The findings in this section suggests the social networks are useful to create and use social capital within HEEEs for higher education institutions. In this case, to better connect with relevant stakeholder groups and build affiliations that serve HEEEs. Social capital dimensions for each contextual factor are discussed next.

9.4 Social capital dimensions for entrepreneurial orientation

Within the first HEEE factor of entrepreneurial orientation, stakeholders discussed the senior leadership, entrepreneurial culture and entrepreneurship chair(s) as mechanisms to operationalise HEEEs. In this section, the three dimensions of social capital are discussed in relation to entrepreneurial orientation and its three mechanisms.

9.4.1 Structural social capital for entrepreneurial orientation

For entrepreneurial orientation, internal and external stakeholders emphasise that institutional management, educators and staff representatives primarily make up the internal actors of the social network. It is likely that an institution will be unable to co-create an HEEE without senior leadership or implement it without support from staff (Rice, Fetters & Greene 2014). Their internal drive is considered as the commitment to build a robust HEEE using leadership from the senior management and through a succession of staff responsible for entrepreneurship within the institution. The social structure inside the institution and network ties between the internal stakeholders play a crucial role in driving the institution from within towards entrepreneurship. The first HEEE factor emerging from this study; entrepreneurial orientation, is the only factor involving internal stakeholders while the rest include a combination of internal and external stakeholders.

Interviewee AI3, an angel investor working with young entrepreneurs: "... if you look at the US universities and compare with Sri Lanka. What do we lack? From my observations and experiences, we lack people from the top enforcing entrepreneurship or bringing about that change into the institute. If senior management does not value or embrace entrepreneurship, how can we expect the future generation to be aware of entrepreneurship and what entrepreneurship is about?"

Interviewee DH2, a head of school: "We have two entrepreneurship pillars appointed for the role of fostering the entrepreneurial spirit among students. These pillars are academics teaching entrepreneurship but also have a zest for entrepreneurship. Basically, their role is to promote entrepreneurship our students. The e-club is one of their responsibilities as well. Monthly they must report what has been achieved and discuss upcoming initiatives."

Interviewee AC2, an academic in charge of an entrepreneurial initiative: "I feel in any institute, the top management has a very critical role to play. Because they are the people who actually set the culture organisation for the whole institute. So, if the leaders or the top management doesn't cherish an entrepreneurial culture, if they are not really valuing the concept of entrepreneurship, I don't think it is possible to implement entrepreneurship. So, if they have a shared mindset and mission, obviously, that would be spread over to the institute, because they are the people who set the tone for it. So that is one party. And obviously the next party, I can say it's the faculty members of the teaching faculty, because they are the interface between the students and the top management."

9.4.2 Cognitive social capital for entrepreneurial orientation

In actively driving entrepreneurship within the institution, a shared sense of vision and mission plays a significant role by bringing everyone together towards a common goal and purpose in the institution. The involved employees, from senior management to selected employees, may adopt shared values causing a chain reaction and spreading the entrepreneurial culture throughout the institution. These shared attitudes and beliefs will continue to strengthen relationships as information relating to entrepreneurial initiatives flows through various channels among employees. Common knowledge and understanding can develop member perceptions, interpretations and representations for outcomes desired from the HEEE. Cognitive social capital for entrepreneurial orientation relates to shared goals, purpose, values,

culture, language and information that forms the background environment of the HEEE and behaviour of the people in the institution managing the HEEE.

Interviewee SP4, an external mentor in entrepreneurship: "Especially in a country like Sri Lanka, it is not just about teaching entrepreneurship as a module or course. I believe that institutes should focus on developing that spirit, as well as that mindset, and you know, encourage an entrepreneurial culture within the student population. Of course, these shared values should come from the top and spread within the institute. This is mainly because entrepreneurship can happen at any point in their careers and life ..."

Interviewee AI5, an angel investor: "At university level, the culture must support it. If one joins a higher education institution today if that entrepreneurial culture does not exist, then there is no way for that student to really learn about entrepreneurship, be interested in such a career or develop the necessary capabilities. So, the culture of these institutions must drive their ecosystems. If you look at some of the popular universities that have successfully produced entrepreneurs and successful entrepreneurs are the ones that have that entrepreneurial culture nested within."

Interviewee AC1, an academic teaching entrepreneurship: "So you look at the institute and whether we are encouraging our students to become entrepreneurs or not, I think it's starts from our mission and corporate objectives. We have one objective focused on entrepreneurial development. From that, I will say we intend to support our students undertake self-employment if they wish too ..."

9.4.3 Relational social capital for entrepreneurial orientation

Having established the structural and cognitive dimensions for entrepreneurial orientation, this section emphasises relational social capital. To actively drive the institution towards entrepreneurship, the relationships within the institution and with the wider entrepreneurial ecosystem becomes significant. Table 9.2 presents some of the obligations that stakeholders perceive higher education institutions can commit to. Senior management should recognise and celebrate entrepreneurial achievements that encourage relevant behaviour among employees. Using entrepreneurial culture, collective entrepreneurship goals can be associated with individual goals where each employee shares the group norms. Stakeholders also pointed out that higher education institutions can facilitate connections with a variety of internal and external stakeholders developing social relationships within the HEEE. The relational dimension for entrepreneurial orientation relates primarily to creating trust among

stakeholders, sharing norms, and strengthening the quality of relationships for the benefit of HEEEs.

Interviewee EE6, a social entrepreneur: "How do institutions make a robust ecosystem and get people to serve that ecosystem? It can be done by bringing the separate people across different levels together. First, they need to solidify the internal aspect by building and coordinating the ecosystem at institution level. And culture can help with this a lot. Process wise it calls for sharing information, forming capabilities, and structuring the interactions needed to orient the institution."

Interviewee AE9, an alumni entrepreneur of a tech company: "Most institutions still operate as separate with very limited interactions with the outside or limited in their purpose. But that's not a successful approach for something like an ecosystem. Institutions can energise using the external stakeholders. The institution and its people need to put in effort to collaborate with stakeholders from the environment who can support and relationships need to be built on trust and shared values."

HEEE factor & mechanisms	Structural: actors of the social system	Cognitive: system of meaning	Relational: obligations and expectations
Entrepreneurial orientation Senior leadership Entrepreneurial culture Entrepreneurship chair(s)	 Institutional management Educators Staff representatives 	 Shared goals and purpose including vision and mission Shared values and culture Shared language 	 Recognise and celebrate entrepreneurship Coordinate the ecosystem diffusing entrepreneurial culture Facilitate connections with a variety of internal and external stakeholders

Table 9.2 – Social capital dimensions for entrepreneurial orientation

9.5 Social capital dimensions for entrepreneurship and innovation education

For the second HEEE factor, including E&I education, stakeholders emphasised entrepreneurship curriculum, practical pedagogy and real-world assessments as mechanisms within HEEEs. Below, the social capital dimensions are discussed in relation to E&I education and these mechanisms.

9.5.1 Structural social capital for entrepreneurship and innovation education

A combination of deans/heads of school, educators and entrepreneurs emerged as the actors relevant for E&I education. It was the viewpoint of multiple stakeholders that entrepreneurship as a field of study can break the traditional mode of an institution's academic operations and open it to the wider environment. This change highlighted the importance of involving practitioners, entrepreneurs who have launched a startup or are (or have been) involved in the entrepreneurial process, in various aspects of education. Only a combination of academics and practitioners can create the unique learning experience that entrepreneurship calls for. By joining these forces, higher education institutions can become more relevant to the growing student population keen on developing their entrepreneurship capabilities.

Interviewee SP2, a mentor promoting student entrepreneurship: "Entrepreneurs learn from other entrepreneurs. That is the golden rule. That's why individuals are keen to be in such a room where they can learn from people who have done it and feel comfortable asking questions. Majority of the time, students don't ask questions in a normal classroom setting but when its driven by someone they can relate to learning becomes easier and interesting I would say ..."

Interviewee AI6, an angel investor working with young entrepreneurs: "Even if it's a simple cupcake business, potential entrepreneurs need to have the basic knowledge of doing a business. From an education institute standpoint, the focus must be on providing education that gives students this essential knowledge. A lot of students come to us and what I see is that majority of them fail or show signs of failure because they fail to validate an idea whether there is an opportunity in that respective market. When tracing the problem, its goes to back their higher education of being highly academic."

9.5.2 Cognitive social capital for entrepreneurship and innovation education

Cognitive social capital includes shared knowledge and understanding from deans/heads of school, educators and entrepreneurs that flows within the social system. For E&I education, students are likely to learn better through subject and practical knowledge and develop stronger capabilities from a combined force of educators and practitioners. Such shared knowledge and understanding can be passed down beyond traditional lectures involving storytelling that situate the deeper understanding of the marketplace and industry. When education is taught in this way, it can cause students to start to create connections and relate shared knowledge to their own experiences and perceptions. These types of knowledge and understanding strongly

influence decisions and actions and work together for mutual benefit. In sharing subject and practical knowledge with students, entrepreneurs can bring industry insights related to entrepreneurship and make institutions more aware of the outer word; this in turn helps to develop the institutional portfolio of study programs and curriculums.

Interviewee EE1, a second-generation entrepreneur and co-founder of a global brand: "I've always looked at education, especially higher education, as 'you learn to learn'. Education gives you the skill set, and it gives you the knowledge. For example, what they learn gives students the knowledge to structure a start-up, to structure their thoughts, to structure business plans, and it gives students a widened knowledge about the different aspects or different spectrums within a business. Most importantly, it gives students the necessary tools to keep learning, absorbing information, and present information and so on, which is integral to running a business or creating a brand."

Interviewee AI1, an angel investor working with start-ups: "Entrepreneurs can be invited to get involved in structuring of study programs in entrepreneurship and selecting the content that needs to be essentially taught in these programs. They [entrepreneurs] can bring in the practical insights of what knowledge and skills students should develop, being at the receiving end of the supply chain. This exercise will bring a good balance of people including academics and practitioners to benefit students who may become potential entrepreneurs."

9.5.3 Relational social capital for entrepreneurship and innovation education

The above discussion recognised the actors of the social system and the meaning of the system for the contextual factor, E&I education. Next the affective element that involves obligations and expectations is explored (refer Figure 9.3). Such obligations strengthen reciprocity as higher education institutions strive to produce entrepreneurial talent in the entrepreneurial ecosystem. When higher education institutions collaborate with stakeholders beyond educators for E&I education, they develop relationships between the HEEE and their wider entrepreneurial ecosystems. Curriculum development must draw the participation of alumni and they are also a resource for various other activities including guest lecturing (Fetters, Greene & Rice 2010). Instead of creating entrepreneurship curriculums and limiting input to internal views, external stakeholders such as angel investors can be invited to be part of curriculum and assessment development by sharing their input on future graduate capabilities for entrepreneurship and innovation. In addition to involving alumni entrepreneurs in sharing practical knowledge, students should be given opportunities to apply their learned theoretical knowledge in real-world assessments associated with start-ups.

Interviewee AE3, a young entrepreneur of a fitness start-up: "during curriculum development or deciding on new [study] programs, the tendency is that these institutes they will look at another institute or university, and simply co-create more or less the same program. In doing that, they [higher education institutions] don't pay attention to the cultural nuances and what it really means for the students when developing a program or curriculum. Instead doing this, they can have a group of practitioners that they turn to for input on curriculum development. Because building a start-up in the Silicon Valley is definitely not the same in building a company in Sri Lanka for multiple reasons starting from the limited resources ..."

Interviewee EE4, an award-winning entrepreneur of a social venture: "*it is surprising* to me how some of these entrepreneurship contents are taught in degrees by people who have not experienced entrepreneurship. For example, valuations of new business is taught to a class of undergraduate students by someone who has not even raised capital for a start-up. So, it's one part for academics to come and represent the theory [in a classroom] but it's very different to how theory is applied in the practical world. Again, the focus has to be show students the real world experience, the practical knowledge in addition to the academic knowledge."

HEEE factor & mechanisms	Structural: actors of the social system	Cognitive: system of meaning	Relational: obligations and expectations
Entrepreneurship and innovation education Entrepreneurship curriculum Practical pedagogy Real-world based assessments	 Deans/heads of school Educators Entrepreneurs Angel networks 	• Shared knowledge and understanding through theoretical knowledge, practical knowledge and industry insights	 Share insights on future knowledge, skills and abilities for curriculum/program development Deliver a combination of theoretical and practical knowledge Create opportunities for practical application of knowledge

Table 9.3 – Social capital dimensions for entrepreneurship and innovation education

9.6 Social capital dimensions for entrepreneurship and innovation research

In the third HEEE factor involving E&I research, stakeholders highlighted research-based coursework, research-based projects and research conferences as mechanisms to implement research within HEEEs. Below, social capital's structural, cognitive and relational dimensions are discussed concerning E&I research and these mechanisms.

9.6.1 Structural social capital for entrepreneurship and innovation research

Within this factor, knowledge and understanding can be accumulated through research-based coursework, research-based projects and research conferences involving the contributions of various stakeholders. Interviewees identified educators, entrepreneurship associations, entrepreneurs and research companies as actors significant for E&I research within an HEEE. This is another HEEE factor that depends more on external ties than internal actors as the focus is inclined towards applied research. While educators can promote opportunities for research, other external stakeholders can step in to facilitate research. Applied research requires access to the community's knowledge associated with entrepreneurship such as entrepreneurial opportunities (Brush 2014).

Interviewee SP3, a professional providing advisory input to entrepreneurs: "I think there is a fair extent of research and some data is already available. So, associations like Chambers of Commerce can play a role in providing insights to students into a different industries, like emerging sectors or import sectors. In the same way even private research companies can come forward to share some their knowledge with these students for entrepreneurial purposes."

Interviewee EE5, an entrepreneur of a fashion start-up: "Lecturers can also take initiative to familiarise students with applied research. They can identify some data that is needed for an assignment and approach research companies to provide some information. It is naturally tough for individual students to reach these places on their own but for academic purposes it is possible. So, I think the start-up ecosystem is very helpful and there is a positive attitude towards entrepreneurs. It's only a matter of making the right connections to bring these benefits for students.

9.6.2 Cognitive social capital for entrepreneurship and innovation research

Cognitive social capital for E&I research relates to shared knowledge and understanding through market and business needs, industry insights and scientific knowledge. These

information resources provide meaning to the system for students to investigate and pursue entrepreneurship. Social cognition through research allows students to process and use information in an entrepreneurial context. It can improve capabilities such as research skills, and it further enables students to communicate and speak the same language as potential entrepreneurs. This common lexicon is significant for social interactions when students engage in entrepreneurial initiatives and work with practitioners.

Interviewee AC2, an academic in charge of an entrepreneurial initiative: "It doesn't have to be exact statistics or anything confidential. Just to open their [students] minds to what is currently happening in markets and industries so some customer information, market trends and business needs will help them [students] to think practically. This is the kind of assignments they need to work on rather than hypothetical situations and scenarios."

Interviewee SP3, an external mentor in entrepreneurship: "Research is very very crucial in entrepreneurship. Because information and data are key. Back in the day, we have heard about gut-feeling in taking decisions. But things have changed, and anything related to the product, market, competitors, environment and especially a new start-up is research driven. I have seen start-ups fail when they are built on poor research, especially market research."

9.6.3 Relational social capital for entrepreneurship and innovation research

In the case of relational social capital, developing capabilities in research for E&I and promoting applied research approaches among students can lead to sharing knowledge and understanding of markets and industries (refer Table 9.4 below). Stakeholders claim that this knowledge exchange develops relationships based on mutual trust and respect. In some instances, this interconnection drives entrepreneurship within the HEEE, influencing student E&I capabilities.

Interviewee SP3, an external mentor in entrepreneurship: "So, it's an important skill [research skills] in planning, launching and managing a start-up. I would say research skills is among the top five skills for a successful entrepreneur. For student entrepreneurs, research is the starting point of their start-up."

Interviewee AE1, an alumni entrepreneur founded tech start-ups: "Research plays a major role in entrepreneurship. Students nowadays should apply research through

their assignments and coursework. Institutions should promote applied research approaches among students. Because when you are an entrepreneur, you need to know everything related to your business functions, industry trends, everyday challenges, and even what's happening internationally. That's what improves your knowledge and understanding to run the show when one is in this entrepreneurship game. Then it helps that an entrepreneur apply research in everyday business decisions ..."

HEEE factor & mechanisms	Structural: actors of the social system	Cognitive: system of meaning	Relational: obligations and expectations
Entrepreneurship and innovation research			
Research-based coursework Research-based projects Research conferences	 Educators Entrepreneurs Entrepreneurship associations Research companies 	• Shared knowledge and understanding through market and business needs, industry insights and scientific knowledge	 Develop students' skills and knowledge in research for entrepreneurship and innovation Promote applied research approaches among students

Table 9.4 – Social capital dimensions for entrepreneurship and innovation research

9.7 Social capital dimensions for enterprise experiences

9.7.1 Structural social capital for enterprise experiences

Start-up entrepreneurs, other higher education institutions and private companies can facilitate enterprise experiences for students within the HEEE. These actors emerged as the top three network ties that describe linkages with external stakeholders beyond the institution. External stakeholders expressed their willingness to participate in respective HEEEs of higher education institutions, which established that the local entrepreneurial ecosystem is supportive of HEEEs. Higher education institutions can develop their network members by collaborating on mutual agreements or formalising these relationships through a memorandum of understanding. Developing network ties and connectivity makes it easier for institutions to evolve and engage in mutually beneficial collective actions for the HEEE.

Interviewee AE2, a young entrepreneur with a service-oriented start-up: "higher education institutions can actually do something to break the current practice of the bigger companies coming in, promoting their internship programs and taking the cream of the crop. Then they [institutions] can also attract successful start-ups to the ecosystem. Because currently these start-ups find it challenging to access university students. The benefit of course will be will for the students who get an early start and experience working for a start-up. In return the start-ups get access to good talent to work with and they [student] might spin off some good ideas as well."

Interviewee DH1, a dean of academic affairs: "Today some of the bigger companies like the multinationals are coming forward with various opportunities for students. I think as part of their entrepreneurship and innovation within the company they are investing large amounts of money to attract ideas from young talent. These are excellent opportunities for students if we [higher education institution] connect with the right companies."

9.7.2 Cognitive social capital for enterprise experiences

Students may create and use different types of social capital from activities outside the classroom and higher education institutions. For instance, when participating in entrepreneurship-related competitions, or undertaking an internship or start-up project, students move into distinct environments involving social interactions with entrepreneurs or similar. In these situations, they experience and learn shared language and develop communication skills as a potential entrepreneur. Further, enterprise experiences can orient students as potential entrepreneurs, empowering their efficacy towards entrepreneurship. This can lead to developing an identity as a potential entrepreneur, and drive passion, intentions and decisions. Such experiences also accumulate shared knowledge and understanding through market and practical knowledge.

Interviewee AE3, a young entrepreneur of a fitness start-up: "yes you can learn about entrepreneurship but the best way to learn entrepreneurship is by doing it. It's very hard to learn entrepreneurship only in a controlled environment like a classroom. Alumni or non-alumni entrepreneurs really need to be connected to these ecosystems providing these students interested in entrepreneurship the opportunity to learn in real situations and get to know the language in the real context."

Interviewee AC5, an academic teaching a career planning unit: "something that I have seen overseas but not really common here [Sri Lanka] is where students get work experience during their higher education. For example, if these undergraduate students can take on a placement or an internship with a start-up for a year. They will be embedded in that dynamic environment of a start-up and be involved from the small activities like photocopying, printing to functional tasks like marketing and be exposed to challenges of entrepreneurship."

9.7.3 Relational social capital for enterprise experiences

Stakeholders expressed their views on structural and cognitive social capital for enterprise experiences above. Relational social capital for enterprise experiences involves improving the quality of relationships among actors through obligations (refer Table 9.5). Higher education institutions can provide platforms such as competitions for students to showcase their entrepreneurial potential and capabilities. In addition to internal activities, students can be encouraged to participate in external events at inter-institutional or national levels. Such events should attract external stakeholders from entrepreneurial ecosystems, including representatives of various industries and sectors, to witness students' E&I capabilities. Students can also be exposed to entrepreneurship through enterprise programs, such as internships with start-ups and start-up projects. These social experiences contribute to students seeing themselves as potential entrepreneurs and developing an identity, including traits, abilities and norms.

Interviewee AE4, an alumni entrepreneur: "There are a lot of private companies who are running their own business competitions, hackathons and similar competitions. But they limited within their mandate, meaning within the respective industry the company is in. If its DIMO then it's focused on the auto automobile industry and they interested on pitching for that industry. If they come across a creative idea they like, they fund that idea and support the student, and they will take that idea into that industry. Another company is John Keells. There are platforms and more emerging as well. So I think external players are supportive now and they are getting ideas from outside and grooming it within the organisation."

Interviewee SP1, an external mentor and coach in entrepreneurship: "I am regularly invited to these pitching competitions organised by various institutions and associations. When investors or people like myself we go as the judge board, we witness the students entrepreneurial intention and not just their business idea. I have frequently experienced judges from the panel of judges picking a particular idea and giving a student an offer on the spot. And this is beyond the competition or who is selected as winners. So there are enough opportunities out there for students to benefit from."

HEEE factor & mechanisms	Structural: actors of the social system	Cognitive: system of meaning	Relational: obligations and expectations
mechanisms Enterprise experiences Entrepreneurship- related competitions Internships with start- ups Start-up projects	 • Start-up entrepreneurs • Other institutions • Private companies 	 Meaning Shared knowledge and understanding through market knowledge and practical knowledge Shared language Shared identity 	 and expectations Offer platforms to students for showcasing their entrepreneurial potential and talents Expose students to entrepreneurship through enterprise programs Encourage students to gather enterprise experiences through external participation
			• Build entrepreneurial identity of students

Table 9.5 – Social capital dimensions for enterprise experiences

9.8 Social capital dimensions for entrepreneurial networks

9.8.1 Structural social capital for entrepreneurial networks

As part of the social network in the HEEE, actors including students, other institutions, alumni entrepreneurs, investors and support professionals are essential. To develop the entrepreneurial community, networking opportunities should be created between students, other higher education institutions and external stakeholders. In addition to the interactions of students with educators, regular interactions among students with similar entrepreneurial interests from the same institutions (internal to internal), inter-institutional students keen on entrepreneurship (internal to external) and entrepreneurial practitioners such as alumni entrepreneurs (internal to external) should be encouraged through the HEEE. In terms of entrepreneurial networks, higher education institutions can bring in diversity and link students with various types of stakeholders that could benefit their development. Having different layers of networks together with the idea of inclusion can work well for the HEEE configuration.

Interviewee AI1, an angel investor working with start-ups: "Bringing in people whether it's from the same institute or from outside even entrepreneurs who don't necessarily come from an education background is a rich experience. All that matters is it is people who have built a business, raised capital, given employment, or brought money into the country. They [students] need to socialise with people who have walked the talk ..."

Interviewee EE4, an award-winning entrepreneur of a social venture: "student entrepreneurs inherit from their surroundings, especially from families, friends, their education and other networks. The changing point is usually from another human being or event, it could be something their younger days or during their education. So, I would recommend for institutions to give the right exposure, the right situations and with the right people for the students to learn themselves. This can even start with student communities where peers from the same or various institutes with common goals get together right."

Interviewee AC3, an academic teaching an entrepreneurship-related unit: "Students should meet people beyond the normal campus setup. Of course, the day-to-day people they are in connect with are the deans, the heads, all of us [academics] and their peers and even some alumni. But I think it should be more of the external people so that students are exposed to external influences, outside their campus. I would say maybe investors, entrepreneurs and other experts would be ideal to get involved with students."

9.8.2 Cognitive social capital for entrepreneurial networks

Social capital is created and used primarily in social interactions and entrepreneurial networks as an HEEE factor that allows bonds, bridges and linkages between students and internal and external stakeholders. Storytelling and interactions with peers, alumni entrepreneurs and external stakeholders lead to shared language and narratives. These entrepreneurial networks can increase the likelihood of social support for both students and HEEEs. In addition to shared languages, engaging with diverse stakeholders helps students see themselves as potential entrepreneurs and develop a sense of shared identity as part of the entrepreneurial community. The strong sense of belonging and acceptance as 'one of us' is highly motivating for students. Finally, through various interactions with stakeholders, students gain access to shared knowledge and understanding, including specific and practical knowledge. Students may be motivated to ask specific questions of practitioners or similar in safe environments that are not classroom settings.

Interviewee AE8, an alumni entrepreneur within an online start-up: "Institutions should be able to bring entrepreneurs to tell a great story, you know, be able to inspire students towards entrepreneurship, and then guide them to do what they need to. But majority of the entrepreneurs don't fall into that success category, there are entrepreneurs who have failed, learnt and then succeeded. There are also entrepreneurs who have failed and moved in other directions. These success and failure stories should all be welcome. I personally feel there is a lot to learn from failure than success."

Interviewee SP4, an external mentor in entrepreneurship: "So institutions should identify the right people and connect students with those people. Of course, within the degree one can learn how to write a business plan, how to do a pitch, or how to manage finance and so on. Today this knowledge is largely available, even online. But meeting the right people is something very rare, students can be influenced to become an entrepreneur, trigger a creative idea and learn specific knowledge for their questions. Through such people, students can have opportunity to get membership into this entrepreneurial community."

9.8.3 Relational social capital for entrepreneurial networks

The above sections refer to the network of people and organisations enabling HEEEs to stimulate shared language and narratives, knowledge and understanding, and belongingness and community development. Relational social capital includes parameters that influence relationships within HEEEs (see Table 9.6). Higher education institutions can facilitate networking through the HEEE to create connections with stakeholders relevant to their entrepreneurial journey. Entrepreneurial networks will form start-up communities for networking with relevant stakeholders. Students can interact with entrepreneurs through these entrepreneurial networks, observe their characteristics, and learn from them. Social interactions between students and other actors from the entrepreneurial ecosystem allow building trust, respect and even friendships, which can carry forward for future endeavours. Further, networking inspires togetherness, a sense of belonging and community spirit, building collaboration among students and other stakeholders. This relational social capital results in social development through social bonds with people of shared identity such as peers and entrepreneurs. Students tend to build their self-confidence and entrepreneurial identity through their networks (Longva 2021).

Interviewee PS6, a mentor in entrepreneurship: "I have come across young students who are entrepreneurial and have great business ideas. But they fall short because they don't have the networks. These don't have to be big networks, but they need access to networking opportunities. So, then they can build their networks and absorb relevant information. There is a lot of dynamics in the space of networking and its very high intense.

Interviewee AE6, a co-founder of a men's fashion start-up: "So it is not just general networking but something more specific as well. First thing to build solidarity for his idea, he should build a networking circle around his business idea. Let's see your innovation is related to the medical industry then he should get in contact with medical practitioners, pharmacies and medical research institutes. This is very important, and this is your start-up community. Through this circle the student will have access to various resources."

Interviewee EE3, a second-generation entrepreneur: "Education is not just about going to university and learning the theory of some subject. But the exposure which is way beyond. Meeting people from the outside world, interacting with them, getting to know about the society, people's behaviour, lifestyles and learning about environments. It's really important for students to learn about the society and learn from the broader community. This is the kind of social learning and attitude that students should receive to prepare themselves for something as entrepreneurship."

HEEE factor & mechanisms	Structural: actors of the social system	Cognitive: system of meaning	Relational: obligations and expectations
Entrepreneurial networks			
Peer engagement Alumni networks External stakeholders	 Students Other institutions Alumni entrepreneurs Angel networks Support service professionals 	 Shared knowledge and understanding through specific knowledge and practical knowledge Shared language and narratives Shared identity 	 Facilitate networking Provide social learning and resources Build start-up communities Build entrepreneurial identity of students

Table 9.6 – Social capital dimensions for entrepreneurial networks

9.9 Social capital dimensions for entrepreneurial support

9.9.1 Structural social capital for entrepreneurial support

When focusing on the structural configuration of the HEEE, with particular emphasis on entrepreneurial support, actors such as entrepreneurs, investors, private companies and not-forprofit organisations are of interest. Similar to entrepreneurial networks, higher education institutions can take into account the importance of network ties with external stakeholders for HEEEs. However, institutions can strengthen internal networks by expanding their role in entrepreneurial support as well. Through the structural configuration of HEEEs, higher education institutions can facilitate entrepreneurial support and draw on various benefits, including resources. HEEEs built on stronger ties with external stakeholders, with frequent interactions and closer interrelationships, can increase the level of entrepreneurial support that students can gain.

Interviewee SP5, an external mentor in youth entrepreneurship: "I would say a mentoring process, more like an overall mentoring program that involves experienced entrepreneurs or entrepreneurship experienced professionals like investors engaging with these young students. But this word mentor is highly misunderstood. When we reach institutions, they say they already have mentors so what the difference. So, in entrepreneurship, a mentor is an individual who can elevate these students and show them their potential to become an entrepreneur. And there are different ways to mentor such a student who is interested in entrepreneurship."

Interviewee AE5, an alumni entrepreneur of a revolutionary start-up: "I see the possibility for these educators take on the mentoring role. That way institutions can two ways of mentoring their students, internal and external mentors for their students is a more powerful approach. After all, these academics have close and ongoing contact with students, they have a trustworthy relationship and they can play a crucial role in improving their readiness. I am yet to see this kind of mentoring approach among private institutions but it can happen."

Interviewee AI2, an angel investor: "These institutes can't offer all the support and they don't have all the networks in place. So, they [institutes] should open up to the start-up ecosystem and attract the support for their students. There are a few active angel investors taking various initiatives ..."

9.9.2 Cognitive social capital for entrepreneurial support

Participating in mentor programs, incubators and accelerators allows students to repeat contact and dialogue with external stakeholders, building social capital through shared language. Higher education institutions are responsible for sustaining entrepreneurial support mechanisms and ensuring their purposes are effectively met. Stakeholders with mutual interest are willing to develop students' capabilities and support their entrepreneurial journey. This engagement between students and stakeholders may reciprocate through shared goals and purpose. Through entrepreneurial support, stakeholders share knowledge and understanding for personal development, passing traits and values to students as offspring within the HEEE.

Interviewee AI4, an angel investor: "So this mentor is entrepreneurial, and they can help to enhance these students' capacity. It should be an engagement approach building the internal and external capacities of the mentee, that is the student. It is essentially personal development sharing knowledge on personality traits, social cues, and communication tips. This is what will help them get out there with confidence."

Interviewee PS1, an external mentor and coach in entrepreneurship: "You see accelerator programs gives students expert advice. Whether it's on business pitching, lean models, managing teams, attracting resources and others. Students who get this knowledge and experience can improve their scalable start-ups, ideas and concepts to market-ready product and services with more successful start-ups. Experts involved in such programs come from their own areas and fields which makes them experts in that area."

9.9.3 Relational social capital for entrepreneurial support

Stakeholders significant for entrepreneurial support and shared understandings are reported in Table 9.7 below. Relational social capital relates to the nature and quality of relationships held among students and different stakeholders. Entrepreneurs and experts in entrepreneurship can play the role of mentors by developing soft skills among students, including their personality, attitudes and abilities such as working with others. Through platforms like accelerators, entrepreneurs and investors are also willing to share advice and improve students' practical knowledge on starting a new venture. In addition, private companies and not-for-profit organisations can support students as potential entrepreneurs by providing access to resources including financial funding or infrastructure support. The nature and quality of these relationships are expected to develop over time through continuous engagement and result in interpersonal trust and obligations between students and stakeholders, the HEEE and wider entrepreneurial ecosystem.

Interviewee AC5, an academic teaching a career planning unit: "During their education journey, students look for people they can relate to and learn from. Of course, we are there as lecturers. I have come across a couple of students who are interested in becoming entrepreneurs. So, they look for that person they can talk to, someone like a

mentor who can answer their questions, give them direction and learn everything entrepreneurship. Sometimes it is not the lecturers they look for. They have specific questions like how to become an entrepreneur, how it is like being an entrepreneur, how to improve themselves for it and things like that."

Interviewee AI5, an angel investor: "If I am given the opportunity, I would like to mentor students who are aspiring to become entrepreneurs or even advice students in an incubator. In my capacity and others in similar positions we have a lot of knowledge to share. Students creating their start-ups can really benefit input on their business idea, model, market fit and start-up to see if they need any improvements or how they can tackle any challenges. Of course, it is not limited to this, mentoring can also be about their self-development."

Interviewee EE5, an entrepreneur of a fashion start-up: "Students or young entrepreneurs lack all types of resources. I remember when I started out. And I think the best way around it is to collaborate and source those resources from your network and support system. The institutions, mentors and even outside organisations can bring together the resources you need for mutual benefit or sometime for no benefit at all."

HEEE factor & mechanisms	Structural: actors of the social system	Cognitive: system of meaning	Relational: obligations and expectations
Entrepreneurial support			
Mentor programs Incubator Accelerator	 Educators Entrepreneurs Angel networks Private companies Financial institutions Not-for-profit organisations 	 Shared goals and purpose Shared knowledge and understanding for personal development Shared language through expert advice and specialised knowledge 	 Mentor students on becoming successful entrepreneurs Advice students on their business ideas, models and start-ups Provide resources for creating start-ups

 Table 9.7 – Social capital dimensions for entrepreneurial support

9.10 Stakeholder engagement in HEEEs

9.10.1 Structural configuration of HEEEs

The structural configuration of HEEEs includes factors and actors, in this case as it relates to actors being the social network. HEEEs, similar to entrepreneurial ecosystems, are multi-

stakeholder environments that consist of many stakeholders (Feld 2012; Fetters, Greene & Rice 2010). Designing HEEEs with relevant stakeholders is important as developing student E&I capabilities can benefit from stakeholders beyond the institution and from the wider entrepreneurial ecosystem (Bischoff, Volkmann & Audretsch 2018). In this research, higher education institutions are challenged by weak ties, structural holes, limited connectivity and low network stability of members in HEEEs.

An HEEE is directly influenced by its social network and higher education institutions along with their students can benefit from developing bonds, bridges and linkages with relevant ecosystem actors. It is also important to consider the inclusion of stakeholders beyond the institution in the HEEE as they are the receiving end of talent; that is, potential student entrepreneurs. With respect to the structural configuration of HEEEs, network members internally include institutional management, educators, selected employees who are staff representatives, and students. Externally, the social network can involve other higher education institutions, entrepreneurs, investors, start-up support professionals, private companies, financial institutions, not-for-profit organisations and parents. Higher education institutions need to fill the structural holes in the social structure of an HEEE by including a combination of internal and external members in the social network.

Both stakeholder groups, internal and external, recognise that co-creating and evolving an HEEE in a resource-constrained environment depends on strong connections, aligned stakeholders and teamwork involving the collaboration of stakeholders from the HEEE and the wider entrepreneurial ecosystem. A dominant point is the importance of actors in the entrepreneurial ecosystem being involved in HEEE factors such as E&I education and vice versa. The nature of these connections may vary according to the roles and relationships of each member. Unlike HEEEs in developed countries (Lahikainen et al. 2019), it is apparent that HEEEs are not self-sustaining in a resource-constrained environment and need to bridge with their entrepreneurial ecosystem, increasing their resource repertoire. However, these various network members and their synergies can lead to better E&I capabilities for students and greater network stability in the HEEE.

Involving the wider entrepreneurial ecosystem through collaborations such as knowledge sharing has allowed the development of students' capabilities and the creation of new start-ups (Carayannis & Campbell 2009). Close interactions between students and educators, companies and institutions create a favourable environment, seeding students with

knowledge, skills and experiences. In this research, external stakeholders urged higher education institutions to work together with stakeholders in the wider entrepreneurial ecosystem to stimulate students E&I capabilities. The social network a student inherits during their higher education journey might have a pivotal impact on their access to knowledge that facilitates their capabilities to identify and exploit opportunities as potential entrepreneurs. Creating these relationships with other stakeholders results in building social capital and fostering better connectivity for students.

Embedding HEEEs with relevant stakeholders can address specific needs, including resource constraints. Social embeddedness enables access to latent resources and resources otherwise not available (Jack & Anderson 2002). Embeddedness is a configurating element that involves understanding the nature of the structure, enacting this structure with new ties and sustaining both the structure and links in order to bring together the environment (Whittington 1992). In principle, embeddedness is actors embedded in ongoing systems of social relations and networks (Granovetter 1985). For example, HEEEs lack resident entrepreneurs who are available onsite at higher education institutions in developed countries, and in such resource-constrained environments alumni entrepreneurs can fill in and engage with students. When it comes to providing seed funds for start-up projects, HEEEs can seek support from financial institutions to sponsor new business ideas when they lack funding for such initiatives.

9.10.2 Shared value within HEEEs

While the structural dimension involves ecosystem actors, ties and configuration, the cognitive dimension relates to shared value. Concerning cognitive social capital, HEEEs lack common goals and shared meaning with their wider entrepreneurial ecosystem for the sustainability of HEEEs. Stakeholders suggested a variety of cognitive social capital including shared goals and purpose, shared knowledge and understanding, shared values and culture, shared language and narratives, and shared identity through social interactions within HEEEs. Shared knowledge and understanding are the most common types of cognitive social capital when operationalising the six contextual HEEE factors. All these cognitive schemes can be understood as shared value and can be envisaged as part of the embeddedness discussed above in the structural configuration of HEEEs.

Cognitive social capital establishes a better foundation for HEEEs by communicating the shared knowledge of functioning among stakeholders. Multiple stakeholders with mutual interests must work in sync, collaborating, coordinating and co-existing to drive students E&I capabilities as a goal of HEEEs and the wider entrepreneurial ecosystem. Including external stakeholders and drawing on their shared value can make the HEEE and the wider entrepreneurial ecosystem stronger, resulting in individual and societal impacts.

With relevant stakeholders, internal and external, sharing their value in operationalising the HEEE factors that may lead to students developing their capabilities for E&I in a resourceconstrained environment, such shared value can be twofold. Students in a resource-constrained environment are drawn towards entrepreneurship by internal capabilities and external support, and not necessarily by the desire for entrepreneurship (Lin et al. 2013). The effectuation process explains that at the starting point, entrepreneurs ask themselves questions such as 'Who are we?', 'What do we know?' and 'Whom do we know?' considering the current situation and resources at hand (Sarasvathy & Dew 2005). First, interactions with diverse stakeholders and building social capital can benefit students through 'whom they know' to pursue entrepreneurship in the long term. Students may seek out investors, with whom they then interact and create a relationship to assess their business idea and evaluate a business model. In a resource-constrained environment, where educators may lack entrepreneurial experience, students turn to practitioners outside their institution. Further, by asking 'whom they know', entrepreneurs convince themselves of a common future (Wiltbank et al. 2006). To develop selfefficacy, students may seek verbal encouragement and emotional support from individuals who are entrepreneurs. Especially in a resource-constrained environment, students rely on whom they know for their entrepreneurial endeavours.

Second, capabilities are driven by knowledge and theory traditionally taught in higher education. However, a variety of stakeholders shape the flow of E&I capabilities obtained by students (Spigel 2017). Being exposed to stakeholders beyond the institution might generate better value for students, such as more practical and social learning that is otherwise limited in the institution. For instance, entrepreneurs can pass on their personalities and behaviours to students as potential entrepreneurs to be innovative, create value and cope with challenges through innovation. Further, exposure to the open environment will add to students' knowledge of tools and techniques for innovation. An entrepreneur must create, manage and assume the risk of a start-up, embracing the total innovative process (Cunningham & Lischeron 1991). Networking influences individuals and uses key people to achieve objectives (Santandreu-Mascarell, Garzon & Knorr 2013). In a resource-constrained environment, a well-connected HEEE can support creative thinking to develop students' E&I capabilities.

9.10.3 Relationships within HEEEs

Having established the structural configuration and shared value in HEEEs, this section relates to the relationships among stakeholders within HEEEEs. An HEEE with a focus on developing students' E&I capabilities can benefit from strong connections, shared goals and a high level of trust. The feelings of trust and respect that need to be shared by diverse stakeholders became prominent regarding exchanging resources within the HEEE. When working together, particularly across different yet connected contexts, higher education institutions need to pay attention to what and how stakeholders think and feel. Stakeholders within HEEEs need to interact with a variety of relevant stakeholders, strengthening existing relationships and forming new relationships. The quality and nature of relationships can be understood in the embeddedness discussed above regarding structural and cognitive social capital.

The co-creation, success and sustainability of HEEEs are deeply connected with the effectiveness of internal and external stakeholder relationships. In addition to economic development, higher education institutions are considered a driving force for community development (Reggiani 2017). Higher education institutions are a significant domain of the wider entrepreneurial ecosystem for their role in developing entrepreneurial talent (Isenberg 2010). Likewise, HEEEs affect their immediate environment and can build entrepreneurial communities. Entrepreneurial communities focus on developing entrepreneurs and take a systems approach to community development (Lichtenstein, Lyons & Kutzhanova 2004). Such communities include stakeholders who are actively engaged in developing students for entrepreneurship through know-how, support and resources.

Higher education institutions in resource-constrained environments can alleviate deficiencies in social capital through community engagement (Vohara et al. 2004). Events, projects and competitions taking place through the HEEE can act to bring cross-community groups together, including external stakeholders from the wider entrepreneurial ecosystem. This engagement can build connections with trust and respect, leveraging students through the HEEE and building connectivity that might not exist otherwise. Programs within HEEEs that deliberately leverage such opportunities offer the ability to access social capital in other localities (Bedő, Erdős & Pittaway 2020).

9.11 Chapter summary

An analysis of data and presentation of findings addressing the third research question on how diverse stakeholders can engage within the six contextual factors of HEEEs that could influence

students' E&I capabilities in a resource-constrained environment is included in this chapter. A three-level content analysis focusing on the data leading to theoretical dimensions was conducted at the beginning of this study. While stakeholder collaboration in entrepreneurship education has been investigated, stakeholder engagement within HEEEs is understudied. This study sheds light on social networks including 12 key stakeholders, contributing to the composition and configuration of HEEEs. Creating social capital considering its structural, cognitive and relational dimensions is revealed when operationalising HEEEs. Social capital increases the likelihood of becoming a potential entrepreneur, progressing through the start-up process, and avoiding start-up failure. This section brings together factors and actors in HEEEs with social capital that has the potential to strengthen the HEEE. In doing so, this study contributes to social capital dimensions by contextualising theory to HEEEs in a resource-constrained environment. The chapter ends with a section on the structural configuration, shared value and relationships for stakeholder engagement in HEEEs.

CHAPTER 10: DISCUSSION AND CONCLUSION

10.1 Chapter overview

This final chapter provides a discussion of the findings associated with the research questions, highlights the contributions to theory and implications for practice, and concludes with limitations of the study and future research directions (refer to Figure 10.1). This chapter demonstrates 'what' was known in the literature prior to this study and 'how' and 'why' thinking should be different because of this research. That difference constitutes the value added to the literature contributed by this research. In this case, the main objective of this chapter is to discuss key and new findings.



Figure 10.1 – Chapter 10 outline

10.2 Findings for each research question

Findings from the study reveal that higher education institutions in a developing economy, their HEEEs and their challenges and opportunities are different compared to developed economies. Therefore, context matters (Langowitz & Minniti 2007; Boettke & Coyne 2009; Autio et al. 2014) even in HEEEs and in this case, a resource-constrained environment, namely Sri Lanka as an emerging economy. This section discusses the findings relating to the research questions, outlining their relationships to theory and the HEEE literature, and explains the results.

10.2.1 Beyond student start-ups, entrepreneurial mindset and intention

The first research question inquired how diverse stakeholders anticipate the continued development of HEEEs in a resource-constrained environment. The objective was to understand the perceptions of internal and external stakeholders about the future of HEEEs in a resource-constrained environment through their lived experiences. Three overarching findings emerged from the qualitative analysis of interviews with deans/heads of school, academics, alumni entrepreneurs, expert entrepreneurs, angel investors and support professionals. These findings illustrate the relationship between HEEEs and students' E&I capabilities in a resource-constrained environment, through a view of the continued development.

Finding 1: For the continued development of HEEEs, higher education institutions need to focus on capability development, particularly E&I capabilities of students (see 6.3).

Finding 2: An HEEE is a sub-ecosystem of the wider entrepreneurial ecosystem affecting the overall performance of E&I (see 6.4.2).

Finding 3: There are different roles that HEEEs can play, including influential, developmental, networking, entrepreneurial and regional, to serve their purpose to students and ecosystem actors in the context of resource constrained environments (see 6.4.1).

Current literature considers the student somewhat at the heart of the ecosystem and discusses HEEEs that lead to start-ups, foster entrepreneurial mindset and encourage entrepreneurial intention (Longva 2021; Webber, Kitagawa & Plumridge 2020). The emergence of HEEEs has focused on start-ups for the most part (Longva 2021; Meyer et al. 2017; Wright, Siegel & Mustar 2017; Miller & Acs 2017; Rice, Fetters & Greene 2014). More recently, studies began exploring HEEEs at the individual level including entrepreneurial mindset and intention (Webber et al. 2020; Guerrero et al. 2020; Secundo et al. 2020). With this limited view of HEEEs, this study recognises the continued development of HEEEs to develop students' E&I capabilities. The concept of E&I capabilities represents skills, expertise, acumen and knowledge, terms interchangeably used in entrepreneurship literature (Liu, Kulturel-Konak & Konak 2021). E&I capabilities involve the ability of an entrepreneur to start and grow a new venture using a combination of resources (Gumsay & Bohne 2018). The importance of E&I capabilities is based on the rationale that youth with entrepreneurial aspirations require pre-venture capabilities to sustain start-ups. A recent report found that 'lack of skills and expertise' is a top reason to why start-ups fail in Sri Lanka
(PricewaterhouseCoppers 2020). Alumni entrepreneurs revealed 'skills and expertise' as the most important facilitating factor of successful start-ups. Angel investors and support professionals pointed out that 'lack of market research' and 'insufficient business acumen' are key issues that drive start-ups to fail and that these are capabilities that should be primarily developed during the education journey. Unlike prior HEEE studies that primarily focused on students' start-ups, entrepreneurial mindset and intention, this study advances the HEEE literature by establishing the importance of developing students' E&I capabilities. In this case, higher education institutions support students' development, and the success of this transformation would be profound for the wider entrepreneurial ecosystem (Matriz et al. 2020; Birch et al. 2017).

The main theme explored in entrepreneurship education is reinforced in this study, the contention that capability development matters (Longva 2021; Webber, Kitagawa & Plumridge 2020; Guerrero, Urbano & Gajón 2020; Meyer et al. 2017; Wright, Siegel & Mustar 2017; Miller & Acs 2017; Rice, Fetters & Greene 2014). Higher education institutions must produce entrepreneurial capital and be catalysts for economic and societal development (Guerrero, Cunningham & Urbano 2015). E&I capabilities refer to the collection of knowledge, skills and attitude that contributes to one's entrepreneurship-specific human capital profile (Ucbasaran, Westhead & Wright 2007). However, HEEEs should focus on achieving a series of entrepreneurial outcomes rather than focusing on a single outcome - whether it is E&I capabilities from this study or start-ups, entrepreneurial mindset or intention in previous studies. This study contributes to the missing link identified when mapping current HEEE studies against the start-up process for students by Duruflé, Hellmann and Wilson (2018). This means that in the future higher education institutions can rethink the co-creation of their HEEEs and focus on a sequence of entrepreneurial outcomes. Ideally, HEEEs when co-creating and evolving their HEEEs can align to the three stages proposed by Duruflé, Hellmann and Wilson (2018) where there are activities focused at (1) creating entrepreneurial mindset and intentions; (2) developing entrepreneurial knowledge and skills (capabilities); and (3) building start-ups.

An entrepreneur's capabilities are known as inputs presumed to be related to outputs such as start-ups (Ucbasaran, Westhead & Wright 2008). This understanding comes from the systems perspective found in entrepreneurial ecosystems research (Stam 2015; Feld 2012; Isenberg 2011; Cohen 2006; Neck et al. 2004). This trend of HEEE literature focused primarily on student start-ups raises concerns about the primary purpose and tradition of higher education institutions that produce students for future work. In this case, higher education institutions

need to focus on developing the E&I capabilities of students, the potential entrepreneurs of the future, who may become successful entrepreneurs and transit into the economy and society. In moving forward, higher education institutions should focus on capability development through their HEEEs, taking the responsibility of a higher education institution. The role of entrepreneurship education and experiences extends beyond creating start-ups and into developing employability skills (Ustav & Venesaar 2018). Students with E&I capabilities and the desire to become an entrepreneur may take the ideal path of recognising an innovative business idea and creating a start-up. Other students may choose to work in an existing start-up or even work in organisations as an intrapreneur, that is, an entrepreneurial employee, creating new innovations such as business models, brands, products/services and even operational practices. Given that almost every industry has embraced entrepreneurship and innovation to varying degrees, students with E&I capabilities become relevant to all organisations.

Every HEEE delivers entrepreneurship education and supports new venture creation (Rice, Fetters & Greene 2014). However, such ecosystems not only cater to students aspiring to become start-up founders and those interested in learning about entrepreneurship but can also support students who are keen on intrapreneurship and want to include an entrepreneurial role by developing innovative ideas in a business. In developing countries, like Sri Lanka, where higher education institutions are primarily focused on nurturing jobseekers and catering to the job market, these institutions can rethink their ecosystem to promote employability and entrepreneurship.

A recent study involving 18,000 participants in 15 countries revealed that entrepreneurship is a future citizen skill (McKinsey & Company 2021). Higher education institutions are expected to nurture students into graduates who are capable of coping in an uncertain world as the responsibility in human capital (Barnett 2018). The catastrophic effects of the COVID-19 pandemic created a unique opportunity for entrepreneurship education along with challenges to higher education institutions (Liguori & Winkler 2020). The pandemic demonstrated the need for higher education institutions to develop students with E&I capabilities, now more than ever, that will enable them to strive in an uncertain world with disruptions in economic and social environments. HEEEs have the ability to develop the 'means' of students, which Sarasvathy (2022) recognised as 'Who I am' (identity), 'What I know' (knowledge) and 'Whom I know' (network) and that lead decisions and actions that potential entrepreneurs are involved in. Although HEEE literature has not made explicit reference to effectuation theory, existing studies demonstrate that scholars have explored students more about what they know (Longva 2021; Webber, Kitagawa & Plumridge 2020; Guerrero, Urbano & Gajón 2020; Meyer et al. 2017; Wright, Siegel & Mustar 2017; Miller & Acs 2017). In this study, it is clear that HEEEs should provide the very input for the entrepreneurial process. HEEEs can not only deliver entrepreneurship education and support new venture creations (Rice, Fetters & Greene 2014), but HEEEs should nurture 'talent' in the form of potential innovators and entrepreneurs with relevant E&I capabilities for the entrepreneurial ecosystem to harness.

Within less than a decade to achieve the United Nation SDGs (UNICEF 2022), higher education institutions must acknowledge their role in contributing to sustainable development through 'quality education' in their economy. In contrast, a majority of HEEE studies emphasise the role of higher education institutions in contributing to either economic growth or economic growth and social transformation. For instance, economic growth in Norway (Longva 2021) and the US (Miller & Acs 2017), and economic growth and social transformation in the UK (Webber, Kitagawa & Plumridge 2020). There is relatively little HEEE research that has focused on SDG4 and quality education, which aims to substantially increase the number of youths with relevant skills for entrepreneurship. In resource-constrained environments, like Sri Lanka, higher education institutions and their HEEEs along with their national priorities and global movements.

Based on a research call by Webber, Kitagawa and Plumridge (2020), this study took a holistic approach of exploring HEEEs in their broader environment, being consistent with the interconnected nature of ecosystems. Prior studies have explored HEEEs and their composition by considering them as a standalone ecosystem (Lahikainen et al. 2019) and as a result isolating these ecosystems of higher education institutions from the bigger picture to some extent. A recent literature review on entrepreneurial ecosystems suggested that research should examine the complex system nature of ecosystems (Wurth, Stam & Spigel 2021). This study is consistent with this thinking and extends the co-creation of HEEEs by positioning them as a sub-ecosystem of entrepreneurial ecosystems and promoting their system nature. HEEEs can function more effectively when open to their wider entrepreneurial ecosystems and

collaborating with ecosystem actors. This study differs by showing that HEEEs depend on their respective entrepreneurial ecosystem and vice versa for effective functioning, particularly in a resource-constrained environment.

When operating together as one system, the entrepreneurial ecosystem can support HEEEs to adapt to new conditions and changes as entrepreneurship is dynamic and promotes resourcefulness. Further, where there are scarce resources, HEEEs can draw resources in the form of personnel, funding, infrastructure and services in support. The connection with the entrepreneurial ecosystem makes the HEEE sustainable (Theodoraki, Messeghem & Rice 2018) and allows both ecosystems to contribute to one another. Higher education institutions need to take a more integrated approach to co-creating their HEEEs by understanding the connection needed with their wider entrepreneurial ecosystem and that HEEEs are stronger together with their entrepreneurial ecosystem.

Findings reveal that HEEEs in the private sector hold minimal linkages with ecosystem actors from the wider entrepreneurial ecosystem, which are mostly one-time or ad hoc. These higher education institutions can learn from public universities that have co-created their HEEEs with stronger relationships with the entrepreneurial ecosystem. One example is the University Business Linkage by the University of Colombo, an effort to guide students to establish new ventures including start-ups with the support of collaborations and partnerships with industry and beyond (University of Colombo 2016). Another instance is the Center for Entrepreneurship and Innovation powered by the University of Sri Jayewardenapura with the purpose of promoting the E&I of the nation together with the broader community (University of Sri Jayewardenapura 2021). The Center for Entrepreneurship and Innovation actively operates with ecosystem actors to offer students entrepreneurship education and entrepreneurial support including training, information, incubation, access to networks and more (University of Sri Jayewardenapura 2021). Initiatives such as the University Business Linkage and the Center for Entrepreneurship and Innovation demonstrate formalised and stable relationships between public universities and the wider entrepreneurial ecosystem working together towards E&I among students. Private higher education institutions can mirror such initiatives for their HEEEs and make bonds with key ecosystem actors in a more regular and formal way.

Further, HEEEs should move beyond engaging students within the HEEE and expose them to the entrepreneurial ecosystem. Exposing students to the real world of E&I will help them grow and adapt. Involving the wider entrepreneurial ecosystem and engaging students with diverse ecosystem actors at different levels such as industries, governments and societies enables the development of E&I capabilities (Carayannis & Campbell 2009). The close interaction between students and ecosystem actors creates a favourable learning environment that enables students to recognise problems, identify opportunities and generate ideas through deeper knowledge and experiences while easing the challenges they face in resourceconstrained environments. Further, students can draw inspiration, motivation, practical knowledge and even behaviours from ecosystem actors that they may lack within the HEEE in developing E&I capabilities. Such exploration and involvement will prepare students and potential entrepreneurs to smoothly enter the wider entrepreneurial ecosystem with some knowledge and networks.

Higher education institutions in more resource-constrained environments tend to pay less attention to external influences when co-creating their HEEEs and place greater emphasis on the internal environment, particularly strategies priorities. This study establishes the importance of HEEEs being open to the broader environment to embrace external influences as opposed to being insular and having boundaries. Such awareness and understanding of external influences would bridge the disengagement between HEEEs and the wider entrepreneurial ecosystem when addressing the needs of key stakeholders such as parents who have long term influence on their children (Gallage, Laferriere & Selvarajah 2022). Entrepreneurship literature asserts that parental influence is the most important factor that influences a child's development and choice of career, and entrepreneurial parents increase the likelihood of children starting an entrepreneurial career (Lindquist et al. 2015). Particularly in a resource-constrained environment, higher education institutions need to sense external influences, take targeted actions within their HEEEs and effectively use their resources to satisfy key stakeholders.

As a sub-ecosystem of the wider entrepreneurial ecosystem in which it is embedded, HEEEs should meet the social and economic needs of the ecosystem actors involved. It is important for higher education institutions to acknowledge and pay attention to other needs of academics, alumni entrepreneurs, parents and other ecosystem actors. Academics should be offered support by educating them and offering them avenues for development, as some lack experience in entrepreneurship and may not hold entrepreneurship-specialised education. Extending from the view that HEEEs are sub-ecosystems of respective entrepreneurial ecosystems, this study demonstrates the interdependency between HEEEs and entrepreneurial ecosystems for E&I, and thereby contributes to the HEEE literature on the role that these ecosystems play in developing students' E&I capabilities. Higher education institutions represent a key domain of the entrepreneurial ecosystem that conducts teaching, research and entrepreneurship towards economic and social development (Isenberg 2010; Kirby, Guerrero & Urbano 2011; Guerrero, Urbano & Cunningham 2014). HEEEs are natural incubators for potential entrepreneurs are a significant input for the entrepreneurial ecosystem. These institutions and their HEEEs must understand their role extend beyond preparing students for E&I, and includes entrepreneurial mindset and intention, such as in recent HEEE studies including Webber, Kitagawa and Plumridge (2020), Guerrero et al. (2020) and Secundo et al. (2020).

The role of higher education institutions has been focused on teaching and research. Higher education institutions and their HEEEs are dealing with various challenges limiting the development into a fully-fledged ecosystem. To strengthen as an entrepreneurial institution and evolve their HEEE, different roles can be played serving the purpose to students and ecosystem actors. These roles extended the third mission of contribution to the society, involving influential, developmental, networking; entrepreneurial; and regional. By label, these roles sound typical however the contribution can be found in the uniqueness related to the resourceconstrained context encompassed in these roles.

As higher education institutions in resource-constrained environment are principally focused on nurturing corporate-fit graduates, influencing students for entrepreneurship while developing their E&I capabilities is the much-needed transformation. Next, the role of networking was suggested for not only sharing resources but for social learning and community belonging that benefit students. To evolve as a HEEE, higher education institutions must assist their students to create start-ups through entrepreneurial support including access to incubators and accelerator programs. Fundamental to these roles are HEEEs being open to their regional milieu and national priorities and accepting the responsibility of leading human capital development. In this case, it is students with E&I capabilities who have the knowledge and skills related to the startup process, judgmental abilities associated with entrepreneurial action and social attitudes for networking (Alsos et al. 2022).

Higher education institutions and their HEEEs suffer from challenges that stem from the resource-constrained environment. The scarcity of resources for these HEEEs include two main scarcities: financial resources and human resources. By addressing these scarcities, HEEEs can effectively function developing students E&I capabilities. If higher education institutions orchestrate HEEEs by playing these five roles, they can combine existing and new resources to develop students' E&I capabilities. Creating this resource-rich environment is a key endeavor for higher education institutions and should engage with the wider entrepreneurial ecosystems that facilitates the five roles.

10.2.2 Beyond the common HEEE factors

The first part of the second research question investigated what diverse stakeholders perceive as the contextual factors of HEEEs that could influence students' E&I capabilities in a resource-constrained environment and three overarching findings surfaced. These findings contribute to illustrating the relationship between HEEEs and students' E&I capabilities in a resource-constrained environment. This study offers context-specific qualitative research concerning HEEE factors of higher education institutions in Sri Lanka, extending from the results of empirical studies in developed economies. The emergence of HEEEs in resourceconstrained environments is highlighted in a literature-based study by Bedő, Erdős and Pittaway (2020) that suggested a conceptual framework towards student start-ups using Stam's (2015) entrepreneurial model. With little empirical evidence on HEEEs in a resourceconstrained environment, this study is a study to explore the composition of HEEEs for developing students' E&I capabilities in developed countries.

Finding 4: Six contextual factors – entrepreneurial orientation, E&I education, E&I research, enterprising experiences, entrepreneurial networks and entrepreneurial support – can influence students' E&I in a resource-constrained environment (see 7.3).

Finding 5: Contextual factors are different to those commonly accepted in developed economies and are specific to institutional nature and geographic context (see 7.3 and 7.4).

Finding 6: Students involved in activities of HEEEs as a learning environment translates into students' E&I capabilities (see 7.5).

Despite how the co-creation of HEEEs begins, internal and external forces facilitate their development process (Rice, Fetters & Greene 2014). The lack of existing HEEE studies involving the combined efforts of internal and external factors led this study to explore a more

holistic HEEE that took into account factors within the institution and outside in the broader environment. One study by Webber, Kitagawa and Plumridge (2020) explored entrepreneurship education, extracurricular activities and enterprise experiences while another study by Guerrero, Urbano and Gajón (2020) focused on educational programs, business incubators and other infrastructures. A recent study investigated a combination of internal and external factors extending curricular activities, co-curricular activities, enterprise experience, infrastructure, industry, incubators and public support systems (Longva 2021). The findings from this study illustrates a set of six contextual factors along with 18 operational mechanisms that may have a positive influence on developing students' E&I capabilities enabling higher education institutions in resource-constrained environments to operate their HEEEs more effectively. Existing HEEE studies have not explicitly identified the six contextual factors found in this study, although some similarities such as entrepreneurship education exist (Longva 2021; Meyer et al. 2020; Webber, Kitagawa & Plumridge 2020; Wright, Siegel & Mustar 2017; Miller & Acs 2017; Rice, Fetters & Greene 2014). The commonly identified 'entrepreneurship education' factor among current HEEE studies is enhanced as 'entrepreneurship and innovation education' in this study, acknowledging the difference that innovation makes in entrepreneurship. Business models and product ideas can be new and good but this is not the same as innovative ideas that have something more. Such businesses and products with that 'wow' factor have the difference of 'innovation'. Education in E&I would ensure that students learn how to use innovative techniques to create value in a start-up while developing E&I capabilities.

Entrepreneurship education and entrepreneurial support are both a substantial opportunity for higher education institutions and a significant responsibility. While student start-ups are most prominent among HEEEs in developed economies in resource rich environments (Longva 2021; Meyer et al. 2017; Wright, Siegel & Mustar 2017; Miller & Acs 2017; Rice, Fetters & Greene 2014), HEEEs in resource-constrained environments need to work actively in developing students' E&I capabilities. The challenge is how these higher education institutions identify the contextual factors within their HEEE composition to drive E&I (Cunningham, Lehmann & Menter 2021). An early study found that relative strengths of HEEE factors may vary, however all higher education institutions share common factors (Rice, Fetters & Greene 2014). Through this study it became evident that HEEE factors may be similar in HEEEs in developed and developing countries, there are other HEEE factors that are

unique to the institution and the context. There is uniqueness among HEEE factors because of the nature of the institution and circumstances of the context. In an emerging economy, most higher education institutions in Sri Lanka do offer an entrepreneurship program or have resources for a separate entrepreneurship department/division or chair or offer access to start-up funding, links to angel investors or incubators for students. This understanding from this study contributes to the clarification of HEEE factors as highlighted by Hsieh and Kelley (2020) and the emerging knowledge body of HEEEs in resource-constrained environments explored by Bedő, Erdős and Pittaway (2020).

Higher education institutions and their HEEEs in resource-constrained environments depend on support from individuals and organisations in the entrepreneurial ecosystem to operationalise some HEEE factors. In this case, institutions should shift from traditional ways of working, limited to the institution, into more inter-institutional and collaborative partnerships to build sustainable HEEEs. In developed economies, there is evidence of how some HEEEs collaborate with various ecosystem actors including alumni entrepreneurs and industry for better access to resources and services (Fetters, Greene & Rice 2010; Hancock 2011). Similarly, the University of Peradeniya (2021) has liaised with the British Council to promote social entrepreneurship among all students and support potential social entrepreneurs.

Further, some of the top public universities including the University of Colombo, University of Peradeniya, University of Sri Jayewardenepura, University of Kelaniya, University of Moratuwa and University of Ruhuna were found to be collaborating to some extent with one another in contribution to the National Innovation System (Weerasinghe, Jayewardene & Deshani 2016). By contrast, existing HEEE studies have not found higher education institutions working with similar institutions in joint efforts. Developing countries confront the challenges of their resource-constrained environment in their own ways, differently to developed economies. HEEEs in resource-constrained environments can develop stronger collaboration with other higher education institutions to improving the ability to overcome challenges and share resources for functioning effectively. In this respect, this study brings an understanding of how HEEE factors can work differently in some environments and not others. This study calls for a closer integration of an HEEE in the respective entrepreneurial ecosystem to work with more stakeholders including other higher education institutions. Therefore, having boundaries or not sharing resources restricts the success and survival of HEEEs. In this sense, HEEEs as an 'open-system' orchestration would work better in resourceconstrained environments, as opposed to the common closed system.

Entrepreneurship literature has established the system nature of ecosystems (Stam 2015; Feld 2012; Isenberg 2011; Cohen 2006; Neck et al. 2004). According to the open system developed by Ludwig von Bertanlanffy (1956), parts are independent and the interactions between parts become more complex. This means that each factor within the HEEE needs to function collaboratively with the other factors; there might not be a direct impact or interdependence, but everything is connected to the value proposition. A student may devote psychological and physical energy to studying, taking part in activities, attending events and interacting with peers, academics and other ecosystem actors during their higher education journey. What this student learnt in the classroom (E&I education) may be applied during an internship with a start-up or while participating in an entrepreneurial networks), the student may have been inspired about a related topic or issue that they discuss with their expert mentor (entrepreneurial support) for better clarity. Through the HEEE, students will have access to various activities and may participate continuously during their journey; participation in one activity may have an impact in another, translating to their capability development.

Entrepreneurial ecosystems are formed by various domains or organisations working together and creating value for the economy and society. HEEEs are largely known to support entrepreneurship development through a variety of related activities in the context of a specific higher education institution (Fetters, Greene & Rice 2010; Belitski 2019). In this study, the HEEE is viewed as the 'learning environment' that affects the evolution of students, which can also be understood as students' entrepreneurial development. In this sense, students adapt to the environment; embrace new dynamics; build relationships; and develop their mindset, intention and capabilities with the support of the HEEEs and their ecosystem actors. Thus, it is important to understand the co-evolution between students, the higher education institution and ecosystem actors. This means the influence of growth and evolution of each other through shared relationships. On the one hand, students develop their E&I capabilities through close interactions with the higher education institution and ecosystem actors. On the other hand, higher education institutions can co-create and evolve their HEEEs through reciprocal relationships between students and ecosystem actors. Findings point out that evolution between students, the higher education institution and ecosystem actors is a result of working together within the HEEE. In addition to sharing the evolution of each other, there is also the transition from one ecosystem to another. Initially, students are part of the HEEE as their learning

environment and then they may progress as graduates to become entrepreneurs or intrapreneurs in the wider entrepreneurial ecosystem.

Diverse stakeholders perceived six contextual factors of HEEEs that could influence students' E&I capabilities in a resource-constrained environment: entrepreneurial orientation, E&I education, E&I research, enterprising experiences, entrepreneurial networks and entrepreneurial support. These contextual factors are similar but different contextual factors are specific to institutional nature and geographic context as the composition of HEEEs. Students involved in various activities of HEEEs as a learning environment translates into their capability development. When illustrating the relationship between HEEEs and students' E&I capabilities in a resource-constrained environment, contextual factors go beyond the commonly accepted factors.

10.2.3 Beyond factors of higher education entrepreneurial ecosystems

The final part of the second research question explored how specific contextual factors of HEEEs could influence students' E&I capabilities in a resource-constrained environment. From this exploration, three key findings emerged. Existing research has presented exclusive lists of factors that represent the composition of HEEEs for respective higher education institutions in developed economies. As suggested by Longva (2021), future studies need to go beyond identifying HEEE factors, and this study takes a step further by offering insights on the 'how' aspect, demonstrating ways in which such contextual factors can be implemented within the HEEE from the perspective of key internal and external stakeholders. With little research on how higher education institutions can operationalise HEEE factors, this study extends the literature by adding knowledge on mechanisms and extending the approach to be taken when configuring HEEEs.

Finding 7: Mechanisms that emerged in this study are practical solutions addressing problems in implementing contextual factors of HEEEs in a resource-constrained environment.

Finding 8: The approach to operationalising contextual factors needs to put 'people' at the centre of the process empathising the needs of ecosystem actors.

Finding 9: Higher education institutions do not see the whole-picture perspective of their HEEE as a value chain.

Some efforts have started to emerge on how a higher education institution and its HEEE can operationalise towards achieving desired outcomes, either student start-ups, entrepreneurial mindsets or intentions. One such conceptual study suggested overarching ways such as knowledge creation and spill over; human capital creation and acquisition; social capital connectivity; funding intermediation; economic diversity and fluidity; enabling infrastructure; incubation and catalytic change; and cultural change for operationalising an HEEE towards student start-ups (Bedő, Erdős & Pittaway 2020). While these emerging mechanisms are institution focused and broad, this empirical study explored specific mechanisms that may work within each contextual factor, addressing any challenges in a resource-constrained environment.

When examining the influence of entrepreneurial ecosystems on the entrepreneurial process, it was found that the context, in this case the HEEE, can be a driver as well as a barrier (Guerrero et al. 2020). Higher education institutions in developing countries may have hindering factors that differ or do not exist for HEEEs in developed economies. These challenges are context based and could be the opposite among higher education institutions in developed economies. While management drive, institutional strategies and focus on preparing future entrepreneurs were facilitating attributes of a Finnish higher education institution (Lahikainen et al. 2019) and the National University of Singapore (Po, Singh & Wong 2010), these tend to be more hindering factors of a typical higher education institution in Sri Lanka. Findings reveal that higher education institutions face a lack of institutional direction where institutional priorities do not align with national priorities such as entrepreneurship education and future youth with entrepreneurial capabilities. Extending from this, most higher education institutions have weak relationships with the entrepreneurial ecosystem, which results in failing to understand the role it plays in the shared value proposition. Next, few higher education institutions have strategic objectives oriented towards being entrepreneurial; however, the operations taking place do not work in their favour. These institutions tend to limit resources for operationalising entrepreneurial initiatives of HEEEs and, for instance, this leads to poor entrepreneurial exposure for students. Also, the dyadic relationships between higher education institutions and multinational companies restrict the institutions from understanding the potential of start-ups in benefiting their HEEEs, building networks and collaborating to build students' E&I capabilities. Last, higher education institutions make the criticism that only a small number of students are keen on an entrepreneurship major, and it seems that there almost no attention on the view of commercialising through HEEEs.

Previous studies demonstrate HEEEs as an entrepreneurial initiative well supported by the senior management of higher education institutions and pay less attention to examining contextual factors of senior leadership. One of the early studies proposed senior leadership vision, engagement and sponsorship as key success factors, having found that the senior leadership commitment sustains the HEEE as a robust ecosystem (Rice, Fetters & Greene 2014). When taking on the third mission of teaching, research and entrepreneurship, the leadership team of the National University of Singapore committed to a vision of 'global knowledge enterprise' and reoriented the core function of education in preparing students through an emerging HEEE to improve entrepreneurship education and stimulate student start-ups (Ho, Singh & Wong 2010).

Contrasting to existing HEEE studies based on developed economies, empirical findings reveal the lack of commitment and enthusiasm from senior management towards their HEEEs. Higher education institutions in the Sri Lankan private sector do not seem to be convinced by entrepreneurship just yet, according to their current programs, initiatives and actions. Although there is an increasing wave of entrepreneurship among public universities and in the broader society, where even media companies are promoting entrepreneurial reality shows such as "Startup 2021" (TV Derana 2022), senior management of private institutions are found to pay less priority to entrepreneurship education and investing in entrepreneurial initiatives such as HEEEs. Senior management seems to fail to understand the changing career preference towards entrepreneurship held by students, the preference of entrepreneur parents as well as the importance of enhancing entrepreneurial capabilities in students to support their future, long-term career potential, and they overlook students who may want to work in the start-up industry. It appears that higher education institutions are not strongly committed to their HEEEs due to the lack of understanding held about the dynamics in their broader environment. Senior leadership was suggested as a key success factor for HEEEs in early literature (Rice, Fetters & Greene 2014) and in this study, senior leadership is framed as a way to operationalise (i.e. a mechanism) the contextual factor of 'entrepreneurial orientation'. Unlike HEEE studies in developed economies, this study demonstrates the importance of the internal drive from senior management as a mechanism in co-creating HEEEs for higher education institutions in a resource-constrained environment.

The argument on whether entrepreneurship can be taught is becoming obsolete given the increasing number of entrepreneurship degrees, specialisations and courses along with the success of entrepreneurial graduates and growing demand for entrepreneurship educators around the world (Hagg & Kurczewska 2021). Entrepreneurship education is very much in vogue across developed economies where many higher education institutions offer courses and classes in entrepreneurship to the broader student population. Almost every HEEE study in the literature found programs in entrepreneurship or entrepreneurship and innovation offered for undergraduates (Fetters, Greene & Rice 2010; Miller & Acs 2017; Meyer et al. 2020; Guerrero, Urbano & Gajón 2020; Webber, Kitigawa & Plumridge 2020; Longva 2021). Some public universities in Sri Lanka, such as the University of Sri Jayewardenepura (2021) which offers a Bachelor of Science in Entrepreneurship, are unlike private higher education institutions in Sri Lanka that fall short in offering entrepreneurship education programs and influencing students' entrepreneurial careers. Similar to Hartmann (2021), these higher education institutions perceive that there are more pains than gains associated with offering entrepreneurship education, especially related to costs. With this perspective, students are exposed in the direction of stable careers as employees, limiting the development of E&I capabilities which are applicable to a corporate career, self-employment and personal identity development in multiple walks of life. Given that E&I education is a key factor of HEEEs, students should have access to entrepreneurship courses and classes that may influence some of them towards an entrepreneurial career and others to work in start-ups.

Incubators (Longva 2021; Webber, Kitagawa & Plumridge 2020; Guerrero, Urbano & Gajón 2020; Meyer et al. 2017; Wright, Siegel & Mustar 2017) and accelerators (Wright, Siegel & Mustar 2017; Miller & Acs 2017) are common factors for entrepreneurial support in building the composition of HEEEs. HEEEs in developed economies such as the US offer entrepreneurial support to students through start-up funds, in-house incubators and accelerator programs (Rice, Fetters & Greene 2014). Findings from this study highlighted that internal stakeholders see no pathway for the creation of incubators and accelerators in private sector HEEEs, whereas external stakeholders believed that entrepreneurial support through incubators and accelerators is essential but was limited. Particularly in Sri Lanka, financing incubators and accelerators is costly for higher education institutions in the private sector and as a result this limits the entrepreneurial support available for students. In a resource-constrained environment, these factors need to be achieved differently from how they are implemented by higher education institutions in a developed economy.

Mechanisms for providing the benefits from incubators and accelerators appear possible through support and collaboration with ecosystem actors. Higher education institutions can collaborate with other private institutions or public universities to share incubators or accelerators, leading to higher student participation. The Sri Lanka Institute of Information Technology, a leading private higher education institution is actively managing a technology-based incubator that provides a variety of resources and services such as business acumen, funds, access to networks and communication facilities (SLIIT 2020). The University of Peradeniya (2021) launched a Social Enterprise Incubation Laboratory to offer students mentoring, training and access to relevant agencies for a social entrepreneurship start-up. These are possibilities for students from higher education institutions to access such incubators of other private institutions or public universities. Furthermore, support from the public system can offer students access to facilities such as hosting incubator and accelerator programs for HEEEs, regional incubators or national accelerator programs. In contrast to owning, managing and running such facilities and programs in more developed economies, higher education institutions in resource-constrained environments can operationally address challenges by sharing infrastructure, resources and services available in the broader entrepreneurial ecosystem and environment. Findings in this study reveal that entrepreneurial support can take forms such as mentoring with experts, and access to shared incubators and private accelerator programs in resource-constrained environments.

Entrepreneurship, including start-ups, are recognised as a driver for economic growth and social transformation (Audretsch et al. 2021; Webber, Kitagawa & Plumridge 2020; Klofsten et al. 2019). Given that entrepreneurship is considered a societal phenomenon (Davidsson 2003), looking beyond unicorns (ventures valued at USD1 billion), decacorns (ventures valued at USD10 billion) and foreign direct investments as benefits at the national level, a meaningful purpose of entrepreneurship in an emerging economy is financial independence. The Sri Lanka Startup Report 2019 showed that financial motivation is the second top reason for embarking on a start-up (PricewaterhouseCoopers 2019). Financial independence can be achieved through entrepreneurship education (O'Connor 2013) in the case that youth create their own start-ups as assets that generate income to pay for living expenses without depending on a job or others and equity that can grow wealth. This means that HEEEs can respond to economic and social needs by developing students' E&I capabilities and as a result contributing towards financial independence of youth in an emerging economy. In a resource-constrained environment, developing students with E&I capabilities can support youth to progress beyond living paycheck to paycheck and improve their financial independence, through which they can improve their quality of life.

Entrepreneurship involves entrepreneurial individuals forming or transforming organisations that create value including financial or social benefits (Gartner 1990). Value is understood as a promise of tangible and intangible benefits that a business offers for its customers (Lanning & Michaels 1988). For higher education institutions, a core value proposition is to offer quality higher education for students that likely has long-term sustainability and impact (O'Brien, Cooney & Blenker 2019). Higher education institutions becoming entrepreneurial creates and maximises value through HEEEs. Ecosystems such as HEEEs must create a differentiated value proposition to attract students, the end consumer (Adner, Oxley & Silverman 2013). To do so, higher education institutions may want to support students beyond educating them to secure a corporate job, such as developing students with E&I capabilities to become entrepreneurs and move up the economic ladder, unlocking their financial independence, in a resource-constrained environment. A higher education institution vision and value proposition must express that the institution deeply cares about students' career aspirations and their development for the future. They must improve the current promise of the ability to secure a stable job with a competitive salary through a first degree to include the career option of self-employment, founding start-ups and becoming entrepreneurs. Selling a value proposition of such nature and promoting the betterment of students will help in cocreating a sustainable HEEE in a resource-constrained environment. This new thinking brings an integral element of value proposition from strategy and marketing (Lanning & Michaels 1988) to HEEE literature.

Aligning to such a more holistic value proposition would require higher education institutions to take on a dichotomous role by catering for students who prefer a corporate career and others who desire an entrepreneurial career. This means providing high quality learning experiences for their students and contributing towards the United Nations SDG4 of quality education (United Nations 2020). In a resource-constrained environment, many higher education institutions may not see the benefit of entrepreneurship and tend to avoid wider efforts to focus on students' alternative career aspirations. Higher education institutions must begin to believe that to create value, their institutions must be strategically entrepreneurial (Covin 2002) and holistic, better understanding the value chain they are a part of. In this case, higher education institutions need to understand the diverse ecosystem actors, from potential students and parents to current students, alumni, industries, start-ups and the wider ecosystem in which the HEEE operates. This understanding of the value chain emphasises the role higher

education institutions and their HEEEs play in developing youth with relevant skills including in E&I for entrepreneurship and employment.

Ecosystems include a set of actors that need to be brought together in alignment for a value proposition to materialise in the environment (Adner 2017). Similarly, in HEEEs, senior management of institutions must bring ecosystem actors together and embrace the same ethos in developing students for E&I. Externally, ecosystem actors from diverse backgrounds with individual goals can work together and amplify their contributions to the value proposition that the HEEE is working towards. Internally, senior management can train their employees to think and work in teams towards the value proposition. In co-creating and evolving HEEEs, higher education institutions must understand the holistic transformation and management involved. Entrepreneurship and innovation should be a way of being, behaving and interacting among each other within the HEEE and in the wider entrepreneurial ecosystem.

Internal stakeholders shared their challenges and external stakeholders offered insights on the ways in which the six contextual factors can be implemented within HEEEs. Mechanisms of operationalising contextual factors can be practical solutions addressing problems prevailing in a resource-constrained environment. Further, the approach to operationalising contextual factors needs to put 'people' at the centre of the process, with empathy to understand the needs of diverse ecosystem actors. When co-creating and evolving HEEEs, higher education institutions can view this as holistically seeing the whole-picture perspective in any value chain. Findings from this study add understanding and knowledge to the literature, beyond the composition of factors to the configuration of HEEEs.

10.2.4 Beyond the composition of HEEEs

The third and final research question studied how diverse stakeholders can engage within the factors of HEEEs that could influence developing students' E&I capabilities in a resource-constrained environment. Three overarching findings emerged illustrating ways ecosystem actors and social capital influence the development of students' E&I capabilities within HEEEs in a resource-constrained environment through stakeholder engagement. Developed capabilities are referred to as resources, particularly related to social capital (de Brito, Lenz & Pacheco 2022).

Finding 10: Higher education institutions and their HEEEs lack the social network of internal and external stakeholders within the contextual factors (see 9.3).

Finding 11: Developing social capital enables HEEEs to function effectively through people, knowledge and relationships (see 9.4 to 9.9).

Finding 12: Ecosystem actors embedded within the contextual factors of HEEE create, share and capture value towards a shared value proposition (see 9.10).

HEEEs are primarily defined by the ecosystem actors within a specific context that may contribute to the function of delivering entrepreneurship education and supporting start-ups (Fetters, Greene & Rice 2010; Rice, Fetters & Greene 2014; Bock et al. 2020). HEEE studies investigate factors relevant to geographic contexts although they are not explicitly included in the definitions of the concept (Miller & Acs 2017; Rice, Fetters & Greene 2014). This study reveals the pivotal role that ecosystem actors, such as alumni entrepreneurs, play in orchestrating the HEEE factors with a collective effort in a resource-constrained context towards developing students for E&I. Social networks facilitate exchange between stakeholders and in this case, it is the exchange of resources. This advances the understanding the definition of HEEEs as a combination of factors and actors, similar to the definition of entrepreneurial ecosystems by Stam and Spigel (2016).

When examining the existing HEEE definitions, it is evident that scholars have highlighted ecosystem actors such as 'multidimensional enterprises' (Fetters, Greene & Rice 2010) to 'a set of actors' (Bock et al. 2020) within a specific context that may contribute to delivering entrepreneurship education and supporting start-ups. However, the extant HEEE literature explores factors as the composition of HEEEs with less attention to the ecosystem actors identified in the definitions of the concept. The corollary of the insight, the African proverb "It takes a village to raise a child", applies here; in the sense that it takes a community of individuals and organisations to develop an entrepreneur and their start-up in a conducive environment (Lyons 2002). Further, as John Donne, the English poet wrote, "no man is an island, entire of itself" (Doanward & Rasciute 2016); in this case, a potential entrepreneur who is disconnected from others and isolated from the environment cannot thrive on their own. To fully understand HEEEs, it is essential to recognise and appreciate their network and the network in which it is embedded. Broadly, no HEEE or potential entrepreneur is self-sufficient. In order to develop students' E&I capabilities, HEEEs need to collaborate over the identified six contextual factors with ecosystem actors, who can contribute to the value proposition. Advancing the understanding of the HEEE composition established by existing studies, this study encompassed factors with relevant actors.

Most top 12 reasons start-ups fail point at the entrepreneur rather than the market, competition and investors, and founders are held responsible (CB Insights 2021). Sri Lanka as an emerging economy suffers from a high start-up failure rate and the top associated reasons are insufficient business acumen, lack of market research and lack of commitment of entrepreneurs (PricewaterhouseCoopers 2020). Social networks are a mechanism for facilitating entrepreneurship (Hills et al 1997; Singh et al 1999) and engaging with the environment (Johannisson 1988; Weick 1969). A variety of stakeholders shape the development of E&I capabilities obtained by students (Spigel 2017).

Socially proximal groups in the HEEE and the entrepreneurial ecosystem can develop students for E&I, and potentially increase the number of successful start-ups by improving their E&I capabilities. Not only can students gain access to infrastructure and resources through their networks, students also learn better and develop stronger commitment through the inclusion of relevant stakeholders from the entrepreneurial ecosystem. Through social learning from relevant ecosystem actors, students can learn various knowledge and skills of business acumen and develop the core market research skills needed as E&I capabilities to build a successful start-up. The capability of learning contributes to the potential long-term survival of small businesses, including start-ups (de Brito, Lenz & Pacheco 2022). For this capability development to happen, HEEEs can embedded relevant ecosystem actors within the contextual factors to promote social learning.

Higher education institutions have low connectedness where they lack institutional frameworks and support mechanisms in promoting social interactions among their students (de Silva, Uyarra & Oakey 2012). These higher education institutions tend to operate within closed environments to a large extent and any external relationships are more of an ad hoc nature. Within the HEEE, this means that they lack the strengths that local relationships and a sense of community for the ecosystem. This low connectedness hinders the ability of social learning for students where they can acquire new knowledge and behaviours by observing others, such as young entrepreneurs and expert mentors. Social interactions with relevant external stakeholders will offer students exposure beyond the classroom and books. In contrast to some higher education institutions in the private sector, public universities such as the Department of Entrepreneurship of the University of Sri Jayewardenepura (2021) have strong ties with entrepreneurs and regularly invite them to share inspirations, motivations, knowledge and experience with students and staff via an online forum. This is consistent with the Innovation and Entrepreneurship Development Unit of the University of Colombo (2021) that has active

dialogue with entrepreneurs to create awareness and share knowledge on starting and doing small businesses. Given the collective culture in Sri Lanka (Hofstede Insights 2022), relationship building should come naturally, with a high degree of interdependence in the community. Social interactions with diverse stakeholders have been found to play a critical role in the cognitive development of students (Okita 2012). In this case, HEEEs offering opportunities for social interactions and learning can develop students' E&I capabilities through support and long-term commitment from a collectivist society where people perceive that they belong in groups and have strong relationships.

Building relationships and engaging with ecosystem actors results in social capital that is greatly associated with successful venture outcomes. Exposing students to entrepreneurial ecosystems and offering opportunities to engage with relevant stakeholders can benefit their resource repertoire. External stakeholders claim that it is necessary for students who are potential entrepreneurs to build their social capital in the early stages, particular in a resourceconstrained environment. Students build their social capital within HEEEs during their higher education journey (Longva 2021). While higher education institutions and their HEEEs build people (structural), knowledge (cognitive) and relationships (relational) through social capital, students can benefit from whom they have met and get to know. To start as an entrepreneur and pursue entrepreneurship, students tend to convince themselves of a common future with successful entrepreneurs they know (Wiltbank et al. 2006) and turn to who they know in the network (Sarasvathy & Dew 2005). In this case, social capital may develop students' selfefficacy and E&I capabilities with HEEEs playing an important role in entrepreneurial careers, both in the short term and long term as students' progress in their careers.

In developing countries, HEEEs operate with resources drawn from the respective higher education institution and from external sources. To function effectively, HEEEs in resource-constrained environments need to develop their resource repertoire, in addition to the resources of students. Instead of calling for resources in general for the HEEE, each HEEE factor can benefit from the engagement of both internal and external stakeholders. Ecosystem actors can be embedded within the six identified contextual factors, creating, sharing and capturing value. However, this study found signs of social barriers between higher education institutions and ecosystem actors that create costs and losses, depleting energy in working together within the HEEE. In moving forward, attracting relevant ecosystem actors and leveraging stakeholder engagement within the HEEE can bridge social barriers and foster connectedness. Being embedded within the entrepreneurial ecosystem, including a sense of

community and local relationships between ecosystem actors, is strongly associated with startup performance; high connectedness of the ecosystem results in higher start-up performance (Genome Global Startup Ecosystem Report 2018). Strengthening the connectedness of an HEEE may have a ripple effect and improve the low connectedness rating and the health of the entrepreneurial ecosystem in Sri Lanka (Genome Global Startup Ecosystem Report 2021). Higher education institutions can benefit from involving stakeholders who matter the most to the development of E&I capabilities in their respective HEEEs and connect with the wider entrepreneurial ecosystem. How and the degree to which HEEEs are connected to the entrepreneurial ecosystem can influence developing students' E&I capabilities and the performance of both ecosystems. In a resource-constrained environment, connecting ecosystems and embracing as a community can result in a competitive advantage.

Developing an entrepreneurial community can assist entrepreneurs in the creation, growth and survival of their start-ups (Koven & Lyons 2003). Even in the most entrepreneurial regions in Europe, some higher education institutions lack an entrepreneurial community to support their students and start-ups (Belitski 2019). In HEEE studies, entrepreneurial clubs and other societies in the form of networks can act as entrepreneurial communities (Wright, Siegel & Mustar 2017; Miller & Acs 2017). Relationships can be created not just within HEEEs but between HEEEs of different higher education institutions and even with the wider entrepreneurial ecosystem. HEEE communities are shaped by risk-taking management, courageous educators, supportive entrepreneurs, tenacious mentors and potential student entrepreneurs in these ecosystems. Contextual factors and ecosystem actors of HEEEs working together can lead to individual and societal impacts in a resource-constrained environment. When connecting and collaborating with stakeholders in the entrepreneurial ecosystem, this leads to developing an entrepreneurial community and improving the wellbeing of HEEEs. The more HEEEs are exposed to the unity of their contextual factors and ecosystem actors, the more this becomes inclusive and harmonious, going beyond the factors and strongly evolving into the wider entrepreneurial ecosystem.

Diverse stakeholders can engage within the factors of HEEEs that could influence students' E&I capabilities in a resource-constrained environment. To do so, higher education institutions need to embrace the social element of HEEEs and understand the network of internal and external stakeholders within the identified six contextual factors. Developing social capital enables HEEEs to function effectively through people, knowledge and relationships. When ecosystem actors embed the contextual factors of HEEEs, they create, share and capture value towards a shared value proposition. These empirical findings illustrate the relationship between HEEEs and students' E&I capabilities in a resource-constrained environment, beyond the composition of HEEEs.

10.3 Contributions and implications

10.3.1 Contributions to theory

This study generates significant contributions to the HEEE literature and to the intersection of entrepreneurship and the entrepreneurial ecosystem. In reference to theoretical contributions by Locke and Biddle (1997), this study fills in some 'incompleteness' and 'inadequate' aspects in the existing body of HEEE knowledge. In some instances, the concept of HEEE is in its initial stage of development and extant literature is neither mature nor comprehensive. In other instances, HEEE literature has overlooked different views that are significant for a more nuanced understanding.

Previous literature indicates that HEEEs are not only emerging but also an underdeveloped theoretical stream (Longva 2020; Hsieh & Kelley 2020). The first contribution of this study is related to the outcome of HEEEs, particularly through the views of internal and external stakeholders. Despite the exponential growth of HEEE studies, most studies focus on students' start-ups and more recently students' entrepreneurial mindset and intention, paying less attention to the capabilities needed to bridge the two. In a context which the literature stream on HEEEs is primarily focused on student start-ups, this study emphasizes the relevance of developing E&I capabilities of students. HEEEs need to shift focus to developing E&I capabilities of students and reduce the widening gap between entrepreneurial mindset and intentions and student start-ups. This understanding brings E&I capabilities to HEEE literature advocating for human capital development of entrepreneurs in resource-constrained environments (de Brito, Lenz & Pacheco 2022; Mair et al. 2012). More precisely, developing E&I capabilities of students can help to improve start-up failure caused due to lack of entrepreneurial skills and expertise as found in a report by PricewaterhouseCoppers (2020). Further, HEEEs focusing on E&I capabilities will also help students across multiple occupation roles from self-employment and hybrid entrepreneurship to intrapreneurship (Alsos et al. 2022). This holistic view of HEEEs contributes to a more advanced understanding of the concept that embraces the systematic perspective of ecosystems.

The literature highlights a lack of knowledge about the concept of HEEEs and the need for the composition to be clarified (Hsieh & Kelley 2020). A common interest in the literature

is the key success factors that form the HEEE composition of a higher education institution in an emerging economy (Fetters, Greene & Rice 2010). Several studies have attempted to understand the HEEE factors that lead to students' start-ups, entrepreneurial mindset and intention. However, the literature overlooks understanding for those higher education institutions who may want to promote entrepreneurship in resource-constrained environments (Roundy 2017) and co-creating HEEEs in such environments is inherently challenging as they lack resources (Bedő, Erdős & Pittaway 2020). The HEEE composition including entrepreneurial orientation, E&I education, E&I research, enterprise experiences, entrepreneurial networks and entrepreneurial support emerged from this study as a set of six contextual factors that students may involve in and as a result develop their E&I capabilities. These contextual factors are unique to some extent to the nature of higher education institutions and are shaped by the specific challenges and circumstances of the developing country. As a result, the HEEE factors may differ according to the context and will not remain common across higher education institutions as found by Rice, Fetters and Greene (2014). Co-creating HEEEs in resource-constrained environments differ from developed countries confirming that context matters, consistent with Welter and Lasch (2008).

This study brings another theoretical contribution related to ecosystem actors. First, literature on HEEEs has mainly focused on composition by determining what factors work for respective higher education institutions, mostly case by case (Miller & Acs 2017; Rice, Fetters & Greene 2014) or in a region/country (Webber, Kitagawa & Plumridge 2020). Second, some studies recognise ecosystem actors such as industry and public systems that can add value to the HEEE (Longva 2021). Factors and actors are explored and examined more individually rather than how factors can involve actors or vice versa, while there is a need to address all parts (factors and actors) of the ecosystem (Volkmann et al. 2021). Ecosystem actors shape the flow of entrepreneurial knowledge and skills obtained by students (Spigel 2017). Therefore, all stakeholders should be involved in HEEEs (Rice, Fetters & Greene 2014) whereas higher education institutions in resource-constrained environments lack networks and collaborations for various reasons (Bedő, Erdős & Pittaway 2020). In addition to limited resource allocations for entrepreneurship, higher education institutions in Sri Lanka also lack strong networking relationships with stakeholders (Weernasinge, Jayewardane & Deshani 2016). Similar to Barki et al. (2020), this study also establishes the importance of social capital in overcoming challenges and crafting ecosystems in more resource-constrained environments.

A final and important contribution is linked to the approach of designing HEEEs in resource-constrained environments. Previous studies have focused on co-creating HEEEs by exploring their composition, without reaching consensus on the approach of design. The literature lacks understanding of how a higher education institution's internal design serves the purposes of an HEEE (Brush 2014). Higher education institutions need to understand value creation through their HEEEs. While value creation tools are not as common in the education environment, various tools from other disciplines can be contextualised (Lackéus 2015). Illuminating the findings, this study brings a design approach to literature for designing HEEEs in order that they co-create, grow and scale while addressing challenges in a resource-constrained environment.

10.3.2 Implications for practice

From a practical point of view, the objective of these implications is to encourage higher education institutions to co-create and evolve their HEEEs, aligned with the wider entrepreneurship ecosystem, to build a sustainable HEEE. Given the study focused on a resource-constrained environment, Sri Lanka, most of the implications below relate specifically to higher education institutions in developing countries. In addition, there are some practical implications applicable for all HEEEs including in developed economies.

Industries around the world are facing a paradigm shift in organisation and taking on the ecosystem perspective to business models to achieve goals such as creating value, growing core businesses and generating revenue (McKinsey & Company 2021). Higher education is one such industry where some institutions are launching entrepreneurial initiatives such as HEEEs for business enhancement. Enhancing higher education institutions requires change and traditional institutions are increasingly being replaced with ecosystems where HEEEs comprising diverse stakeholders within and outside work together in shaping the E&I capabilities of the next generation.

The understanding advocated in the literature refers to the tip of the iceberg that involves delivering entrepreneurship education and supporting new ventures in HEEEs. What we pay less attention to is the bottom of the iceberg that involves designing the co-creation of the HEEE. Empirical findings pose important implications for higher education institutions on how to co-create an HEEE and design a learning environment to develop E&I capabilities of students. A key argument brought forward by Isenberg (2016) is that ecosystems are selforganising and self-sustaining and such ecosystems are affected, influenced and facilitated rather than co-created. Despite this argument, almost all HEEE studies refer to co-creating or developing HEEEs. Going with most views, HEEEs are co-created and emerge, similar to any ecosystem. Based on the findings from the research questions, practical implications for higher education institutions in a resource-constrained environment, particularly Sri Lanka, are that they need to design an HEEE that suits the nature of the institution and the geographic context. Higher education institutions need to undertake a design approach and make key choices to increase the effectiveness of their HEEEs. These HEEE design choices must be consistent with respective entrepreneurial ecosystems and offer a coherent configuration.

Co-creating an HEEE as a learning environment requires higher education institutions to take responsibility as the orchestrator focused on developing students' E&I capabilities. As the orchestrator of their HEEE, higher education institutions need to uphold their responsibility to national priorities and take accountability at global level. Education is recognised as an integral element of the United Nations Sustainable Development Goals, particularly SDG4 'quality education', and a key enabler for all the other SGDs. Higher education institutions can champion sustainable development by supporting, promoting and contributing to high quality education. Although Sri Lanka claims that the nation is implementing the SDGs to improve sustainability (Ministry of Sustainable Development 2018), most higher education institutions are yet to align their services and operations to relevant targets, perhaps not realising that the education sector plays a pivotal role in global sustainability. Further, business schools of private higher education institutions in Sri Lanka are not listed as signatory members of the Principles for Responsible Management Education (PRME) Community of Practice (PRME 2022). In this case, these higher education institutions are less connected with the global movement of quality education through responsible management. Being involved in such communities would enable higher education institutions to understand the new shifts in higher education and be inspired by these principles. For instance, the first principle aims to "develop the capabilities of students to be future generators of sustainable value for business and society at large and to work for an inclusive and sustainable global economy" (PRME 2022). Embracing these principles in the business model will encourage higher education institutions to commit and play an influential role in quality education through their HEEE, develop students E&I capabilities and co-create an evolving HEEE for the future. Higher education that is transformative would enable youth with relevant skills for entrepreneurship and employment, leading to a better world and reducing the inequality of education across the globe.

Designing an HEEE requires a system perspective, staying true to the nature of ecosystems. Systems such as entrepreneurial ecosystems and HEEEs can be more or less systemic and change over time as factors and actors interact towards entrepreneurial performance (Malecki 2017). Understanding factors that facilitate student start-ups, entrepreneurial mindset and intentions contributing to the composition of HEEEs as in current HEEE studies are early work (Longva 2021; Webber, Kitagawa & Plumridge 2020; Guerrero, Urbano & Gajón 2020; Meyer et al. 2017; Wright, Siegel & Mustar 2017; Miller & Acs 2017; Rice, Fetters & Greene 2014). Ecosystem actors are equally important to factors where respective stakeholders engage in the six identified contextual factors. However, this is complicated as these ecosystem actors tend to have their own goals and agenda that need to jointly be involved in the HEEEs for developing students E&I capabilities. Higher education institutions can champion their HEEE through a set of interdependent actors and factors, taking a system perspective.

Entrepreneurial ecosystems are regarded as contexts that can be co-created using design science approaches (O'Shea et al. 2019) and design principles can complement research on entrepreneurial ecosystems (Wurth et al. 2021). Given that HEEEs are a multi-stakeholder environment and based on the empirical evidence from this study, a human-centred strategy would be an effective approach to co-create an HEEE. Multiple stakeholders expressed the need for higher education institutions to understand the needs of key stakeholders. This involves designing the learning environment to develop E&I capabilities of students, while addressing the dynamics and challenges in a resource-constrained environment. The challenges faced by higher education institutions and their HEEEs are not problems that neatly fit within one discipline. These challenges are complex, interconnected and dynamic, and current solutions can become future problems, similar to any 'wicked' problem (Dorst 2015; Rittel & Webber 1973). Design thinking is a human-centred and interdisciplinary approach for solving wicked and ill-structured problems using an iterative prototyping method (Plattner, Meinel & Leifer 2010; 2015). Entrepreneurship education has benefited from using design thinking to disrupt teaching models (de Waal & Maritz 2021; Nielsen & Strovang 2015). Similarly, HEEEs could be at a competitive advantage by using design thinking to co-create and evolve, while addressing challenges in a resource-constrained environment.

Using an approach such as design thinking to co-create, evolve and scale an HEEE can leverage higher education institutions to integrate the needs of key stakeholders including students, parents and ecosystem actors and the requirements for business success, when designing the learning environment to develop E&I capabilities. Such a strategy would draw together members from the various functions in the institution in understanding the different input from ecosystem actors and the broader environment on co-creating an HEEE. An interfunctional team using suitable tools and developing solutions that address problems in a free-thinking, learning and creative environment follows a design thinking process (Plattner, Meinel & Leifer 2020). This interconnectedness and interdependency would contribute to an impact that is broader and more effective than being institutionally driven and designed. Some practical implications for higher education institutions in resource-constrained environments are discussed in the sequence of discovery, design, development and delivery aligned to the phases of design thinking.

Design thinking takes the form of a seven-step process of understand, observe, define viewpoint, ideate, prototype, test and reflect (Plattner, Meinel & Leifer 2020). Higher education institutions can condense these seven steps to four practical phases that map with strategic planning – discovery, design, development and deliver. First, the higher education institution must discover by understanding the problem and challenges, observing the needs of key stakeholders, taking stock of resources and defining how they might arrive at a solution. To understand the status quo of the HEEE, higher education institutions can undertake an annual review of activities, performance and outcomes, then advance into relationships with ecosystem actors including students, parents, alumni, entrepreneurs and angel investors. A start would be to conduct a series of focus group discussions with key stakeholders including students to understand their needs, expectations and feedback on current HEEE activities. Next, it would be ideal to invite key stakeholders representing the entrepreneurial ecosystem to a graduate profiling meeting to take input and recommendations that would assist higher education institutions to achieve a closer alignment and integration of the HEEE with the wider entrepreneurial ecosystem. Discussing review results and recommendations with ecosystem stakeholders would be worthwhile to determine how they can play a role and contribute to achieving the value proposition that leads to the wider entrepreneurial ecosystem. Insights from this phase will enable higher education institutions to understand the HEEE in its entirety to improve and position for success and sustainability.

10.4 Limitations of the study

While limitations are common in any research, efforts were made to minimise limitations throughout the research process ensuring that the study contributes to the body of knowledge

and practice. This section is a broad disclaimer acknowledging all the limitations of the study including the place where the research is based, specific reference to time and special circumstances that warrant the generalisation of findings (Wolcott 1990).

The first limitation involves the geographic context of this study. Since the emergence of the concept, HEEE studies have been context specific. While a large proportion of studies have been based on universities in developed economies where the concept was born, this study took the initiative to explore HEEEs of private higher education institutions in a resource-constrained environment by focusing on an emerging economy, Sri Lanka. Higher education institutions in the Sri Lankan private sector play a pivotal role in supplying to the demand that public universities are constrained to satisfy. Some 150,000 students out of 250,000 qualify for higher education and approximately 92,000 want to pursue higher education (University Grants Commission 2022). However, public universities can only absorb 45,000 students, which is almost 50% of the demand, while private institutions cater to a little more than 10%, approximately 11,000 students (Sri Lanka Export Development Board 2018). Findings from this study are generalisable to some extent and applicable to developing countries with similar characteristics including private higher education institutions. Scholars can consider using these findings to test in similar developing countries and resource-constrained contexts.

The second limitation of this study involves the special circumstances in which this study was conducted. Most HEEE studies emerged before the global pandemic. COVID-19 was declared in March 2020. One of the most recent HEEE studies published in mid-2021 has been in the publication process since end of 2020 and there is no indication of whether data was collected before the pandemic. Data collection for this study on HEEEs was during the pandemic. However, participants did not discuss the topic or relate to the pandemic during the two rounds of interviews (September to November 2020 and August to September 2021).

A limitation to keep in mind when interpreting findings from this study is that it has no representation of government bodies. Given the current nature of private higher education institutions in Sri Lanka and the low impact that the government domain has on these private businesses, government bodies were excluded as an ecosystem actor in data collection. The unit of analysis in this study was represented by higher education institutions and external stakeholders that hold interest or worked with HEEEs in some way. However, with emerging political instability and possibilities of a leadership change in Sri Lanka during the last stage of this study, there could be no need to rule out the perceptions of government representatives in a future study. It could be worthwhile to take a policy lens on private higher education institutions including their HEEEs that can contribute to national priorities of E&I, including youth entrepreneurship and SMEs.

The study is also subject to the limitation that arises from scoping to draw broader perspectives from internal and external stakeholders, before investigating students. While the study focuses on students and the central theory framing is student centric, the study explores HEEEs through perspectives from deans/heads of schools, academics, alumni entrepreneurs, expert entrepreneurs, angel investors and support professionals. Students, the subjects experiencing HEEEs yet lacking real-world experiences, were understood to have limited external validity (Peterson 2001) and were seen as more suitable for investigation in a future study to advance the findings from this study.

The limitations of this study prevent the full exploration of HEEEs and future research directions could explore some of the limitations deemed relevant, including government bodies for a policy perspective on HEEEs and students' voice on the composition of HEEEs.

10.5 Future research directions

While there is much attention to entrepreneurship education and entrepreneurial ecosystems in the literature, it is unfortunate that there is less attention paid to HEEEs and how higher education institutions can develop their HEEEs. Future research directions are derived through findings where new questions emerge and limitations are described. Scholars may relate to these recommendations as research gaps and questions that might be of interest for further investigation, contributing to HEEE literature.

When exploring HEEEs in resource-constrained environments, some challenges that impede the development of HEEEs in Sri Lanka came to light in this study. However, exploring these challenges restricting HEEEs was not the main focus of this study and thus did not provide deeper insights into why these barriers exist in an emerging economy. Facilitating and hindering factors among higher education institutions in Finland has led to a more balanced understanding of HEEEs (Lahikainen, Peltonen & Ruskovaara 2019). Less focus has been put into understanding what is challenging HEEEs with resource constraints in developing countries within extant literature. Given that this study focused on Sri Lanka, scholars may want to investigate a comparative study of the factors acting as barriers towards HEEEs at institutional (micro) and national (macro) levels, in two or more developing countries, illustrating ways HEEEs are impacted uniquely in geographically dispersed yet similar contexts. Such a study could provide an opportunity to understand if similar or different hindering factors are found in HEEEs of developing countries. This empirical evidence would contribute a holistic perspective of HEEEs in more resource-constrained environments.

Given the nature of higher education institutions in the private sector and the limited government influence on them embodied in Sri Lanka, representation of ministries was excluded from the sample and a policy lens was not pursued in this study. In other Asian countries, governments are making efforts to accelerate entrepreneurship education and improve the quality of higher education through policies (Yu 2018; Yu et al. 2017). Asian developing economies such as China, south-east economies such as Indonesia, Malaysia, Vietnam and Thailand, as well as South Asian developing economies including India and Bangladesh are steadily increasing E&I outcomes through various education-oriented policies (Weerasinghe & Jayawardane 2019). Policymakers should emphasise human capital development through education and training so that entrepreneurs are better prepared for running their start-ups in a resource-constrained environment, resulting in more successful ventures (de Brito, Lenz & Pacheco 2022). Through the evidence here, this is an important consideration for relevant government bodies in Sri Lanka to regulate private higher education in order to accelerate E&I for sustainable development. As a result, regulating private sector institutions will contribute to the SDG4 on quality education that focuses on increasing the number of youth with relevant skills for entrepreneurship (United Nations 2020). In the most recent national review of the SDGs, improvement areas are mapped to public universities and vocational education, overlooking private higher education institutions that have the potential to prepare students with relevant skills for entrepreneurship (Ministry of Sustainable Development 2018). For the way forward, future research that sets out policy, precautions and practices would be worthwhile to support policy planning and implementation that manifests outcomes including E&I capabilities among students in private higher education institutions.

This study specifically explored and understood the composition of HEEEs focused on developing E&I capabilities in a resource-constrained environment. Findings reveal six contextual factors and 18 mechanisms that higher education institutions can apply in their HEEEs. Further examination can verify specific factors and mechanisms that are effective for a particular higher education institution or a group of similar institutions in a given region. There are not many empirical studies examining the interdependences of such factors (Roundy, Bradshaw & Brockman 2018; Mack & Mayer 2016). Scholars may want to examine the

strength, impact and interdependences of HEEE factors among students, as an extension of this study, through a longitudinal study. Experimental research is ideal in this case where students' change can be assessed and evaluated as before-and-after scenarios. Given that there is a growing interest in uplifting the quality of education, aligned to SDG4, it would be a worthwhile project to focus on advancing HEEEs in Colombo, Sri Lanka. Similar projects, including Accelerating Higher Education Expansion and Development (AHEAD) and Improving Relevance and Quality of Undergraduate Education (IRQUE) funded by the World Bank, have supported the higher education sector, including private institutions, in Sri Lanka. Understanding the strength, impact and interdependences of HEEE factors through change in students' perspectives and behaviour would offer insights for higher education institutions on ways resources can be effectively utilised within contextual factors and mechanisms for high impact. This would also contribute a resourcing perspective to HEEE literature.

The final research question of this study paid attention to stakeholder engagement within the HEEE through a social capital perspective. Based on the future direction set by Alvedalen and Boschma (2017), entrepreneurial ecosystem studies including Neumeyer and Santos (2018; 2019) and Pittz, White and Zoller (2019) have applied social network analysis (SNA) to underline social network theory to explore social clusters and their connectivity. Scholars exploring HEEEs are yet to employ SNA techniques to examine relationship data, including strength of the networking relationships among higher education institutions and ecosystem actors. Such a study would advance the understanding of stakeholder engagement and social capital in HEEEs. Building relationships and social capital has been seen as an increasingly important resource for communities (Powell & Baker 2011), including students and higher education institutions. The SNA approach missing in the literature thus far might offer a much-needed analytical method to determine who needs to be part of the HEEE and the interplay between the six contextual factors, advancing the findings of this study; for instance, the strength of ecosystems actors that emerged in this study and the relationship effects for the configuration of HEEEs.

Extant literature including this study pay attention to the composition of HEEEs by discovering contextual factors and ecosystem actors. The findings of this study reveal the interference of 'challenges' in the environment and the importance of the 'people' involved in HEEEs in resource-constrained environments. As a result, higher education institutions could benefit by taking a human-centred approach to co-creating and evolving their HEEEs, such as design thinking. Design thinking has disrupted entrepreneurship education where teaching

models have been created for delivering higher education programs (Nielsen & Strovang 2015; de Waal & Maritz 2021). Using design thinking as an approach to co-create and evolve HEEEs may help higher education institutions to create better value, develop E&I capabilities for students, collaborate with ecosystem actors and overcome resource constraints. Studies can empirically explore design thinking and examine how design principles can be applied to advance HEEEs. In doing so, future research can examine the ways in which user-centricity, empathy, interdisciplinary teams, collaboration, ideation and other design thinking principles (Plattner, Meinel & Leifer 2020) become applicable for HEEEs. Qualitative research, particularly interviews with ecosystem actors, would lead to a deeper understanding of design thinking principles in the HEEE context and contribute to exploring the design perspective of HEEEs.

10.6 Chapter summary

This chapter clearly articulates the findings relating to the research questions, explains the significance of the findings, discusses contributions to theory and implications for practice, outlines the limitations and presents future research directions. The growth in HEEE literature has led to a multidisciplinary and fragmented canon. This study contributes to illustrating the relationship between HEEEs and students' E&I capabilities in a resource-constrained environment. The research contributes to theory by extending perspectives on how HEEE can develop students' E&I capabilities in a resource-constrained environment and contributions are made primarily to the domain of HEEE literature in the field of entrepreneurship. Implications for higher education institutions were explained regarding how they can continue the development of their HEEEs, pay attention to E&I capabilities of students and contribute to the entrepreneurial ecosystem. Limitations of the study are discussed in relation to the geographic context, specific reference to timeframe and special circumstances that warrant the generalisation of findings. Future research is proposed to take new perspectives including design, leadership, resourcing and balanced aligning to insights and queries that have raised from this study. The study and this chapter close with this summary.

REFERENCES

- Abayasekara, A. (2018). Building a More English-Literate Sri Lanka: The Need to Combat Inequities. Ips.lk. Retrieved from https://www.ips.lk/talkingeconomics/2018/04/23/building-a-more-english-literate-srilanka-the-need-to-combatinequities/#:~:text=Although%20teaching%20English%20as%20a,above%2015%20y ears%20of%20age.
- Abney, A., Cook, L., Fox, A., & Stevens, J. (2018). Intercollegiate Social Media Education Ecosystem. *Journal Of Marketing Education*, 41(3), 254-269. <u>https://doi.org/10.1177/0273475318786026</u>
- Abreu, M., Demirel, P., Grinevich, V., & Karataş-Özkan, M. (2016). Entrepreneurial practices in research-intensive and teaching-led universities. *Small Business Economics*, 47(3), 695-717. <u>https://doi.org/10.1007/s11187-016-9754-5</u>
- Abreu, M., & Grinevich, V. (2013). The nature of academic entrepreneurship in the UK: Widening the focus on entrepreneurial activities. *Research Policy*, 42(2), 408-422. <u>https://doi.org/10.1016/j.respol.2012.10.005</u>
- Acs, Z., & Amorós, J. (2008). Entrepreneurship and competitiveness dynamics in Latin America. *Small Business Economics*, 31(3), 305-322. <u>https://doi.org/10.1007/s11187-008-9133-y</u>
- Acs, Z., Audretsch, D., & Lehmann, E. (2013). The knowledge spillover theory of entrepreneurship. *Small Business Economics*, 41(4), 757-774. <u>https://doi.org/10.1007/s11187-013-9505-9</u>
- Acs, Z., Desai, S., & Hessels, J. (2008). Entrepreneurship, economic development and institutions. *Small Business Economics*, 31(3), 219-234. https://doi.org/10.1007/s11187-008-9135-9
- Acs, Z. J., & Audretsch, D. B. (2003). Handbook of Entrepreneurship Research An Interdisciplinary Survey and Introduction. New York: Springer. <u>https://doi.org/10.1007/b105789</u>
- Acs, Z. J., Audretsch, D. B., Braunerhjelm, P., & Carlsson, B. (2012). Growth and entrepreneurship. *Small Business Economics*, 39(2), 289-300. <u>https://doi.org/10.1007/s11187-010-9307-2</u>
- Acs, Z. J., Audretsch, D. B., Lehmann, E. E., & Licht, G. (2016). National systems of entrepreneurship. *Small Business Economics*, 46(4), 527-535. <u>https://doi.org/10.1007/s11187-016-9705-1</u>
- Acs, Z. J., Autio, E., & Szerb, L. (2014). National Systems of Entrepreneurship: Measurement issues and policy implications. *Research Policy*, 43(3), 476-494. <u>https://doi.org/https://doi.org/10.1016/j.respol.2013.08.016</u>

- Acs, Z. J., Stam, E., Audretsch, D. B., & O'Connor, A. (2017). The lineages of the entrepreneurial ecosystem approach. *Small Business Economics*, 49(1), 1-10. https://doi.org/10.1007/s11187-017-9864-8
- Adejimola, A., & Olufunmilayo, T. (2009). Spinning off an entrepreneurship culture among Nigerian University Students: *Prospects and challenges*, 3(3), 80-88. <u>https://academicjournals.org/journal/AJBM/article-full-text-pdf/D1EA54015677</u>
- Ajzen, I. (1991). The theory of planned behavior. *Organizational Behavior and Human* Decision Processes, 50(2), 179-211. <u>https://doi.org/10.1016/0749-5978(91)90020-T</u>
- Al Mamun, A., Fazal, S. A., & Muniady, R. (2019). Entrepreneurial knowledge, skills, competencies and performance. *Asia Pacific Journal of Innovation and Entrepreneurship*, 13(1), 29-48. <u>https://doi.org/10.1108/APJIE-11-2018-0067</u>
- Al-Atabi, M., & Deboer, J. (2014). Teaching entrepreneurship using Massive Open Online Course (MOOC). *Technovation*, 34(4), 261-264. <u>https://doi.org/10.1016/j.technovation.2014.01.006</u>
- Al-Dajani, H., Dedoussis, E., Watson, E., & Tzokas, N. (2014). Graduate Entrepreneurship Incubation Environments: A Framework of Key Success Factors. *Industry and Higher Education*, 28(3), 201-213. <u>https://doi.org/10.5367/ihe.2014.0205</u>
- Alexander, P. A., & Judy, J. E. (1988). The interaction of domain-specific and strategic knowledge in academic performance. *Review of Educational Research*, 58(4), 375–404.
- Aldridge, T. T., & Audretsch, D. (2011). The Bayh-Dole Act and scientist entrepreneurship. *Research Policy*, 40(8), 1058-1067. <u>https://doi.org/10.1016/j.respol.2011.04.006</u>
- Algieri, B., Aquino, A., & Succurro, M. (2013). Technology transfer offices and academic spin-off creation: the case of Italy. *The Journal of Technology Transfer*, 38(4), 382-400. <u>https://doi.org/10.1007/s10961-011-9241-8</u>
- Ali, I., Ali, M., & Badghish, S. (2019). Symmetric and asymmetric modeling of entrepreneurial ecosystem in developing entrepreneurial intentions among female university students in Saudi Arabia. *International Journal of Gender and Entrepreneurship*, 11(4), 435-458. <u>https://doi.org/10.1108/IJGE-02-2019-0039</u>
- Alibekova, G., Tleppayev, A., Medeni, T. D., & Ruzanov, R. (2019). Determinants of Technology Commercialization Ecosystem for Universities in Kazakhstan. *The Journal of Asian Finance, Economics and Business,* 6(4), 271-279. <u>https://doi.org/10.13106/jafeb.2019.vol6.no4.271</u>
- Allahar, H., & Sookram, R. (2019). Emergence of university-centred entrepreneurial ecosystems in the Caribbean. *Industry and Higher Education*, 33(4), 246-259. https://doi.org/10.1177/0950422219838220

- Allahar, H., & Sookram, R. (2019). A University Business School as an Entrepreneurial Ecosystem Hub. *Technology Innovation Management Review*, 9(11), 15-25. <u>https://doi.org/10.22215/timreview/1280</u>
- Allen, K., & Lieberman, M. (2010). The Development of University-based Entrepreneurship Ecosystems: University of Southern California. In M. L. Fetters., P. G. Greene., M. P. Rice & J. S. Butler, *The Development of University-based Entrepreneurship Ecosystems Global Practices* (76-95). Cheltenham: Edward Elgar Publishing.
- Álvarez, C., Amorós, J., & Urbano, D. (2014). Regulations and Entrepreneurship: Evidence from Developed and Developing Countries. *Innovar*, 24(1), 81-89.
- Alvedalen, J., & Boschma, R. (2017). A critical review of entrepreneurial ecosystems research: towards a future research agenda. *European Planning Studies*, 25(6), 887-903. <u>https://doi.org/10.1080/09654313.2017.1299694</u>
- Amadi-Echendu, A. P., Phillips, M., Chodokufa, K., Visser, T., McGreal, R., & Conrad, D. (2016). Entrepreneurial Education in a Tertiary Context: A Perspective of the University of South Africa. *International Review of Research in Open and Distributed Learning*, 17(4), 21-35. <u>https://doi.org/10.19173/irrodl.v17i4.2482</u>
- Anderson, A. R., & Jack, S. L. (2002). The articulation of social capital in entrepreneurial networks: a glue or a lubricant? *Entrepreneurship & Regional Development*, 14(3), 193-210. <u>https://doi.org/10.1080/08985620110112079</u>
- Aparicio, S., Urbano, D., & Audretsch, D. (2016). Institutional factors, opportunity entrepreneurship and economic growth: Panel data evidence. *Technological Forecasting & Social Change*, 102(C), 45-61. https://doi.org/10.1016/j.techfore.2015.04.006
- Arend, R. J., Sarooghi, H., & Burkemper, A. C. (2016). Effectuation, not being pragmatic or process theorizing, remains ineffectual: responding to the commentaries. *The Academy of Management Review*, 41(3), 549-556. <u>https://doi.org/10.5465/amr.2016.0086</u>
- Arenius, P., & Clercq, D. (2005). A Network-based Approach on Opportunity Recognition. An International Journal, 24(3), 249-265. <u>https://doi.org/10.1007/s11187-005-1988-6</u>
- Astin, A. W. (1999). Student Involvement: A Developmental Theory for Higher Education. *Journal of College Student Development*, 25(4), 297-308. <u>https://www.researchgate.net/publication/220017441_Student_Involvement_A_Development_Theory_for_Higher_Education</u>
- Atiase, V. Y., Mahmood, S., Wang, Y., & Botchie, D. (2018). Developing entrepreneurship in Africa: investigating critical resource challenges. *Journal of Small Business and Enterprise Development*, 25(4), 644-666. <u>https://doi.org/10.1108/JSBED-03-2017-0084</u>
- Audretsch, D. B. (2014). From the entrepreneurial university to the university for the entrepreneurial society. *Journal of Technology Transfer*, 39(3), 313-321.

- Audretsch, D., Belitski. M., Guerrero, M., & Siegel, D. (2022). Assessing The Impact of The U.K.'S Research Excellence Framework on The Relationship between University Scholarly Output and Education and Regional Economic Growth. Academy Management Learning and Education, 21(3), 394-421.
- Audretsch, D., Kuratko, D., & Link, A. (2015). Making sense of the elusive paradigm of entrepreneurship. *Small Business Economics*, 45(4), 703-712. https://doi.org/10.1007/s11187-015-9663-z
- Audretsch, D., Mason, C., Miles, M. P., & O'Connor, A. (2018). The dynamics of entrepreneurial ecosystems. *Entrepreneurship & Regional Development*, 30(3), 471-474. <u>https://doi.org/10.1080/08985626.2018.1436035</u>
- Audretsch, D. B., & Belitski, M. (2017). Entrepreneurial ecosystems in cities: establishing the framework conditions. *The Journal of Technology Transfer*, 42(5), 1030-1051. https://doi.org/10.1007/s10961-016-9473-8
- Audretsch, D. B., Belitski, M., & Cherkas, N. (2021). Entrepreneurial ecosystems in cities: The role of institutions. *PLoS One*, 16(3), 1-22. <u>https://doi.org/10.1371/journal.pone.0247609</u>
- Audretsch, D. B., Cunningham, J. A., Kuratko, D. F., Lehmann, E. E., & Menter, M. (2019). Entrepreneurial ecosystems: economic, technological, and societal impacts. *The Journal of Technology Transfer*, 44(2), 313-325. <u>https://doi.org/10.1007/s10961-018-9690-4</u>
- Auld, G., Diker, A., Bock, M., Boushey, C., Bruhn, C., & Cluskey, M. et al. (2007). Development of a Decision Tree to Determine Appropriateness of NVivo in Analyzing Qualitative Data Sets. *Journal of Nutrition Education and Behavior*, 39(1), 37-47. <u>https://doi.org/10.1016/j.jneb.2006.09.006</u>
- Autio, E., Kenney, M., Mustar, P., Siegel, D., & Wright, M. (2014). Entrepreneurial innovation: The importance of context. *Research Policy*, 43(7), 1097-1108.
- Azizi, M., & Mahmoudi, R. (2019). Learning outcomes of entrepreneurship education: Entrepreneurship education for knowing, doing, being, and living together. *Journal of Education for Business*, 94(3), 148-156. <u>https://doi.org/10.1080/08832323.2018.1502139</u>
- Bacigalupo, M., Kampylis, P., Punie, Y., Van den Brande, G. (2016). EntreComp: The Entrepreneurship Competence Framework. Luxembourg: Publication Office of the European Union.
- Backs, S., Günther, M., & Stummer, C. (2019). Stimulating academic patenting in a university ecosystem: an agent-based simulation approach. *The Journal of Technology Transfer*, 44(2), 434-461. <u>https://doi.org/10.1007/s10961-018-9697-x</u>
- Bacq, S., Ofstein, L. F., Kickul, J. R., & Gundry, L. K. (2017). Perceived entrepreneurial munificence and entrepreneurial intentions: A social cognitive perspective. *International Small Business Journal*, 35(5), 639-659. <u>https://doi.org/10.1177/0266242616658943</u>
- Bae, T. J., Qian, S., Miao, C., & Fiet, J. O. (2014). The Relationship between Entrepreneurship Education and Entrepreneurial Intentions: A Meta–Analytic Review. *Entrepreneurship Theory and Practice*, 38(2), 217-254. <u>https://doi.org/10.1111/etap.12095</u>
- Bagchi-Sen, S., & Smith, H. L. (2012). The Role of the University as an Agent of Regional Economic Development. *Geography Compass*, 6(7), 439-453. <u>https://doi.org/10.1111/j.1749-8198.2012.00497.x</u>
- Balan, P., Maritz, A., & McKinlay, M. (2018). A structured method for innovating in entrepreneurship pedagogies. *Education* + *Training*, 20(7-8), 819-840. https://doi.org/10.1108/ET-05-2017-0064
- Ballereau, V., Sinapi, C., Toutain, O., & Juno-Delgado, E. (2015). Developing a Business Model The Perception Of Entrepreneurial Self-Efficacy Among Students In The Cultural And Creative Industry. In O. Kuhlke., Schramme. A & Kooyman. R, *Cultural Entrepreneurship in Theory, Pedagogy and Practice* (164-177). Utrecht, Eburon Academic Publishers.
- Baraldi, E., & Ingemansson Havenvid, M. (2016). Identifying new dimensions of business incubation: A multi-level analysis of Karolinska Institute's incubation system. *Technovation*, 50-51, 53-68. <u>https://doi.org/10.1016/j.technovation.2015.08.003</u>
- Barki, E., de Campos, J. G. F., Lenz, A. K., Kimmitt, J., Stephan, U., & Naigeborin, V. (2020). Support for social entrepreneurs from disadvantaged areas navigating crisis: Insights from Brazil. *Journal of Business Venturing Insights*, 14(e00205), 1-9.
- Baron, R. A., & Henry, R. A. (2010). How entrepreneurs acquire the capacity to excel: insights from research on expert performance. *Strategic Entrepreneurship Journal*, 4(1), 49-65. <u>https://doi.org/10.1002/sej.82</u>
- Baron, R.A. (2014). *Essentials of Entrepreneurship: Evidence and Practice*. Cheltenham: Edward Elgar.
- Basu, R. (2014). Entrepreneurship Education in India: A Critical Assessment and a Proposed Framework. *Technology Innovation Management Review*, 4(8), 5-10. <u>https://doi.org/10.22215/timreview/817</u>
- Batjargal, B. (2007). Internet entrepreneurship: Social capital, human capital, and performance of Internet ventures in China. *Research Policy*, 36(5), 605-618. <u>https://doi.org/10.1016/j.respol.2006.09.029</u>
- Bauman, A., & Lucy, C. (2021). Enhancing entrepreneurial education: Developing competencies for success. *The International Journal of Management Education*, 19(1), 1-10. <u>https://doi.org/10.1016/j.ijme.2019.03.005</u>

- Bazan, C., Gaultois, H., Shaikh, A., Gillespie, K., Frederick, S., & Amjad, A. et al. (2020). A systematic literature review of the influence of the university's environment and support system on the precursors of social entrepreneurial intention of students. *Journal of Innovation and Entrepreneurship*, 9(4), 1-28. https://doi.org/10.1186/s13731-020-0116-9
- Bedő, Z., Erdős, K., & Pittaway, L. (2020). University-centred entrepreneurial ecosystems in resource-constrained contexts. *Journal of Small Business and Enterprise Development*, 27(7), 1149-1166. <u>https://doi.org/10.1108/JSBED-02-2020-0060</u>
- Belitski, M. (2019). Entrepreneurship Ecosystems in Higher Education. In Innovate Higher Education to Enhance Graduate Employability: Rethinking the Possibilities. Oxfordshire: Taylor & Francis Group.
- Belitski, M., & Heron, K. (2016). Expanding entrepreneurship education ecosystems. *Journal* of Management Development, 36(2), 163-177. <u>https://doi.org/10.1108/JMD-06-2016-0121</u>
- Benneworth, P., Pinheiro, R., & Sánchez-Barrioluengo, M. (2016). One size does not fit all! New perspectives on the university in the social knowledge economy. *Science and Public Policy*, 43(6), 731-735. <u>https://doi.org/10.1093/scipol/scw018</u>
- Bezerra, É., Borges, C., & Andreassi, T. (2017). Universities, local partnerships and the promotion of youth entrepreneurship. *Journal of Lifelong Learning*, 63(5), 703-724. <u>https://doi.org/10.1007/s11159-017-9665-y</u>
- Bhat, S., & Khan, R. (2014). Entrepreneurship education ecosystem: an assessment study of J&K state. *International Journal of Economics, Commerce and Management*, 2(4), 1-8. <u>https://papers.ssrn.com/sol3/papers.cfm?abstract_id=2426797</u>
- Bhatia, A., & Levina, N. (2020). The Diverse Rationalities of Entrepreneurship Education: Epistemic Stance Perspective. Academy of Management Learning & Education, 19(3), 323-344. <u>https://doi.org/10.5465/amle.2019.0201</u>
- Bird, B. (1988). Implementing Entrepreneurial Ideas: The Case for Intention. *The Academy of Management Review*, 13(3), 442-453. <u>https://doi.org/10.2307/258091</u>
- Bischoff, K. (2021). A study on the perceived strength of sustainable entrepreneurial ecosystems on the dimensions of stakeholder theory and culture. *Small Business Economics*, 56(3), 1121-1140. https://doi.org/10.1007/s11187-019-00257-3
- Bischoff, K., Volkmann, C., & Audretsch, D. (2018). Stakeholder collaboration in entrepreneurship education: an analysis of the entrepreneurial ecosystems of European higher educational institutions. *The Journal of Technology Transfer*, 43(1), 20-46. <u>https://doi.org/10.1007/s10961-017-9581-0</u>

- Blenker, P., Dreisler, P., Faergemann, H., & Kjeldsen, J. (2008). A framework for developing entrepreneurship education in a university context. *International Journal of Entrepreneurship and Small Business*, 5(1), 45-63. <u>https://doi.org/10.1504/ijesb.2008.015953</u>
- Blenker, P., Frederiksen, S. H., Korsgaard, S., Müller, S., Neergaard, H., & Thrane, C. (2012). Entrepreneurship as Everyday Practice: Towards a Personalized Pedagogy of Enterprise Education. *Industry and Higher Education*, 26(6), 417-430. <u>https://doi.org/10.5367/ihe.2012.0126</u>
- Bock, C., Dilmetz, D., Selznick, B., Zhang, L., & Mayhew, M. (2020). How the university ecosystem shapes the innovation capacities of undergraduate students – evidence from Germany. *Industry And Innovation*, 28(3), 307-342. <u>https://doi.org/10.1080/13662716.2020.1784710</u>
- Bolzani, D., Munari, F., Rasmussen, E., & Toschi, L. (2020). Technology transfer offices as providers of science and technology entrepreneurship education. *The Journal of Technology Transfer*, 46(2), 335-365. <u>https://doi.org/10.1007/s10961-020-09788-4</u>
- Bramwell, A., & Wolfe, D. A. (2008). Universities and regional economic development: The entrepreneurial University of Waterloo. *Research Policy*, 37(8), 1175-1187. <u>https://doi.org/10.1016/j.respol.2008.04.016</u>
- Brandão, C. (2014). P. Bazeley and K. Jackson. Qualitative Data Analysis with NVivo. *Qualitative Research in Psychology*, *12*(4), 492-494. <u>https://doi.org/10.1080/14780887.2014.992750</u>
- Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative research* in psychology, 3(2), 77-101. <u>https://doi.org/10.1191/1478088706qp063oa</u>
- Breznitz, S., & Zhang, Q. (2019). Determinants of graduates' entrepreneurial activity. *Small Business Economics*, 55(4), 1039-1056. <u>https://doi.org/10.1007/s11187-019-00171-8</u>
- Breznitz, S. M., & Zhang, Q. (2019). Fostering the growth of student start-ups from university accelerators: An entrepreneurial ecosystem perspective. *Industrial and Corporate Change*, 28(4), 855-873. <u>https://doi.org/10.1093/icc/dtz033</u>
- Brito, R. P. D., Lenz, A. K., & Pacheco, M. G. M. (2022). Resilience building among small businesses in low-income neighborhoods. *Journal of Small Business Management*, 1-36.
- Brinckmann, J., Grichnik, D., & Kapsa, D. (2010). Should entrepreneurs plan or just storm the castle? A meta-analysis on contextual factors impacting the business planning– performance relationship in small firms. *Journal of Business Venturing*, 25(1), 24-40. <u>https://doi.org/10.1016/j.jbusvent.2008.10.007</u>
- Brown, R., & Mason, C. (2017). Looking inside the spiky bits: a critical review and of entrepreneurial ecosystems. *Small Business Economics*, 49(1), 11-30. https://doi.org/10.1007/s11187-017-9865-7

- Brush, C. (2014). Exploring the Concept of an Entrepreneurship Education Ecosystem. In S. Hoshkinson & D. F Kuratko (Eds.), *Innovative Pathways for University Entrepreneurship in the 21st Century (24, 25-39)*. Emerald Group Publishing Limited.
- Brush, C., Edelman, L., Manolova, T., & Welter, F. (2018). A gendered look at entrepreneurship ecosystems. *Small Business Economics*, 53(2), 1-16. <u>https://doi.org/10.1007/s11187-018-9992-9</u>
- Brush, C., & Gatewood, E. (2008). Women growing businesses: Clearing the hurdles. Business Horizons, 51(3), 175-179. <u>https://doi.org/10.1016/j.bushor.2008.01.007</u>
- Bruton, G. D., Ahlstrom, D., & Li, H. L. (2010). Institutional Theory and Entrepreneurship: Where Are We Now and Where Do We Need to Move in the Future? *Entrepreneurship Theory and Practice*, 34(3), 421-440. <u>https://doi.org/10.1111/j.1540-6520.2010.00390.x</u>
- Bull, I., & Willard, G. E. (1993). Towards a theory of entrepreneurship. Journal of Business Venturing, 8(3), 183-195. <u>https://doi.org/https://doi.org/10.1016/0883-</u> 9026(93)90026-2
- Burnette, J. L., Pollack, J. M., Forsyth, R. B., Hoyt, C. L., Babij, A. D., Thomas, F. N., & Coy, A. E. (2020). A Growth Mindset Intervention: Enhancing Students' Entrepreneurial Self-Efficacy and Career Development. *Entrepreneurship Theory and Practice*, 44(5), 878-908. <u>https://doi.org/10.1177/1042258719864293</u>
- Burrell, G., Clark, P., & Morgan, G. (1981). Sociological Paradigms and Organizational Analysis: Elements of the Sociology of Corporate Life. *The British Journal of Sociology*, 32(3), 448-455. <u>https://doi.org/10.2307/589304</u>
- Burton, M. D., Sørensen, J. B., & Dobrev, S. D. (2016). A Careers Perspective on Entrepreneurship. *Entrepreneurship Theory and Practice*, 40(2), 237-247. <u>https://doi.org/10.1111/etap.12230</u>
- Busch, C., & Barkema, H. (2021). From necessity to opportunity: Scaling bricolage across resource-constrained environments. *Strategic Management Journal*, 42(4), 741-773. https://doi.org/10.1002/smj.3237
- Busenitz, L. W., West, G. P., Shepherd, D., Nelson, T., Chandler, G. N., & Zacharakis, A. (2003). Entrepreneurship Research in Emergence: Past Trends and Future Directions. *Journal of Management*, 29(3), 285-308. <u>https://doi.org/10.1016/S0149-2063(03)00013-8</u>
- Burnard, P. (2006). Constructing Grounded Theory: A practical guide through qualitative analysis. *Nurse Researcher*, *13*(4), 84-84. <u>https://doi.org/10.7748/nr.13.4.84.s4</u>
- Byabashaija, W., & Katono, I. (2011). The Impact of College Entrepreneurial Education On Entrepreneurial Attitudes And Intention To Start A Business in Uganda. *Journal of Developmental Entrepreneurship*, 16(1), 127-144. <u>https://doi.org/10.1142/S1084946711001768</u>

- Bygrave, W. D. (1993). Theory building in the entrepreneurship paradigm. *Journal of Business Venturing*, 8(3), 255-280. <u>https://doi.org/10.1016/0883-9026(93)90031-Y</u>
- Callahan, J. L. (2010). Constructing a Manuscript: Distinguishing Integrative Literature Reviews and Conceptual and Theory Articles. *Human Resource Development Review*, 9(3), 300-304. <u>https://doi.org/10.1177/1534484310371492</u>
- Campanella, F., & Serino, L. (2019). Gender and Financial Constraints: An Empirical Investigation in Italy. *International Journal of Financial Research*, 10(2), 109-120. <u>https://doi.org/10.5430/ijfr.v10n2p109</u>
- Cao, Z., & Zhou, M. (2018). Research on the Innovation and Entrepreneurship Education Mode in Colleges and Universities Based on Entrepreneurial Ecosystem Theory. *Kuram ve Uygulamada Egitim Bilimleri*, 18(5), 1612-1619. <u>https://doi.org/10.12738/estp.2018.5.060</u>
- Carayannis, E., Grigoroudis, E., Campbell, D., Meissner, D., & Stamati, D. (2018). 'Mode 3' universities and academic firms: thinking beyond the box trans-disciplinarity and nonlinear innovation dynamics within coopetitive entrepreneurial ecosystems. *International Journal of Technology Management*, 77(1-3), 145-185. https://doi.org/10.1504/IJTM.2018.091714
- Carayannis, E. G., & Campbell, D. F. J. (2009). 'Mode 3' and 'Quadruple Helix': toward a 21st century fractal innovation ecosystem. *International Journal of Technology Management*, 46(3-4), 201-234. <u>https://doi.org/10.1504/IJTM.2009.023374</u>
- Carlos, B., Hannah, G., Arifusalam, S., Katie, G., Sean, F., Ali, A., Simon, Y., Chantel, F., James, R., & Nafisa, B. (2020). A systematic literature review of the influence of the university's environment and support system on the precursors of social entrepreneurial intention of students. *Journal of innovation and entrepreneurship*, 9(1), 1-28. <u>https://doi.org/10.1186/s13731-020-0116-9</u>
- Carlsson, B., Braunerhjelm, P., McKelvey, M., Olofsson, C., Persson, L., & Ylinenpää, H. (2013). The evolving domain of entrepreneurship research. *Small Business Economics*, 41(4), 913-930. <u>https://doi.org/10.1007/s11187-013-9503-y</u>
- Carpenter, S. (2018). Ten Steps in Scale Development and Reporting: A Guide for Researchers. *Communication methods and measures*, 12(1), 25-44. https://doi.org/10.1080/19312458.2017.1396583
- Cavallo, A., Ghezzi, A., & Balocco, R. (2019). Entrepreneurial ecosystem research: present debates and future directions. *International Entrepreneurship and Management Journal*, 15(4), 1291-1321. <u>https://doi.org/10.1007/s11365-018-0526-3</u>
- Capelleras, J., Contin-Pilart, I., Larraza-Kintana, M., & Martin-Sanchez, V. (2018). Entrepreneurs' human capital and growth aspirations: the moderating role of regional entrepreneurial culture. *Small Business Economics*, 52(1), 3-25. <u>https://doi.org/10.1007/s11187-017-9985-0</u>

- Cervantes, M. (2017). Higher Education Institutions in the Knowledge Triangle. *Foresight* and STI Governance, 11(2), 27-42. <u>https://doi.org/10.17323/2500-2597.2017.2.27.42</u>
- Ceptureanu, E. G. (2016). How to enhance entrepreneurship in universities. Bucharest university of economic studies case study. *Analele Universității din Oradea-Științe Economice*, 25(1), 881-888.
- Chander, N., Siow, M. L., Ramachandran, S., Kunasekaran, P., & Rathakrishnan, T. (2020). Conceptualizing Inclusive Learning and Development: A Framework towards Entrepreneurial Competency Practices for Sustainability. *Sustainability (Basel, Switzerland)*, 12(17), 1-20. https://doi.org/10.3390/su12176905
- Chandler, G. N., DeTienne, D. R., McKelvie, A., & Mumford, T. V. (2011). Causation and effectuation processes: A validation study. *Journal of Business Venturing*, 26(3), 375-390. <u>https://doi.org/10.1016/j.jbusvent.2009.10.006</u>
- Chang, V., & Guetl, C. (2007). E-Learning Ecosystem (ELES) A Holistic Approach for the Development of more Effective Learning Environment for Small-and-Medium Sized Enterprises (SMEs). Proceedings of 2007 Inaugural IEEE-IES Digital Ecosystem and Technologies Conference. (pp. 420-425). IEEE.
- Chang, Y.-C., Yang, P. Y., Martin, B. R., Chi, H.-R., & Tsai-Lin, T.-F. (2016). Entrepreneurial universities and research ambidexterity: *A multilevel analysis*. *Technovation*, 54, 7-21. https://doi.org/10.1016/j.technovation.2016.02.006
- Chavoushi, Z. H., Zali, M. R., Valliere, D., Faghih, N., Hejazi, R., & Dehkordi, A. M. (2020). Entrepreneurial alertness: a systematic literature review. *Journal of Small Business* and Entrepreneurship, 33(2), 123-152. <u>https://doi.org/10.1080/08276331.2020.1764736</u>
- Chen, S. (2019). Research on Innovation Ecosystem in Colleges and Universities-Focus on Chongqing University of Arts and Sciences. *Frontiers in Educational Research*, 2(8), 36-41.
- CIA. (2020). Sri Lanka. Central Intelligence Agency. Retrieved from https://www.cia.gov/library/publications/the-world-factbook/geos/ce.html
- Clauss, T., Moussa, A., & Kesting, T. (2018). Entrepreneurial university: a stakeholder-based conceptualisation of the current state and an agenda for future research. *International Journal of Technology Management*, 77(1-3), 109-144. <u>https://doi.org/10.1504/IJTM.2018.091726</u>
- Clouse, V. G. H. (1990). A controlled experiment related entrepreneurial education to students' start-up decisions. *Journal of Small Business Management*, 28(2), 45-53.

- Collins, C. S., & Stockton, C. M. (2018). The Central Role of Theory in Qualitative Research. *International Journal of Qualitative Methods*, 17(1), 1-10. <u>https://doi.org/10.1177/1609406918797475</u>
- Collins, L., Hannon, P. D., & Smith, A. (2004). Enacting entrepreneurial intent: the gaps between student needs and higher education capability. *Education* + *Training*, 46(8/9), 454-463. <u>https://doi.org/10.1108/00400910410569579</u>
- Collinson, E., & Shaw, E. (2001). Entrepreneurial marketing a historical perspective on development and practice. *Management Decision*, 39(9), 761-766. https://doi.org/10.1108/EUM000000006221
- Colombo, M., Dagnino, G., Lehmann, E., & Salmador, M. (2019). The governance of entrepreneurial ecosystems. *Small Business Economics*, 52(2), 419-428. <u>https://doi.org/10.1007/s11187-017-9952-9</u>
- Comfort, O. C., & Diseph, O. P. (2018). Availability and Utilization Of Instructional Resources For Teaching Entrepreneurship In Business Education. Retrieved from <u>http://www.erint.savap.org.pk/PDF/Vol.7.4/ERInt.2018-7.4-06.pdf</u>
- Cope, J. (2005). Toward a Dynamic Learning Perspective of Entrepreneurship. *Entrepreneurship Theory and Practice*, 29(4), 373-397. <u>https://doi.org/10.1111/j.1540-6520.2005.00090.x</u>
- Corbett, A., Kreiser, P., Marino, L., & Wales, W. (2021). The Future of Entrepreneurial Orientation research. In *Entrepreneurial Orientation: Epistemological Theoretical, and Empirical Perspectives*, (22, pp.1-16). Emerald Publishing Limited.
- Cornelius, B., Landström, H., & Persson, O. (2006). Entrepreneurial Studies: The Dynamic Research Front of a Developing Social Science. *Entrepreneurship Theory and Practice*, 30(3), 375-398. <u>https://doi.org/10.1111/j.1540-6520.2006.00125.x</u>
- Costin, Y., O'Brien, M. P., & Hynes, B. (2019). Developing Cognitive and Non-Cognitive Entrepreneurial Competences through Business Simulation Games. *Journal of Enterprising Culture*, 27(4), 471-498. <u>https://doi.org/10.1142/S0218495819500171</u>
- Creswell, J., & Creswell, J. (2018) *Research design: Qualitative, Quantitative and Mixed Methods Approaches.* London: SAGE Publications.
- Cronin, M., & George, E. (2020). The Why and How of the Integrative Review. *Organizational Research Methods*. 1-25.
- Crow, M. M., Whitman, K., & Anderson, D. M. (2020). Rethinking academic entrepreneurship: university governance and the emergence of the academic enterprise. *Public Administration Review*, 80(3), 511-515.
- Crotty, M. (1998). The foundations of social research: meaning and perspective in the research process. London: SAGE Publications.

- Cumberland, D., Peake, W., Kerrick, S., & Lapadat-Tapolsky, M. (2020). LaunchIt: A Case Study of Entrepreneurship Education. *Small Business Institute*, 34.
- Cunningham, J., Lehmann, E., & Menter, M. (2021). The organizational architecture of entrepreneurial universities across the stages of entrepreneurship: a conceptual framework. *Small Business Economics*, *59*(1), 11-27.
- Davey, T., Plewa, C., & Struwig, M. (2011). Entrepreneurship perceptions and career intentions of international students. *Education* + *Training*, 53(5), 335-352. <u>https://doi.org/10.1108/00400911111147677</u>
- Davidsson, P. Researching Entrepreneurship (Vol. 5). Boston: Springer. https://doi.org/10.1007/b100548
- Davidsson, P. (1991). Continued entrepreneurship: Ability, need, and opportunity as determinants of small firm growth. *Journal of Business Venturing*, 6(6), 405-429. https://doi.org/https://doi.org/10.1016/0883-9026(91)90028-C
- Davidsson, P. (2015). Entrepreneurial opportunities and the entrepreneurship nexus: A reconceptualization. *Journal of Business Venturing*, 30(5), 674-695. <u>https://doi.org/10.1016/j.jbusvent.2015.01.002</u>
- Davidsson, P., & Honig, B. (2003). The role of social and human capital among nascent entrepreneurs. *Journal of Business Venturing*, 18(3), 301-331. <u>https://doi.org/10.1016/S0883-9026(02)00097-6</u>
- De Clercq, D., Lim, D. S. K., & Oh, C. H. (2013). Individual-Level Resources and New Business Activity: The Contingent Role of Institutional Context. *Entrepreneurship Theory and Practice*, 37(2), 303-330. <u>https://doi.org/10.1111/j.1540-</u> <u>6520.2011.00470.x</u>
- De Jager, H. J., Mthembu, T. Z., Ngowi, A. B., & Chipunza, C. (2017). Towards an Innovation and Entrepreneurship Ecosystem: A Case Study of the Central University of Technology, Free State. *Science, Technology & Society*, 22(2), 310-331. <u>https://doi.org/10.1177/0971721817702292</u>
- de Moraes, G., Fischer, B., Campos, M., & Schaeffer, P. (2020). University Ecosystems and the Commitment of Faculty Members to Support Entrepreneurial Activity. *Brazilian Administration Review*, 17(2), 1-26. <u>https://doi.org/10.1590/1807-</u> <u>7692bar2020190013</u>
- de Silva, L. R., Uyarra, E., & Oakey, R. (2012). Academic Entrepreneurship in a Resource-Constrained Environment: Diversification and Synergistic Effects. In D. B Audretsch., E. E Lehmann., A. N Link & A. Stanrnecker (Eds.), *Technology transfer in a global economy (pp.73-97)*. Boston: Springer.
- Deakins, D., Sullivan, R., & Whittam, G. (2002). Developing support for entrepreneurial learning: evidence from start-up support programs. *International Journal of Entrepreneurship and Innovation Management*, 2(4-5), 323-338. <u>https://doi.org/10.1504/ijeim.2002.000488</u>

- Dean, M. A., Shook, C. L., & Payne, G. T. (2007). The Past, Present, and Future of Entrepreneurship Research: Data Analytic Trends and Training1. *Entrepreneurship Theory and Practice*, 31(4), 601-618. <u>https://doi.org/10.1111/j.1540-6520.2007.00190.x</u>
- Decker-Lange, C., Lange, K., Dhaliwal, S., & Walmsley, A. (2021). Exploring Entrepreneurship Education Effectiveness at British Universities – An Application of the World Café Method. *Entrepreneurship Education and Pedagogy*, 5(1), 113-136. <u>https://doi.org/10.1177/2515127420935391</u>
- Degl'Innocenti, M., Matousek, R., & Tzeremes, N. G. (2019). The interconnections of academic research and universities' "third mission": Evidence from the UK. *Research Policy*, 48(9), 1-13. <u>https://doi.org/10.1016/j.respol.2019.05.002</u>
- Delanoë-Gueguen, S., & Theodoraki, C. (2021). From incubator to full internal entrepreneurship education ecosystem: the example of TBS. In H. Neck & Y. Liu (Eds.), *Innovation in global entrepreneurship education: teaching entrepreneurship in* practice (pp. 18-31). Cheltenham: Edward Elgar Publishing.
- Denzin, N & Lincoln, Y. (2017). *The SAGE handbook of qualitative research*. London: SAGE Publications.
- Dess, G. G., & Lumpkin, G. T. (2005). The role of entrepreneurial orientation in stimulating effective corporate entrepreneurship. *The Academy of Management executive*, 19(1), 147-156. <u>https://doi.org/10.5465/ame.2005.15841975</u>
- Devetag, M. G., Zazzerini, G., Tuan, N. Q., & Hung, D. Q. (2020). Developing Entrepreneurial Competencies in Vietnam: Evidence from the Bac Ninh Province. *Journal of Entrepreneurship and Innovation in Emerging Economies*, 6(2), 329-346. <u>https://doi.org/10.1177/2393957520924987</u>
- Dew, N., Read, S., Sarasvathy, S. D., & Wiltbank, R. (2009). Effectual versus predictive logics in entrepreneurial decision-making: Differences between experts and novices. *Journal of Business Venturing*, 24(4), 287-309. https://doi.org/10.1016/j.jbusvent.2008.02.002
- Di Gregorio, D., & Shane, S. (2003). Why do some universities generate more start-ups than others? *Research Policy*, 32(2), 209-227. <u>https://doi.org/10.1016/S0048-7333(02)00097-5</u>
- Dickson, P. H., Solomon, G. T., & Weaver, K. M. (2008). Entrepreneurial selection and success: does education matter? *Journal of Small Business and Enterprise Development*, 15(2), 239-258. <u>https://doi.org/10.1108/14626000810871655</u>
- DiMaggio, P. J. (1995). Comments on "What Theory is Not". Administrative science quarterly, 40(3), 391-397. <u>https://doi.org/10.2307/2393790</u>
- Discua Cruz, A., Howorth, C., & Hamilton, E. (2013). Intrafamily Entrepreneurship: The Formation and Membership of Family Entrepreneurial Teams. *Entrepreneurship Theory and Practice*, 37(1), 17-46. <u>https://doi.org/10.1111/j.1540-6520.2012.00534.x</u>

- Dissanayake, M. (2020). Explaining Mismatch between Labor Supply and Demand in Sri Lanka. The Word Bank Group. Retrieved from <u>http://documents1.worldbank.org/curated/en/601981591360828247/pdf/Explaining-Mismatch-Between-Labor-Supply-and-Demand-in-Sri-Lanka-Youth-and-Gender-Qualitative-Study.pdf</u>
- Donald, W. E., Ashleigh, M. J., & Baruch, Y. (2018). Students' perceptions of education and employability. *Career Development International*, 23(5), 513-540. https://doi.org/10.1108/CDI-09-2017-0171
- Douglas, E. J., & Shepherd, D. A. (2002). Self-Employment as a Career Choice: Attitudes, Entrepreneurial Intentions, and Utility Maximization. *Entrepreneurship Theory and Practice*, 26(3), 81-90. <u>https://doi.org/10.1177/104225870202600305</u>
- Draycott, M., & Rae, D. (2011). Enterprise education in schools and the role of competency frameworks. *International Journal of Entrepreneurial Behaviour & Research*, 17(2), 127-145. <u>https://doi.org/10.1108/13552551111114905</u>
- Drazin, R., 1985. Innovation and entrepreneurship: Practice and principles, by Peter F. Drucker. New York: Harper & amp; Row, 277 pp., \$19.95. *Human Resource Management*, 24(4), 509-512.
- Drucker, P. F. (1985). Innovation and entrepreneurship. Oxford: Butterworth-Heinemann.
- Du, K., & O'Connor, A. (2018). Entrepreneurship and advancing national level economic efficiency. Small Business Economics, 50(1), 91-111. <u>https://doi.org/10.1007/s11187-017-9904-4</u>
- Duruflé, G., Hellmann, T., & Wilson, K. (2018). Catalysing entrepreneurship in and around universities. Oxford review of economic policy, 34(4), 615-636. <u>https://doi.org/10.1093/oxrep/gry015</u>
- Easterby-Smith, M., Thorpe, R., Jackson, P., & Lowe, A. (2008). *Management Research*. London: SAGE Publications.
- Edwards-Schachter, M., García-Granero, A., Sánchez-Barrioluengo, M., Quesada-Pineda, H., & Amara, N. (2015). Disentangling competences: Interrelationships on creativity, innovation and entrepreneurship. *Thinking skills and creativity*, 16(2015), 27-39. <u>https://doi.org/10.1016/j.tsc.2014.11.006</u>
- Elsbach, K. D., & Knippenberg, D. v. (2020). Creating High-Impact Literature Reviews: An Argument for 'Integrative Reviews'. *Journal of Management Studies*, 57(6), 1277-1289. <u>https://doi.org/10.1111/joms.12581</u>
- Engle, R. L., Dimitriadi, N., Gavidia, J. V., Schlaegel, C., Delanoe, S., Alvarado, I., He, X., Buame, S., & Wolff, B. (2010). Entrepreneurial intent: A twelve-country evaluation of Ajzen's model of planned behavior. *International Journal of Entrepreneurial Behaviour & Research*, 16(1), 35-57. <u>https://doi.org/10.1108/13552551011020063</u>

- Ertuna, Z., & Gurel, E. (2011). The moderating role of higher education on entrepreneurship. Education + Training, 53(5), 387-402. <u>https://doi.org/10.1108/00400911111147703</u>
- Espinoza, C., Mardones, C., Sáez, K., & Catalán, P. (2019). Entrepreneurship and regional dynamics: the case of Chile. *Entrepreneurship & Regional Development*, 31(9-10), 1-13. <u>https://doi.org/10.1080/08985626.2019.1565421</u>
- Etzkowitz, H. (2002). Incubation of incubators: innovation as a triple helix of universityindustry-government networks. *Science and Public Policy*, 29(2), 115-128. <u>https://doi.org/10.3152/147154302781781056</u>
- Etzkowitz, H. (2011). The triple helix: science, technology and the entrepreneurial spirit. *Journal of Knowledge-based Innovation in China*, 3(2), 76-90. <u>https://doi.org/10.1108/1756141111138937</u>
- Etzkowitz, H. (2013). Anatomy of the entrepreneurial university. *Social Science Information*, 52(3), 486-511. <u>https://doi.org/10.1177/0539018413485832</u>
- Etzkowitz, H. (2014). The Entrepreneurial University Wave: From Ivory Tower to Global Economic Engine. *Industry and Higher Education*, 28(4), 223-232. <u>https://doi.org/10.5367/ihe.2014.0211</u>
- Etzkowitz, H., & Leydesdorff, L. (2000). The dynamics of innovation: from National Systems and "Mode 2" to a Triple Helix of university–industry–government relations. *Research Policy*, 29(2), 109-123. <u>https://doi.org/10.1016/S0048-7333(99)00055-4</u>
- Eze, J. F., & Nwali, A. C. (2012). Capacity Building For Entrepreneurship Education: The Challenge For The Developing Nations. *American Journal of Business Education*, 5(4), 401-408. <u>https://doi.org/10.19030/ajbe.v5i4.7117</u>
- Fallah Haghighi, N., Mahmoudi, M., & Bijani, M. (2018). Barriers to Entrepreneurship Development in Iran's Higher Education: A Qualitative Case Study. A Quarterly Review of Education, 49(3), 353-375. <u>https://doi.org/10.1007/s10780-018-9330-9</u>
- Fayolle, A. (2013). Personal views on the future of entrepreneurship education. *Entrepreneurship & Regional Development*, 25(7-8), 692-701. <u>https://doi.org/10.1080/08985626.2013.821318</u>
- Fayolle, A., & Gailly, B. (2008). From craft to science: Teaching models and learning processes in entrepreneurship education. *Journal of European Industrial Training*, 32(7), 569-593. <u>https://doi.org/10.1108/03090590810899838</u>
- Fayolle, A., Gailly, B., & Lassas-Clerc, N. (2006). Assessing the impact of entrepreneurship education programmes: a new methodology. *Journal of European Industrial Training*, 30(9), 701-720. <u>https://doi.org/10.1108/03090590610715022</u>
- Fayolle, A., Verzat, C., & Wapshott, R. (2016). In quest of legitimacy: The theoretical and methodological foundations of entrepreneurship education research. *International Small Business Journal*, 34(7), 895-904. <u>https://doi.org/10.1177/0266242616649250</u>

- Feinstein, A. H., & Cannon, H. M. (2002). Constructs of Simulation Evaluation. *Simulation & Gaming*, 33(4), 425-440. <u>https://doi.org/10.1177/1046878102238606</u>
- Feldman, M., Siegel, D. S., & Wright, M. (2019). New developments in innovation and entrepreneurial ecosystems. *Industrial and Corporate Change*, 28(4), 817-826.
- Feld, B. (2012). Startup Communities: Building an Entrepreneurial Ecosystem in Your City. New Jersey: John Wiley & Sons.
- Fellnhofer, K. (2018). Narratives boost entrepreneurial attitudes: Making an entrepreneurial career attractive? *European Journal of Education*, 53(2), 218-237. <u>https://doi.org/10.1111/ejed.12274</u>
- Ferrandiz, J., Fidel, P., & Conchado, A. (2018). Promoting entrepreneurial intention through a higher education program integrated in an entrepreneurship ecosystem. *International Journal of Innovation Science*, 10(1), 6-21.
- Ferreira, J. (2020). Emerging Perspectives on Regional Academic Entrepreneurship. Higher Education Policy, 33(2), 367-395. <u>https://doi.org/10.1057/s41307-018-0099-3</u>
- Ferreira, J., Paço, A., Raposo, M., Hadjichristodoulou, C., & Marouchou, D. (2021). International entrepreneurship education: Barriers versus support mechanisms to STEM students. *Journal of International Entrepreneurship*, 19(1), 130-147. <u>https://doi.org/10.1007/s10843-020-00274-4</u>
- Fetters, M. (2019). The Mixed Methods Research Workbook Activities for Designing, Implementing, and Publishing Projects. London: SAGE Publications.
- Fetters, M. L., Greene, P. G., & Rice, M. P. (2010). The Development of University-based Entrepreneurship Ecosystems: Babson College. Cheltenham: Edward Elgar Publishing.
- Fiet, J. (2001). The pedagogical side of entrepreneurship theory. *Journal of Business Ventures*, 16(2), 101-117. <u>https://doi.org/10.1016/S0883-9026(99)00042-7</u>
- Fiet, J. (2001). The theoretical side of teaching entrepreneurship. *Journal of Business Ventures*, 16(1), 1-24. <u>https://doi.org/10.1016/S0883-9026(99)00041-5</u>
- Fiet, J., & Patel, P. (2008). Entrepreneurial Discovery as Constrained, Sytematic Search. Small Business Economics, 30(3), 215-229. <u>https://doi.org/10.1007/s11187-006-9010-5</u>
- Finkle, T. A., Menzies, T. V., Kuratko, D. F., & Goldsby, M. G. (2013). An examination of the financial challenges of entrepreneurship centers throughout the world. *Journal of Small Business & Entrepreneurship*, 26(1), 67-85. https://doi.org/10.1080/08276331.2012.761803

- Fischer, B. B., Queiroz, S., & Vonortas, N. S. (2018). On the location of knowledge-intensive entrepreneurship in developing countries: lessons from São Paulo, Brazil. *Entrepreneurship & Regional Development*, 30(5-6), 612-638. <u>https://doi.org/10.1080/08985626.2018.1438523</u>
- Fisher, S., Graham, M., & Compeau, M. (2008). Understanding the learning outcomes of undergraduate entrepreneurship education. In R. Harrison & C. Leitch (Eds.), *Entrepreneurial Learning: Conceptual Frameworks and Applications*. Oxfordshire: Taylor & Francis Group. <u>https://doi.org/10.4324/9780203931929</u>
- Fox, J., Pittaway, L., & Uzuegbunam, I. (2018). Simulations in Entrepreneurship Education: Serious Games and Learning Through Play. *Entrepreneurship Education and Pedagogy*, 1(1), 61-89. <u>https://doi.org/10.1177/2515127417737285</u>
- Frank, H., & Landström, H. (2016). What makes entrepreneurship research interesting? Reflections on strategies to overcome the rigour-relevance gap. *Entrepreneurship and Regional Development*, 28(1-2), 51-75. <u>https://doi.org/10.1080/08985626.2015.1100687</u>
- Galan-Muros, V., & Davey, T. (2019). The UBC ecosystem: putting together a comprehensive framework for university-business cooperation. *The Journal of Technology Transfer*, 44(4), 1311-1346. <u>https://doi.org/10.1007/s10961-017-9562-3</u>
- Galán-Muros, V., Sijde, P., Groenewegen, P., & Baaken, T. (2017). Nurture over nature: How do European universities support their collaboration with business? *The Journal* of Technology Transfer, 42(1), 184-205. <u>https://doi.org/10.1007/s10961-015-9451-6</u>
- Galvao, A. (2019). Triple helix and its evolution: a systematic literature review. *Journal of Science and Technology Policy Management*, 10(3), 812-833. https://doi.org/10.1108/JSTPM-10-2018-0103
- Galvão, A., Marques, C., & Ferreira, J. (2019). Evaluation of an entrepreneurship training programme: a proposal for new guidelines. *Education* + *Training*, *61*(2), 136-152. <u>https://doi.org/10.1108/et-11-2018-0228</u>
- Galvão, A., Marques, C., Ferreira, J., & Braga, V. (2020). Stakeholders' role in entrepreneurship education and training programmes with impacts on regional development. *Journal of Rural Studies*, 74(2020), 169-179. <u>https://doi.org/10.1016/j.jrurstud.2020.01.013</u>
- Gamage, S. (2012). Comparing Choices in Higher Education. Retrieved from <u>https://lirneasia.net/2012/06/higher-education-landscape-in-sri-lanka-presented-at-the-</u> <u>ceylon-chamber-of-commerce/</u>
- Gamage, S., & Wijesooriya, T. (2011). Mapping the Higher Education Landscape in Sri Lanka. Retrieved from <u>https://lirneasia.net/wp-content/uploads/2012/06/HE-HEI_Survey2012June27_PPT.pdf</u>

- Garba, M., & Abubakar, S. (2019). The Influence of Entrepreneurial Learning Environment and Intrinsic Learners' Need on Entrepreneurship Education. Open Journal of Business and Management, 07(03), 1244-1261. https://doi.org/10.4236/ojbm.2019.73087
- Gartner, W. (1985). A Conceptual Framework for Describing the Phenomenon of New Venture Creation. *Academy Of Management Review*, *10*(4), 696-706. <u>https://doi.org/10.5465/amr.1985.4279094</u>
- Gartner, W. B. (1988). "Who is an Entrepreneur?" Is the Wrong Question [Article]. American Journal of Small Business, 12(4), 11-32. https://doi.org/10.1177/104225878801200401
- Gartner, W. B., Shaver, K. G., Gatewood, E., & Katz, J. A. (1994). Finding the Entrepreneur in Entrepreneurship. *Entrepreneurship Theory and Practice*, 18(3), 5-9. <u>https://doi.org/10.1177/104225879401800301</u>
- Gartner, W. B., & Vesper, K. H. (1994). Experiments in entrepreneurship education: Successes and failures. *Journal of Business Venturing*, 9(3), 179-187. <u>https://doi.org/10.1016/0883-9026(94)90028-0</u>
- Gehman, J., Glaser, V. L., Eisenhardt, K. M., Gioia, D., Langley, A., & Corley, K. G. (2018). Finding Theory–Method Fit: A Comparison of Three Qualitative Approaches to Theory Building. *Journal of Management Inquiry*, 27(3), 284-300. <u>https://doi.org/10.1177/1056492617706029</u>
- Geuna, A., & Muscio, A. (2009). The Governance of University Knowledge Transfer: A Critical Review of the Literature. A Review of Science, Learning and Policy, 47(1), 93-114. <u>https://doi.org/10.1007/s11024-009-9118-2</u>
- Giacomin, O., Janssen, F., & Shinnar, R. S. (2016). Student entrepreneurial optimism and overconfidence across cultures. *International Small Business Journal*, 34(7), 925-947. <u>https://doi.org/10.1177/0266242616630356</u>
- Gibb, A. (2002). In pursuit of a new 'enterprise' and 'entrepreneurship' paradigm for learning: creative destruction, new values, new ways of doing things and new combinations of knowledge. International *Journal of Management Reviews*, 4(3), 233-269. <u>https://doi.org/10.1111/1468-2370.00086</u>
- Gibb, A. (2005). The future of entrepreneurship education–Determining the basis for coherent policy and practice. *The dynamics of learning entrepreneurship in a cross-cultural university context*, 2(2005), 44-67.
- Gibb, A., Haskins, G., & Robertson, I. (2013). Leading the Entrepreneurial University: Meeting the Entrepreneurial Development Needs of Higher Education Institutions. In: Altmann, A., Ebersberger, B. (eds) Universities in Change. Innovation, Technology, and Knowledge Management. New York: Springer.

- Gibb, A. (1996). Entrepreneurship and small business management: can we afford to neglect them in the twenty-first century business school?. *British Journal of management*, 7(4), 309-321.
- Gioia, D., Corley, K., & Hamilton, A. (2012). Seeking Qualitative Rigor in Inductive Research. Organizational Research Methods, 16(1), 15-31. <u>https://doi.org/10.1177/1094428112452151</u>
- Giones, F., Brem, A., Pollack, J. M., Michaelis, T. L., Klyver, K., & Brinckmann, J. (2020). Revising entrepreneurial action in response to exogenous shocks: Considering the COVID-19 pandemic. *Journal of Business Venturing Insights*, 14(2020), 1-7. <u>https://doi.org/10.1016/j.jbvi.2020.e00186</u>
- Gnyawali, D., & Fogel, D. (1994). Environments for Entrepreneurship Development: Key Dimensions and Research Implications. *Entrepreneurship Theory and Practice*, 18(4), 43-62. <u>https://doi.org/10.1177/104225879401800403</u>
- Godley, A., Morawetz, N., & Soga, L. (2019). The complementarity perspective to the entrepreneurial ecosystem taxonomy. *Small Business Economics*, *56*(2), 723-738. <u>https://doi.org/10.1007/s11187-019-00197-y</u>
- Golden-Biddle, K., & Locke, K. (2007). *Composing qualitative research* (2nd ed.). London: SAGE Publications
- González-Serrano, M. H., Valantine, I., Hervás, J. C., Pérez-Campos, C., & Moreno, F. C. (2018). Sports university education and entrepreneurial intentions. *Education* + *Training*, 60(5), 389-405.
- Gordashnikova, O., Kocherjagina, N., Odintsova, T., & Fedorchuk, Y. (2020). Methods Toolkit for Estimating the University Ecosystem as an Innovative Form for Upgrading Education Quality. *Calitatea*, 21(174), 76-80.
- Gorman, G., Hanlon, D., & King, W. (1997). Some Research Perspectives on Entrepreneurship Education, Enterprise Education and Education for Small Business Management: A Ten-Year Literature Review. *International Small Business Journal*, 15(3), 56-77. <u>https://doi.org/10.1177/0266242697153004</u>
- Gomesde, L., Facin, A, & Salerno M. (2018). Unpacking the innovation ecosystem construct: Evolution, gaps and trends. *Technological Forecasting and Social Change*, 136 (2018), 30–48.
- Grant, C., & Osanloo, A. (2014). Understanding, Selecting, and Integrating a Theoretical Framework in Dissertation Research: Creating the Blueprint for Your "House". *Administrative issues journal: education, practice, and research,* 4(2), 12-26. <u>https://doi.org/10.5929/2014.4.2.9</u>
- Grbich, C. (2013). Qualitative data analysis: an introduction (2nd ed.). London: SAGE Publications.

- Greene, F. J., Mole, K. F., & Storey, D. J. (2004). Does More Mean Worse? Three Decades of Enterprise Policy in the Tees Valley. *Urban Studies*, 41(7), 1207-1228.
- Grégoire, D. A., & Cherchem, N. (2020). A structured literature review and suggestions for future effectuation research. *Small Business Economics*, 54(3), 621-639.
- Grimaldi, R. (2011). Special issue: 30 years after Bayh-Dole; reassessing academic entrepreneurship. Amsterdam: Elsevier.
- Guba, E. (1990). The paradigm dialog. London: SAGE Publications
- Guelich, U., & Bosma, N. (2018). *Youth Entrepreneurship in Asia and the Pacific 2019*. United Nations Development Programme.
- Guerrero, M., Liñán, F., & Cáceres-Carrasco, R. (2020). The influence of ecosystems on the entrepreneurship process: a comparison across developed and developing economies. *Small Business Economics*, 57(4), 1733-1759. <u>https://doi.org/10.1007/s11187-020-00392-2</u>
- Guerrero, M., Rialp, J., & Urbano, D. (2008). The impact of desirability and feasibility on entrepreneurial intentions: A structural equation model. *International Entrepreneurship and Management Journal*, 4(1), 35-50. <u>https://link.springer.com/article/10.1007/s11365-006-0032-x</u>
- Guerrero, M., & Urbano, D. (2012). The development of an entrepreneurial university. *The Journal of Technology Transfer*, 37(1), 43-74. <u>https://doi.org/10.1007/s10961-010-</u> <u>9171-x</u>
- Guerrero, M., & Urbano, D. (2014). Academics' start-up intentions and knowledge filters: an individual perspective of the knowledge spillover theory of entrepreneurship. *Small Business Economics*, 43(1), 57-74. <u>https://www.jstor.org/stable/43553734</u>
- Guerrero, M., Urbano, D., Cunningham, J., & Organ, D. (2012). Entrepreneurial universities in two European regions: a case study comparison. *The Journal of Technology Transfer*, 39(3), 415-434. <u>https://doi.org/10.1007/s10961-012-9287-2</u>
- Guerrero, M., Urbano, D., & Fayolle, A. (2016). Entrepreneurial activity and regional competitiveness: evidence from European entrepreneurial universities. *The Journal of Technology Transfer*, 41(1), 105-131. https://link.springer.com/article/10.1007/s10961-014-9377-4
- Guerrero, M., Urbano, D., Fayolle, A., Klofsten, M., & Mian, S. (2016). Entrepreneurial universities: emerging models in the new social and economic landscape. *Small Business Economics*, 47(3), 551-563. <u>https://doi.org/10.1007/s11187-016-9755-4</u>
- Guerrero, M., Urbano, D., & Gajón, E. (2020). Entrepreneurial university ecosystems and graduates' career patterns: do entrepreneurship education programmes and university business incubators matter?. *Journal of Management Development*, 39(5), 175-202. <u>https://doi.org/10.1108/jmd-10-2019-0439</u>

- Guerrero, M., Urbano, D., & Gajón, E. (2020). Entrepreneurial university ecosystems and graduates' career patterns: do entrepreneurship education programmes and university business incubators matter?. *Journal of Management Development*, 39(5), 753-775. <u>https://doi.org/10.1108/jmd-10-2019-0439</u>
- Gunawardane, H. N. R., & Weerasinghe, R. N. (2021). Undergraduate entrepreneurial intention of University of Sri Jayewardenepura. *Journal of Women Entrepreneurship and Business Management*, 2(1), 7-15.
- Gulzar, F., & Fayaz, A. (2016). Fostering Entrepreneurship Education among Students–A Study. *Advances in Economics and Business Management*, 3(4), 345-348. <u>https://www.sciencedirect.com/science/article/pii/S2405844022005023</u>
- Gunasekara, S. (2020). The importance of business incubators for emergence of new businesses. Retrieved from <u>http://www.ft.lk/columns/The-importance-of-business-incubators-for-emergence-of-new-businesses/4-694137</u>
- Gunatlilaka, R., Mayer, M., & Vodopivec, M. (Eds.). (2010). *The challenge of youth employment in Sri Lanka*. World Bank publications.
- Gürol, Y., & Atsan, N. (2006). Entrepreneurial characteristics amongst university students: Some insights for entrepreneurship education and training in Turkey. *Education* + *Training*, 48(1), 25-38. <u>https://doi.org/10.1108/00400910610645716</u>
- Haase, H., & Lautenchlager, A. (2012). The "teachability dilemma" of entrepreneurship. *IEEE Engineering Management Review*, 40(2), 131-147. <u>https://doi.org/10.1109/emr.2012.6210522</u>
- Haibin, L. (2018). Research on the Construction Strategy of Entrepreneurship Education Ecosystem in University. *China Higher Education Research* (2), 7.
- Hager, P., & Gonczi, A. (1996). What is competence? *Medical Teacher*, 18(1), 15–18.
- Haichun, L., & Tingting, X. (2017). Analysis on the Functions of the Innovation and Entrepreneurship Education Institutions and Their Relationships in Colleges and Universities. *Journal of National Academy of Education Administration*, (10), 5.
- Haifeng, Q., & Zoltan, J. A. (2013). An absorptive capacity theory of knowledge spillover entrepreneurship. *Small Business Economics*, 40(2), 185-197. <u>https://doi.org/10.1007/s11187-011-9368-x</u>
- Haijun, W., Yang, X., Sibo, W., Shuo, L., & Zhanyue, G. (2019). Research on the Implementation of Innovation and Entrepreneurship Education in Adult Higher Education. Asia-Pacific Conference on Advance in Education, Learning and Teaching, Malaysia.
- Hall, J., Daneke, G., & Lenox, M. (2010). Sustainable development and entrepreneurship: Past contributions and future directions. *Journal Of Business Venturing*, 25(5), 439-448. <u>https://doi.org/10.1016/j.jbusvent.2010.01.002</u>

- Hancock, S. (2011). The development of university-based entrepreneurship ecosystems: global practices. *Studies In Higher Education*, 36(3), 373-374. https://doi.org/10.1080/03075079.2011.580982
- Hannon, P. (2013). Why is the Entrepreneurial University Important?. *Journal of Innovation Management*, 1(2), 10-17. <u>https://doi.org/10.24840/2183-0606_001.002_0003</u>
- Hannon, P. (2018). On becoming and being an entrepreneurship educator: a personal reflection. *Entrepreneurship & Regional Development*, 30(7-8), 698-721. https://doi.org/10.1080/08985626.2018.1464259
- Hannon, P., Collins, L., & Smith, A. (2005). Exploring Graduate Entrepreneurship. *Industry And Higher Education*, 19(1), 11-23. <u>https://doi.org/10.5367/000000053123600</u>
- Harmeling, S. S., & Sarasvathy, S. D. (2013). When Contingency is a Resource: Educating Entrepreneurs in the Balkans, the Bronx, and Beyond. *Entrepreneurship Theory and Practice*, 37(4), 713-744. <u>https://doi.org/10.1111/j.1540-6520.2011.00489</u>
- Harris, M., & Gibson, S. (2008). Examining the entrepreneurial attitudes of US business students. *Education* + *Training*, 50(7), 568-581. <u>https://doi.org/10.1108/00400910810909036</u>
- Harrison, R., & Leitch, C. (2005). Entrepreneurial Learning: Researching the Interface between Learning and the Entrepreneurial Context. *Entrepreneurship Theory and Practice*, 29(4), 351-371. <u>https://doi.org/10.1111/j.1540-6520.2005.00089.x</u>
- Hayter, C. (2010). In search of the profit-maximizing actor: motivations and definitions of success from nascent academic entrepreneurs. *The Journal of Technology Transfer*, 36(3), 340-352. <u>https://doi.org/10.1007/s10961-010-9196-1</u>
- Hayter, C. (2013). Conceptualizing knowledge-based entrepreneurship networks: perspectives from the literature. *Small Business Economics*, 41(4), 899-911. <u>https://doi.org/10.1007/s11187-013-9512-x</u>
- Hayter, C. (2015). Public or private entrepreneurship? Revisiting motivations and definitions of success among academic entrepreneurs. *The Journal of Technology Transfer*, 40(6), 1003-1015. <u>https://doi.org/10.1007/s10961-015-9426-7</u>
- Hayter, C. (2016). A trajectory of early-stage spinoff success: the role of knowledge intermediaries within an entrepreneurial university ecosystem. *Small Business Economics*, 47(3), 633-656. <u>https://doi.org/10.1007/s11187-016-9756-3</u>
- Hayter, C., Lubynsky, R., & Maroulis, S. (2016). Who is the academic entrepreneur? The role of graduate students in the development of university spinoffs. *The Journal of Technology Transfer*, 42(6), 1237-1254. <u>https://doi.org/10.1007/s10961-016-9470-y</u>
- Hayter, C., Nelson, A., Zayed, S., & O'Connor, A. (2018). Conceptualizing Academic Entrepreneurship Ecosystems: A Review, Analysis and Extension of the Literature. *The Journal of Technology Transfer*, 43(4), 1039-1082.

- Hayward, M., Hunt, R., & Miller, D. (2022). How vulnerability enriches family firm relationships: A social exchange perspective. *Journal of Family Business Strategy*, 13(1), 1-7. <u>https://doi.org/10.1016/j.jfbs.2021.100450</u>
- Heaton, S., Siegel, D. S., & Teece, D. J. (2019). Universities and innovation ecosystems: A dynamic capabilities perspective. *Industrial and Corporate Change*, 28(4), 921-939. <u>https://doi.org/10.1093/icc/dtz038</u>
- Hechavarria, D., Renko, M., & Matthews, C. (2012). The nascent entrepreneurship hub: goals, entrepreneurial self-efficacy and start-up outcomes. *Small Business Economics*, 39(3), 685-701. <u>https://doi.org/10.1007/s11187-011-9355-2</u>
- Hechavarria, D. M., & Ingram, A. (2014). A review of the entrepreneurial ecosystem and the entrepreneurial society in the United States: An exploration with the global entrepreneurship monitor dataset. *Journal of Business and Entrepreneurship*, 26(1), 1-35.
- Henderson, S., & Segal, E. (2013). Visualizing Qualitative Data in Evaluation Research. New Directions for Evaluation, 13(139), 53-71. <u>https://doi.org/10.1002/ev.20067</u>
- Hennink, M., Hutter, I., & Bailey, A. (2020). Qualitative Research Methods. London: SAGE Publications.
- Henry, C., Hill, F. M., & Leitch, C. M. (2004). The Effectiveness of Training for New Business Creation: A Longitudinal Study. *International Small Business Journal*, 22(3), 249-271. <u>https://doi.org/10.1177/0266242604042378</u>
- Hills, G. (1988). Variations in University entrepreneurship education: An empirical study of an evolving field. *Journal Of Business Venturing*, 3(2), 109-122. <u>https://doi.org/10.1016/0883-9026(88)90021-3</u>
- Hindle, K. (2002). A Grounded Theory for Teaching Entrepreneurship Using Simulation Games. Simulation & Gaming, 33(2), 236-241. <u>https://doi.org/10.1177/1046878102332012</u>
- Hindle, K., & Klyver, K. (2007). Exploring the relationship between media coverage and participation in entrepreneurship: Initial global evidence and research implications. *International Entrepreneurship and Management Journal*, 3(2), 217-242. <u>https://doi.org/10.1007/s11365-006-0018-8</u>
- Ho, C. S. M., Lu, J., & Bryant, D. A. (2020). The impact of teacher entrepreneurial behaviour: a timely investigation of an emerging phenomenon. *Journal of Educational Administration*, 58(6), 697-712. <u>https://doi.org/10.1108/JEA-08-2019-0140</u>
- Ho, Y.-P., Singh, A., & Wong, P.-K. (2010). *The Development of University-Based Entrepreneurship Ecosystems: National University of Singapore*. Edward Elgar Publishing Limited, UK.

- Hoang, H., & Antoncic, B. (2003). Network-based research in entrepreneurship: A critical review. Journal of Business Venturing, 18(2), 165-187. https://doi.org/10.1016/S0883-9026(02)00081-2
- Hockerts, K. (2017). Determinants of Social Entrepreneurial Intentions. *Entrepreneurship Theory and Practice*, 41(1), 105-130. <u>https://doi.org/10.1111/etap.12171</u>
- Holcomb, T. R., Ireland, R. D., Holmes, R. M., & Hitt, M. A. (2009). Architecture of Entrepreneurial Learning: Exploring the Link among Heuristics, Knowledge, and Action. *Entrepreneurship Theory and Practice*, 33(1), 167-192. <u>https://doi.org/10.1111/j.1540-6520.2008.00285.x</u>
- Holmgren, C., & From, J. (2005). Taylorism of the Mind: Entrepreneurship Education from a Perspective of Educational Research. *European Educational Research Journal*, 4(4), 382-390. <u>https://doi.org/10.2304/eerj.2005.4.4.4</u>
- Hoppe, M. (2016). Policy and entrepreneurship education. *Small Business Economics*, 46(1), 13-29. <u>https://doi.org/10.1007/s11187-015-9676-7</u>
- Hornaday, R. (1992). Thinking about entrepreneurship: A fuzzy set approach. *Journal of Small Business Management*, 30(4), 12-23.
- Hruskova, M., & Mason, C. (2020). Mapping Entrepreneurship Support Provision in Entrepreneurial Ecosystems. In 2020 Australian Centre for Entrepreneurship Research Exchange Conference, Adelaide, Australia.
- Hsieh, R., & Kelley, D. (2019). A Study of Key Indicators of Development for University-Based Entrepreneurship Ecosystems in Taiwan. *Entrepreneurship Research Journal*, 10(2), 1-17. <u>https://doi.org/10.1515/erj-2018-0331</u>
- Huang-Saad, A., Duval-Couetil, N., & Park, J. (2018). Technology and talent: capturing the role of universities in regional entrepreneurial ecosystems. *Journal of Enterprising Communities: People and Places in the Global Economy*, 12(2), 92-116. <u>https://doi.org/10.1108/JEC-08-2017-0070</u>
- Huang-Saad, A., Fay, J., & Sheridan, L. (2017). Closing the divide: accelerating technology commercialization by catalyzing the university entrepreneurial ecosystem with I-CorpsTM. *The Journal of Technology Transfer*, 42(6), 1466-1486. <u>https://doi.org/10.1007/s10961-016-9531-2</u>
- Huber, L., Sloof, R., & van Praag, M. (2012). The Effect of Early Entrepreneurship Education: Evidence from a Randomized Field Experiment. *IDEAS Working Paper Series*. <u>https://doi.org/10.2139/ssrn.2042058</u>
- Huebscher, J., & Lendner, C. (2010). Effects of Entrepreneurship Simulation Game Seminars on Entrepreneurs' and Students' Learning. *Journal of Small Business & Entrepreneurship*, 23(4), 543-554. <u>https://doi.org/10.1080/08276331.2010.10593500</u>

- Huezo-Ponce, L., Fernández-Pérez, V., & Rodríguez-Ariza, L. (2020). Emotional competencies and entrepreneurship: modeling universities. *International Entrepreneurship and Management Journal*, 17(3), 1497-1519. <u>https://doi.org/10.1007/s11365-020-00683-w</u>
- Huff, A. S. (2008). Designing Research for Publication. London: SAGE Publications.
- Huyghe, A., Knockaert, M., & Obschonka, M. (2016). Unraveling the "passion orchestra" in academia. *Journal of Business Venturing*, 31(3), 344-364. https://doi.org/10.1016/j.jbusvent.2016.03.002
- Hytti, U. (2021). *A research agenda for the entrepreneurial university*. Northampton: Edward Elgar Publishing.
- ICTA. (2020). Current Projects. Retrieved from <u>https://www.icta.lk/current-projects/</u>
- Ilonen, S., & Heinonen, J. (2018). Understanding affective learning outcomes in entrepreneurship education. *Industry & Higher Education*, 32(6), 391-404. <u>https://doi.org/10.1177/0950422218805177</u>
- International Labour Organization (2018). Decent Work in Sri Lanka and the Maldives. Retrieved from <u>https://www.ilo.org/colombo/dw/lang--en/index.htm</u>
- International Labour Organization. (2021). Youth unemployment in Asia-Pacific. Retrieved from <u>https://www.ilo.org/global/about-the-ilo/multimedia/maps-and-charts/enhanced/WCMS_650040/lang--en/index.htm</u>
- Ireland, R. D., Hitt, M. A., & Sirmon, D. G. (2003). A Model of Strategic Entrepreneurship: The Construct and its Dimensions. *Journal of Management*, 29(6), 963-989. <u>https://doi.org/10.1016/S0149-2063_03_00086-2</u>
- Isabelle, D. A. (2020). Gamification of Entrepreneurship Education. *Decision Sciences Journal of Innovative Education*, 18(2), 203-223. <u>https://doi.org/10.1111/dsji.12203</u>
- Iscaro, V., Castaldi, L., & Sepe, E. (2017). ExperimentaLab: A virtual platform to enhance entrepreneurial education through training. *Industry and Higher Education*, 31(1), 13-22. <u>https://doi.org/10.1177/0950422216684059</u>
- Isenberg, D. (2010). How to start an entrepreneurial revolution?. Harvard Business Review 88(6), 41-50.
- Isenberg, D. (2016). Applying the Ecosystem Metaphor to Entrepreneurship: Uses and Abuses. *The Antitrust Bulletin*, 61(4), 564-573. https://doi.org/10.1177/0003603X16676162
- Jack, S., Dodd, S. D., & Anderson, A. R. (2008). Change and the development of entrepreneurial networks over time: a processual perspective. *Entrepreneurship and Regional Development*, 20(2), 125-159. <u>https://doi.org/10.1080/08985620701645027</u>

- Jain, R. K. (2011). Entrepreneurial Competencies: A Meta-analysis and Comprehensive Conceptualization for Future Research. *Vision*, 15(2), 127-152. <u>https://doi.org/10.1177/097226291101500205</u>
- Jang, Y. (2013). Modeling student entrepreneurship: a longitudinal study. *Journal of Entrepreneurship Education*, 16(SI), 93-114. <u>https://www.proquest.com/openview/c4f91d96f6933a440c6e4103f044b04e/1?pq-origsite=gscholar&cbl=28224</u>
- Jayathilake, H. (2020). The Impact of Unemployed Youth on National Development in Sri Lanka. *Journal of Emerging Issues in Economics, Finance and Banking*, 9(1), 2985-3005.
- Jena, R. (2020). Measuring the impact of business management student's attitude towards entrepreneurship education on entrepreneurial intention: A case study. *Computers In Human Behavior*, 107(2020), 1-10. <u>https://doi.org/10.1016/j.chb.2020.106275</u>
- Jirawan, C., Thawesaengskulthai, N., & Jarotwan Koiwanit. (2022). Development of a University Innovation Ecosystem Assessment Model for Association of Southeast Asian Nations Universities. *Industry & Higher Education*. 1-15.
- Johnson, D., Bock, A. J., & George, G. (2019). Entrepreneurial dynamism and the built environment in the evolution of university entrepreneurial ecosystems. *Industrial and Corporate Change*, 28(4), 941-959. <u>https://doi.org/10.1093/icc/dtz034</u>
- Jones, C., Penaluna, K., & Penaluna, A. (2019). The promise of andragogy, heutagogy and academagogy to enterprise and entrepreneurship education pedagogy. *Education* + *Training*, 61(9), 1170-1186. <u>https://doi.org/10.1108/ET-10-2018-0211</u>
- Jones, P., Forbes-Simpson, K., Maas, G., & Newbery, R. (2015). Beta: An Experiment in Funded Undergraduate Start-up. *Industry And Higher Education*, 29(5), 405-418. <u>https://doi.org/10.5367/ihe.2015.0271</u>
- Jones, P., Pickernell, D., Fisher, R., & Netana, C. (2017). A tale of two universities: Graduates perceived value of entrepreneurship education. *Education* + *Training*, 59(7/8), 689–705.
- Jones, P., & Jones, A. (2014). Attitudes of Sports Development and Sports Management undergraduate students towards entrepreneurship. *Education* + *Training*, 56(8/9), 716-732. <u>https://doi.org/10.1108/ET-06-2014-0060</u>
- Juhl, J., & Buch, A. (2018). Transforming academia: The role of education. Educational Philosophy and Theory, 51(8), 803-814. <u>https://doi.org/10.1080/00131857.2018.1508996</u>
- Juvonen, P., & Kurvinen, A. (2018). A Perfect Ecosystem for Learning? Modern Thoughts for Organizing Higher Education. Advances In Science, Technology and Engineering Systems Journal, 3(1), 82-93. <u>https://doi.org/10.25046/aj030110</u>

- Karataş-Özkan, M. (2011). Understanding relational qualities of entrepreneurial learning: Towards a multi-layered approach. *Entrepreneurship & Regional Development*, 23(9-10), 877-906. <u>https://doi.org/10.1080/08985626.2011.577817</u>
- Karatas-Ozkan, M., Anderson, A., Fayolle, A., Howells, J., & Condor, R. (2014).
 Understanding Entrepreneurship: Challenging Dominant Perspectives and Theorizing Entrepreneurship through New Postpositivist Epistemologies. *Journal Of Small Business Management*, 52(4), 589-593. <u>https://doi.org/10.1111/jsbm.12124</u>
- Karwowska, E. (2019). Creating shared value by the university. *Social Responsibility Journal*, 17(1), 30-47. <u>https://doi.org/10.1108/SRJ-05-2019-0172</u>
- Kasturiratne, D., Lean, J., & Phippen, A. (2012). International enterprise education in Sri Lanka: a blended approach. *Education* + *Training*, 54(4), 306-318. <u>https://doi.org/10.1108/00400911211236172</u>
- Katono, I. W., Priilaid, D., & Human, G. (2015). Cultural predictions of entrepreneurial orientation and the moderating role of entrepreneurial competencies on graduate entrepreneurial intentions: a cross sectional survey of Uganda. In 45th Atlantic Schools of Business conference, University of New Brunswick.
- Katz, J. A. (2003). The chronology and intellectual trajectory of American entrepreneurship education: 1876–1999. *Journal of Business Venturing*, 18(2), 283-300. https://doi.org/10.1016/S0883-9026(02)00098-8
- Kautonen, T., van Gelderen, M., & Fink, M. (2015). Robustness of the Theory of Planned Behavior in Predicting Entrepreneurial Intentions and Actions. *Entrepreneurship Theory and Practice*, 39(3), 655-674. <u>https://doi.org/10.1111/etap.12056</u>
- Keat, O. Y., Selvarajah, C., & Meyer, D. (2011). Inclination towards entrepreneurship among university students: an empirical study of Malaysian university students. *International Journal of Business and Social Science*, 2(4), 206.
- Kelmar, J. H. (1992). Business Plans for Teaching Entrepreneurial Behaviour. *Education* + *Training*, 34(1), 30-32. <u>https://doi.org/10.1108/00400919210007511</u>
- Kim, S., Ko, H.-J., & Lee, Y. (2015). A Study of Entrepreneurship Education of University in Science and Engineering for Vitalization of Technology-based Startup. *Journal of Engineering Education Research*, 18(2), 3-7. <u>https://doi.org/10.18108/jeer.2015.18.2.3</u>
- King, N., & Horrocks, C. (2010). Interviews in Qualitative Research. London: SAGE Publications.
- Kirby, D. (2004). Entrepreneurship education: can business schools meet the challenge? *Education* + *Training*, 46(8/9), 510-519. <u>https://doi.org/10.1108/00400910410569632</u>

- Kirby, D., & Ibrahim, N. (2012). An enterprise revolution for Egyptian universities. Education, Business and Society: *Contemporary Middle Eastern Issues*, 5(2), 98-111. <u>https://doi.org/10.1108/17537981211251142</u>
- Kirby, D., & Ibrahim, N. (2013). Entrepreneurship Education Policies in the MENA Region (1): Challenges and Opportunities. *American Journal of Entrepreneurship*, 6(2), 1-15.
- Klandt, H. (2004). Entrepreneurship Education and Research in German-Speaking Europe. *Academy of Management Learning & Education*, 3(3), 293-301. <u>https://doi.org/10.5465/AMLE.2004.14242226</u>
- Kleinhempel, J., Beugelsdijk, S., & Klasing, M. (2020). The Changing Role of Social Capital During the Venture Creation Process: A Multilevel Study. *Entrepreneurship Theory* and Practice, 46(2), 297-330. <u>https://doi.org/10.1177/1042258720913022</u>
- Klofsten, M., Fayolle, A., Guerrero, M., Mian, S., Urbano, D., & Wright, M. (2019). The entrepreneurial university as driver for economic growth and social change - Key strategic challenges. *Technological Forecasting & Social Change*, 141(2019), 149-158. <u>https://doi.org/10.1016/j.techfore.2018.12.004</u>
- Klyver, K., & Foley, D. (2012). Networking and culture in entrepreneurship. *Entrepreneurship & Regional Development*, 24(7-8), 561-588. <u>https://doi.org/10.1080/08985626.2012.710257</u>
- Klyver, K., Steffens, P., & Lomberg, C. (2020). Having your cake and eating it too? A twostage model of the impact of employment and parallel job search on hybrid nascent entrepreneurship. *Journal of Business Venturing*, 35(5), 106042. <u>https://doi.org/10.1016/j.jbusvent.2020.106042</u>
- Kodithuwakku, S. S., & Rosa, P. (2002). The entrepreneurial process and economic success in a constrained environment. *Journal of Business Venturing*, 17(5), 431-465. https://doi.org/10.1016/S0883-9026(01)00074-X
- Kolvereid, L. (1996). Organizational Employment versus Self-Employment: Reasons for Career Choice Intentions. *Entrepreneurship Theory and Practice*, 20(3), 23-31. <u>https://doi.org/10.1177/104225879602000302</u>
- Kolvereid, L. (1996). Prediction of Employment Status Choice Intentions. *Entrepreneurship Theory and Practice*, 21(1), 47-58. <u>https://doi.org/10.1177/104225879602100104</u>
- Kolvereid, L. (2018). Entrepreneurship among parents. *Journal of Innovation and Entrepreneurship*, 7(1), 1-14. <u>https://doi.org/10.1186/s13731-018-0089-0</u>
- Komulainen, K., Naskali, P., Korhonen, M., & Keskitalo-Foley, S. (2011). Internal Entrepreneurship--A Trojan Horse of the Neoliberal Governance of Education? Finnish Pre- and In-Service Teachers' Implementation of and Resistance towards Entrepreneurship Education. *Journal for Critical Education Policy Studies*, 9(1), 341.

- Kopera, S., Wszendybył-Skulska, E., Cebulak, J., & Grabowski, S. (2018). Interdisciplinarity in Tech Startups Development – Case Study of 'Unistartapp' Project. *Foundations of Management*, 10(1), 23-32. <u>https://doi.org/10.2478/fman-2018-0003</u>
- Korsgaard, S., Hunt, R. A., Townsend, D. M., & Ingstrup, M. B. (2020). COVID-19 and the importance of space in entrepreneurship research and policy. *International Small Business Journal*, 38(8), 697-710. <u>https://doi.org/10.1177/0266242620963942</u>

Kozlinska, I., Mets, T., & Rõigas, K. (2020). Measuring Learning Outcomes of Entrepreneurship

Education Using Structural Equation Modeling. *Administrative Sciences*, 10(3), 1-17 https://doi.org/10.3390/admsci10030058

- Kraaijenbrink, J., Bos, G., & Groen, A. (2010). What do students think of the entrepreneurial support given by their universities? *International Journal of Entrepreneurship and Small Business*, 9(1), 110-125.
- Krieger, A., Stuetzer, M., Obschonka, M., & Salmela-Aro, K. (2021). The growth of entrepreneurial human capital: origins and development of skill variety. *Small Business Economics*, 59(2), 654-664.
- Kristel, M., Allen, A., James, A. C., & Ekaterina, A. (2018). Entrepreneurial academics and academic entrepreneurs: a systematic literature review. *International Journal of Technology Management*, 77(1), 9-37. <u>https://doi.org/10.1504/IJTM.2018.091710</u>
- Kristiansen, S., & Indarti, N. (2004). Entrepreneurial Intention Among Indonesian and Norwegian Students. *Journal of Enterprising Culture*, 12(1), 55-78. <u>https://doi.org/10.1142/S021849580400004X</u>
- Kriz, W. C., & Auchter, E. (2016). 10 Years of Evaluation Research Into Gaming Simulation for German Entrepreneurship and a New Study on Its Long-Term Effects. *Simulation* & Gaming, 47(2), 179-205. <u>https://doi.org/10.1177/1046878116633972</u>
- Krueger, N. F., Reilly, M. D., & Carsrud, A. L. (2000). Competing models of entrepreneurial intentions. *Journal of Business Venturing*, 15(5), 411-432. <u>https://doi.org/10.1016/S0883-9026(98)00033-0</u>
- Kumar, S., & Das, S. (2019). An extended model of theory of planned behaviour. Journal of Entrepreneurship in Emerging Economies, 11(3), 369-391. <u>https://doi.org/10.1108/JEEE-09-2018-0089</u>
- Kuratko, D. F. (2014). Entrepreneurship: theory, process, practice (9th edition. ed.). Boston: Cengage learning.
- Kuratko, D. F., & Morris, M. H. (2018). Examining the Future Trajectory of Entrepreneurship. *Journal of Small Business Management*, 56(1), 11-23. <u>https://doi.org/10.1111/jsbm.12364</u>

- Kwong, C., & Thompson, P. (2016). The When and Why: Student Entrepreneurial Aspirations. *Journal of Small Business Management*, 54(1), 299-318. <u>https://doi.org/10.1111/jsbm.12146</u>
- Kyndt, E., & Baert, H. (2015). Entrepreneurial competencies: Assessment and predictive value for entrepreneurship. *Journal of Vocational Behavior*, 90, 13-25. <u>https://doi.org/10.1016/j.jvb.2015.07.002</u>
- Lackéus, M. (2015). Entrepreneurship in education what, why, when and how. Entrepreneurship 360 OECD Background Paper, Local Economic and Employment Development Programme, Paris.
- Lahikainen, K., Kolhinen, J., Ruskovaara, E., & Pihkala, T. (2019). Challenges to the development of an entrepreneurial university ecosystem: The case of a Finnish university campus. *Industry and Higher Education*, 33(2), 96-107.
- Lahikainen, K., Peltonen, K., Hietanen, L., & Oikkonen, E. (2021). Calling for student engagement in an entrepreneurial university. In U. Hytti (Ed.), A Research Agenda for the Entrepreneurial University. Cheltenham: Edward Elgar Publishing.
- Landry, R., Amara, N., & Rherrad, I. (2006). Why are some university researchers more likely to create spin-offs than others? Evidence from Canadian universities. *Research Policy*, 35(10), 1599-1615.
- Landström, H., Harirchi, G., & Åström, F. (2012). Entrepreneurship: Exploring the knowledge base. *Research Policy*, 41(7), 1154-1181. <u>https://doi.org/https://doi.org/10.1016/j.respol.2012.03.009</u>
- Langowitz, N., & Minniti, M. (2007). The Entrepreneurial Propensity of Women. Entrepreneurship Theory and Practice, 31(3), 341-364.
- Laura, C. D. (2010). From Rational to Wise Action: Recasting Our Theories of Entrepreneurship. *Journal of Business Ethics*, 92(4), 513-530. <u>https://doi.org/10.1007/s10551-009-0170-5</u>
- Lazzeroni, M., & Piccaluga, A. (2003). Towards the Entrepreneurial University. Local Economy: *The Journal of the Local Economy Policy Unit*, 18(1), 38-48.
- Lee, C. K., Wiklund, J., Amezcua, A., Bae, T. J., & Palubinskas, A. (2021). Business failure and institutions in entrepreneurship: a systematic review and research agenda. *Small Business Economics*, 58(2021), 1-27. <u>https://doi.org/10.1007/s11187-021-00495-4</u>
- Lerner, J. (2010). The future of public efforts to boost entrepreneurship and venture capital. *Small Business Economics*, 35(3), 255-264. <u>https://doi.org/10.1007/s11187-010-9298-</u> Z
- Leydesdorff, L. (2000). The triple helix: an evolutionary model of innovations. *Research Policy*, 29(2), 243-255. <u>https://doi.org/10.1016/S0048-7333(99)00063-3</u>

- Li, M., He, L., & Zhao, Y. (2020). The triple helix system and regional entrepreneurship in China. *Entrepreneurship & Regional Development*, 32(7-8), 508-530. https://doi.org/10.1080/08985626.2019.1666168
- Li, Q., & Zhang, L. (2019). Design of Innovative and Entrepreneurial Talents Cultivation System for Accounting Major. *Journal of Educational Institute of Jilin Province*, 9(9).
- Lichtenstein, G. A., & Lyons, T. S. (2001). The Entrepreneurial Development System: Transforming Business Talent and Community Economies. *Economic Development Quarterly*, 15(1), 3-20. <u>https://doi.org/10.1177/089124240101500101</u>
- Lichtenstein, G. A., Lyons, T. S., & Kutzhanova, N. (2004). Building Entrepreneurial Communities: The Appropriate Role of Enterprise Development Activities. *Journal of the Community Development Society*, 35(1), 5-24. <u>https://doi.org/10.1080/15575330409490119</u>
- Liễu, T. T. B., & Barth, J. R. (2014). Entrepreneurial skills for women's business success lessons around the world and implications for Vietnam. *Journal of Science: Economics and Business*, 30(2), 55-64.
- Liguori, E., Bendickson, J., Solomon, S., & McDowell, W. C. (2019). Development of a multi-dimensional measure for assessing entrepreneurial ecosystems. *Entrepreneurship & Regional Development*. 31(1-2), 7-21. <u>https://doi.org/10.1080/08985626.2018.1537144</u>
- Liguori, E., & Winkler, C. (2020). From Offline to Online: Challenges and Opportunities for Entrepreneurship Education Following the COVID-19 Pandemic. *Entrepreneurship Education and Pedagogy*, 3(4), 346-351. <u>https://doi.org/10.1177/2515127420916738</u>
- Lin, H., Ma, R., & Lv, J. (2018). Research on the Integration of Legal Education of Entrepreneurship and Professional Education for Science and Engineering Students. 2018 2nd International Conference on Education Science and Economic Management (ICESEM 2018), 1042-1045.
- Lin, S. & Xu., Zhengda. (2017). The factors that influence the development of entrepreneurship education: Based on the case of China. *Management Decision*, 55(7), 1351-1370.
- Lin, X., Carsrud, A., Jagoda, K., & Shen, W. (2013). Determinants of Entrepreneurial Intentions: Applying Western Model to the Sri Lanka Context. *Journal of Enterprising Culture*, 21(02), 153-174. <u>https://doi.org/10.1142/S0218495813500076</u>
- Lina, D. (2018). The Role of the Universities in Shaping the Entrepreneurial Ecosystem. A Qualitative Approach. *Management of Sustainable Development*, 10(2), 57-61. <u>https://doi.org/10.2478/msd-2019-0009</u>
- Liñán, F., & Chen, Y. W. (2009). Development and Cross–Cultural Application of a Specific Instrument to Measure Entrepreneurial Intentions. *Entrepreneurship Theory and Practice*, 33(3), 593-617. <u>https://doi.org/10.1111/j.1540-6520.2009.00318.x</u>

- Liñán, F., & Fayolle, A. (2015). A systematic literature review on entrepreneurial intentions: Citation, thematic analyses, and research agenda. *International Entrepreneurship and Management Journal*, 11(4), 907-933. <u>https://doi.org/10.1007/s11365-015-0356-5</u>
- Linder, C., Lechner, C., & Pelzel, F. (2020). Many Roads Lead to Rome: How Human, Social, and Financial Capital Are Related to New Venture Survival. *Entrepreneurship: Theory & Practice*, 44(5), 909-932. <u>https://doi.org/10.1177/1042258719867558</u>
- Lindquist, M. J., Sol, J., & Van Praag, M. (2015). Why Do Entrepreneurial Parents Have Entrepreneurial Children? *Journal of Labor Economics*, 33(2), 269-296. <u>https://doi.org/10.1086/678493</u>
- Link, A. N., & Sarala, R. M. (2019). Advancing conceptualisation of university entrepreneurial ecosystems: The role of knowledge-intensive entrepreneurial firms. *International Small Business Journal*, 37(3), 289-310. <u>https://doi.org/10.1177/0266242618821720</u>
- Little, A. W. (2011). Education policy reform in Sri Lanka: the double-edged sword of political will. *Journal of Education Policy: The Policies, Politics and Progress of Access to Basic Education*, 26(4), 499-512. <u>https://doi.org/10.1080/02680939.2011.555005</u>
- Liu, C.-Q., Gamble, J. H., & Chen, I. H. (2019). An Empirical Study of the Influence of Continuing Education on Youth Entrepreneurship in Taiwan. <u>https://dx.doi.org/10.2139/ssrn.3496092</u>
- Liu, H., Kulturel-Konak, S., & Konak, A. (2021). A measurement model of entrepreneurship education effectiveness based on methodological triangulation. *Studies in Educational Evaluation*, 70(2021), 1-13. <u>https://doi.org/10.1016/j.stueduc.2021.100987</u>
- Liu, H., Kulturel-Konak, S & Konak, A. (2021). Key Elements and Their Roles in Entrepreneurship Education Ecosystem: Comparative Review and Suggestions for Sustainability. Sustainability, 13(19), 1-28.
- Locke, K., & Golden-Biddle, K. (1997). Constructing Opportunities for Contribution: Structuring Intertextual Coherence and "Problematizing" in Organizational Studies. *Academy of Management Journal*, 40(5), 1023-1062. <u>https://doi.org/10.2307/256926</u>
- Lockett, A., Wright, M., & Franklin, S. (2003). Technology Transfer and Universities' Spin-Out Strategies. *Small Business Economics*, 20(2), 185-200. <u>https://doi.org/10.1023/A:1022220216972</u>
- Loi, M. (2018). Dynamics of Entrepreneurship Education: How Human and Social Capital Interact With Learning. In N. Zakaria & L. A Kaushal (Eds.), *Global Entrepreneurship and New Venture Creation in the Sharing Economy* (pp. 37-53). Hershey: IGI Global.

- Longva, K. K. (2021). Student venture creation: developing social networks within entrepreneurial ecosystems in the transition from student to entrepreneur. *International Journal of Entrepreneurial Behaviour & Research*, 27(5), 1264-1284. <u>https://doi.org/10.1108/IJEBR-09-2020-0661</u>
- Lorber, L. (2017). Universities, knowledge networks and local environment for innovationbased regional development: Case study of the university of maribor. *Geograficky Casopis*, 69(4), 361-383.
- Lourenço, F., & Jayawarna, D. (2011). Enterprise education: the effect of creativity on training outcomes. *International Journal of Entrepreneurial Behaviour & Research*, 17(3), 224-244. <u>https://doi.org/10.1108/13552551111130691</u>
- Lourenço, F., Taylor, T., & Taylor, D. (2013). Integrating "education for entrepreneurship" in multiple faculties in "half-the-time" to enhance graduate entrepreneurship. *Journal of Small Business and Enterprise Development*, 20(3), 503-525. <u>https://doi.org/10.1108/JSBED-04-2013-0052</u>
- Luker, K. (2008). Salsa Dancing into the Social Sciences: Research in an Age of Info-Glut. Cambridge: Harvard University Press.
- Lumpkin, G. T., & Dess, G. G. (1996). Clarifying the Entrepreneurial Orientation Construct and Linking It to Performance. *The Academy of Management Review*, 21(1), 135-172. <u>https://doi.org/10.2307/258632</u>
- Lumpkin, G. T., & Dess, G. G. (2015). Entrepreneurial Orientation. New Jersey: John Wiley & Sons.
- Lyons, T. S. (2015). Entrepreneurship and community development: what matters and why? *Community development*, 46(5), 456-460. https://doi.org/10.1080/15575330.2015.1085194
- Lv, M., Zhang, H., Georgescu, P., Li, T., & Zhang, B. (2022). Improving Education for Innovation and Entrepreneurship in Chinese Technical Universities: A Quest for Building a Sustainable Framework. *Sustainability*, 14(2), 1-19.
- Mahon, K., Heikkinen, H. L. T., & Huttunen, R. (2019). Critical educational praxis in university ecosystems: enablers and constraints. *Pedagogy, Culture & Society*, 27(3), 463-480. <u>https://doi.org/10.1080/14681366.2018.1522663</u>
- Malecki, E. J. (2018). Entrepreneurship and entrepreneurial ecosystems. *Geography Compass*, 12(3), 1-21. <u>https://doi.org/10.1111/gec3.12359</u>
- Man, T. W. Y. (2007). Understanding entrepreneurial learning: A competency approach. International Journal of Entrepreneurship and Innovation, 8(3), 189-198. <u>https://doi.org/10.5367/00000007781698554</u>

- Manolova, T. S., Eunni, R. V., & Gyoshev, B. S. (2008). Institutional Environments for Entrepreneurship: Evidence from Emerging Economies in Eastern Europe. *Entrepreneurship Theory and Practice*, 32(1), 203-218. <u>https://doi.org/10.1111/j.1540-6520.2007.00222.x</u>
- Maritz, A. (2017). Illuminating the black box of entrepreneurship education programmes: Part 2. *Education* + *Training*, 59(5), 471-482. <u>https://doi.org/10.1108/ET-02-2017-0018</u>
- Maritz, A. (2020). A Multi-Disciplinary Business Approach to Covid-19: La Trobe Business School Perspectives. *International Journal of Organizational Innovation*, 13(1), 45-62.
- Maritz, A., & Brown, C. R. (2013). Illuminating the black box of entrepreneurship education programs. *Education* + *Training*, 55(3), 234-252. https://doi.org/10.1108/00400911311309305
- Maritz, A., Jones, C., & Shwetzer, C. (2015). The status of entrepreneurship education in Australian universities. *Education* + *Training*, 57(8/9), 1020-1035. <u>https://doi.org/10.1108/ET-04-2015-0026</u>
- Maritz, A., Nguyen, Q., & Bliemel, M. (2019). Boom or bust? Embedding entrepreneurship in education in Australia. *Education* + *Training*, 61(6), 737-755. https://doi.org/10.1108/ET-02-2019-0037
- Maritz, A., Perenyi, A., de Waal, G., & Buck, C. (2020). Entrepreneurship as the Unsung Hero during the Current COVID-19 Economic Crisis: Australian Perspectives. *Sustainability*, 12(2022), 1-9. <u>https://doi.org/10.3390/su12114612</u>
- Maroufkhani, P., Wagner, R., & Wan Ismail, W. K. (2018). Entrepreneurial ecosystems: a systematic review. *Journal of Enterprising Communities: People and Places in the Global Economy*, 12(4), 545-564. <u>https://doi.org/10.1108/JEC-03-2017-0025</u>
- Marques, C. S., Santos, G., Galvão, A., Mascarenhas, C., & Justino, E. (2018).
 Entrepreneurship education, gender and family background as antecedents on the entrepreneurial orientation of university students. *International Journal of Innovation Science*, 10(1), 58-70. <u>https://doi.org/10.1108/IJIS-07-2017-0067</u>
- Martin, B. C., McNally, J. J., & Kay, M. J. (2013). Examining the formation of human capital in entrepreneurship: A meta-analysis of entrepreneurship education outcomes. *Journal of Business Venturing*, 28(2), 211-224.
- Marzocchi, C., Kitagawa, F., & Sánchez-Barrioluengo, M. (2019). Evolving missions and university entrepreneurship: academic spin-offs and graduate start-ups in the entrepreneurial society. *The Journal of Technology Transfer*, 44(1), 167-188.
- Masakure, O., Henson, S., & Cranfield, J. (2009). Performance of microenterprises in Ghana: a resource-based view. *Journal of Small Business and Enterprise Development*, 16(3), 466-484. <u>https://doi.org/10.1108/14626000910977170</u>

- Mason, C., Anderson, M., Kessl, T., & Hruskova, M. (2020). Promoting student enterprise: Reflections on a university start-up programme. *Local economy*, 35(1), 68-79. <u>https://doi.org/10.1177/0269094219894907</u>
- Mathibe, D., & Croucamp, P. (2019). Policy and strategy framework for black, womenowned small and medium businesses, with reference to the Johannesburg and Tshwane metropoles. *Journal of Public Administration*, 54(4-1), 714-727.
- Matlay, H. (2006). Researching entrepreneurship and education: Part 2: What is entrepreneurship education and does it matter? *Education* + *Training*, 48(8), 704-718. <u>https://doi.org/10.1108/et.2006.00448eaa.001</u>
- Matlay, H. (2009). Entrepreneurship education in the UK. Journal of Small Business and Enterprise Development, 16(2), 355-368. https://doi.org/10.1108/14626000910956100
- Matlay, H., & Carey, C. (2007). Entrepreneurship education in the UK: A longitudinal perspective. *Journal of Small Business and Enterprise Development*, 14(2), 252-263
- Matt, M., & Schaeffer, V. (2018). Building Entrepreneurial Ecosystems Conducive to Student Entrepreneurship: New Challenges for Universities. *Journal of Innovation Economics & Management*, 25(09), 9-32.
- McKeown, J., Millman, C., Reddy Sursani, S., Smith, K., & Martin, L. M. (2006). Graduate entrepreneurship education in the United Kingdom. *Education* + *Training*, 48(8/9), 597-613. <u>https://doi.org/10.1108/00400910610710038</u>
- McMullan, W. E., & Gillin, L. M. (1998). Entrepreneurship Education: Developing technological start-up entrepreneurs: A case study of a graduate entrepreneurship programme at Swinburne University. *Technovation*, 18(4), 275-286. <u>https://doi.org/10.1016/S0166-4972(97)00119-3</u>
- McMullan, W. E., & Long, W. A. (1987). Entrepreneurship education in the nineties. *Journal* of Business Venturing, 2(3), 261-275. <u>https://doi.org/https://doi.org/10.1016/0883-9026(87)90013-9</u>
- Mehta, K., Zappe, S., Brannon, M. L., & Zhao, Y. (2016). An Educational and Entrepreneurial Ecosystem to Actualize Technology-Based Social Ventures. *Advances in Engineering Education*, 5(1), 1-38.
- Mehtap, S., Pellegrini, M. M., Caputo, A., & Welsh, D. H. B. (2017). Entrepreneurial intentions of young women in the Arab world: Socio-cultural and educational barriers. *International Journal of Entrepreneurial Behaviour and Research*, 23(6), 880-902. <u>https://doi.org/10.1108/IJEBR-07-2017-0214</u>
- Mertens, D. (2014). *Research and evaluation in education and psychology: Integrating diversity with quantitative, qualitative and mixed methods*. London: SAGE Publications.

- Meissner, D., Polt, W., & Cervantes, M. (2020). New challenges for universities in the knowledge triangle. *Journal of Technology Transfer*, 45(3), 806-819. https://doi.org/10.1007/s10961-018-9699-8
- Meng-meng, S. (2017). On the Ecosystem of Entrepreneurial Education from the Perspective of Symbiosis Theory. *Vocational Education Research in Xinjiang*, (1), 23-31.
- Meoli, M., Paleari, S., & Vismara, S. (2019). The governance of universities and the establishment of academic spin-offs. *Small Business Economics*, *52*(2), 485-504.
- Meyer, M. H., Lee, C., Kelley, D., & Collier, G. (2020). An Assessment and Planning Methodology for University-Based: Entrepreneurship Ecosystems. *The Journal of Entrepreneurship*, 29(2), 259-292. <u>https://doi.org/10.1177/0971355720930571</u>
- Middleton, W., Antonio, P.-M., Nigel, L., Carla, Q.-P., & Sarah, J. (2019). The university as an entrepreneurial learning space: The role of socialized learning in developing entrepreneurial competence. *International Journal of Entrepreneurial Behaviour & Research*, 26(5), 887-909. <u>https://doi.org/10.1108/IJEBR-04-2018-0263</u>
- Miller, D. (2011). Miller (1983) Revisited: A Reflection on EO Research and Some Suggestions for the Future. *Entrepreneurship Theory and Practice*, 35(5), 873-894. <u>https://doi.org/10.1111/j.1540-6520.2011.00457.x</u>
- Miller, D., & Acs, Z. (2017). The campus as entrepreneurial ecosystem: the University of Chicago. *Small Business Economics*, 49(1), 75-95. <u>https://doi.org/10.1007/s11187-017-9868-4</u>
- Minichiello, V. (2008). *In-depth interviewing: principles, techniques, analysis* (3rd ed.). Melbourne: Pearson Education.
- Ministry of Sustainable Development. (2018). Sri Lanka Voluntary National Review on the Status of Implementing Sustainable Development Goals. Retrieved from <u>https://sustainabledevelopment.un.org/content/documents/19677FINAL_SriLankaVN</u> <u>R_Report_30Jun2018.pdf</u>
- Minniti, M., & Lévesque, M. (2008). Recent developments in the economics of entrepreneurship. *Journal of Business Venturing*, 23(6), 603-612. <u>https://doi.org/https://doi.org/10.1016/j.jbusvent.2008.01.001</u>
- Mir Shahid, S., & Alarifi, G. (2021). Social entrepreneurship education: A conceptual framework and review. *The International Journal of Management Education*, 19(3), 100533. <u>https://doi.org/https://doi.org/10.1016/j.ijme.2021.100533</u>
- Mitchelmore, S., & Rowley, J. (2010). Entrepreneurial competencies: a literature review and development agenda. *International Journal of Entrepreneurial Behaviour & Research*, 16(2), 92-111. <u>https://doi.org/10.1108/13552551011026995</u>
- Mitra, J., Abubakar, Y., & Sagagi, M. (2011). Knowledge creation and human capital for development: the role of graduate entrepreneurship. *Education* + *Training*, 53(5), 462-479.

- Mitton, D. G. (1989). The Compleat Entrepreneur. *Entrepreneurship Theory and Practice*, 13(3), 9-20.
- Moggi, S., & Dameri, R. P. (2021). Circular business model evolution: Stakeholder matters for a self-sufficient ecosystem. *Business Strategy and the Environment*, *30*(6), 2830-2842.
- Molas-Gallart, J., & Castro-Martínez, E. (2007). Ambiguity and conflict in the development of 'Third Mission' indicators. *Research Evaluation*, 16(4), 321-330.
- Morris, M., Kuratko, D., & Pryor, C. (2013). Building Blocks for the Development of University-Wide Entrepreneurship. *Entrepreneurship Research Journal*, 4(1), 45-68.
- Morris, M. H., Neumeyer, X., Jang, Y., & Kuratko, D. F. (2018). Distinguishing Types of Entrepreneurial Ventures: An Identity-Based Perspective. *Journal of Small Business Management*, 56(3), 453-474.
- Morris, M. H., Shirokova, G., & Tsukanova, T. (2017). Student entrepreneurship and the university ecosystem: a multi-country empirical exploration. *European Journal International Management*, 11(1), 65-85.
- Mueller, S. (2011). Increasing entrepreneurial intention: effective entrepreneurship course characteristics. *International Journal of Entrepreneurship and Small Business*, 13(1), 55-74.
- Mukesh, H. V., & Rajasekharan Pillai, K. (2020). Role of Institutional Ecosystem in Entrepreneurship Education: An Empirical Reiteration. *The Journal of Entrepreneurship*, 29(1), 176-205.
- Nabi, G., & Holden, R. (2008). Graduate entrepreneurship: intentions, education and training. *Education* + *Training*, 50(7), 545-551.
- Nabi, G., Holden, R., & Walmsley, A. (2010). From student to entrepreneur: towards a model of graduate entrepreneurial career-making. *Journal of Education and Work*, 23(5), 389-415. <u>https://doi.org/10.1080/13639080.2010.515968</u>
- Nabi, G., & Liñán, F. (2011). Graduate entrepreneurship in the developing world: intentions, education and development. *Education* + *Training*, 53(5), 325-334.
- Nagel, R., Holland, K., Gipson, K., Henriques, J., & Paterson, K. (2020). Creating an ecosystem that fosters innovation and entrepreneurial mindset at an undergraduate institution through pathways to innovation. *Advances in Engineering Education*, 08(01), 1-13.
- Naia, A., Baptista, R., Januário, C., & Trigo, V. (2014). A Systematization of the Literature on Entrepreneurship Education: Challenges and Emerging Solutions in the Entrepreneurial Classroom. *Industry and Higher Education*, 28(2), 79-96. <u>https://doi.org/10.5367/ihe.2014.0196</u>

- Nanda, R., & Sørensen, J. B. (2010). Workplace peers and entrepreneurship. *Management Science*, 56(7), 1116-1126.
- Naudé, W. (2009). Entrepreneurship, developing countries, and development economics: new approaches and insights. *Small Business Economics*, 34(1), 1. https://doi.org/10.1007/s11187-009-9198-2
- Naumann, C. (2017). Entrepreneurial Mindset: A Synthetic Literature Review. *Entrepreneurial Business and Economics Review*, 5(3), 149-172. <u>https://doi.org/10.15678/EBER.2017.050308</u>
- Neck, H., & Greene, P. (2011). Entrepreneurship Education: Known Worlds and New Frontiers. *Journal of Small Business Management*, 49(1), 55-70. https://doi.org/10.1111/j.1540-627X.2010.00314.x
- Neck, H. M., Meyer, G. D., Cohen, B., & Corbett, A. C. (2004). An Entrepreneurial System View of New Venture Creation. *Journal of Small Business Management*, 42(2), 190-208. <u>https://doi.org/10.1111/j.1540-627X.2004.00105.x</u>
- Neergaard, H., & Ulhoi, J. P. (2007). *Handbook of Qualitative Research Methods in Entrepreneurship*. Cheltenham: Edward Elgar Publishing.
- Nelles, J., & Vorley, T. (2010). From policy to practice: engaging and embedding the third mission in contemporary universities. *International Journal of Sociology and Social Policy*, 30(7/8), 341-353. <u>https://doi.org/10.1108/01443331011060706</u>
- Neuman, L. (2014). *Social methods: qualitative and quantitative approaches*. Pearson Education
- Neumeyer, X., Santos, S. C., Caetano, A., & Kalbfleisch, P. (2019). Entrepreneurship Ecosystems and Women Entrepreneurs: A Social Capital and Network Approach. *Small Business Economics*, 53(2), 475-489.
- Newman, A., Schwarz, S., & Borgia, D. (2014). How does microfinance enhance entrepreneurial outcomes in emerging economies? The mediating mechanisms of psychological and social capital. *International Small Business Journal*, 32(2), 158-179. <u>https://doi.org/10.1177/0266242613485611</u>
- Neergaard, H., & Christensen, D. R. (2017). Breaking the waves: Routines and rituals in entrepreneurship education. *Industry and Higher Education*, 31(2), 90–100.
- Nicotra, M., Romano, M., Del Giudice, M., & Schillaci, C. E. (2017). The causal relation between entrepreneurial ecosystem and productive entrepreneurship: a measurement framework. *The Journal of Technology Transfer*, 43(3), 640-673. <u>https://doi.org/10.1007/s10961-017-9628-2</u>
- Noor Hazlina, A., Hasliza Abdul, H., Thurasamy, R., Simona, P., & Armando, P. (2018). The ecosystem of entrepreneurial university: the case of higher education in a developing country. *International Journal of Technology Management*, 78(1/2), 52-69. https://doi.org/10.1504/IJTM.2018.093935

- Notani, A. S. (1998). Moderators of Perceived Behavioral Control's Predictiveness in the Theory of Planned Behavior: A Meta-Analysis. *Journal of Consumer Psychology*, 7(3), 247-271. <u>https://doi.org/10.1207/s15327663jcp0703_02</u>
- Novela, S., Syarief, R., Fahmi, I., & Arkeman, Y. (2021). Creating a University-Based Entrepreneurial Ecosystem in Indonesia. *Academic Journal of Interdisciplinary Studies*, 10(1), 174.
- Nylund, P. A., Brem, A., & Agarwal, N. (2021). Innovation ecosystems for meeting sustainable development goals: The evolving roles of multinational enterprises. *Journal of Cleaner Production*, 281, 125329.
- O'Connor, A., & Reed, G. (2018). Theorizing the university governance role in an entrepreneurial ecosystem. In O'Connor, F. Sussan, A., E. Stam, and D. B Audretsch (Eds.), *Entrepreneurial ecosystems. (pp.73-97)*. Boston: Springer.
- O'Gorman, C., Byrne, O., & Pandya, D. (2008). How scientists commercialise new knowledge via entrepreneurship. *The Journal of Technology Transfer*, 33(1), 23-43.
- O'Brien, E., Cooney, T. M., & Blenker, P. (2019). Expanding university entrepreneurial ecosystems to under-represented communities. *Journal of Entrepreneurship and Public Policy*, 8(3), 384-407.
- Obschonka, M., Fritsch, M., & Stuetzer, M. (2021). *The geography of entrepreneurial psychology*. Cheltenham: Edward Elgar Publishing.
- Obschonka, M., Hakkarainen, K., Lonka, K., & Salmela-Aro, K. (2017). Entrepreneurship as a twenty-first century skill: entrepreneurial alertness and intention in the transition to adulthood. *Small Business Economics*, 48(3), 487-501.
- Obschonka, M., Silbereisen, R. K., & Schmitt-Rodermund, E. (2010). Entrepreneurial intention as developmental outcome. *Journal of Vocational Behavior*, 77(1), 63-72.
- O'Connor, A. (2013). A conceptual framework for entrepreneurship education policy: Meeting government and economic purposes. Journal of Business Venturing, 28(4), 546-563. <u>https://doi.org/10.1016/j.jbusvent.2012.07.003</u>
- O'Connor, J., Fenton, M., & Barry, A. (2012). Entrepreneurship Education: Ireland's Solution to Economic Regeneration? *Industry and Higher Education*, 26(3), 241-249.
- Olutuase, S. O., Brijlal, P., Yan, B., & Ologundudu, E. (2018). Entrepreneurial orientation and intention: Impact of entrepreneurial ecosystem factors. *Journal of Entrepreneurship Education*, 21(3), 1-14.
- Onwuegbuzie, A. J., & Leech, N. L. (2004). Enhancing the Interpretation of "Significant" Findings: The Role of Mixed Methods Research. *Qualitative report*, 9(4), 770-792.
- Oosterbeek, H., van Praag, M., & Ijsselstein, A. (2010). The impact of entrepreneurship education on entrepreneurship skills and motivation. *European Economic Review*, 54(3), 442-454. <u>https://doi.org/10.1016/j.euroecorev.2009.08.002</u>

- Ordu, P., & Abdulkarim, M. (2018). Institution-Community Entrepreneurship Education Ecosystem (ICEEE) Model for Effective Entrepreneurial Graduates' development in Tertiary Institution. *Nigerian Journal of Business Education*, 4(1), 156-169.
- Ordu, P., & Abdulkarim, M. A. (2019). Impact Of Institution-Community Entrepreneurship Education Ecosystem (ICEEE) On Business Education Graduates' jobs Creation In Omoku–Rivers State. *European Journal of Education Studies*. <u>http://dx.doi.org/10.46827/ejes.v0i0.2698</u>
- Ortiz-Medina, L., Fernández-Ahumada, E., Lara-Vélez, P., Taguas, E. V., Gallardo-Cobos, R., Del Campillo, M. C., & Guerrero-Ginel, J. E. (2016). Designing an accompanying ecosystem to foster entrepreneurship among agronomic and forestry engineering students. Opinion and commitment of university lecturers. *European Journal of Engineering Education*, 41(4), 393-410. https://doi.org/10.1080/03043797.2015.1079815
- Ozdemir, P. (2018). Internal And External Interactions of Entrepreneurial Education Ecosystems of Universities: A Study at Turkish Universities. *The Online Journal of Quality in Higher Education*, 5(3), 49-59.
- Pandit, D., Joshi, M. P., & Tiwari, S. R. (2018). Examining Entrepreneurial Intention in Higher Education: An Exploratory Study of College Students in India. *The Journal of Entrepreneurship*, 27(1), 25-46. <u>https://doi.org/10.1177/0971355717738595</u>
- Paul, J., & Criado, A. R. (2020). The art of writing literature review: What do we know and what do we need to know? *International Business Review*, 29(4), 1-7. <u>https://doi.org/10.1016/j.ibusrev.2020.101717</u>
- Pauna, D., & Frank, E. J. (2017). Entrepreneurship Education: A Look at Two Universities. *Global Journal of Entrepreneurship*, 1(1), 80-100.
- Perry, C. (1998). A Structured Approach for Presenting Theses. *Australasian Marketing Journal*, 6(1), 63-85. <u>https://doi.org/10.1016/S1441-3582(98)70240-X</u>
- Peterman, N. E., & Kennedy, J. (2003). Enterprise Education: Influencing Students' Perceptions of Entrepreneurship. *Entrepreneurship Theory and Practice*, 28(2), 129-144. <u>https://doi.org/10.1046/j.1540-6520.2003.00035.x</u>
- Philippe, L. (2007). Revisiting the Third Mission of Universities: Toward a Renewed Categorization of University Activities? *Higher Education Policy*, 20(4), 441. <u>https://doi.org/10.1057/palgrave.hep.8300169</u>
- Philpott, K., Dooley, L., O'Reilly, C., & Lupton, G. (2011). The entrepreneurial university: Examining the underlying academic tensions. *Technovation*, 31(4), 161-170. <u>https://doi.org/10.1016/j.technovation.2010.12.003</u>
- Pihlström, S. (2015). The Bloomsbury Companion to Pragmatism. London: Continuum International Publishing.
- Pillai, K. R., Upadhyaya, P., Balachandran, A., & Nidadavolu, J. (2019). Versatile Learning Ecosystem: A Conceptual Framework. *Higher Education for the Future*, 6(1), 85-100. <u>https://doi.org/10.1177/2347631118802653</u>
- Piperopoulos, P., & Dimov, D. (2015). Burst bubbles or build steam? Entrepreneurship education, entrepreneurial self-efficacy, and entrepreneurial intentions. *Journal of Small Business Management*, 53(4), 970-985. https://doi.org/10.1111/jsbm.12116
- Pirnay, F., Surlemont, B., Nlemvo, F., x00E, x00E, & ric. (2003). Toward a Typology of University Spin-offs. Small Business Economics, 21(4), 355-369. <u>https://doi.org/10.1023/A:1026167105153</u>
- Pittaway, L., & Cope, J. (2007). Entrepreneurship Education: A Systematic Review of the Evidence. *International Small Business Journal*, 25(5), 479-510. <u>https://doi.org/10.1177/0266242607080656</u>
- Pollack, J. M., Burnette, J. L., & Hoyt, C. L. (2012). Self-Efficacy in the Face of Threats to Entrepreneurial Success: Mind-Sets Matter. *Basic and Applied Social Psychology*, 34(3), 287-294. <u>https://doi.org/10.1080/01973533.2012.674452</u>
- Poschke, M. (2013). 'Entrepreneurs out of necessity': a snapshot. *Applied Economics Letters*, 20(7), 658-663. <u>https://doi.org/10.1080/13504851.2012.727968</u>
- Poschke, M. (2013). Who becomes an entrepreneur? Labor market prospects and occupational choice. *Journal of Economic Dynamics and Control*, 37(3), 693-710. https://doi.org/10.1016/j.jedc.2012.11.003
- Powell, P., & Walsh, A. (2018). Mutualising the university: achieving community impact through an ecosystem. *Journal of Lifelong Learning*, 64(5), 563-583. <u>https://doi.org/10.1007/s11159-018-9739-5</u>
- Pratt, M. G. (2009). For the lack of a boilerplate: Tips on writing up (and reviewing) qualitative research. *Academy of Management Journal*, 52(5), 856-862. <u>https://doi.org/10.5465/AMJ.2009.44632557</u>
- Pratt, M. G., Sonenshein, S., & Feldman, M. S. (2020). Moving Beyond Templates: A Bricolage Approach to Conducting Trustworthy Qualitative Research. Organizational Research Methods, 25(2), 211-238. <u>https://doi.org/10.1177/1094428120927466</u>
- Preedy, S., Jones, P., Maas, G., & Duckett, H. (2020). Examining the perceived value of extracurricular enterprise activities in relation to entrepreneurial learning processes. *Journal of Small Business and Enterprise Development*, 27(7), 1085-1105. <u>https://doi.org/10.1108/JSBED-12-2019-0408</u>
- Premand, P., Brodmann, S., Almeida, R., Grun, R., & Barouni, M. (2016). Entrepreneurship Education and Entry into Self-Employment Among University Graduates. *World Development*, 77(C), 311-327. <u>https://doi.org/10.1016/j.worlddev.2015.08.028</u>

PricewaterhouseCoopers. (2019). Sri Lanka Startup Report 2019. PricewaterhouseCoopers.

- Priyadharshini, J., & Muthusamy, S. (2016). Technopreneurial Education and its Impact in India. *Asian Journal of Multidimensional Research*, 5(10-11), 1-10.
- Purbasari, R., Wijaya, C., & Rahayu, N. (2019). The Entrepreneurial Ecosystem as A Network-Rich System: A Systematic Mapping Study. Academy of Entrepreneurship Journal, 25(2), 1-17.
- Qian, H., Acs, Z. J., & Stough, R. R. (2013). Regional systems of entrepreneurship: the nexus of human capital, knowledge and new firm formation. *Journal of Economic Geography*, 13(4), 559-587. <u>https://doi.org/10.1093/jeg/lbs009</u>
- Qiang, W. L. L. (2012). Comparative Study of Chinese and Foreign University Entrepreneurship Education System. *Science Mosaic*, 3(3), 185-194.
- QualityAssuranceAgency. (2018). Enterprise and Entrepreneurship Education: Guidance for UK Higher Education Providers. The Quality Assurance Agency for Higher Education, Gloucester, United Kingdom.
- Quan, L., & Zhou, H. (2018). Evaluation of Innovation and Entrepreneurship Education Capability in Colleges and Universities Based on Entropy TOPSIS-A Case Study. *Educational Sciences: Theory & Practice*, 18(5), 994-1004.
- Rae, D. (2014). Graduate Entrepreneurship and Career Initiation in the 'New Era' Economy. Journal of General Management, 40(1), 79-95. https://doi.org/10.1177/030630701404000105
- Rae, D., Martin, L., Antcliff, V., & Hannon, P. (2012). Enterprise and entrepreneurship in English higher education: 2010 and beyond. *Journal of Small Business and Enterprise Development*, 19(3), 380-401. <u>https://doi.org/10.1108/14626001211250090</u>
- Rahim, H. L., Kadir, M. A. B. A., Abidin, Z. Z., Junid, J., Kamaruddin, L. M., Lajin, N. F. M., Buyong, S. Z., & Bakri, A. A. (2015). Entrepreneurship education in Malaysia: A critical review. *Journal of Technology Management and Business*, 2(2), 1-12.
- Rais, S. S., Albina, R. S., Rovzat, A. A., Isita, V. M., Tatyana, V. Y., Lolita, A. D., Fariza, V. M., Venera, M. N., & Linar, G. A. (2016). Innovative Entrepreneurship in Education: A New Look in the Students Training Content and Existing Problems. *International Review of Management And Marketing*, 6(2S), 51-56.
- Ramos-Rodríguez, A.-R., Medina-Garrido, J.-A., Lorenzo-Gómez, J.-D., & Ruiz-Navarro, J. (2010). What you know or who you know? The role of intellectual and social capital in opportunity recognition. *International Small Business Journal*, 28(6), 566-582. https://doi.org/10.1177/0266242610369753
- Ranga, M., Perälampi, J., & Kansikas, J. (2016). The new face of university-business cooperation in Finland. Science and Public Policy, 43(5), 601-612. <u>https://doi.org/10.1093/scipol/scw044</u>

- Ranjan, R. K., & Gautam, A. (2019). Entrepreneurship Education in India: A Critical Assessment. *Asian Journal of Management*, 10(4), 400-404.
- Rasmussen, E., & Borch, O. J. (2010). University capabilities in facilitating entrepreneurship: A longitudinal study of spin-off ventures at mid-range universities. *Research Policy*, 39(5), 602-612
- Rasmussen, E., & Wright, M. (2015). How can universities facilitate academic spin-offs? An entrepreneurial competency perspective. The Journal of Technology Transfer, 40(5), 782-799.
- Rasmussen, E. A., & Sørheim, R. (2006). Action-based entrepreneurship education. *Technovation*, 26(2), 185-194.
- Ratten, V. (2020). Coronavirus (Covid-19) and the entrepreneurship education community. *Journal of Enterprising Communities: People and Places in the Global Economy*, 14(5), 753-764. <u>https://doi.org/10.1108/JEC-06-2020-0121</u>
- Ratten, V., & Jones, P. (2020). Covid-19 and entrepreneurship education: Implications for advancing research and practice. *The International Journal of Management Education*, 19(1), 1-10. <u>https://doi.org/10.1016/j.ijme.2020.100432</u>
- Ratten, V., & Miragaia, D. (2020). Entrepreneurial passion amongst female athletes. Journal of Small Business & Entrepreneurship: Entrepreneurial Decision Making and Behaviour, 32(1), 59-77.
- Ratten, V., & Usmanij, P. (2020). Entrepreneurship education: Time for a change in research direction? *The International Journal of Management Education*, 19(1), 1-8. https://doi.org/10.1016/j.ijme.2020.100367
- Rauch, A., & Hulsink, W. (2015). Putting Entrepreneurship Education Where the Intention to Act Lies: An Investigation Into the Impact of Entrepreneurship Education on Entrepreneurial Behavior. Academy of Management Learning & Education, 14(2), 187-204.
- Redford, D. T., & Fayolle, A. (2014). Stakeholder management and the entrepreneurial university. In *Handbook on the entrepreneurial university*. Cheltenham: Edward Elgar Publishing.
- Regele, M., & Neck, H. (2012). The Entrepreneurship Education Subecosystem in The United States: Opportunities to Increase Entrepreneurial Activity. *Journal of Business* and Entrepreneurship, 23(2), 25-47.
- Reggiani, I. (2017). Beehive challenges to reinforce Asian University entrepreneurship. *Formamente*, 12(2017), 123-136.
- Reuber, A. R., & Fischer, E. M. (1993). The learning experiences of entrepreneurs. *Frontiers* of Entrepreneurship Research, 234-245.

- Reyes, C. N. (2016). Framing the entrepreneurial university: the case of the National University of Singapore. *Journal of Entrepreneurship in Emerging Economies*, 8(2), 134-161. <u>https://doi.org/10.1108/JEEE-09-2015-0046</u>
- Ricci, R., Colombelli, A., & Paolucci, E. (2019). Entrepreneurial activities and models of advanced European science and technology universities. *Management Decision*, 57(12), 3447-3472. <u>https://doi.org/10.1108/MD-11-2018-1237</u>
- Rice, M. P., Fetters, M. L., & Greene, P. G. (2014). University-based entrepreneurship ecosystems: A global study of six educational institutions. *International Journal of Entrepreneurship and Innovation Management*, 18(05), 481–501.
- Rideout, E. C., & Gray, D. O. (2013). Does Entrepreneurship Education Really Work? A Review and Methodological Critique of the Empirical Literature on the Effects of University-Based Entrepreneurship Education. *Journal of Small Business Management*, 51(3), 329-351. <u>https://doi.org/10.1111/jsbm.12021</u>
- Rizzo, U. (2015). Why do scientists create academic spin-offs? The influence of the context. The Journal of Technology Transfer, 40(2), 198-226. <u>https://doi.org/10.1007/s10961-014-9334-2</u>
- Ronstadt, R. (1985). The educated entrepreneurs: A new era of entrepreneurial education is beginning. *American Journal of Small Business*, 10(1), 7–23.
- Roberts, E. B., & Eesley, C. E. (2011). Entrepreneurial impact: The role of MIT. *Foundations* and *Trends in Entrepreneurship*, 7(1-2), 1-149. <u>https://doi.org/10.1561/0300000030</u>
- Robson, P. J. A., Wijbenga, F., & Parker, S. C. (2009). Entrepreneurship and Policy: Challenges and Directions for Future Research. *International Small Business Journal*, 27(5), 531-535. <u>https://doi.org/10.1177/0266242609338753</u>
- Rocco, T. S., & Plakhotnik, M. S. (2009). Literature Reviews, Conceptual Frameworks, and Theoretical Frameworks: Terms, Functions, and Distinctions. *Human Resource Development Review*, 8(1), 120-130. <u>https://doi.org/10.1177/1534484309332617</u>
- Rodríguez-López, Á., & Souto. (2020). Empowering entrepreneurial capacity: training, innovation and business ethics. *Eurasian Business Review*, 10(1), 23-43. https://doi.org/10.1007/s40821-019-00133-w
- Ronstadt, R. (1985). The educated entrepreneurs: a new era of entrepreneurial education is beginning. *American Journal of Small Business*, 11(4), 37-54.
- Ronstadt, R., Vesper, K. H., & McMullan, W. E. (1988). Entrepreneurship: Today Courses, Tomorrow Degrees? *Entrepreneurship Theory and Practice*, 13(1), 7-13. <u>https://doi.org/10.1177/104225878801300102</u>
- Rothaermel, F. T., Agung, S. D., & Jiang, L. (2007). University entrepreneurship: a taxonomy of the literature. *Industrial and Corporate Change*, 16(4), 691-791. https://doi.org/10.1093/icc/dtm023

- Roy, R., & Das, N. (2016). Cultivating evidence-based entrepreneurship education (EBEE): A review on synchronization process behind entrepreneurial spirit. *Business and Economics Review*, 25(2), 98-114.
- Saeed, S., Muffatto, M., & Yousafzai, S. (2014). A Multi-level Study of Entrepreneurship Education among Pakistani University Students. *Entrepreneurship Research Journal*, 4(3), 297-321.
- Saeed, S., Yousafzai, S. Y., Yani-De-Soriano, M., & Muffatto, M. (2015). The role of perceived university support in the formation of students' entrepreneurial intention. *Journal of Small Business Management*, 53(4), 1127-1145. <u>https://doi.org/10.1111/jsbm.12090</u>
- Saji, B. S., & Nair, A. R. (2018). Effectiveness of innovation and entrepreneurship education in UAE higher education. *Academy of Strategic Management Journal*, 17(4), 1-12.
- Sam, C., & Sijde, P. (2014). Understanding the concept of the entrepreneurial university from the perspective of higher education models. *The International Journal of Higher Education Research*, 68(6), 891-908. <u>https://doi.org/10.1007/s10734-014-9750-0</u>
- Sambo, W. (2018). A Conceptual Study of an Intrapreneurship Ecosystem at South African Universities. *Problemy Zarzadzania*, 2018(73), 192-215. <u>https://doi.org/10.7172/1644-9584.73.12</u>
- Sánchez, J. C. (2013). The impact of an entrepreneurship education program on entrepreneurial competencies and intention. *Journal of Small Business Management*, 51(3), 447-465. <u>https://doi.org/10.1111/jsbm.12025</u>
- Santandreu-Mascarell, C., Garzon, D., & Knorr, H. (2013). Entrepreneurial and innovative competences, are they the same? *Management Decision*, 51(5), 1084-1095. https://doi.org/10.1108/MD-11-2012-0792
- Saras, D. S. (2001). Causation and Effectuation: Toward a Theoretical Shift from Economic Inevitability to Entrepreneurial Contingency. *The Academy of Management Review*, 26(2), 243-263. <u>https://doi.org/10.2307/259121</u>
- Sarasvathy, S. D. (2004). The questions we ask and the questions we care about: reformulating some problems in entrepreneurship research. *Journal of Business Venturing*, 19(5), 707-717. <u>https://doi.org/10.1016/j.jbusvent.2003.09.006</u>
- Saunders, M., & Lewis, P. (2017). Doing Research in Business and Management. London: Pearson Education Limited.
- Sautet, F. (2013). Local and Systemic Entrepreneurship: Solving the Puzzle of Entrepreneurship and Economic Development. *Entrepreneurship Theory and Practice*, 37(2), 387-402. <u>https://doi.org/10.1111/j.1540-6520.2011.00469.x</u>

- Schlaegel, C., & Koenig, M. (2014). Determinants of Entrepreneurial Intent: A Meta-Analytic Test and Integration of Competing Models. *Entrepreneurship Theory and Practice*, 38(2), 291-332. <u>https://doi.org/10.1111/etap.12087</u>
- Schmidt, J., & Molkentin, K. (2015). Building And Maintaining a Regional Inter-University Ecosystem for Entrepreneurship: Entrepreneurship Education Consortium. *Journal of Entrepreneurship Education*, 18(1), 157-168.

Schumpeter, J. A. (1934). The theory of economic development. Harvard University Press.

Secundo, G., Del Vecchio, P., Simeone, L., & Schiuma, G. (2020). Creativity and stakeholders' engagement in open innovation: Design for knowledge translation in technology-intensive enterprises. *Journal of Business Research*, *119*, 272-282.

Secundo, G., Mele, G., Del Vecchio, P., & Degennaro, G. (2020). Knowledge spillover creation in university-based entrepreneurial ecosystem: the role of the Italian "Contamination Labs". *Knowledge Management Research & Practice*, 19(1), 137-151.

- Sekaran, U., & Bougie, R. (2016). Research Methods for Business: A Skill Building Approach. New Jersey: John Wiley & Sons.
- Selvarajah, C., Le, T. D., & Sukunesan, S. (2019). The Vietnam project: developing conceptual knowledge on cross-cultural skills for training in SME internationalization. Asia Pacific Business Review, 25(3), 338-366. <u>https://doi.org/10.1080/13602381.2019.1598076</u>
- Senarath, S., Patabendige, S., & Amarathunga, S. (2017). Sri Lankan Graduate Labour Market: A Status Mismatch. *Journal of Economics, Business and Management*, 5(1), 63-68. <u>http://www.joebm.com/vol5/487-MM0032.pdf</u>
- Shane, S. (2003). A General Theory of Entrepreneurship: The Individual-opportunity Nexus. Cheltenham: Edward Elgar Publishing
- Shane, S. (2004). Academic entrepreneurship: university spinoffs and wealth creation. Cheltenham: Edward Elgar Publishing
- Shane, S., & Venkataraman, S. (2000). The Promise of Entrepreneurship as a Field of Research. *The Academy of Management Review*, 25(1), 217-226. <u>https://doi.org/10.2307/259271</u>
- Sharma, L. (2017). Teachers' perspective on institutional barriers to academic entrepreneurship a case of Uttarakhand state, India. *Teachers and Teaching*, 23(4), 436-450. <u>https://doi.org/10.1080/13540602.2016.1205017</u>
- Shil, M., Shahriar, M., Sultana, S., Rahman, S.-N., & Zayed, N. (2020). Introduction to university-based entrepreneurship ecosystem (U-BEE): a model case study from Bangladesh. *International Journal of Entrepreneurship*, 24(1), 1-9.

- Shinnar, R. S., Giacomin, O., & Janssen, F. (2012). Entrepreneurial Perceptions and Intentions: The Role of Gender and Culture. *Entrepreneurship Theory and Practice*, 36(3), 465-493. <u>https://doi.org/10.1111/j.1540-6520.2012.00509.x</u>
- Shirokova, G., Osiyevskyy, O., Morris, M. H., & Bogatyreva, K. (2017). Expertise, university infrastructure and approaches to new venture creation: assessing students who start businesses. *Entrepreneurship & Regional Development*, 29(9-10), 912-944. <u>https://doi.org/10.1080/08985626.2017.1376516</u>
- Shook, C. L., Priem, R. L., & McGee, J. E. (2003). Venture Creation and the Enterprising Individual: A Review and Synthesis. *Journal of Management*, 29(3), 379-399. <u>https://doi.org/10.1016/S0149-2063(03)00016-3</u>
- Siegel, D. S., & Wright, M. (2015). Academic Entrepreneurship: Time for a Rethink? *British Journal of Management*, 26(4), 582-595. <u>https://doi.org/10.1111/1467-8551.12116</u>
- Silveyra, G., Herrero, Á., & Pérez, A. (2021). Model of Teachable Entrepreneurship Competencies (M-TEC): Scale development. *The International Journal of Management Education*, 19(1), 100392. <u>https://doi.org/10.1016/j.ijme.2020.100392</u>
- Sittarage, N. (2018). English Language Education in Sri Lanka Link with the Learners' Motivational Factors. Humanising Language Teaching. Retrieved from <u>https://www.hltmag.co.uk/aug18/english-language-education-in-sri-lanka</u>
- Slone, D. J. (2009). Visualizing qualitative information. *Qualitative report*, 14(3), 489-497.
- Smith, H. L., & Bagchi-Sen, S. (2012). The research university, entrepreneurship and regional development: Research propositions and current evidence. *Entrepreneurship* & Regional Development, 24(5-6), 383-404. <u>https://doi.org/10.1080/08985626.2011.592547</u>
- Snyder, H. (2019). Literature review as a research methodology: An overview and guidelines. Journal of Business Research, 104, 333-339. https://doi.org/10.1016/j.jbusres.2019.07.039
- Solesvik, M., & Westhead, P. (2019). Fostering of Entrepreneurship Competencies and Entrepreneurial Intentions in a Weak Ecosystem. *Foresight and STI Governance*, 13(04), 60–68. <u>https://doi.org/10.17323/2500-2597.2019.4.60.68</u>
- Solesvik, M. Z., Westhead, P., Kolvereid, L., & Matlay, H. (2012). Student intentions to become self-employed: the Ukrainian context. *Journal of Small Business and Enterprise Development*, 19(3), 441-460. https://doi.org/10.1108/14626001211250153
- Solomon, G. (2007). An examination of entrepreneurship education in the United States. Journal of Small Business and Enterprise Development, 14(2), 168-182. https://doi.org/10.1108/14626000710746637

- Solomon, G. T., Weaver, K. M., & Fernald, L. W. (1994). A Historical Examination of Small Business Management and Entrepreneurship Pedagogy. *Simulation & Gaming*, 25(3), 338-352. <u>https://doi.org/10.1177/1046878194253003</u>
- Song, Y., Dana, L. P., & Berger, R. (2021). The entrepreneurial process and online social networks: forecasting survival rate. *Small Business Economics*, 56(3), 1171-1190. <u>https://doi.org/10.1007/s11187-019-00261-7</u>
- Soto-Simeone, A., Sirén, C., & Antretter, T. (2020). New Venture Survival: A Review and Extension. *International Journal of Management Reviews*, 22(4), 378-407. https://doi.org/10.1111/ijmr.12229
- Souitaris, V., Zerbinati, S., & Al-Laham, A. (2007). Do entrepreneurship programmes raise entrepreneurial intention of science and engineering students? The effect of learning, inspiration and resources. *Journal of Business Venturing*, 22(4), 566-591. <u>https://doi.org/10.1016/j.jbusvent.2006.05.002</u>
- Soyupak, O., & Bağlı, H. H. (2019). How Design Enhances Startup Ecosystem: Design Education Centered Interventions. *Online Journal of Art & Design*, 07(04).
- Spigel, B. (2016). Developing and governing entrepreneurial ecosystems: the structure of entrepreneurial support programs in Edinburgh, Scotland. *International Journal of Innovation and Regional Development*, 7(2), 141-160.
- Spigel, B. (2017). The Relational Organization of Entrepreneurial Ecosystems. *Entrepreneurship Theory and Practice*, 41(1), 49-72. <u>https://doi.org/10.1111/etap.12167</u>
- Spigel, B., & Harrison, R. (2018). Toward a process theory of entrepreneurial ecosystems. *Strategic Entrepreneurship Journal*, 12(1), 151-168. <u>https://doi.org/10.1002/sej.1268</u>
- Stam, E. (2015). Entrepreneurial Ecosystems and Regional Policy: A Sympathetic Critique. *European Planning Studies*, 23(9), 1759-1769. <u>https://doi.org/10.1080/09654313.2015.1061484</u>
- StartupGenome. (2021). The Global Startup Ecosystem Report 2020. Retrieved from <u>https://startupgenome.com/report/gser2021</u>
- Stewart, A., & Nel, D. (2009). Blended and online learning: student perceptions and performance. *Interactive Technology and Smart Education*, 6(3), 140-155. <u>https://doi.org/10.1108/17415650911005366</u>
- Sun, L.-Y., Aryee, S., & Law, K. S. (2007). High-Performance Human Resource Practices, Citizenship Behavior, and Organizational Performance: A Relational Perspective. *Academy of Management Journal*, 50(3), 558-577. <u>https://doi.org/10.5465/AMJ.2007.25525821</u>

- Sutter, C., Bruton, G. D., & Chen, J. (2019). Entrepreneurship as a solution to extreme poverty: A review and future research directions. *Journal of Business Venturing*, 34(1), 197-214. <u>https://doi.org/10.1016/j.jbusvent.2018.06.003</u>
- Sutton, R. I., & Staw, B. M. (1995). What Theory is Not. *Administrative Science Quarterly*, 40(3), 371-384. <u>https://doi.org/10.2307/2393788</u>
- Swales, J., & Najjar, H. (1987). The Writing of Research Article Introductions. *Written* Communication, 4(2), 175-191. <u>https://doi.org/10.1177/0741088387004002004</u>
- Tanja Zdolšek, D., & Karin, Š. (2018). Conceptual Research Model for Studying Students' Entrepreneurial Competencies. Nase Gospodarstvo, 64(4), 23-33. <u>https://doi.org/10.2478/ngoe-2018-0020</u>
- Taylor, D. W., & Thorpe, R. (2004). Entrepreneurial learning: a process of co-participation. Journal of Small Business and Enterprise Development, 11(2), 203-211. <u>https://doi.org/10.1108/14626000410537146</u>
- Teixeira, A. A. C., Teixeira, A. A. C., Forte, R. P., & Forte, R. P. (2017). Prior education and entrepreneurial intentions: the differential impact of a wide range of fields of study. *Review of Managerial Science*, 11(2), 353-394. <u>https://doi.org/10.1007/s11846-015-0188-2</u>
- Tejero, A., Pau, I., & Leon, G. (2019). Analysis of the Dynamism in University-Driven Innovation Ecosystems Through the Assessment of Entrepreneurship Role. *IEEE access*, 7, 89869-89885.<u>https://doi.org/10.1109/ACCESS.2019.2926988</u>
- Theodoraki, C., Messeghem, K., & Rice, M. P. (2018). A social capital approach to the development of sustainable entrepreneurial ecosystems: an explorative study. *Small Business Economics*, 51(1), 153-170. <u>https://doi.org/10.1007/s11187-017-9924-0</u>
- Thompson, E. R. (2009). Individual Entrepreneurial Intent: Construct Clarification and Development of an Internationally Reliable Metric. *Entrepreneurship Theory and Practice*, 33(3), 669-694. <u>https://doi.org/10.1111/j.1540-6520.2009.00321.x</u>
- Thomsen, B., Muurlink, O., & Best, T. (2018). The political ecology of university-based social entrepreneurship ecosystems. *Journal of Enterprising Communities: People and Places in the Global Economy*, 12(2), 199-219. <u>https://doi.org/10.1108/JEC-08-2017-0068</u>
- Thrane, C., Blenker, P., Korsgaard, S., & Neergaard, H. (2016). The promise of entrepreneurship education: Reconceptualizing the individual–opportunity nexus as a conceptual framework for entrepreneurship education. *International Small Business Journal*, 34(7), 905-924. <u>https://doi.org/10.1177/0266242616638422</u>
- Tillmar, M. (2006). Swedish tribalism and Tanzanian entrepreneurship: preconditions for trust formation. *Entrepreneurship & Regional Development*, 18(2), 91-107. <u>https://doi.org/10.1080/08985620500531956</u>

- Tkachev, A., & Kolvereid, L. (1999). Self-employment intentions among Russian students. Entrepreneurship & Regional Development, 11(3), 269-280. https://doi.org/10.1080/089856299283209
- Trần, T. B. L. (2014). Entrepreneurial Skills for Women's Business Success Lessons Around the World and Implications for Vietnam. *VNU Journal of Economic and Business*, 30(2), 55-64.
- Tok, M. E., Olawuyi, D. S., & D'Alessandro, C. (2018). Advancing Clean Technology Entrepreneurship in the Middle East and North African (mena) Region: Law, Education and Policy Imperatives. *Global Journal of Comparative Law*, 7(1), 115-133.
- Tomy, S., & Pardede, E. (2020). An entrepreneurial intention model focusing on higher education. *International Journal of Entrepreneurial Behavior & Research*, 26(7), 1423-1447. <u>https://doi.org/10.1108/IJEBR-06-2019-0370</u>
- Toronto, C., & Remington, R. (2020). A Step-by-Step Guide to Conducting an Integrative Review. Switzerland: Springer International Publishing.
- Torraco, R. J. (2005). Writing Integrative Literature Reviews: Guidelines and Examples. *Human Resource Development Review*, 4(3), 356-367.
- Torraco, R. J. (2016). Writing Integrative Literature Reviews: Using the Past and Present to Explore the Future. *Human Resource Development Review*, 15(4), 404-428.
- Torres Valdés, R. M., Lorenzo Álvarez, C., Castro Spila, J., & Santa Soriano, A. (2019). Relational university, learning and entrepreneurship ecosystems for sustainable tourism. *Journal of Science and Technology Policy Management*, 10(4), 905-926.
- Toutain, O., Mueller, S., Bornard, F., Cohendet, P., Mazouz, B., Dejardin, M., Luc, S., & Thurik, R. (2019). Decoding Entrepreneurship Education Ecosystems (EEE): A Cross-European Study in Primary, Secondary Schools and Vocational Training. *Management International*, 23(5), 47-65.
- Towers, N., Santoso, A. S., Sulkowski, N., & Jameson, J. (2020). Entrepreneurial capacitybuilding in HEIs for embedding entrepreneurship and enterprise creation – a tripartite approach. *International Journal of Retail & Distribution Management*, 48(8), 881-899.
- Tracy, S. J. (2010). Qualitative Quality: Eight "Big-Tent" Criteria for Excellent Qualitative Research. *Qualitative inquiry*, 16(10), 837-851.
- Trivedi, R. (2016). Does university play significant role in shaping entrepreneurial intention? A cross-country comparative analysis. *Journal of Small Business and Enterprise Development*, 23(3), 790-811. <u>https://doi.org/10.1108/JSBED-10-2015-0149</u>

- Trivedi, R. H. (2017). Entrepreneurial-intention constraint model: A comparative analysis among post-graduate management students in India, Singapore and Malaysia. *International Entrepreneurship and Management Journal*, 13(4), 1239-1261. <u>https://doi.org/10.1007/s11365-017-0449-4</u>
- Trowler, P., & Trowler, V. (2010). Student engagement evidence summary. The Higher Education Academy. Retrieved from <u>https://eprints.lancs.ac.uk/id/eprint/61680/1/Deliverable_2._Evidence_Summary._No</u> v 2010.pdf
- Ucbasaran, D., Westhead, P., & Wright, M. (2008). Opportunity Identification and Pursuit: Does an Entrepreneur's Human Capital Matter? *Small Business Economics*, 30(2), 153-173. <u>https://doi.org/10.1007/s11187-006-9020-3</u>
- Unicef. (2020). South Asia. Retrieved from <u>https://www.unicef.org/rosa/press-releases/more-half-south-asian-youth-are-not-track-have-education-and-skills-necessary</u>
- United Nations. (2020). Member State Sri Lanka. Retrieved from, https://sustainabledevelopment.un.org/memberstates/srilanka
- University of Grants Commission (2019). Admission to Undergraduate Courses of the Universities in Sri Lanka. Retrieved from <u>https://www.ugc.ac.lk/downloads/admissions/Handbook_2019_20/ENGLISH%20HA</u> <u>NDBOOK.pdf</u>
- University of Grants Commission (2020). University Information. Retrieved from <u>https://www.ugc.ac.lk/en/universities-and-institutes/university.html</u>
- Urbano, D., Aparicio, S., & Audretsch, D. (2019). Twenty-five years of research on institutions, entrepreneurship, and economic growth: what has been learned? *Small Business Economics*, 53(1), 21-49. <u>https://doi.org/10.1007/s11187-018-0038-0</u>.
- Uzzi, B. (1997). Social Structure and Competition in Interfirm Networks: The Paradox of Embeddedness. *Administrative science quarterly*, 42(1), 35-67. <u>https://doi.org/10.2307/2393808</u>
- van de Heyde, V., & Siebrits, A. (2019). The ecosystem of e-learning model for higher education. *South African Journal of Science*, 115(5/6). <u>https://doi.org/10.17159/sajs.2019/5808</u>
- van Gelderen, M., Brand, M., van Praag, M., Bodewes, W., Poutsma, E., & van Gils, A. (2008). Explaining entrepreneurial intentions by means of the theory of planned behaviour. *Career Development International*, 13(6), 538-559. <u>https://doi.org/10.1108/13620430810901688</u>
- van Praag, C. M., & Versloot, P. H. (2007). What is the value of entrepreneurship? A review of recent research. *Small Business Economics*, 29(4), 351-382. <u>https://doi.org/10.1007/s11187-007-9074-x</u>

- Vanevenhoven, J. (2013). Advances and Challenges in Entrepreneurship Education. Journal of Small Business Management, 51(3), 466-470. <u>https://doi.org/10.1111/jsbm.12043</u>
- van't Land, H., Corcoran, A., & Iancu, D.-C. (2021). The Promise of Higher Education: Essays in Honour of 70 Years of IAU. Switzerland: Springer International Publishing.
- Vega-Gómez, F. I., Miranda González, F. J., & Pérez-Mayo, J. (2020). Analyzing the Effects of Institutional- and Ecosystem-Level Variables on University Spin-Off Performance. SAGE open, 10(2). <u>https://doi.org/10.1177/2158244020931116</u>
- Vesper, K. H., & Gartner, W. B. (1997). Measuring progress in entrepreneurship education. Journal of Business Venturing, 12(5), 403-421. <u>https://doi.org/10.1016/S0883-9026(97)00009-8</u>
- Volkmann, C., Fichter, K., Klofsten, M., & Audretsch, D. B. (2021). Sustainable entrepreneurial ecosystems: an emerging field of research. *Small Business Economics*, 56(3), 1047-1055. <u>https://doi.org/10.1007/s11187-019-00253-7</u>
- Von Bertalanffy, L. (1972). The History and Status of General Systems Theory. *Academy of Management Journal*, 15(4), 407-426. <u>https://doi.org/10.2307/255139</u>
- Von Graevenitz, G., Harhoff, D., & Weber, R. (2010). The effects of entrepreneurship education. *Journal of Economic Behavior and Organization*, 76(1), 90-112. https://doi.org/10.1016/j.jebo.2010.02.015
- Vozikis, G., Solomon, G., Winkel, D., Morris, M. H., Webb, J. W., Fu, J., & Singhal, S. (2013). A Competency-Based Perspective on Entrepreneurship Education: Conceptual and Empirical Insights. *Journal of Small Business Management*, 51(3), 352. <u>https://doi.org/10.1111/jsbm.12023</u>
- Wadee, A. A., & Padayachee, A. (2017). Higher Education: Catalysts for the Development of an Entrepreneurial Ecosystem, or ... Are We the Weakest Link? *Science, Technology* and Society, 22(2), 284-309. <u>https://doi.org/10.1177/0971721817702290</u>
- Walmsley, A. (2018). Entrepreneurship in Tourism. New York: Routledge
- Walsh, A., & Powell, P. (2018). Supporting student innovation through an engagement, employability and employment ecosystem. *Higher Education, Skills and Work-Based Learning*, 8(1), 15-28. <u>https://doi.org/10.1108/HESWBL-08-2017-0054</u>
- Walter, S. G., & Block, J. H. (2016). Outcomes of entrepreneurship education: An institutional perspective. *Journal of Business Venturing*, 31(2), 216-233. <u>https://doi.org/10.1016/j.jbusvent.2015.10.003</u>
- Walter, S. G., & Dohse, D. (2012). Why mode and regional context matter for entrepreneurship education. *Entrepreneurship & Regional Development* 24(9-10), 807-835. <u>https://doi.org/10.1080/08985626.2012.721009</u>

- Walter, S. G., Parboteeah, K. P., & Walter, A. (2013). University Departments and Self-Employment Intentions of Business Students: A Cross-Level Analysis. *Entrepreneurship Theory and Practice*, 37(2), 175-200. <u>https://doi.org/10.1111/j.1540-6520.2011.00460.x</u>
- Watchravesringkan, K., Hodges, N. N., Yurchisin, J., Hegland, J., Karpova, E., Marcketti, S., & Yan, R. n. (2013). Modeling Entrepreneurial Career Intentions among Undergraduates: An Examination of the Moderating Role of Entrepreneurial Knowledge and Skills. *Family and Consumer Sciences Research Journal*, 41(3), 325-342.
- Webber, D. J., Kitagawa, F., & Plumridge, A. (2020). Student entrepreneurial propensities in the individual-organisational-environmental nexus. *Economic Issues*, 25(1), 31-59.
- Welter, F. (2011). Contextualizing Entrepreneurship—Conceptual Challenges and Ways Forward. *Entrepreneurship Theory and Practice*, 35(1), 165-184. <u>https://doi.org/10.1111/j.1540-6520.2010.00427.x</u>
- Wennberg, K., Wiklund, J., & Wright, M. (2011). The effectiveness of university knowledge spillovers: Performance differences between university spinoffs and corporate spinoffs. *Research Policy*, 40(8), 1128-1143. <u>https://doi.org/10.1016/j.respol.2011.05.014</u>
- Wennekers, S., & Thurik, R. (1999). Linking Entrepreneurship and Economic Growth. Small Business Economics, 13(1), 27-56. <u>https://doi.org/10.1023/A:1008063200484</u>
- Weerasinghe, R. N., Jayewardane, A. K. W., & Deshani, A. L. (2016, December). Systematic View of Innovation: Role of Universities in National Innovation System of Sri Lanka. In University of Sri Jayewardenepura, Sri Lanka, 13th International Conference on Business Management (ICBM).
- Weerasinghe, R. N., & Jayawardane, A. K. (2018). University-Industry Partnerships for Innovation: Sri Lankan Experiences. Annals of Spiru Haret University, Economic Series, 9(1), 25-45.
- Weerasinghe, R. N., & Jayawardane, A. K. W. (2019). The Art of Crafting Actionable National Innovation Policy: The Case of Sri Lanka. *Journal of Economics and Business*, 2(4), 1226-1237.
- Weerasinghe, R. N. (2020). Entrepreneurial Education and Entrepreneurial Intention among Ordinary Level Students in Kelaniya Education Zone in Sri Lanka. *Educational Research*, 2(4), 122-131.
- Whittemore, R., & Knafl, K. (2005). The integrative review: updated methodology. *Journal* of Advanced Nursing, 52(5), 546-553. <u>https://doi.org/10.1111/j.1365-2648.2005.03621.x</u>
- Whittington, R. (1992). Putting giddens into action: social systems and managerial agency. Journal of *Management Studies*, 29(6), 693-712. <u>https://doi.org/10.1111/j.1467-6486.1992.tb00685.x</u>

- Wickramaratne, A., Kiminami, A., & Yagi, H. (2014). Entrepreneurial Competencies and Entrepreneurial Orientation of Tea Manufacturing Firms in Sri Lanka. Asian Social Science, 10(18), 50-62. <u>https://doi.org/10.5539/ass.v10n18p50</u>
- Wickramasinghe, V. (2018). Higher education in state universities in Sri Lanka. *International Journal of Educational Management*, 32(3), 463-478. <u>https://doi.org/10.1108/IJEM-01-2017-0028</u>
- Wieser, D. (2020). Integrating technology into the learning process of higher education: A creative inquiry. *Industry & Higher Education*, 34(3), 138-150. https://doi.org/10.1177/0950422219895773
- Wilson, F., Kickul, J., & Marlino, D. (2007). Gender, Entrepreneurial Self–Efficacy, and Entrepreneurial Career Intentions: Implications for Entrepreneurship Education1. *Entrepreneurship Theory and Practice*, 31(3), 387-406. <u>https://doi.org/10.1111/j.1540-6520.2007.00179.x</u>
- Wilson, K. E., Vyakarnam, S., Volkmann, C., Mariotti, S., & Rabuzzi, D. (2009). Educating the next wave of entrepreneurs: Unlocking entrepreneurial capabilities to meet the global challenges of the 21st century. In *World Economic Forum: A Report* of the Global Education Initiative.
- Winkler, C., Saltzman, E., & Yang, S. (2018). Improvement of practice in entrepreneurship education through action research: The case of coworking at a nonresidential college. *Entrepreneurship Education and Pedagogy*, 1(2), 139-165.
- Wong, P. K., Ho, Y. P., & Autio, E. (2005). Entrepreneurship, Innovation and Economic Growth: Evidence from GEM data. *Small Business Economics*, 24(3), 335-350. <u>https://doi.org/10.1007/s11187-005-2000-1</u>
- Wraae, B., & Thomsen, J. (2019). Introducing a New Framework for Understanding Learning in an Entrepreneurship Education Ecosystem. *Journal of Higher Education Theory* and Practice, 19(2), 170-184.
- Wraae, B., & Walmsley, A. (2020). Behind the scenes: spotlight on the entrepreneurship educator. *Education* + *Training*, 62(3), 255-270. <u>https://doi.org/10.1108/ET-01-2019-0009</u>
- Wright, M. (2014). Academic entrepreneurship, technology transfer and society: where next? *The Journal of Technology Transfer*, 39(3), 322-334.
- Wright, M., Lockett, A., Clarysse, B., & Binks, M. (2006). University spin-out companies and venture capital. *Research Policy*, 35(4), 481-501.
- Wright, M., Siegel, D., & Mustar, P. (2017). An emerging ecosystem for student start-ups. *The Journal of Technology Transfer*, 42(4), 909-922.
- Wright, M., & Stigliani, I. (2013). Entrepreneurship and growth. *International Small Business Journal*, 31(1), 3-22. <u>https://doi.org/10.1177/0266242612467359</u>

- Wurth, B., Stam, E., & Spigel, B. (2021). Toward an Entrepreneurial Ecosystem Research Program. *Entrepreneurship Theory and Practice*, 46(3), 729-778. <u>https://doi.org/10.1177/1042258721998948</u>
- Xiaozhou, X., & Hao, N. (2018). Towards 2050: Vision and Strategies of Building an Innovation and Entrepreneurship Education Ecosystem. *China Higher Education Research*, (1), 29.
- Xie, J., Song, C., & Chen, S. (2014). An Elementary Research on Green Entrepreneurship Education Of University Students in Ecological Perspective. *Journal of Ningbo* University of Technology, (4), 24.
- Xuan Truong, N. (2020). Factors Affecting Entrepreneurial Decision of Nascent Entrepreneurs Belonging Generation Y in Vietnam. *The Journal of Asian Finance*, Economics and Business, 7(8), 407-417.
- Yan, L., & Qiang, Y. (2020). Application of "Trinity" Experiential Teaching in Innovation and Entrepreneurship Education in Chinese Universities. *Curriculum and Teaching*, 3(2020), 39-45.
- Yang, J. (2013). The theory of planned behavior and prediction of entrepreneurial intention among Chinese undergraduates. *Social Behavior and Personality: An International Journal*, 41(3), 367-376. <u>https://doi.org/10.2224/sbp.2013.41.3.367</u>
- Yan-lan, L., & Zai-xi, Z. (2010). Singapore Nanyang Technological University Innovation Education Review from the Perspective of Humanistic Education. *Journal of Yichun College*, (7), 49.
- Zhaoxin, H., & Yannan, L. (2015). How to Reform Entrepreneurship Education in the Era of Innovation and Entrepreneurship. *Research in Educational Development*, (23), 10.
- Yi, G., & Uyarra, E. (2018). Process Mechanisms for Academic Entrepreneurial Ecosystems: Insights from a Case Study in China. *Science, Technology & Society*, 23(1), 85-106.
- Yousheng, L., & Dubin, C. (2015). A Study of the Whole Career Education from the Perspective of the University Entrepreneurial Talent Training. *Journal of China University of Petroleum*, (3), 2902-2906. <u>https://doi.org/10.1145/3482632.3487537</u>
- Yousheng, L., & Dubin, C. (2017). University Entrepreneurial Education Efficiency Evaluation Based on DEA. *Journal of China University of Petroleum*, (2), 710-714.
- Yu, C. W. (2018). Understanding the ecosystems of Chinese and American entrepreneurship education. *Journal of Entrepreneurship Education*, 21(2), 1-18.
- Yu, M.-C., Goh, M., Kao, H.-Y., & Wu, W.-H. (2017). A comparative study of entrepreneurship education between Singapore and Taiwan. *Management Decision*, 55(7), 1426-1440.

- Yusof, M., Siddiq, M., & Nor, L. (2009). An Integrated Model of a University's Entrepreneurial Ecosystem. *Journal of Asia Entrepreneurship and Sustainability*, 5(1), 57-77.
- Zhang, S. X., Foo, M.-D., & Vassolo, R. S. (2021). The ramifications of effectuation on biases in entrepreneurship – Evidence from a mixed-method approach. *Journal of Business Venturing Insights*, 15(2021), 1-8. <u>https://doi.org/10.1016/j.jbvi.2021.e00238</u>
- Zhang, W., Zhao, W., Gao, Y., & Xiao, Z. (2020). How do managerial ties influence the effectuation and causation of entrepreneurship in China? The role of entrepreneurs' cognitive bias. *Asia Pacific Business Review*, 26(5), 613-641.
- Zhaoxin, H., & Yannan, L. (2015). How to Reform Entrepreneurship Education in the Era of Innovation and Entrepreneurship. *Research in Educational Development*, (23), 10.
- Zhaoxin, H., & Zhiqiang, W. (2017). On the Construction of Entrepreneurship Education Ecosystem in Universities. *Educational Research*, (4), 6.
- Zobnina, M., Korotkov, A., & Rozhkov, A. (2019). Structure, Challenges and Opportunities for Development of Entrepreneurial Education in Russian Universities. *Foresight and STI Governance*, 13(4), 69-81.

APPENDIXES

Appendix A: Research ethics

A1 Ethics approval email



donotreply@infonetica.net To: Christopher Selvarajah Cc: RES Ethics; NILUSHA GALLAGE; Richard Laferriere

Letter.pdf V 147 KB $\Im \ \leftarrow \ \ll \ \rightarrow \ \cdots$

Thu 17/09/2020 6:03 PM

Dear All,

Ref: 20202915-5183 : Developing Graduate Entrepreneurs: A model of Entrepreneurship Education and Experience for Private Higher Education Institutes in Colombo, Sri Lanka Your ethics application has been approved. Please see the attachment for details of the approval. Please contact the Swinburne <u>Research Ethics Office</u> if you have any queries. Regards, Ms Leah Barham **Research Ethics Office** Swinburne University of Technology

P: +61 3 9214 8145 | E: resethics@swin.edu.au

A2 Ethics approval certificate

Swinburne University of Technology Human Research Ethics Committee

Approval certificate



17/09/2020

The ethics application for your project Developing Graduate Entrepreneurs: A model of Entrepreneurship Education and Experience for Private Higher Education Institutes in Colombo, Sri Lanka has been approved.

Chief Investigator: Christopher Selvarajah

Ref: 20202915-5183

Approved Duration: 17/09/2020 to 17/09/2022

I refer to the ethical review of the above project protocol by Swinburne's Human Research Ethics Committee (SUHREC) or its sub-committees.

I am pleased to advise that, as submitted to date, the project may proceed in line with standard on-going ethics clearance conditions outlined below.

- The approved duration is as shown above unless an extension request is subsequently approved.
- All human research activity undertaken under Swinburne auspices must conform to Swinburne and external regulatory standards, including the National Statement on Ethical Conduct in Human Research (2018) and with respect to secure data use, retention and disposal.
- The named Swinburne Chief Investigator/Supervisor remains responsible for any personnel appointed to or associated with the project being made aware of ethics
 clearance conditions, including research and consent procedures or instruments approved. Any change in Chief Investigator/Supervisor, and addition or removal of other
 personnel/students from the project, requires timely notification and SUHREC endorsement.
- The above project has been approved as submitted for ethical review by or on behalf of SUHREC. Amendments to approved procedures or instruments ordinarily require
 prior ethical appraisal/clearance from SUHREC for approval. SUHREC must be notified immediately or as soon as possible thereafter of (a) any serious or unexpected
 adverse effects on participants and any redress measures; (b) proposed changes in protocols; and (c) unforeseen events which might affect continued ethical
 acceptability of the project.
- At a minimum, an annual report on the progress of the project is required as well as at the conclusion (or abandonment) of the project.
- A duly authorised external or internal audit of the project may be undertaken at any time.
- Please forward this approval certificate to relevant members of the project team.

This research project was approved during COVID-19 restrictions. The conduct of the research during this period should reflect any changes in relation to university and government COVID-19 mandates in the relevant jurisdictions. To ensure you have accommodated these mandates please refer to the Swinburne Ethics COVID-19 website <u>here</u>.

The following investigators have been approved to work on the project:

Chief Investigator

Christopher Selvarajah

Associate Investigators

Richard Laferriere

Student Investigators

Nilusha Gallage

Please contact the Swinburne Research Ethics Office if you have any queries.

Regards, Ms Leah Barham on behalf of Research Ethics Office Swinburne University of Technology P: +61 3 9214 8145 | E: resethics@swin.edu.au

Page 1 of 1

A3 Change of chief investigator

20/08/2021

Ref: 20212915-8256 : Developing Graduate Entrepreneurs: A model of Entrepreneurship Education and Experience for Private Higher Education Institutes in Colombo, Sri Lanka



Approved Duration: 17/09/2020 to 17/09/2022

Chief Investigator: Viet Le

I refer to your request to modify the approved protocol for the above project. The request was put to a SUHREC/SHESC delegate for consideration.

I am pleased to advise that, as modified to date, the project may continue in line with standard ethics clearance conditions previously communicated and reprinted below. Please note that information on self-auditing, progress/final reporting and modifications/additions to approved protocols can now be found on the Research Ethics Internet pages.

Please contact the Research Ethics Office if you have any queries about on-going ethics clearance, citing the project number. A copy of this correspondence should be retained as part of project record-keeping and forwarded to relevant members of the project team.

This modification was approved during COVID-19 restrictions. The conduct of the research during this period should reflect any changes in relation to university and government COVID-19 mandates in the relevant jurisdictions. To ensure you have accommodated these mandates please refer to the Swinburne Ethics COVID-19 website here.

As before, best wishes for the project.

Yours sincerely,

Ms Leah Barham

Research Ethics Office

Swinburne University of Technology

P: +61 3 9214 8145 | E: resethics@swin.edu.au

A4 Final ethics report acceptance email

Astrid Nordmann To: Viet Le; Nilusha Gallage Cc: RES Ethics Dear Viet and Nilusha This report has now been approved. Kind regards Astrid Dr Astrid Nordmann | Research Ethics Coordinator Research Services | Swinburne University of Technology Ph +61 3 9214 3845] anordmann@swin.edu.au Level 1, Swinburne Place South

From: donotreply@infonetica.net <donotreply@infonetica.net> Sent: Wednesday, 6 July 2022 8:20 AM To: Christopher Selvarajah <<u>cselvarajah@swin.edu.au</u>> Cc: RES Ethics <<u>resethics@swin.edu.au</u>> Subject: Acknowledgement of Final Report for 20222915-10250

24 Wakefield St, Hawthorn VIC 3122, Australia

www.swinburne.edu.au

Dear Viet,

The Final Report for project 20222915-10250 : Developing Graduate Entrepreneurs: A model of Entrepreneurship Education and Experience for Private Higher Education Institutes in Colombo, Sri Lanka has been processed and satisfies the reporting requirements set under the terms of ethics clearance.

Regards, Dr Astrid Nordmann **Research Ethics Office Swinburne University of Technology** P: +61 3 9214 3845 | E: <u>resethics@swin.edu.au</u>

Appendix B: Qualitative data collection

B1 Interview protocol for internal stakeholders

INTERVIEW PROTOCOL FOR INTERNAL STAKEHOLDERS

Basic Information:

Title: Graduate Entrepreneurship: A model of Entrepreneurship Education and Experience for Private Higher education providers in Sri Lanka

Date:	Time:	Interviewer: NG	
Interviewee:	Institution:	Transcription Code:	

Introduction:

I am Nilusha Gallage and I am pursuing my PhD at Swinburne University, Melbourne. Previously I was an academic at a leading private higher education institute in Sri Lanka. The purpose of this study is to explore how institutes can facilitate its students from various fields of study within its environment to become entrepreneurs. The interview will include eight to ten questions and should take 30–45 minutes. Feel free to ask me any questions you may have now or at the end of the interview. Before beginning this interview, I request your consent to participate in this interview and if you agree please on the reverse of this document.

Opening Question: Is entrepreneurship encouraged among students at your institution? And tell me about the student population in terms of the various disciplines.

Content Questions:

- 1. What is your opinion of the higher education entrepreneurial ecosystems at your institution? ecosystems in the private higher education sector?
- 2. How do you perceive the future development of the higher education entrepreneurial ecosystem at your institution? What should the institutions focus on as student related outcome(s) achieving through their ecosystem?
- 3. Does the institution offer a degree in entrepreneurship? What is the availability of entrepreneurship education study programs for undergraduates in private higher education?
- 4. What types of entrepreneurial support is available for students within the higher education entrepreneurial ecosystems?
- 5. How are stakeholders currently involved in higher education entrepreneurial ecosystems? Who do you see as key stakeholders of HEEEs? How can stakeholders engage within factors/activities you mentioned earlier?
- 6. How do you see yourself supporting higher education entrepreneurial ecosystems in the future?

Closing Question: In this interview, we discussed regarding higher education entrepreneurial ecosystems including entrepreneurship education and entrepreneurial support. Is there anything else that you would like to state or share your comments related to the same?

Closing Instructions:

Thank you for participating in this research. It was interesting and important to know your opinions and perceptions. I assure the confidentiality of this interview. Once the PhD thesis is submitted, the abstract will be shared with you and when publications leading from the research are published, the links will be communicated as well. Feel free to ask any questions you may have.

B2 Interview protocol for external stakeholders

INTERVIEW PROTOCOL FOR EXERNAL STAKEHOLDERS

Basic Information:

Title: Graduate Entrepreneurship: A model of Entrepreneurship Education and Experience for Private Higher education providers in Sri Lanka

Date:	Time:	Interviewer: NG
Interviewee:	Institution:	Transcription Code:

Introduction:

I am Nilusha Gallage and I am pursuing my PhD at Swinburne University, Melbourne. Previously I was an academic at a leading private higher education institute in Sri Lanka. The purpose of this study is to explore how institutes can facilitate its students from various fields of study within its environment to become entrepreneurs. The interview will include eight to ten questions and should take 30–45 minutes. Feel free to ask me any questions you may have now or at the end of the interview. Before beginning this interview, I request your consent to participate in this interview and if you agree please on the reverse of this document.

Opening Question: Do young entrepreneurs benefit from higher education? And what is the role of private institution in nurturing graduate entrepreneurs?

Content Questions:

- 1. What is your opinion of current higher education entrepreneurial ecosystems among private higher education institutions?
- 2. How do you perceive the future development of these higher education entrepreneurial ecosystems? What should higher education institutions focus on as student related outcome(s) achieving through their ecosystem?
- 3. How is the availability of entrepreneurship education study programs for undergraduates in private higher education?
- 4. What types of entrepreneurial support is available for students within the higher education entrepreneurial ecosystems?
- 5. How are stakeholders currently involved in higher education entrepreneurial ecosystems? Who do you see as key stakeholders of HEEEs? How can stakeholders engage within factors/activities you mentioned earlier?
- 6. How do you see yourself engaging with higher education entrepreneurial ecosystems in the future?

Closing Question: In this interview, we discussed regarding higher education entrepreneurial ecosystems including entrepreneurship education and entrepreneurial support. Is there anything else that you would like to state or share your comments related to the same?

Closing Instructions:

Thank you for participating in this research. It was interesting and important to know your opinions and perceptions. I assure the confidentiality of this interview. Once the PhD thesis is submitted, the abstract will be shared with you and when publications leading from the research are published, the links will be communicated as well. Feel free to ask any questions you may have.

B3 Participant information statement

Project Title

Developing Graduate Entrepreneurs: A model of entrepreneurship education and experience for Private Higher Education Institutes in Colombo, Sri Lanka

Researcher	Position	Affiliation
Professor Christopher Selvarajah	Chief Investigator	Swinburne University of
Dr Richard Laferriere	Associate Supervisor	Technology, Melbourne
Nilusha Gallage	Student investigator	

Investigators and Other Project Personnel

Introduction to Project and Invitation to Participate

This research project is a part of the doctoral studies undertaken by the student investigator, Nilusha Gallage, and supervised by Professor Christopher Selvarajah and Dr Richard Laferriere at Swinburne University in Melbourne. The project intends to explore in the context of private higher education institutes in Colombo, Sri Lanka; How can entrepreneurship education help undergraduates to become graduate entrepreneurs? To pursue this project, we would like to invite you, as a member of the private higher education sector and/or a stakeholder of the entrepreneurship education system, to participate in an online interview relating to entrepreneurship education. In doing so we solicit participants' perceptions and experiences on private higher education and/or entrepreneurship education in Sri Lanka.

For this project, we seek participation from individuals and organisations relevant to the study. If you are a top management representative of a private higher education institute, please authorise faculty/academics and mentors to participant in this research. You can do this by providing the names and email addresses of nominees in the consent form, which will be forwarded via email if you are willing to take part in this research. If you are an individual such as an entrepreneur or a top management representative of an incubator organisation, you are invited to participate in the project. Each participant, including nominees, is invited to voluntary participation in this research.

Project Rationale

To address the national challenge of youth unemployment and underemployment in Sri Lanka, this research explores the entrepreneurial development needs of undergraduate students towards becoming entrepreneurial after completing their first degree at Private Higher Education Institutes. This project will investigate how undergraduate education shapes the entrepreneurial intention of aspiring young graduates. With your insights, we will be able to develop and propose a *Model of Entrepreneurship Education and Experience for Private Higher Education Providers* that will be useful for academic researchers, industry practitioners, policymakers and other stakeholders. The findings of the project will be produced as a thesis and publications such as academic journal articles and book chapters.

Researcher's interests

This project is being undertaken to satisfy the requirements for the completion of a PhD qualification currently being undertaken by the Student Investigator, Nilusha Gallage, at Swinburne University of Technology in Melbourne, Australia.

Participation in this project

This research will be one of the first studies in Sri Lanka to include perspectives from five stakeholders in a study on entrepreneurship education. Therefore, the research will involve the participation of two groups: (1) faculty/academics and mentors from institutes and (2) alumnus, expert entrepreneurs, and top management of incubator organisations. If you represent one of these positions, you are invited to voluntary participation in this study.

After understanding the importance of this project, if you decide to participate, you are expected to take part in an online interview with the student investigator on a date and time that is convenient to you during the data collection period. It is anticipated that the interview will take approximately 30 to 45 minutes depending on the willingness and availability of the interviewee. This could also be shorter or longer according to your experience and discussion in the research area. You are not required to supply any confidential data during the interview or company records for this study. The interviewee's consent to participate will be requested before conducting the interview where the consent form will be forwarded via email to be signed and reverted before the interview. Upon your consent, the interview will be conducted through Skype or Zoom by the student investigator, and information on connecting to the interview will be shared well in advance after consent to participate is received and a date/time is scheduled. The semi-structured interview involves 10 questions on entrepreneurship education in the Sri Lankan context and follows up questions that may arise during the interview. The online interview will be recorded with the participant's consent and recordings will be held only with the research team. All collected data will remain private confidential and are used only for the study.

The participation in this interview is voluntary, with no implicit or explicit coercion to participate. Even if you are nominated to participate, it is your independent choice to participate. Your participation will contribute towards the research findings in entrepreneurship education which can create an economic and social impact in the future for Sri Lanka. Participating private higher education institutes will be invited to phase two data collection which will involve gathering students' perceptions of entrepreneurship education via a survey questionnaire.

Risks and Benefits

The project intends to capture your genuine perceptions and opinions on entrepreneurship education provided by private higher education institutes in Sri Lanka. It is a time relevant topic to engage and share your experiences as an individual/organisation representing the entrepreneurship education ecosystem. This research will establish evidence-based findings on entrepreneurial development needs that can contribute to support aspiring young entrepreneurs in starting a business venture after completing the first degree. Suggested managerial directions will aid to reform private higher education in Sri Lanka for the future of undergraduate education. The start-up ventures that will emerge may result in an economic and societal benefit for the country and its community. Especially, the recommendations of actions will contribute to resolving the youth unemployment and underemployment that Sri Lanka is

currently suffering. These findings will still relate to developing graduate entrepreneurs post COVID-19.

There are no known risks associated with the participant of this research in either the short or long term. To avoid deception, the interviewee/participant is provided with all related information through this explanatory statement. The interviewee will be requested to inform consent of participation before the interview. Should the participant decide not to undertake in this research, there is no risk to bear. This confirms that participation in this interview is voluntary, with no implicit or explicit coercion to participate.

Participants are invited to answer questions based on an individual reflection of entrepreneurship education in a private conversation with the student investigator. If any adverse event may occur during or after the interview, you may contact your immediate supervisor or the student investigator will direct suitable assistance to avoid any psychological risk. As professionals in respective fields, all your genuine feedback is accepted and kept strictly confidential. In case, if any negative information is shared by the participant during the interview this information will be managed accordingly posing no risk or harm to anyone or any organisation. All interview responses are de-identified and collectively analysed which means there is no possibility to identify or divulge the source of negative comments. Further, the project will maintain confidentially when discussing and publishing findings which means it will not state that 'Participant's Name' from 'Organisation's Name' made this comment. If any quotes are shared in findings, the specifics of the participant will not be disclosed maintaining research ethics. Secure data management will be practiced at all points of the project including data collection and analysis.

Free Consent and Withdrawal from Participation

Participation in this research requires informed 'consent' from respondents. Management representatives of institutes require to provide consent and authorise employees as representatives to participate as an institute. Nominated employees, alumnus, entrepreneurs and top management members of the incubator organisations will be invited to provide written consent to participate in the study as an individual. Therefore, there are two consent forms – one for authorising employees and the other for individuals. The appropriate consent form will be sent via email to you in the second email. The participant will be requested to sign off consent before confirming the online interview and participating in the interview. The signed consent of participation must be returned to the researcher via email and will be securely retained by the researcher where only the research team will have access to the signed consent form, contact information and interview data.

If you decide to participate in the research, you have the right to avoid any question(s) that they may not wish to answer, pause, or withdraw from the study at any time. You may terminate the interview during the interview and/or withdraw the interview data from the research project during three months after the completion of your interview. You can contact the student investigator and communicate that you wish to be excluded from the interview data by completing the withdrawal of the consent form.

Privacy and Confidentiality

This research involving humans has been reviewed and approved by the Swinburne's Human Research Ethics Committee (SUHREC) in line with the National Statement on Ethical Conduct in Human Research (2008). All information related to participating in this research and interview data will be kept strictly confidential. All data is subject to standards of the Swinburne University Code of Ethics 2008.

All possible measures are taken to maintain privacy and confidentiality while minimising any possible risks for the participants of this study.

- 1. Prospective research participants for data collection are contacted by the student investigator and no contact details (such as telephone numbers and email addresses) are made available to Swinburne or any external party. This contact information is recorded and stored in access-controlled storage.
- 2. Each participant, even if nominated by the employer, has the right to consent given that this is a voluntary study. Participants are informed of their consent in the email/LinkedIn invitation, participant information sheet and consent form itself.
- 3. All communication between participant and researcher including interview responses will be kept private and confidential.
- 4. If a participant decides to accept the invitation and engage in the online interview, the electronically signed consent form will be retained by the researcher and only the research team will have access to the secured data storage.
- 5. Online interviews will be a private conversation between participant and researcher on a secure video-conferencing meeting at convenience.
- 6. Recorded interviews will be transcribed by the researcher after de-identifying the participant.
- 7. Analysis of interview data will take place collectively using relevant software with password protection, which means the study will not assess findings individually.
- 8. If any negative information is shared by the participant during the interview this information will be managed accordingly posing no risk or harm to anyone or any organisation. Interview data is deidentified when transcribed and collectively analysed giving no possible to identify the source.
- 9. Secure data management will be practiced at all points of the project including data collection and analysis. This will include the contact information of participants, their signed consent forms and interview data.
- 10. There will be no loose hard copy data such as interview notes.
- 11. The interview data will be retained for a minimum of five years after the research outcomes.
- 12. There is no involvement of any external company during the data collection, preparation, or analysis. All data will only be accessible by the research team listed above.
- 13. The study will preserve the confidentiality of participants when discussing and publishing findings.

Research output

This qualitative study including online interviews will be used as the basis for the next phase of quantitative research among the student population. From this project, the research output will be a PhD. Thesis which is equivalent to 70,000 - 100,000 words, and a submission presentation fulfilling the academic requirement at Swinburne University. The research findings will also be published in academic journals and a book chapter. When research findings are published, the source in the form of a link or PDF document will be shared among participated individuals, institutes and organisations.

Collected interview data may be used for future research in building a longitudinal study within the same research problem or for an emerging research gap related to entrepreneurship education. If you consent and participate in this study, you may be contacted for future related research as well. You have the opportunity to express your interest in this on the consent form. However, it will be voluntary participation giving you the ability to decide on your participation in future studies.

Support Services

If you require any counselling or support services, please contact your immediate supervisor. Alternatively, you have access to the below helpline: For free telephone counselling service in Sri Lanka dial 1333

Further information about the project

If you would like further information about the project, please contact:

Chief Investigator Prof. Christopher Selvarajah School Business, Law and Entrepreneurship Swinburne University of Technology Email: cselvarajah@swin.edu.au

If you would like further information about the interview or project, please contact:

Student Investigator Nilusha Gallage School Business, Law and Entrepreneurship Swinburne University of Technology

Email: ngallage@swin.edu.au

Concerns/complaints about the project

This project has been approved by or on behalf of Swinburne's Human Research Ethics Committee (SUHREC) in line with the National Statement on Ethical Conduct in Human Research. If you have any concerns or complaints about the conduct of this project, you can contact:

Research Ethics Officer, Swinburne Research (H68), Swinburne University of Technology, P O Box 218, Hawthorn VIC 3122 Australia. +61392143845

B4 Consent instrument for authorising employees



Swinburne University of Technology

	Principal Investigator: Professor Christopher Selvarajah		
1.	On behalf of:	rganisatio	on)
	I hereby authorise the following employee(s) to participate in the project in a representative capacity, the project's particulars having been satisfactorily explained to me:		
	Name of representative(s): and		
2.	In relation to this project, please circle your response to the following:		
•	I agree that s/he can be interviewed by the researcher	Yes	No
•	I agree that the interview can be recorded by electronic device	Yes	No
•	I would like to check any transcription / citation in respect of my organisation's involvement	for accu	racy
		Yes	No
3.	Please circle your response to the following:		
•	I give my permission for the organisation to be named in any publication arising from the res	search.	
		Yes	No
•	I further give my permission for the named researcher(s) to access/analyse organisational rec	ords as	
	requested.	Yes	No
•	I understand the length of time researcher(s) will have access to data/records for analysis		
		Yes	No
4.	I acknowledge that the data collected for the Swinburne project will be used for research pur	poses and	d
	not for direct profit; research purposes may include publishable / peer reviewed outcomes.		
	Name of Person of Authority and Position.		



Swinburne University of Technology

Project Title: Developing Graduate Entrepreneurship: A model of entrepreneurship education and experience for private higher education providers in Colombo, Sri Lanka

Principal Investigator(s): Prof. Christopher Selvarajah

- 1. I consent to participate in the project named above. I have been provided a copy of the project consent information statement to which this consent form relates and any questions I have asked have been answered to my satisfaction.
- 2. In relation to this project, please circle your response to the following:

•	I agree to be interviewed by the researcher	Yes	No
•	I agree to allow the interview to be recorded by electronic device	Yes	No
•	I agree to make myself available for further information if required	Yes	No

- 3. I acknowledge that:
 - (a) my participation is voluntary and that I am free to withdraw from the project at any time without explanation;
 - (b) the Swinburne project is for the purpose of research and not for profit;
 - (c) any identifiable information about me which is gathered in the course of and as the result of my participating in this project will be (i) collected and retained for the purpose of this project and (ii) accessed and analysed by the researcher(s) for the purpose of conducting this project;
 - (d) I understand the length of time researcher/s will have access to this information;
 - (e) my anonymity is preserved and I will not be identified in publications or otherwise without my express written consent.

By signing this document I agree to participate in this project.

Name of Participant:

Signature and Date:

B6 Email invitation

Dear <name>,

RE: Invitation to participate in an interview for research on Higher Education Entrepreneurial Ecosystems

I am Nilusha Gallage and I am pursuing my PhD at Swinburne University, Melbourne. My research is titled as 'Developing Graduate Entrepreneurs: A model of Entrepreneurship and experience for Private Higher education providers in Colombo, Sri Lanka' and the study is supervised by Professor Christopher Selvarajah and Dr Richard Laferriere from the Faculty of Business and Law. Previously I was an academic at a leading private higher education institute in Sri Lanka. You may view my LinkedIn profile on https://www.linkedin.com/in/nilushagallage/ for my background.

The purpose of this research is to explore how private higher education providers located in Colombo can develop their students towards entrepreneurship. Therefore, we would like to know, as a member of the private higher education sector and/or a stakeholder of the ecosystem, how this is addressed. As you are identified as an important stakeholder who can tell us about your perceptions and experiences of private higher education, its ecosystem and/or youth entrepreneurship in Sri Lanka, we invite you or an authorised representative(s) to participate in an online interview at a convenient date and time. The interview includes ten questions and should take no longer than 30–45 minutes of the participant's time.

For more information, I have attached the Consent Information Sheet explaining the project, its rationale, interests, risks and benefits, consent, privacy and confidentiality, research output and contact information. Your participation in this research is voluntary and I look forward to hearing your response. If you are willing to contribute to the study, please respond to this email to arrange a date and time for interviewing at your convenience. If you have any questions at this stage, please feel free to get back to me.

Thanking you in advance.

Best regards,

Student Investigator / PhD Candidate Nilusha Gallage School of Business, Law and Entrepreneurship Swinburne University of Information Technology Email: <u>ngallage@swin.edu.au</u>

Appendix C: Graduate Certificate of Research and Innovation Management

Year	Study Period	Unit	Short Title	Ver	Mark	Grade Code	Grade Description	Credit Points
2021	Semester 2	BUS80017	Quantitative Research Methods	1	78	D	Distinction	12.50
2020	Semester 2	BUS80018	Qualitative Research Methods	1	80	HD	High Distinction	12.50
2019	Winter	ENT60010	Innovation and Impact in Research	2	84	HD	High Distinction	12.50
2019	Winter	INF60016	Project Management for Research	2	90	HD	High Distinction	12.50

Appendix D: Publication information



Swinburne Research

Authorship Indication Form

For HDR students

NOTE

This Authorship Indication form is a statement detailing the percentage of the contribution of each author in each submitted/published 'paper'. This form must be signed by each co-author and the Principal Supervisor. This form must be added to the publication of your final thesis as an appendix. Please fill out a separate form for each published paper to be included in your thesis.

DECLARATION

We hereby declare our contribution to the publication of the 'paper' entitled:

Ecosystem Engagement in Entrepreneurship Education: A View from Sri Lanka

First Author

Name Nilusha Gallage

Signature:

Date: 18/05/2022

Percentage of contribution: 85%

Brief description of contribution to the 'paper' and your central responsibilities/role on project: Conceptualisation, methodology, investigation, data analysis, writing and visualisation

Second Author

Name: Dr Richard Laferriere

Percentage of contribution: 10_%

Brief description of your contribution to the 'paper':

Conceptualisation and validation

Third Author

Name: Professor Christopher Selvarajah

_

Percentage of contribution: 05%

Brief description of your contribution to the 'paper': Supervision

Signature:

Date: 0 5 / 08 / 2022

	p	
ignature:	Agen	-

S

Date: 08 / 07 / 2022

Fourth Author

Name:

_Signature:_____

Percentage of contribution: ____%

Date: __/ __/ ____

Brief description of your contribution to the 'paper':

Principal Supervisor:	
Name: Viet Le	Signature:
Date: _05 / 08 / 2022 _	
In the case of more than four authors contribution of the authors.	s please attach another sheet with the names, signatures and

Authorship indication Form



Swinburne Research

Authorship Indication Form

For HDR students

NOTE

This Authorship Indication form is a statement detailing the percentage of the contribution of each author in each submitted/published 'paper'. This form must be signed by each co-author and the Principal Supervisor. This form must be added to the publication of your final thesis as an appendix. Please fill out a separate form for each published paper to be included in your thesis.

DECLARATION

We hereby declare our contribution to the publication of the 'paper' entitled:

Higher education entrepreneurial ecosystems for students' entrepreneurial development

First Author

Name_Nilusha Gallage

Signature:

Date: 18/05/2022

Percentage of contribution: __90_%

Brief description of contribution to the 'paper' and your central responsibilities/role on project: Conceptualisation, methodology, investigation, data analysis, writing and visualisation

Second Author

Name: Dr Viet Le

Percentage of contribution: <u>05</u>%

Brief description of your contribution to the 'paper': Validation and supervision

Third Author

Name: Dr Richard Laferriere

Percentage of contribution: __05__%

Signature:

Date: 06/07/22

Signature:

Date: 05 / 08/ 2022

Brief description of your contribution to the 'paper': Supervision

Fourth Author

Name:

_Signature:____

Percentage of contribution: ____%

Date:	/	/		
-------	---	---	--	--

Brief description of your contribution to the 'paper':

Principal Supervisor:	
Name: Dr Viet Le	_Signature:
Date: _05/ 08_/22	
In the case of more than four authors please attach a contribution of the authors.	nother sheet with the names, signatures and

Authorship Indication Form

Appendix E: Change of thesis title

SWIN	Swinburne Research
BUR * NE *	Change of Thesis Title
PLEASE READ BEFORE COMPLETING FORM	
A Change of Thesis Title involves altering the wording of the re topic. This refinement does not involve a change in topic or rese	esearch title in order to more accurately and more clearly reflect the nature of the arch area or necessitate any other change in candidature.
If the topic of your thesis is being changed, do not complete	e this form. Please complete the Change of Research Project Topic instead.
Higher Degree by Research candidates must obtain approval for completed form should be emailed to <u>HDRcandidature@swin.ed</u>	r a requested title change from their Principal Supervisor. Upon approval, the <u>lu.au</u> .
This form will be kept on your student record in accordance with Policy can be viewed at: <u>http://www.swinburne.edu.au/corporate</u> .	University Privacy Policy. The Swinburne University of Technology Privacy /registrar/privacy_collection.htm
SECTION A: PERSONAL DETAILS	
Sumame or Family Name: Gallage	Given Names: Nilusha Gayani
Student ID: 102364009	Program of Enrolment: Doctor of Philosophy
Faculty of Enrolment: 🛛 FBL 🗌 FHAD	
Current Title: Developing graduate entrepreneurs: A model of en Lanka Proposed New Title:	ntrepreneurship education for private higher education providers in Colombo, Sri
An exploratory study of multiple stakeholder views on higher edu	ucation entrepreneurial ecosystems in a resource-constrained environment
Reason for change in title: The current title was tentatively set at title reflects the nature of study more accurately and clearly. It is	t the beginning of the doctoral study and after completing the study, the proposed not a change of topic.
I confirm that this change to the wording of the research title is re topic. This requested change does not involve a change in project	equested in order to more accurately and more clearly reflect the nature of the ct topic or research area or necessitate any other change in candidature.
Candidature Signature:	
Date: 21 Jun 2022	
SECTION C. PRINCIPAL SUPERVISOR APPROVAL	
Name of Principal Supervisor: I confirm that this change to the wording of the research title is a topic. This approved change does not involve a change in projec	pproved in order to more accurately and more clearly reflect the nature of the ct topic or research area or necessitate any other change in candidature.
Principal Supervisor Signature:	
Frincipal Supervisor Signature:	

Updated June 2019